



POLITECNICO DI TORINO

PH.D. PROGRAM IN "PURE AND APPLIED MATHEMATICS" DIPARTIMENTO DI MATEMATICA "G. PEANO" DIPARTIMENTO DI MATEMATICA "G.L. LAGRANGE"

PH. D. THESIS

ON THE NODAL SET OF SOLUTIONS OF DEGENERATE - SINGULAR AND NONLOCAL EQUATIONS

CANDIDATE: GIORGIO TORTONE ADVISOR: PROF. SUSANNA TERRACINI

> ACADEMIC YEAR 2017-2018 XXX PH. D. CYCLE

Men fear thought as they fear nothing else on earth, more than ruin, more even than death. Thought is subversive and revolutionary.

— Bertrand Russel

Guarda che luna, guarda che ..

Fred Buscaglione

Esatto! — Dogui Nicheli

ACKNOWLEDGEMENTS

I obviously start by showing my gratitude to my advisor Prof. Susanna Terracini for having supported me along this beautiful trip around beautiful problems and hard works. She always encouraged me and really made me love math with her passions and visions of mathematics. I will never forget your advise to find something, in the research as in the life, for which you are ready to fight and also for which you are willing to keep you awake at night.

I would like to thanks for all the stupid question, useful discussion and all the swearwords screamed in front of some laptop with Stefano Vita. We shared travels, conferences, "indie" songs, rap music videos and worries about the future.

This work would not be here without many suggestions and discussions with many colleagues who I met in these years, in particular I would like to mention the uncountable questions asked to Dario Mazzoleni and the constant support of my co-advisor Alessandro Zilio. A special thanks goes to all the new colleagues I met in these three years: Alessandro Audrito, Gian Marco Canneori, Eleonora Cinti, Gabriele Cora, Alessandro Iacopetti, Yannick Sire, Nicola Soave, Davide Zucco and all the others.

I will always be grateful for the "cinepattoni deeliziosi" and coffees with Filippo, the insuperable not politically correct jokes of Gabriele and the patience of the new kid Gian Marco. Obviously thanks for the mad train rides with Alice, the marocchino con Pistacchio with Sara, who more than everyone got in touch with my craziness, the wisest advises of Nietta and the discussions on the couch with Saverio.

Many thanks to my family that beard all my change of programs, my nervous piano sessions and keep to support me. Last but not least, thanks to Martina for the infusions, the glasses of still white wine, the infinite amount of "taglieri" and the beautiful unpredictability that always kept me smiling.

This work has been partially supported by the ERC Advanced Grant 2013 n. 339958 "*Complex Patterns for Strongly Interacting Dynamical Systems - COMPAT*", and by the GNAMPA project "*Aspetti non-locali in fenomeni di segregazione - Non-local aspects in segregation phenonema*".

CONTENTS

INTRODUCTION iii	
1 NODAL SET OF SEGREGATED CRITICAL CONFIGURATIONS 1	
1.1 Introduction 1	
1.2 Almgren's type monotonicity formula 5	
1.3 Compactness of the blow-up sequences 13	
1.4 Hausdorff dimension estimates for regular and singular sets 30	
1.5 Regularity and flatness of the regular set 38	
1.6Singular limit of a competition-diffusion problem45	
2 ON <i>s</i> -HARMONIC FUNCTIONS ON CONES 51	
2.1 Introduction 51	
2.2 Homogenous <i>s</i> -harmonic functions on cones 56	
2.3 Characteristic exponent $\gamma_s(C)$: properties and asymptotic behaviour 63	
2.3.1 Distributional semigroup property 64	
2.3.2 Monotonicity of $s \mapsto \gamma_s(C)$ 74	
2.3.3 Asymptotic behavior of $\frac{C(n,s)}{2s-\gamma_s(C)}$ 77	
2.4 The limit for $s \nearrow 1$ 83	
2.5 Uniform estimates in $C^{0,\alpha}$ on annuli 89	
3 NODAL SET OF SOLUTIONS OF DEGENERATE - SINGULAR EQUATIONS	95
3.1 Introduction 95	
3.2 Decomposition of L_a -harmonic functions 99	
3.3 Almgren type monotonicity formula 105	
3.4 Compactness of blow-up sequences 116	
3.5 Uniqueness and continuity of tangent maps and tangent fields 131	
3.6 Hausdorff dimension estimates for the nodal set 149	
3.7 Regularity of the Regular and Singular strata 155	
3.8 Fractional power of elliptic operator in divergence form 166	
3.9 Measure estimates of nodal sets of <i>s</i> -harmonic functions 188	
REFERENCES 191	

ii CONTENTS

INTRODUCTION

In this thesis we are concerned with the study of the nodal set of nontrivial solutions of different types of elliptic equations strictly related to a problem of singularly perturbed systems of nonlocal elliptic equations.

The manuscript is divided in three different parts, corresponding to the main problems treated during the last years. The first one is devoted to the study of the nodal set of a segregated critical configuration arising as singular limit of a system of elliptic nonlocal equations with strongly competing interaction terms, while in the second one we consider the problem of *s*-harmonic functions on cones when the parameter *s* approaches 1, wondering whether solutions of the problem do converge to harmonic functions in the same cone or not.

Finally, in the last part we focus the attention on the nodal set of solutions of a class of degeneratesingular elliptic equation trying to understand how the presence of degeneracy and singularity in the coefficient affects the structure and the regularity of the solutions. Moreover, in this last part, we find a remarkable link with the problem of the nodal set of *s*-harmonic functions.

Before moving on, we would like to stress that all the Chapters are not only centred on the research topic of the nodal set of solutions of partial differential equations, but they represent three key points in the analysis of patterns formation through spatial segregation in some models of enhanced anomalous diffusion.

Several physical phenomena can be described by a certain number of densities, populations or probabilities distributed in a domain and subject to laws of diffusion, reaction, and competitive interaction. In the pioneering work [53] of Georgii Gause of the 1932, has been introduced the so called "*competitive exclusion principle*" which states that whenever the competitive interaction is the prevailing phenomenon, the densities can not coexist simultaneously and tend to segregate, hence determining a partition of the domain itself.

As a model problem, let us start with the system of stationary equations

$$\begin{cases} -\Delta u_{i,\beta} = f_{i,\beta}(u_{i,\beta}) - \beta u_{i,\beta} \sum_{j \neq i} g_{ij}(u_{j,\beta}) \\ u_{i,\beta} > 0. \end{cases}$$

In particular, the cases $g_{ij}(t) = \beta_{ij}t$ (Lotka-Volterra competitive interactions) and $g_{ij}(t) = \beta_{ij}t^2$ (focusing-defocusing Gross-Pitaevskii system) are of particular interest in the applications to population dynamics [75] and theoretical physics [54, 61] respectively. For the case of standard diffusion, the regularity of solutions and the asymptotic analysis is fairly well understood, starting from [11, 20, 31] for the Lotka-Volterra case and from [17, 26, 30, 33] for the Gross-Pitaevskii system, in a series of recent papers [19, 32, 35, 72, 90], also in the parabolic case [36, 34, 35, 89]. Since, from a modeling point of view, the limiting configurations as $\beta \rightarrow +\infty$ describe an approximation of highly competing systems, one crucial step of this analysis is the study of the qualitative properties of the singular limit. Indeed, in the classic case has been shown that the limit vector $\mathbf{u} = (u_1, \ldots, u_h)$ has densities with mutually disjoint supports, i.e. the segregated states u_i satisfy

$$\begin{cases} -\Delta u_i = f_i(u_i) & \text{in } \{u_i > 0\} \\ u_i \cdot u_j \equiv 0 & \text{for } j \neq i. \end{cases}$$
(1)

The natural subjects of this analysis concern the optimal regularity of the limiting profiles, equilibrium principle at the arising interfaces and regularity of the free boundary itself. In the mentioned papers, the authors in [30, 33] studied singularly perturbed systems relating them to some optimal partition problem for nonlinear eigenvalues. For this latter problem, we remark that in [25] the authors have proved the regularity of free interfaces of optimal partition problems for the eigenvalues of the Laplacian operator with Dirichlet boundary conditions. Moreover, in [32, 29] they proved Lipschitz regularity of the limiting solutions as well as the regularity of the free boundaries in the case of two dimension.

On the other hand, in [71, 72] has been deeply studied that the limits of a system of Gross-Pitaevskii equations relying the proof on elliptic estimates, blow-up technique, the monotonicity formula by Almgren [1] and Alt-Caffarelli-Friedman type formula [2, 3]. We mention the book of Caffarelli-Salsa [21] for a complete picture of the application of the monotonicity formulas for a larger class of free boundaries problem and to [16] for the case of more general nonlinearities.

The common characteristic of all these problems is that in the singular limit, the components of the solutions of these systems group in different blocks and the supports of the different blocks become disjoint. In particular, these are specific cases of free boundary problems and they are strictly connected to the problem of nodal and critical point sets of solution for PDEs, which is itself a research topic that has attracted a great deal of attention in the last decades (see e.g. [15, 41, 56, 57, 58, 66]). The philosophy is that both in the case of energy minimizing solutions and critical ones, the limiting segregated configurations satisfy a reflection law which represents the only interaction between the different densities through the common free boundary. Thanks to this reflection property, the free boundary is locally described as the nodal set of a scalar valued solution of some PDE.

From this perspective, before to compare this results with their nonlocal counterpart, we would like to give more attention to two recent papers [18, 81] which include all the previous cases and summarize the most recent results achieved in this research topic. In the first one, the authors

studied the local structure and the smoothness of singularities of the nodal set of a constrained harmonic maps into a singular space, i.e. given $\Omega \subset \mathbb{R}^n$ a bounded smooth domain and

$$\Sigma = \left\{ x \in \mathbb{R}^h \colon F(x) = 0 \right\} \quad \text{with } F(x) = \sum_{i \neq j} x_i^2 x_j^2,$$

the problem is to find a minimizer $\mathbf{v} = (v_1, \ldots, v_h) \in H^1_0(\Omega, \Sigma)$ such that

$$\int_{\Omega} |\nabla \mathbf{u}|^2 \, \mathrm{d}X = \min\left\{\int_{\Omega} |\nabla \mathbf{v}|^2 \, \mathrm{d}X \colon \mathbf{u} \in H_0^1(\Omega, \Sigma) \text{ such that } \int_{\Omega} \mathbf{u}^2 \mathrm{d}X = 1\right\}$$

We remark that this problem is strictly related to the one of optimal partition for the Dirichlet eigenvalue and contains the class of limiting profile arising as $\beta \to \infty$. In particular they proved a stratification result for the singular set, as was done in the classic case [79], by a convexity argument deeply based on the validity of Weiss type monotonicity formula [91]. Moreover, with a convexity argument, they proved uniqueness of tangent maps as well as the local structures of the singular sets.

Instead, in [81] the authors dealt with the nodal set of segregated critical configurations under a weak reflection law, i.e. they considered the class of functions $\mathbf{u} \in (H^1(\Omega))^h$ whose components are all nonnegative and Lipschitz continuous in the interior of Ω and such that

$$\begin{cases} -\Delta u_i = f_i(u_i) - \mu_i & \text{in } \mathcal{D}'(\Omega) \\ u_i \cdot u_j \equiv 0 & \text{for } j \neq i, \end{cases}$$

where f_i are a suitable collection of differentiable functions and $\mu_i \in \mathcal{M}(\Omega)$ some nonnegative Radon measures, each supported on the nodal set $\Gamma(\mathbf{u}) = \{x \in \Omega : \mathbf{u}(x) = \mathbf{0}\}$. Moreover, they impose the validity of a weak reflection principle based on some Pohožaev type identities, which implies that the absolute value of the gradient is the same when we approach the regular set from opposite sides. The importance of this class is due to the fact that it collects the singular limit to competition-diffusion systems, both those possessing a variational structure and those with Lotka-Volterra type interaction.

In the recent years has been given much attention on the case of anomalous diffusion, when the Gaussian statistics of the classical Brownian motion is replaced by a different one, giving rise to the so called "Lévy jumps". Since such operators are of real interest both in population dynamics (see [59]) and in relativistic quantum electrodynamics (see [64, 65]), we plan to extend the previous analysis of the nodal set of segregated configuration to the nonlocal context. Since the asymptotic analysis and the study of the nodal set in case of fractional Laplacians are



Figure 1: Prototype of segregated configurations in $\mathbb{R}^{n+1}_+ = \mathbb{R}^n_x \times \mathbb{R}^+_y$

very challenging issue, the only known results are contained in [83, 84, 85, 88, 86]. In [83, 84, 86], the authors considered the class of stationary systems of semilinear equations

$$\begin{cases} (-\Delta)^s u_{i,\beta} = f_{i,\beta}(u_{i,\beta}) - \beta u_{i,\beta} \sum_{j \neq i} g_{ij}(u_j) \\ u_{i,\beta} \in H^s(\mathbb{R}^n), \end{cases}$$

focusing on the case $g_{ij}(t) = \beta_{ij}t$ (Lotka-Volterra competitive interactions [86]) and $g_{ij}(t) = \beta_{ij}t^2$ (relativistic Gross-Pitaevskii system [83, 84]). In both cases, they provide some uniform estimates in Hölder spaces with respect to the parameter of competition β . This results can be obtained considering the local realisation of the fractional Laplacian due to the so called Caffarelli-Silvestre extension popularized in [23], which characterize the fractional Laplacian in \mathbb{R}^n as the Dirichlet-to-Neumann map for a variable v depending on one more space dimension. With this formulation, the competition-diffusion problem in \mathbb{R}^n translates into a degenerate-singular elliptic equation in \mathbb{R}^{n+1}_+ with a Neumann type condition on $\partial \mathbb{R}^{n+1}_+ = \mathbb{R}^n$, which allows to introduce the fractional versions of the Alt-Caffarelli-Friedman and Almgren type monotonicity formulas.

As a byproduct, up to subsequences, there is convergence of the above solutions to a limiting profile, which components are segregated. Because of the genuinely nonlocal nature of the problem, many difficulties and technicalities arise in the asymptotic analysis and in the study of the nodal set. First of all, since these problems have been studied using the extension technique, both in the Lotka-Volterra and in the variational case, the segregation occurs only in the *n*-dimensional space and it is natural to expect free boundaries of codimension 2 (see Figure 1). Secondly, the Gross-Pitaevskii competition and the Lotka-Volterra one exhibit deep differences not only from the point of view of the optimal regularity exponent, but also with the one of the

segregated limiting profiles, which is in deep contrast with the local case s = 1, as we previously pointed out. More precisely, in the first case the limiting profiles satisfy a natural extension to the fractional setting of (1), that is

$$\begin{cases} (-\Delta)^s u_i = f_i(u_i) & \text{in } \{u_i > 0\} \\ u_i \cdot u_j \equiv 0 & \text{for } j \neq i, \end{cases}$$

while in the second one

$$\begin{cases} (-\Delta)^s \left(u_i - \sum_{j \neq i} u_j \right) = f_i(u_i) - \sum_{j \neq i} f_j(u_j) & \text{in } \{u_i > 0\} \\ u_i \cdot u_j \equiv 0 & \text{for } j \neq i. \end{cases}$$

Moreover, in the Gross-Pitaevskii case, where the structure of the nodal set is wilder than the Lotka-Volterra one, the nonlocal nature of the problem affects the ideas and techniques developed in [18, 81]. In the latter, the definition of the nonlocal operator $(-\Delta)^s$ does not allow to relate the structure of the free boundary to the nodal set of *s*-harmonic function since, roughly speaking, the linear combination of *s*-harmonic functions with disjoint supports is no more *s*-harmonic in their union. Secondly, the nonlocal counterpart of the formulation via constrained harmonic maps into singular space introduced in [18], is intimately related to the problem of harmonic maps with "partially free boundary". Unfortunately, this strategy turn out to be inefficient since the segregated condition on \mathbb{R}^n translates into the problem of fractional harmonic maps into singular space which implies the occurrence of a "singular partially free boundary" (see [69] for an application in the context of fractional Ginzburg-Landau equations).

Last but not least, we remark that in [83, 84] the most challenging issue lies in the lack of the validity of an exact Alt-Caffarelli-Friedman monotonicity formula, which reflects, at the spectral level, the lack of convexity of the eigenvalues with respect to domain variations.

In Chapter 1 we tried to give a better picture of the limiting profiles in the context of variational competition. In this analysis, the main difficulties are the problem of codimension between the segregation and the degenerate-singular elliptic equation introduced with the extension technique and the lack of validity of a reflection principle that allows to compare our problem to the one of the nodal set of some nonlocal elliptic equation. In order to overcame this problem, we consider the case of planar segregated configurations, in order to exploit the topology of S^1 . Nevertheless, as pointed out in [84], we need to take care of the presence of self-segregation for $s \ge 1/2$. This phenomenon was also discussed in [81] for the local case with critical segregated configurations, but the nonlocal attitude of our problem prevents to apply the same reduction used for the classical Laplacian. We mention that in [35] the authors proved, when we consider segregated profiles arising as limit of competition-diffusion systems, that the self-segregation can be ruled out using an improvement of flatness. In our final result, we split the nodal set into its segregated and self-segregated part, proving a local regularity result near the two strata. Moreover, in the first case, we show the existence of a regular set $\mathcal{R}(\mathbf{u})$ relatively open in $\Gamma(\mathbf{u})$ which satisfy a vanishing Reifenberg flatness condition and a singular set $\mathcal{S}(\mathbf{u})$ which consists in a locally finite collection of singular points. All the result contained in Chapter 1 are obtained in collaboration with Susanna Terracini and Alessandro Zilio.

As already mention, in Chapter 2, we consider the problem of s-harmonic function on cone, i.e. given C an open cone in \mathbb{R}^n with vertex in 0 and $s \in (0, 1)$, we consider the problem of the classification of nontrivial functions which are s-harmonic inside the cone and vanish identically outside, that is

$$\begin{cases} (-\Delta)^s u_s = 0 & \text{in } C, \\ u_s \ge 0 & \text{in } \mathbb{R}^n \\ u_s \equiv 0 & \text{in } \mathbb{R}^n \setminus C. \end{cases}$$
(2)

By [5], it is known that there exists a homogeneous, nonnegative and nontrivial solution of the form

$$u_s(x) = |x|^{\gamma_s} u_s\left(\frac{x}{|x|}\right),$$

where $\gamma_s := \gamma_s(C)$ is a definite homogeneity degree (characteristic exponent of the cone C). This problems is actually deeply connected to the one of Chapter 1, since it consists on the study of such conic *s*-harmonic functions that appear as limiting blow-up profiles and play a major role in many free boundary problems with fractional diffusions and in the study of the geometry of nodal sets, also in the case of partition problems (see, e.g. [7, 14, 39, 51] and the blow-up analysis of Chapter 1). Moreover, as we shall see later, they are strongly involved with the possible extensions of the Alt-Caffarelli-Friedman monotonicity formula to the case of fractional diffusion. The problem of homogeneous *s*-harmonic functions on cones has been deeply studied in [5, 8, 9, 67]. and since not many qualitative properties are known for the *s*-harmonic functions on cones, we decided to focus our attention on the limiting behaviour as $s \nearrow 1$ wondering whether solutions of the problem do converge to a harmonic function in the same cone and, in case, which are the suitable spaces for convergence in order to deduce. In such a way, we wanted to deduce some qualitative properties of the *s*-harmonic function for *s* sufficiently near 1.

We therefore addressed the problem of the asymptotic behavior of the solutions of problem (2) for $s \nearrow 1$, obtaining a rather unexpected result: our analysis shows high sensitivity to the opening solid angle ω of the cone C_{ω} , as evaluated by the value of the homogeneity degree $\gamma(C_{\omega}) = \gamma_1(C_{\omega})$ of the harmonic function on C:

1. in the case of "wide cones", when $\gamma(C) < 2$ (that is, $\theta \in (\pi/4, \pi)$ for spherical caps), our solutions do converge to the harmonic homogeneous function of the cone;

instead, in the case of "*narrow cones*", when γ(C) > 2 (that is, θ ∈ (0, π/4] for spherical caps), then limit of the homogeneity degree will be always two and the limiting profile will be something different, though related through a correction term.



Figure 2: Characteristic exponents of spherical caps of aperture 2θ for s < 1 and s = 1.

This surprisedly result yields different nontrivial improvement in the context of segregated critical configurations. First of all, as shown in [84, 83], estimates in Hölder spaces can be obtained by the use of fractional versions of the Alt-Caffarelli-Friedman and Almgren monotonicity formulas. In particular, one could prove a deep connection with the optimal partition problem among the class \mathcal{P}^s of 2-partitions on S^{n-1}

$$\nu_s^{ACF} := \frac{1}{2} \inf_{(\omega_1, \omega_2) \in \mathcal{P}^2} \sum_{i=1}^2 \gamma_s(\lambda_1^s(\omega_i)),$$

where $\gamma_s(\lambda_1^s(\omega))$ is equal to the characteristic exponent of the cone spanned by $\omega \subset S^{n-1}$. A classical result by Friedland and Hayman, [47], yields $\nu^{ACF} = 1$ (case s = 1), and the minimal value is achieved for two half spheres; this equality is the core of the proof of the classical Alt-Caffarelli-Friedman monotonicity formula.

It [84] was also conjectured that $\nu_s^{ACF} = s$ for every $s \in (0, 1)$. Unfortunately, the exact value of ν_s^{ACF} is still unknown, but as a byproduct of our asymptotic analysis we have

$$\lim_{s \to 1} \nu_s^{ACF} = 1 \; .$$

In the end, we remark that this asymptotic analysis suggests that even the segregated configurations are affected by this unexpect phenomenon since the trace of the blow-up limits introduced in Chapter 1 belong to the class of s-harmonic functions on cones. We believe that this asymptotic

x CONTENTS

result will push our research in a new challenging direction. All these results are obtained in collaboration with Susanna Terracini and Stefano Vita (see [82]).

Finally, in the last Chapter we conclude the thesis dealing with the nodal set of solutions of degenerate-singular elliptic equations, which put an end to our glimpse into the study of segregated configurations and free boundary problems ruled by anomalous diffusion.

In literature, the subject of nodal sets, or level sets in general, is an important research topic for solutions of PDEs. While in some cases, this topics are themselves the primary concern, in many others they provide an important tool in the study of qualitative properties of solutions of PDEs. Initially, in [15, 58] the authors respectively proved an optimal bound on the Hausdorff dimension of the singular set of solutions of linear and superlinear elliptic equations and, in the second work, the first estimate on the (n - 1)-Hausdorff measure of the nodal set in a neighbourhood of a point with vanishing order. We remark that in the second paper, the estimate is explicit and based only on the existence of a finite order of vanishing, which suggests that the validity of a strong unique continuation property is the starting point of this kind of analysis.

Recently in [41, 56, 57, 66] they proved several results on the structure of the singular set and even some estimate on the (n-2)-Hausdorff measure of the singular set.

In all these cases, as pointed out in [48, 49, 66], the class of solution of PDEs, of which we want to study the nodal set, must satisfy a strong unique continuation principle, in order to ensure the existence of a finite vanishing order. Many improvement have been done in this topic, using on one side the Carleman estimates approach and on the other one the monotonicity approach (see e.g. [48, 49]), deeply based on the existence of an Almgren type monotonicity formulas and a geometrical reduction, first introduced in [4]. These last results prove the validity of the strong unique continuation principle for solutions of divergence form elliptic equations of the second order with Lipschitz leading coefficients and suitable lower order terms. In particular, in [68] the author proved the optimality of the Lipschitz condition with an Hölder continuous counterexample.

At the same time, in their pioneering papers [44, 43] the authors introduced a general class of degenerate operators $L = \operatorname{div}(A(X)\nabla \cdot)$ whose coefficient $A(X) = (a_{ij}(X))$ are defined starting from a symmetric matrix valued function such that

$$\lambda\omega(X) \, |\xi|^2 \leq (A(X)\xi,\xi) \leq \Lambda\omega(X) \, |\xi|^2 \,, \quad ext{for some } \lambda,\Lambda>0,$$

where ω may either vanish, or be infinite, or both. In particular they focus the attention on the case $\omega \in A_2$ -Muckenhoupt class, i.e.

$$\sup_{B \subset \mathbb{R}^{n+1}} \left(\frac{1}{|B|} \int_B \omega(X) \mathrm{d}X \right) \left(\frac{1}{|B|} \int_B \omega^{-1}(X) \mathrm{d}X \right) < \infty.$$

While in the recent years these operators are quite commonly used since they are strictly related to the local realisation of fractional powers of operators (see [23, 27, 80] for different application in the extension of fractional operator), the authors initiated this research motivated by some result on the boundary behaviour of harmonic functions in non-tangentially accessible domains in [60].

Inspired by [43, Section 3], given $X = (x, y) \in \mathbb{R}^n_x \times \mathbb{R}_y$ we consider the cases of $\omega(X) = |y|^a$, with $a \in (-1, 1)$, and the associated degenerate-singular operator L_a defined as $L_a = \operatorname{div}(|y|^a \nabla)$, with div and ∇ respectively the divergence and the gradient operator in \mathbb{R}^{n+1} . Now, given Σ the "*characteristic manifold*" associated to our weight, as the set of points where the coefficient either vanishes or blows up, we studied the properties of the nodal set $\Gamma(u)$ of solutions to

$$-L_a u = 0$$
 in $B_1 \subset \mathbb{R}^{n+1}$,

focusing the attention on the restriction of the nodal set $\Gamma(u)$ on the characteristic manifold Σ . Following the philosophy explained in the third paragraph of the Introduction, one motivation of our analysis is the application of this results on a competition-diffusion system with variational competition and degenerate-singular diffusion: one can imagine that the characteristic manifold Σ is playing an active role in the diffusion phenomenon, indeed we expect that the diffusion across the manifold Σ is penalized or encouraged accordingly to the value of $a \in (-1, 1)$.

On the other hand, the choice to study this class of 1-dimensional homogeneous weights, allows to extend our analysis to the cases when Σ is an *n*-dimensional manifold properly embedded in \mathbb{R}^{n+1} and the weights take the form $\omega(X) = \text{dist}(X, \Sigma)^a$ and even then to a wilder class of monomial weights(see for example [12, 62]).

In Chapter 3 we discuss the local properties of L_a -harmonic functions and their nodal set near the characteristic manifold Σ . In particular, using some Almgren and Weiss type monotonicity formulas, we classify the possible blow-up limit and we prove the uniqueness of a nondegenerate tangent map at every point of the nodal set.

The main feature of this class of degenerate-singular equations is that any L_a -harmonic function can be decomposed with respect to the direction orthogonal to the characteristic manifold Σ , in the sense that given u an L_a -harmonic function in $H^{1,a}(B_1)$ there exist $u_e^a \in H^{1,a}(B_1), u_e^{2-a} \in$ $H^{1,2-a}(B_1)$ two unique functions symmetric with respect to Σ respectively L_a and L_{2-a} harmonic in B_1 and locally smooths, such that

$$u(X) = u_e^a(X) + u_e^{2-a}(X)y |y|^{-a}$$
 in B_1 .

Therefore, local properties of the solutions, as their exponent of optimal regularity, their Taylor expansion near nodal set and the structure of the nodal set itself, are fulled comprehended by knowing the local behaviour of their even (symmetric with respect to orthogonal direction of Σ)

and odd (antisymmetric with respect to orthogonal direction of Σ) parts.

This specific "conduct" of the solutions is due to the presence of the characteristic manifold Σ , where either the vanishing or the blowing up of the weights imposes a quantization of the possible ways in which the nodal set can diffuse across Σ .

With the previous decomposition in mind, we restrict our blow-up analysis to the symmetric case and finally we introduce the new notion of "tangent field" Φ^{X_0} of u at a nodal point $X_0 \in \Sigma$, which takes care of the different behaviour of both the symmetric and antisymmetric part of u. Moreover, we introduce the regular $\mathcal{R}(u)$ and the singular part $\mathcal{S}(u)$ as

$$\mathcal{R}(u) = \left\{ X \in \Gamma(u) \colon \left| \nabla_x u(X) \right|^2 + \left| \partial_y^a u(X) \right|^2 \neq 0 \right\}, \quad \mathcal{S}(u) = \Gamma(u) \setminus \mathcal{R}(u)$$

and we developed a blow-up analysis in order to fully understand the structure in \mathbb{R}^{n+1} and its restriction on Σ .

Since our approach seems to be quite flexible, in the last part of Chapter 3 we present an applications of our theory in the context of nonlocal elliptic equations. In particular, inspired by [23, 27, 80], we exploit the local realisation of the fractional Laplacian, and more generally of fractional power of divergence form operator L with Lipschitz leading coefficient, in order to study the structure and the regularity of the nodal set of $(-L)^s$ -harmonic functions, for $s \in (0, 1)$. Moreover, this last Section allows to extend our analysis to fractional powers $(-\Delta_M)^s$ of the Laplace-Beltrami operator on a Riemannian manifold M, also for the case of Lipschitz metric, and moreover to conformal fractional Laplacian on conformally compact Einstein manifolds and asymptotically hyperbolic manifold, thanks to the extension technique developed in [27] and the asymptotic expansion of their geodesic boundary defining function.

As suggested in [86], we would like to stress that our analysis on the nodal set of *s*-harmonic function allows to fully understand the limiting profile arising from the case of Lotka-Volterra competition, showing a different behaviour with respect to the one presented in Chapter 1.

Finally, our results show some purely nonlocal feature on the possible local expansion of *s*-harmonic map near their zero set and on the structure of the nodal set itself. On one side we prove that first term of the Taylor expansion of an $(-L)^s$ -harmonic function is either an homogeneous harmonic polynomial or any possible homogeneous polynomial. In particular, we exhibit the stratification of the singular set S(u), showing the existence of an unexpected stratum $S^s(u)$ contained in a (n-1)-dimensional C^1 manifolds, in deep contrast with the local case. In the end, we prove what could be seen as the nonlocal counterpart of a conjecture that Lin proposed in [66]. Following his strategy, we give an explicit estimate on the (n-1)-Hausdorff measure of the nodal set $\Gamma(u)$ in terms of the Almgren monotonicity formula previously introduced.

This Chapter is part of a bigger project in collaboration with Yannick Sire, Susanna Terracini and Stefano Vita.

Notations and general results. Throughout the manuscript, we will consider $X = (x, y) \in \mathbb{R}^{n+1}$, with $x \in \mathbb{R}^n$ and $y \in \mathbb{R}$. With this notation in mind, we define the subspace $\mathbb{R}^{n+1}_+ = \mathbb{R}^{n+1} \cap \{y > 0\}$ with $\partial \mathbb{R}^{n+1}_+ = \mathbb{R}^N \times \{0\}$. Now, given $D \subset \mathbb{R}^{n+1}_+$ we write



In the picture, we use this notation with $D = B_r(x_0, 0)$ for the (n + 1)-dimensional ball centered in $X_0 = (x_0, 0) \in \mathbb{R}^n \times \{0\}$.

For any vector valued function $\mathbf{u} = (u_1, \ldots, u_h) \in \mathbb{R}^h$, we define $\mathbf{u}^2 = \sum_{i=1}^h u_i^2$, $\nabla \mathbf{u} := (\nabla u_1, \ldots, \nabla u_h)$ and $\partial_{\nu} \mathbf{u} := (\partial_{\nu} u_1, \ldots, \partial_{\nu} u_h)$, for every $\nu \in \mathbb{R}^{n+1}$. In particular, through Chapter 1 we will deeply use the notation $\langle \cdot, \cdot \rangle$ for the scalar product in \mathbb{R}^h .

Through the paper, for $a \in (-1, 1)$ we will always consider the weighted Sobolev spaces $H^{1,a}(B_1)$ deeply studied in [44, 43, 70] as the closure of $C^{\infty}(\overline{B_1})$ with respect the norm

$$||u||_{H^{1,a}(B_1)}^2 = \int_{B_1} |y|^a \, u^2 \mathrm{d}X + \int_{B_1} |y|^a \, |\nabla u|^2 \, \mathrm{d}X.$$

In this setting, we will always denote with $L_a = \operatorname{div}(|y|^a \nabla)$ the divergence form operator associated to the weight $\omega(y) = |y|^a$. More precisely, following the idea in the mentioned [23], for every $u \in H^s(\mathbb{R}^n)$, we consider $v \in H^{1,a}(\mathbb{R}^{n+1}_+)$ satisfying

$$\begin{cases} \operatorname{div}(|y|^a \, \nabla v) = 0 & \text{in } \mathbb{R}^{n+1}_+, \\ v(x,0) = u(x) & \text{in } \mathbb{R}^n , \end{cases}$$

with $a = 1 - 2s \in (-1, 1)$. In this setting, the nonlocal operator $(-\Delta)^s$ translates into

$$(-\Delta)^s \colon H^s(\mathbb{R}^n) \to H^{-s}(\mathbb{R}^n), \quad u \longmapsto -\frac{C(n,s)}{\gamma(n,s)} \lim_{y \to 0^+} y^{1-2s} \partial_y v(x,y).$$

Such an extension exists unique for a suitable class of functions u, and it is given by the formula

$$v(x,y) = \gamma(n,s) \int_{\mathbb{R}^n} \frac{y^{2s} u(\eta)}{(|x-\eta|^2 + y^2)^{n/2+s}} \mathrm{d}\eta \quad \text{where } \gamma(n,s)^{-1} := \int_{\mathbb{R}^n} \frac{1}{(|\eta|^2 + 1)^{n/2+s}} \mathrm{d}\eta \;.$$

In the introduction of Chapter 2 and in Chapter 3, we will give more details on this content.

xiv CONTENTS

NODAL SET OF SEGREGATED CRITICAL CONFIGURATIONS

1.1 INTRODUCTION

Several physical phenomena can be described by a certain number of densities or populations distributed in a domain and subject to laws of diffusion, reaction, and competitive interaction, starting from biological models for competing species in population dynamics [75] to the phase-segregation phenomenon in Bose-Einstein condensation and theoretical physics [54, 61].

In the recent years has been given much attention on the fractional Laplacians, since such operators are of real interest both in population dynamics (see [59]) and in relativistic quantum electrodynamics (see [64, 65]). Inspired by this physical motivations, we plan to extend the known results on the nodal set of segregated configuration to their nonlocal counterpart.

Hence, as pointed out in [83, 84], exploiting the local realization of the fractional Laplacian $(-\Delta)^s$ as a Dirichlet-to-Neumann map (see for instance [23]), several asymptotic results can be proved in the context of competition-diffusion problems with internal reactions, fractional diffusion and strong variational competition.

Theorem 1.1.1 ([84]). Let $\beta > 0$, $(f_{i,\beta})_{\beta}$ be a collection of continuous functions uniformly bounded with respect to β on bounded sets and let $(\mathbf{u}_{\beta})_{\beta} \in H^{1,a}(B_1^+; \mathbb{R}^h)$ be a family of solutions $\mathbf{u}_{\beta} = (u_{1,\beta}, \ldots, u_{h,\beta})$ of the problems

$$\begin{cases} -L_a u_{i,\beta} = 0 & \text{in } B_1^+ \\ -\partial_y^a u_{i,\beta} = f_{i,\beta}(u_{i,\beta}) - \beta u_{i,\beta} \sum_{j \neq i} a_{ij} u_{j,\beta}^2 & \text{on } \partial^0 B_1^+. \end{cases}$$
(3)

Let us assume that

$$\left\|\mathbf{u}_{\beta}\right\|_{L^{\infty}(B_{1}^{+})} \leq M$$

for some constant M > 0 independent on β . Then, there exists $\alpha = \alpha(n, s) > 0$, non depending on β , such that for $\alpha \in (0, \alpha^*)$

$$\|\mathbf{u}_{\beta}\|_{C^{0,\alpha}(\overline{B^+_{1/2}})} \le C_{1/2}$$

with $C = C(M, \alpha)$. Moreover, $(\mathbf{u}_{\beta})_{\beta}$ is relatively compact in $H^{1,a}(B^+_{1/2}) \cap C^{0,\alpha}(\overline{B^+_{1/2}})$, for $\alpha \in (0, \alpha^*)$.

The above result allows to prove its natural global counterpart, either on the whole of \mathbb{R}^n or on domains with suitable boundary conditions.

Theorem 1.1.2 ([84]). Let $\beta > 0$, $(f_{i,\beta})_{\beta}$ be a collection of continuous functions uniformly bounded with respect to β on bounded sets and let $(\mathbf{u}_{\beta})_{\beta} \in H^{s}(\mathbb{R}^{n};\mathbb{R}^{h})$ be a family of solutions $\mathbf{u}_{\beta} = (u_{1,\beta}, \ldots, u_{h,\beta})$ of the problems

$$\begin{cases} (-\Delta)^{s} u_{i,\beta} = f_{i,\beta}(u_{i,\beta}) - \beta u_{i,\beta} \sum_{j \neq i} a_{ij} u_{j,\beta}^{2} & \text{in } \Omega \\ u_{i,\beta} \equiv 0 & \text{in } \mathbb{R}^{n} \setminus \Omega, \end{cases}$$
(4)

where Ω is a bounded domain of \mathbb{R}^n with smooth boundary. Let us assume that

$$\|\mathbf{u}_{\beta}\|_{L^{\infty}(\Omega)} \leq M$$

for some constant M > 0 independent on β . Then, there exists $\alpha^* = \alpha^*(n, s) > 0$, non depending on β , such that for $\alpha \in (0, \alpha^*)$

$$\|\mathbf{u}_{\beta}\|_{C^{0,\alpha}(\mathbb{R}^n)} \le C,$$

with $C = C(M, \alpha)$.

As a byproduct of these results, can be proved that, up to subsequences, we have convergence of the above solutions to a limiting profile, which components are segregated on the boundary $\partial^0 B_1^+$. Actually, if furthermore $f_{i,\beta} \to f_i$, uniformly on compact sets, we can prove that this $\mathbf{u} = (u_1, \ldots, u_h)$ limiting configuration satisfies

$$\begin{cases} -L_a u_i = 0 & \text{in } B_1^+ \\ u_i \left(\partial_y^a u_i + f_i(u_i) \right) = 0 & \text{on } \partial^0 B_1^+ \\ u_i \cdot u_j = 0 & \text{on } \partial^0 B_1^+, \text{ for every } i \neq j \end{cases}$$

Since in the singular limit one finds a vector $\mathbf{u} \in H^{1,a}(B_1^+)$ of limiting profiles with mutually disjoint supports, it is a natural question to understand the regularity and the structure of the nodal set

$$\Gamma(\mathbf{u}) = \{ X \in \partial^0 B_1^+ \colon \mathbf{u}(X) = \mathbf{0} \},\$$

where all the components of u takes zero value. Hence, we focus our attention on the following class of vector valued configurations with segregated supports. At this point we postpone the discussion on the order $\alpha^* \in (0, 1)$ of Hölder regularity of the segregated configurations for several reasons that will be show later.

Definition 1.1.3. Let $s \in (0,1)$, $a = 1 - 2s \in (-1,1)$ and $\alpha^* \in (0,1)$, we define the class $\mathcal{G}^s(B_1^+)$ as the set of vector valued function $\mathbf{u} = (u_1, \ldots, u_h) \in H^{1,a}_{\text{loc}}(\overline{B_1^+}; \mathbb{R}^h)$ whose components are all non negative, continuous functions such that

- (1) $\mathbf{u} \in H^{1,a}(K \cap B_1^+) \cap C^{0,\alpha}(\overline{K \cap B_1^+})$, for every compact set $K \subset B$ and every $\alpha \in (0, \alpha^*)$;
- (2) $u_i \cdot u_j|_{y=0} \equiv 0$ for every $i \neq j$ and $\mathbf{u} \neq \mathbf{0}$ on $B_1 \cap \Sigma$. Moreover, for $i = 1, \ldots, h$ it satisfies

$$\begin{cases} -L_a u_i = 0 & \text{in } B_1^+ \\ u_i \left(\partial_y^a u_i + f_i(u_i) \right) = 0 & \text{on } \partial^0 B_1^+ \end{cases}$$
(5)

where $f_i \colon \mathbb{R}^+ \to \mathbb{R}$ are nonnegative \mathcal{C}^1 functions such that $f_i(s) = O(s)$ for $s \to 0$;

(3) for every $X_0 = (x_0, 0) \in \partial^0 B_1^+$ and $r \in (0, \text{dist}(X_0, \partial B))$, the following Pohožaev type identity holds

$$(1-a-n)\int_{B_{r}^{+}(X_{0})}|y|^{a}|\nabla\mathbf{u}|^{2} dX + r\int_{\partial B_{r}^{+}(X_{0})}|y|^{a}|\nabla\mathbf{u}|^{2} d\sigma + 2n\int_{\partial^{0}B_{r}^{+}(X_{0})}\sum_{i=1}^{h}F_{i}(u_{i})dx - 2r\int_{S_{r}^{n-1}(X_{0})}\sum_{i=1}^{h}F_{i}(u_{i})dx = 2r\int_{\partial^{+}B_{r}^{+}(X_{0})}|y|^{a}(\partial_{r}\mathbf{u})^{2}d\sigma$$
(6)

where
$$\mathbf{F}(s) = (F_1(s), \dots, F_h(s))$$
 with $F_i(s) = \int_0^s f_i(t) dt$ for every $i = 1, \dots, h$.

First, in [83] the authors proved for the case s = 1/2, i.e. a = 0, that for solutions $\mathbf{u} \in \mathcal{G}^{1/2}(B^+)$ the highest possible regularity correspond to the Hölder exponent $\alpha^* = 1/2$. This result is based on a blow-up analysis based on an Almgren type monotonicity formula and an optimal Liouville type theorem for segregated configuration.

Instead, for the general case $s \in (0, 1)$ in [84] the authors proved, with a combination of a blow-up analysis and a Liouville type theorem based on the validity of an Alt-Caffarelli-Friedman type monotonicity formula, that the highest possible regularity of the limiting profile correspond to the Hölder exponent $\alpha^* = \alpha^*(n, s)$ such that

$$\alpha^* = \begin{cases} \nu_s^{ACF}, & 0 < s \le \frac{1}{2}, \\ \min\{\nu_s^{ACF}, 2s - 1\}, & \frac{1}{2} < s < 1, \end{cases}$$

where ν_s^{ACF} corresponds to the exponent associated to the Alt-Caffarelli-Friedman formula (see [83, 84]). The threshold s = 1/2 is due to the presence of the phenomenon of self-segregation of nonlocal problem where $s \in (1/2, 1)$, which consists in the existence of a ball $\tilde{B}^+ \subset \mathbb{R}^{n+1}_+$ centered on the nodal set $\Gamma(\mathbf{u})$ and an index $i = 1, \ldots, h$ such that all the components u_i of \mathbf{u}

with $j \neq i$ are identically zero on the ball make exception of u_i which is not identically zero and such that

$$\partial^0 \tilde{B}^+ \setminus \Gamma(\mathbf{u}) = \{ X \in \partial^0 \tilde{B}^+ \colon u_i(X) > 0 \}$$

More precisely, following the idea of the optimal Liouville exponent in [83], in Section 1.6 we easily improve the previous results finding the following bound

$$\alpha^* = \begin{cases} s, & 0 < s \le \frac{1}{2}, \\ 2s - 1, & \frac{1}{2} < s < 1. \end{cases}$$

In particular, this improvement emphasizes the deep relation between the different α -Hölder regularity of the solutions near the nodal set and the structure of the nodal set itself. Inspired by this connection, we decompose the nodal set into its "segregated" and "self-segregated" part. Finally, under the previous notations, we have

$$\begin{array}{ll} \text{Segregated nodal set} \\ \Gamma(\mathbf{u}) = \partial^0 B_1^+ \setminus \bigcup_{i=1}^h \operatorname{int}\left(\overline{\{u_i > 0\}}\right) \\ \alpha \text{ - Hölder continuous, for } \alpha \in (0,s) \end{array} \qquad \begin{array}{ll} \text{Self-Segregated nodal set} \\ \Gamma(\mathbf{u}) = \bigcup_{i=1}^h \partial\{u_i > 0\} \setminus \operatorname{int}\left(\overline{\{u_i > 0\}}\right) \\ \alpha \text{ - Hölder continuous, for } \alpha \in (0,s) \end{array}$$

Since our main result does not concern the self-segregated portion of the nodal set, we remark that through this Chapter we will always consider the class of segregated profiles $\mathcal{G}^s(B^+)$ locally α -Hölder continuous for every $\alpha \in (0, s)$. Just in Section , we will give more details in the context of self-segregation.

Our approach is deeply based on the validity of an Almgren's type monotonicity formula and on a blow-up analysis of the critical configurations. More precisely, for every $X_0 \in \partial^0 B_1^+$ and r > 0 such that $B_r^+(X_0) \subset B_1^+$, we define the functionals

$$E(X_0, \mathbf{u}, r) = \frac{1}{r^{n-1+a}} \left(\int_{B_r^+(X_0)} |y|^a |\nabla \mathbf{u}|^2 \, \mathrm{d}X - \int_{\partial^0 B_r^+} \langle \mathbf{u}, F(x, \mathbf{u}) \rangle \mathrm{d}x \right)$$
$$H(X_0, \mathbf{u}, r) = \frac{1}{r^{n+a}} \int_{\partial^+ B_r^+(X_0)} |y|^a \, \mathbf{u}^2 \mathrm{d}\sigma,$$

and, the Almgren's monotonicity formula as

$$N(x_0, \mathbf{u}, r) = \frac{E(x_0, \mathbf{u}, r)}{H(x_0, \mathbf{u}, r)}.$$

Unfortunately, as we anticipate in the Introduction of the manuscript, in order to overcame some technical problem due to problem of codimension between the free boundary $\Gamma(\mathbf{u})$ and the space where the degenerate-singular equation is satisfied, we restrict our attention on the planar case n = 2. In this case, we are able to prove

Proposition 1.1.4. Given $s \in (0, 1)$, n = 2 and $u \in \mathcal{G}^s(B_1^+)$, then for $X_0 \in \Gamma(\mathbf{u})$ either

$$N(X_0, \mathbf{u}, 0^+) = s \quad or \quad N(X_0, \mathbf{u}, 0^+) \ge s + \delta,$$

for some universal constant $\delta > 0$. Moreover, the possible values of the Almgren frequency formula $N(X_0, \mathbf{u}, 0^+)$ are a discrete set in [s, 2s) with 2s as point of accumulation.

This result, combined with the convergence of the blow-up sequence both with respect to the strong topologies in $H_{\text{loc}}^{1,a}$, $C_{\text{loc}}^{0,\alpha}$ and to the Hausdorff distance $d_{\mathcal{H}}$, allows to prove the main result of this Chapter.

Theorem 1.1.5. Let $s \in (0,1)$, n = 2 and $\mathbf{u} \in \mathcal{G}^s(B^+)$. Then the nodal set $\Gamma(\mathbf{u})$ splits into its regular and singular part defined by

$$\mathcal{R}(\mathbf{u}) = \{X_0 \in \Gamma(\mathbf{u}) \colon N(X_0, \mathbf{u}, 0^+) = s\},\$$

$$\mathcal{S}(\mathbf{u}) = \{X_0 \in \Gamma(\mathbf{u}) \colon N(X_0, \mathbf{u}, 0^+) > s\},\$$

where $S(\mathbf{u})$ is a locally finite collection of points and $\mathcal{R}(\mathbf{u})$ a set relatively open in $\Gamma(\mathbf{u})$ which satisfies a vanishing Reifenberg flatness condition.

This Chapter is organized as follows. In Section 1.2 we prove that elements in $\mathcal{G}^s(B^+)$ satisfy an Almgren's type monotonicity formula; by exploiting this fact, in Section 1.3 we prove convergence of blow-up sequences as well as some closure properties of the class $\mathcal{G}^s(B_+)$. In Section 1.4 we use the Federer's Reduction Principle in order to prove some Hausdorff estimates for the nodal sets and we introduce the notion of regular and singular set. Moreover, in Section 1.5 we prove that the regular part of the nodal set satisfies a vanishing Reifenberg condition. Finally in Section 1.6 we present some useful remark and the relation between the class $\mathcal{G}^s(B_1^+)$ and the singular limit of competition-diffusion problem with fractional diffusion and variational competition.

Through this Chapter we will substitute the assumption on the dimension n = 2 only in the "clean-up" type result, in order to stress which results hold for every dimensions.

1.2 ALMGREN'S TYPE MONOTONICITY FORMULA

The functions belonging to $\mathcal{G}^s(B_1^+)$ have a very rich structure, mainly thanks to the validity of the Pohožaev identities we are able to prove the validity of the Almgren's monotonicity formula. The most challenging feature of this Section is that the segregation occurs only in the *n*-dimensional space, which it implies that, when dealing with Pohožaev type identities, integrals on the "boundary of the boundary" appear.

Let us recall the definition of the class $\mathcal{G}^{s}(B_{1}^{+})$ that we will use through the paper.

Definition 1.2.1. Let $a \in (-1, 1)$, we define the class $\mathcal{G}^{s}(B_{1}^{+})$ as the set of vector valued function $\mathbf{u} = (u_{1}, \ldots, u_{h}) \in H^{1,a}_{\text{loc}}(\overline{B_{1}^{+}}; \mathbb{R}^{h})$ whose components are all non negative, continuous functions such that

- (1) $\mathbf{u} \in H^{1,a}(K \cap B_1^+) \cap C^{0,\alpha}(\overline{K \cap B_1^+})$, for every compact set $K \subset B$ and every $\alpha \in (0,s)$;
- (2) $u_i \cdot u_j|_{y=0} \equiv 0$ for every $i \neq j$, $\mathbf{u} \neq \mathbf{0}$ on Σ and for every $i = 1, \dots, h$ it satisfies

$$\begin{cases} -L_a u_i = 0 & \text{in } B_1^+ \\ u_i \left(\partial_y^a u_i + f_i(u_i) \right) = 0 & \text{on } \partial^0 B_1^+ \end{cases}$$
(7)

where $f_i \colon \mathbb{R}^+ \to \mathbb{R}$ are nonnegative \mathcal{C}^1 functions such that $f_i(s) = O(s)$ for $s \to 0$;

(3) for every $X_0 = (x_0, 0) \in \partial^0 B_1^+$ and $r \in (0, \text{dist}(X_0, \partial B))$, the following Pohožaev type identity holds

$$(1-a-n)\int_{B_{r}^{+}(X_{0})}|y|^{a}|\nabla\mathbf{u}|^{2}\,\mathrm{d}X+r\int_{\partial B_{r}^{+}(X_{0})}|y|^{a}|\nabla\mathbf{u}|^{2}\,\mathrm{d}\sigma+$$

$$+2n\int_{\partial^{0}B_{r}^{+}(X_{0})}\sum_{i=1}^{h}F_{i}(u_{i})\mathrm{d}x-2r\int_{S_{r}^{n-1}(X_{0})}\sum_{i=1}^{h}F_{i}(u_{i})\mathrm{d}x=2r\int_{\partial^{+}B_{r}^{+}(X_{0})}|y|^{a}(\partial_{r}\mathbf{u})^{2}\mathrm{d}\sigma$$
(8)

where $\mathbf{F}(s) = (F_1(s), \dots, F_h(s))$ with $F_i(s) = \int_0^s f_i(t) dt$ for every $i = 1, \dots, h$.

Now, for every $X_0 \in \partial^0 B_1^+$ and r > 0 such that $B_r^+(X_0) \subset B_1^+$, we define the functionals

$$E(X_0, \mathbf{u}, r) = \frac{1}{r^{n-1+a}} \left(\int_{B_r^+(X_0)} |y|^a |\nabla \mathbf{u}|^2 \, \mathrm{d}X - \int_{\partial^0 B_r^+} \langle \mathbf{u}, F(x, \mathbf{u}) \rangle \mathrm{d}x \right)$$
$$H(X_0, \mathbf{u}, r) = \frac{1}{r^{n+a}} \int_{\partial^+ B_r^+(X_0)} |y|^a \, \mathbf{u}^2 \mathrm{d}\sigma$$

and, whenever the average $H(x_0, \mathbf{u}, r) \neq 0$, the Almgren's frequency formula by

$$N(x_0, \mathbf{u}, r) = \frac{E(x_0, \mathbf{u}, r)}{H(x_0, \mathbf{u}, r)}.$$

Since $\mathbf{u} \in H^{1,a}(B_1; \mathbb{R}^h)$ both $r \mapsto E(X_0, \mathbf{u}, r)$ and $r \mapsto H(X_0, \mathbf{u}, r)$ are locally absolutely continuous functions for $r \in (0, \text{dist}(X_0, \partial B_1^+))$. As usual, integrating by parts on $B_r^+(X_0)$ every component u_i and summing over $i = 1, \ldots, h$ we get

$$E(X_0, \mathbf{u}, r) = \frac{1}{r^{n-1+a}} \int_{\partial^+ B_r^+} |y|^a \langle \mathbf{u}, \partial_r \mathbf{u} \rangle \mathrm{d}\sigma = \frac{r}{2} \frac{d}{dr} H(X_0, \mathbf{u}, r).$$
(9)

The presence of internal reaction terms in the definition of the energy $E(X_0, \mathbf{u}, r)$ has to be dealt with. For this reason, we introduce the following lemmata to provide a crucial estimate in order to bound the Almgren quotient.

Let $a \in (-1,1)$ and $u \in H^{1,a}(B_r^+(X_0))$ for some $X_0 \in \partial^0 B_1^+$ and $r \in (0, \operatorname{dist}(X_0, \partial B_1^+))$. Then, for every $p \in [2, p^{\#}]$, where $p^{\#} = 2n/(n-2s)$ there exists a constant C(n, p, s) such that

$$\left[\frac{1}{r^n}\int_{\partial^0 B_r^+(X_0)} |u|^p \,\mathrm{d}x\right]^{\frac{2}{p}} \le C(n,p,s) \left[\frac{1}{r^{n-1+a}}\int_{B_r^+(X_0)} |y|^a \,|\nabla u|^2 \,\mathrm{d}X + \frac{1}{r^{n+a}}\int_{\partial^+ B_r^+(X_0)} |y|^a \,u^2 \mathrm{d}X\right] \tag{10}$$

This result is a direct consequence of the characterization of the class of trace of $H^{1,a}(B_1^+)$ in [70] and the critical Sobolev exponent for the trace embedding in the context of fractional Sobolev-Slobodeckij spaces $W^{s,2}(K)$, with $s \in (0,1)$ and $K \subset \mathbb{R}^n$.

Lemma 1.2.2. Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$. Then, for every $p \in [2, p^{\#}]$ and $X_0 \in \partial^0 B_1^+$ there exist constants $C > 0, \overline{r} > 0$ such that

$$\left[\frac{1}{r^n}\int_{\partial^0 B_r^+(X_0)} |\mathbf{u}|^p \,\mathrm{d}x\right]^{\frac{2}{p}} \le C\left(E(X_0,\mathbf{u},r) + H(X_0,\mathbf{u},r)\right),$$

for every $r \in (0, \overline{r})$.

Proof. Since $\mathbf{u} \in L^{\infty}(B_1^+)$, and each components of $\mathbf{F} = (f_1, \ldots, f_h)$ is locally Lipschitz continuous with $f_i(0) = 0$, we obtain

$$\begin{aligned} \left| \frac{1}{r^{n-1+a}} \int_{\partial^0 B_r^+} \langle \mathbf{u}, \mathbf{F}(\mathbf{u}) \rangle \mathrm{d}x \right| &\leq \frac{C}{r^{n-1+a}} \int_{\partial^0 B_r^+} \mathbf{u}^2 \mathrm{d}x \\ &\leq C_2 r^{1-a} \left[\frac{1}{r^{n-1+a}} \int_{B_r^+(X_0)} |y|^a \, |\nabla \mathbf{u}|^2 \, \mathrm{d}X + \frac{1}{r^{n+a}} \int_{\partial^+ B_r^+(X_0)} |y|^a \, \mathbf{u}^2 \mathrm{d}X \right], \end{aligned}$$

where we used the trace inequality in the case p = 2. Finally, since $a \in (-1, 1)$ we get

$$E(X_0, \mathbf{u}, r) + H(X_0, \mathbf{u}, r) \ge (1 - C_2 r^{1-a}) \left[\frac{1}{r^{n-1+a}} \int_{B_r^+(X_0)} |y|^a |\nabla \mathbf{u}|^2 \, \mathrm{d}X + \frac{1}{r^{n+a}} \int_{\partial^+ B_r^+(X_0)} |y|^a \, \mathbf{u}^2 \mathrm{d}X \right]$$
(11)

the result follows by taking into account the trace inequality and choosing $\overline{r}>0$ sufficiently small. $\hfill \Box$

Following the same idea in [83] for the case s = 1/2, let introduce for $p \in (2, p^{\#}]$ the auxiliary function

$$\psi(X_0, \mathbf{u}, r) = \left(\frac{1}{r^n} \int_{\partial^0 B_r^+(X_0)} |\mathbf{u}|^2 \, \mathrm{d}X\right)^{1-\frac{2}{p}}$$

which is bounded for $r \in (0, \text{dist}(X_0, \partial B_1^+))$. Under this notations, for $a \in (-1, 1)$ consider

$$\Psi(X_0, \mathbf{u}, r) = C(n, s) \int_0^r t^{-a} \left(1 + \frac{d}{dt} \left(t\psi(X_0, \mathbf{u}, t) \right) dt \right)$$

which is well defined on $r \in (0, \operatorname{dist}(X_0, \partial B_1^+))$ such that $\lim_{r \to 0^+} \Psi(X_0, \mathbf{u}, r) = 0$, since $\psi(X_0, \mathbf{u}, r)$ is bounded for r sufficiently small. In order to simplify the notations, through the Section we will just use the notation $\psi(r)$ and $\Psi(r)$ for the auxiliary functions previously defined.

Lemma 1.2.3. Let $s \in (0,1)$ and $u \in \mathcal{G}^s(B_1^+)$. Then, for every $p \in (2, p^{\#}]$ and $X_0 \in \partial^0 B_1^+$ there exist constants $C > 0, \overline{r} > 0$ such that

$$\frac{1}{r^{n-1}} \int_{S_r^{n-1}} |\mathbf{u}|^p \,\mathrm{d}\sigma \le C \left(E(X_0, \mathbf{u}, r) + H(X_0, \mathbf{u}, r) \right) \frac{d}{dr} \left(r\psi(r) \right),$$

for every $r \in (0, \overline{r})$.

Proof. The proof follows it is the same of [83, Lemma 9.5] make exception in our case is based on the generalized Poincarè inequality (10). Hence, a direct computation yields the identity

$$\frac{d}{dr}\left(r\psi(r)\right) = \psi(r)\left[r\left(1-\frac{2}{p}\right)\frac{\int_{S_r^{n-1}}|\mathbf{u}|^p\,\mathrm{d}\sigma}{\int_{\partial^0 B_r^+}|\mathbf{u}|^p\,\mathrm{d}\sigma} + \left(1-n\left(1-\frac{2}{p}\right)\right)\right],$$

and, since $p \le p^{\#}$ implies $n(1-2/p) \le 1$, we infer

$$\frac{d}{dr}\left(r\psi(r)\right) \ge r\psi(r)\left(1-\frac{2}{p}\right)\frac{\int_{S_r^{n-1}} |\mathbf{u}|^p \,\mathrm{d}\sigma}{\int_{\partial^0 B_r^+} |\mathbf{u}|^p \,\mathrm{d}\sigma}.$$

Finally, recalling the definition of ψ and using Lemma 1.2.2, we deduce

$$\left(E(X_0,\mathbf{u},r)+H(X_0,\mathbf{u},r)\right)\frac{d}{dr}\left(r\psi(r)\right)\geq C\frac{1}{r^{n-1}}\int_{S_r^{n-1}}|\mathbf{u}|^p\,\mathrm{d}\sigma.$$

We are now ready to prove the boundedness of the Almgren quotient, rather than its monotonicity, considering a modified version of the quotient.

Proposition 1.2.4. Given $s \in (0, 1)$, $u \in \mathcal{G}^s(B_1^+)$ and $\Omega^+ \subset \subset B_1^+$, there exist constants $C, \overline{r} > 0$ such that, for every $X_0 \in \partial^0 \Omega^+ \subset \partial^0 B_1^+$ and $r \in (0, \overline{r})$ such that $B_r^+(X_0) \subset B_1^+$, we have that $H(X_0, u, r) > 0$ and $N(X_0, u, r) > 0$ for every $r \in (0, \overline{r})$. Moreover, the map

$$r \mapsto e^{C\Psi(X_0,\mathbf{u},r)} \left(N(X_0,\mathbf{u},r) + 1 \right)$$

is monotone non decreasing on $(0, \overline{r})$, which ensures the existence of limit

$$N(X_0, \mathbf{u}, 0^+) = \lim_{r \to 0^+} N(X_0, \mathbf{u}, r),$$

which is finite and called the Almgren frequency of \mathbf{u} at X_0 .

Proof. Let $X_0 \in \Gamma(\mathbf{u})$ and $\overline{r} > 0$ be such that $\overline{r} < \text{dist}(\Omega^+, B_1^+)$ and Lemma 1.2.2 and Lemma 1.2.3 hold true. First, let us consider the following modified Almgren frequency formula

$$\widetilde{N}(X_0, \mathbf{u}, r) = \frac{E(X_0, \mathbf{u}, r)}{H(X_0, \mathbf{u}, r)} + 1 = N(X_0, \mathbf{u}, r) + 1.$$
(12)

Under this notations, we get by Lemma 1.2.2

$$E(X_0, \mathbf{u}, r) + H(X_0, \mathbf{u}, r) \ge 0 \longrightarrow \widetilde{N}(X_0, \mathbf{u}, r) = \frac{E(X_0, \mathbf{u}, r)}{H(X_0, \mathbf{u}, r)} + 1 \ge 0,$$

whenever $H(X_0, \mathbf{u}, r) \neq 0$. By continuity of $r \mapsto H(X_0, \mathbf{u}, r)$ we can consider a reasonable neighborhood of r where it does not vanish. Since $\mathbf{u} \in L^{\infty}(B_1^+)$, and each components of $\mathbf{F} = (f_1, \ldots, f_h)$ is locally Lipschitz continuous with $f_i(0) = 0$, there exists a positive constant C > 0 such that

$$|\langle \mathbf{u}, \mathbf{F}(x, \mathbf{u}) \rangle| \le C \mathbf{u}^2$$
 and $|\mathbf{F}(x, \mathbf{u})| \le C \mathbf{u}^2$,

for every i = 1, ..., h. Now, taking into account the Pohožaev identity (8), if we differentiate the map $r \mapsto E(X_0, \mathbf{u}, r)$ we obtain

$$\frac{d}{dr}E(X_0, \mathbf{u}, r) = -\frac{n-1+a}{r^{n+a}} \left(\int_{B_r^+} |y|^a |\nabla \mathbf{u}|^2 \, \mathrm{d}X - \int_{\partial^0 B_r^+} \langle \mathbf{u}, \mathbf{F}(\mathbf{u}) \rangle \mathrm{d}x \right) + \\ + \frac{1}{r^{n-1+a}} \int_{\partial^+ B_r^+} |y|^a |\nabla \mathbf{u}|^2 \, \mathrm{d}\sigma - \frac{1}{r^{n-1+a}} \int_{S_r^{N-1}} \langle \mathbf{u}, \mathbf{F}(\mathbf{u}) \rangle \mathrm{d}\sigma \\ = \frac{2}{r^{n-1+a}} \int_{\partial^+ B_r^+} |y|^a |\partial_r \mathbf{u}|^2 \, \mathrm{d}\sigma + R(x_0, \mathbf{u}, r),$$

where the remainder can be estimated as

$$\begin{aligned} |R(X_{0},\mathbf{u},r)| &\leq \frac{n-1+a}{r^{n+a}} \int_{\partial^{0}B_{r}^{+}(X_{0})} |\langle \mathbf{u},\mathbf{F}(\mathbf{u})\rangle| \,\mathrm{d}x + \frac{2n}{r^{n+a}} \int_{\partial^{0}B_{r}^{+}(X_{0})} \sum_{i=1}^{h} |F_{i}(u_{i})| \,\mathrm{d}x + \\ &+ \frac{2}{r^{n+a-1}} \int_{S_{r}^{n-1}(X_{0})} \sum_{i=1}^{h} |F_{i}(u_{i})| \,\mathrm{d}x + \frac{1}{r^{n+a-1}} \int_{S_{r}^{n-1}(X_{0})} |\langle \mathbf{u},\mathbf{F}(\mathbf{u})\rangle| \,\mathrm{d}\sigma \\ &\leq C(n,s) \left[\frac{1}{r^{n+a}} \int_{\partial^{0}B_{r}^{+}(X_{0})} \mathbf{u}^{2} \mathrm{d}x + \frac{1}{r^{n+a-1}} \int_{S_{r}^{n-1}(X_{0})} \mathbf{u}^{2} \mathrm{d}\sigma \right] \\ &\leq C(n,s)r^{-a} \left(E(X_{0},\mathbf{u},r) + H(X_{0},\mathbf{u},r) \right) \left(1 + \frac{d}{dr}(r\psi(r)) \right) \end{aligned}$$

where in the third inequality we used Lemma 1.2.2 and Lemma 1.2.3. Therefore, differentiating the Almgren quotient and using the Cauchy-Schwarz inequality on $\partial^+ B_r^+$, we obtain

$$\begin{split} \frac{d}{dr}\widetilde{N}(X_{0},\mathbf{u},r) &= \frac{\frac{d}{dr}E(X_{0},\mathbf{u},r) + \frac{d}{dr}H(X_{0},\mathbf{u},r)}{E(X_{0},\mathbf{u},r) + H(X_{0},\mathbf{u},r)} - \frac{\frac{d}{dr}H(X_{0},\mathbf{u},r)}{H(X_{0},\mathbf{u},r)} \\ &\geq \frac{2H(X_{0},\mathbf{u},r)}{r^{2n+2a-1}} \left[\int_{\partial^{+}B_{r}^{+}} |y|^{a} |\partial_{r}\mathbf{u}|^{2} \,\mathrm{d}\sigma \int_{\partial^{+}B_{r}^{+}} |y|^{a} \,\mathbf{u}^{2} \mathrm{d}\sigma - \left(\int_{\partial^{+}B_{r}^{+}} |y|^{a} \,\langle \mathbf{u},\partial_{r}\mathbf{u}\rangle \mathrm{d}\sigma \right)^{2} \right] + \\ &- C(n,s)\widetilde{N}(X_{0},\mathbf{u},r)r^{-a} \left(1 + \frac{d}{dr}(r\psi(r)) \right) \\ &\geq - C(n,s)\widetilde{N}(X_{0},\mathbf{u},r)r^{-a} \left(1 + \frac{d}{dr}(r\psi(r)) \right). \end{split}$$

which implies that the function

$$r \mapsto e^{C\Psi(X_0, \mathbf{u}, r)} \widetilde{N}(X_0, \mathbf{u}, r)$$

is nondecreasing as far as $H(X_0, \mathbf{u}, r) \neq 0$. Passing to the logarithmic derivative of $r \mapsto H(X_0, \mathbf{u}, r)$ we infer from (9) that for $r \in (r_1, r_2)$ we get

$$\frac{d}{dr}\log H(X_0, \mathbf{u}, r) = \frac{2}{r}N(X_0, \mathbf{u}, r).$$
(13)

More precisely, we can choose $r_1 = 0, r_2 = +\infty$. On one hand, the above equation provides that, if $\log H(X_0, \mathbf{u}, R) > -\infty$ then $\log H(X_0, \mathbf{u}, r) > -\infty$ for every r > R, so that $r_2 = \operatorname{dist}(X_0, \partial B_1^+)$. Now, on the other hand assume by contradiction that

$$r_1 = \inf \{r \colon H(X_0, \mathbf{u}, r) > 0 \text{ on } (r, r_2)\} > 0.$$

 \square

By the monotonicity result on the modified Almgren quotient (12), we have that

$$N(X_0, \mathbf{u}, r) < e^{C\Psi(2r_1)} \left(N(X_0, \mathbf{u}, 2r_1) + 1 \right) - 1,$$

for every $r_1 < r \le 2r_1$. Hence, integrating (13) between r and $2r_1$, we get

$$\frac{H(X_0, \mathbf{u}, 2r_1)}{H(X_0, \mathbf{u}, r)} \le \left(\frac{2r_1}{r}\right)^{2\left(e^{C\Psi(2r_1)}(N(X_0, \mathbf{u}, 2r_1)+1)-1\right)}$$

and, since $r \mapsto H(X_0, \mathbf{u}, r)$ is continuous we deduce the absurd $H(X_0, \mathbf{u}, r_1) > 0$.

The first Corollary of the Almgren monotonic y result is the following lower bound for the Almgren frequency formula of $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ at $X_0 \in \Gamma(\mathbf{u})$.

Corollary 1.2.5. Given $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$, for any $X_0 \in \Gamma(\mathbf{u})$ we have

$$N(X_0, \mathbf{u}, 0^+) \ge s.$$

Proof. Let $\overline{r} > 0$ be such that Proposition 1.2.4 holds true and suppose by contradiction the existence of $0 < \tilde{r} < \overline{r}$ and $\varepsilon > 0$ such that

$$e^{C\Psi(\widetilde{r})}(N(X_0,\mathbf{u},\widetilde{r})+1) \le 1+s-\varepsilon.$$

By the above bound, we obtain for every $r \in (0, \tilde{r})$ that

$$\frac{d}{dr}\log H(X_0, \mathbf{u}, r) \le 2\frac{e^{C\Psi(\tilde{r})}\left(N(X_0, \mathbf{u}, \tilde{r}) + 1\right) - 1}{r} \le \frac{2(s - \varepsilon)}{r}$$

Integrating this inequality between r and \tilde{r} yields

$$\frac{H(X_0,\mathbf{u},\widetilde{r})}{H(X_0,\mathbf{u},r)} \le \left(\frac{\widetilde{r}}{r}\right)^{2(s-\varepsilon)}$$

which, together with the fact that $\mathbf{u} \in C^{0,\alpha}_{\text{loc}}(B_1^+)$ for every $\alpha \in (0,s)$ and that $\mathbf{u}(X_0) = \mathbf{0}$, implies

$$Cr^{2(s-\varepsilon)} \le H(X_0, \mathbf{u}, r) \le Cr^{2\alpha},$$

for every $\alpha \in (0, s)$. Hence, the contradiction follows for r sufficiently small.

Corollary 1.2.6. For every $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ the map from $X_0 \mapsto N(X_0, \mathbf{u}, 0^+)$ is upper semi-continuous on $\partial^0 B_1^+$.

Proof. Fixed a vector valued function $\mathbf{u} \in \mathcal{G}^s(B_1^+)$, let us take a sequence $X_k \to X$ in $\partial^0 B_1^+$. By Proposition 1.2.4 there exists a constant C > 0 and $\overline{r} > 0$ such that, for $r \in (0, \overline{r})$

$$N(X_k, \mathbf{u}, r) = e^{-C\Psi(r)} e^{C\Psi(r)} \left(N(X_k, \mathbf{u}, r) + 1 \right) - 1 \ge e^{-C\Psi(r)} \left(N(X_k, \mathbf{u}, 0^+) + 1 \right) - 1.$$

By taking the limit superior in k and afterwards the limit as $r \to 0^+$ we obtain

$$N(X, \mathbf{u}, 0^+) \ge \limsup_{k \to \infty} N(X_k, \mathbf{u}, 0^+).$$

Another simple consequence of the monotonicity result is the following comparison property which, with $r_2 = 2r_1$ is the so called doubling property.

Proposition 1.2.7. Given $s \in (0,1)$, $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ and $\Omega^+ \subset B_1^+$, there exists $\overline{C} > 0$ and $\overline{r} > 0$ such that $(r_0)^{2\overline{C}}$

$$H(X_0, \mathbf{u}, r_2) \le H(X_0, \mathbf{u}, r_1) \left(\frac{r_2}{r_1}\right)^{2\epsilon}$$

for every $X_0 \in \partial^0 \Omega^+$ and $0 < r_1 < r_2 \leq \overline{r}$.

Proof. Fixed $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ and $\Omega^+ \subset B_1^+$, let C > 0 and $\overline{r} > 0$ be such that Proposition 1.2.4 holds true for every $X_0 \in \partial^0 \Omega^+$. Hence, given

$$\overline{C} = \sup_{X_0 \in \partial^0 \Omega^+} N(X_0, \mathbf{u}, \overline{r}) < +\infty,$$
(14)

we get

$$\begin{aligned} \frac{d}{dr} \log H(X_0, \mathbf{u}, r) &= \frac{2}{r} N(X_0, \mathbf{u}, r) \\ &\leq \frac{2}{r} \left(e^{-C\Psi(\overline{r})} e^{C\Psi(\overline{r})} \left(N(X_0, \mathbf{u}, \overline{r}) + 1 \right) - 1 \right) \\ &\leq \frac{2}{r} \left((\overline{C} + 1) e^{-C\Psi(\overline{r})} - 1 \right), \end{aligned}$$

for every $0 < r < \overline{r}$. Now, by integrating the previous inequality between r_1 and r_2 , for $0 < r_1 < r_2 \le \overline{r}$, we obtain

$$\frac{H(X_0, \mathbf{u}, r_2)}{H(X_0, \mathbf{u}, r_1)} \le \left(\frac{r_2}{r_1}\right)^{2C}$$

as we previously claimed.

1.3 COMPACTNESS OF THE BLOW-UP SEQUENCES

All techniques presented in this Chapter involve a local analysis of the nodal set of the solution, which will be performed via a blow-up and blow-down procedure. In this Section we study the behaviour of the class $\mathcal{G}^s(B_1^+)$ under rescaling and translations with respect to point on Σ , in order to apply the blow-up analysis of $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ near the nodal set $\Gamma(\mathbf{u})$.

Given $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ let $\mathbf{F} = (F_1, \dots, F_h)$ be the vector valued associated to Definition 1.2.1. For every $\rho, t > 0$ and $X_0 = (x_0, 0) \in \partial^0 B_1^+$ we define the rescaled function

$$\mathbf{v}(X) = \frac{\mathbf{u}(X_0 + tX)}{\rho}, \quad \text{for } X \in B_{X_0,t}^+ := \frac{B_1^+ - X_0}{t}, \tag{15}$$

where obviously the previous equality holds for every component of the vector valued function. It is easy to check that each components of \mathbf{v} solves the system

$$\begin{cases} -L_a v_i = 0 & \text{in } B^+_{X_0,t} \\ v_i \left(\partial^a_y v_i + g_i(v_i) \right) = 0 & \text{on } \partial^0 B^+_{X_0,t} \end{cases}$$
(16)

where

$$g_i(s) = \frac{t^{1-a}}{\rho} f_i(\rho s).$$
 (17)

In this setting, if we define for any $Z_0 \in B^+_{X_0,t}$ and $r \in (0, \text{dist}(Z_0, \partial B^+_{X_0,t}))$

$$E(Z_0, \mathbf{v}, r) = \frac{1}{r^{n-1+a}} \left(\int_{B_r^+(X_0)} |y|^a |\nabla \mathbf{v}|^2 \, \mathrm{d}X - \int_{\partial^0 B_r^+} \langle \mathbf{v}, \mathbf{G}(\mathbf{v}) \rangle \mathrm{d}x \right)$$
$$H(Z_0, \mathbf{v}, r) = \frac{1}{r^{n+a}} \int_{\partial^+ B_r^+(X_0)} |y|^a \, \mathbf{v}^2 \mathrm{d}\sigma$$

and the following identities hold

$$E(Z_0, \mathbf{v}, r) = \frac{1}{\rho^2} E(X_0 + tZ_0, \mathbf{u}, tr) \text{ and } H(Z_0, \mathbf{v}, r) = \frac{1}{\rho^2} H(X_0 + tZ_0, \mathbf{u}, tr)$$
(18)

and hence $N(Z_0, \mathbf{v}, r) = N(X_0 + tZ_0, \mathbf{u}, tr)$.

Proposition 1.3.1. Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ be fixed. Then, for every $\rho, t > 0$ and $X_0 \in \partial^0 B_1^+$, given \mathbf{v} as in (1.3.1) we have $\mathbf{v} \in \mathcal{G}^s(B_{x_0,t}^+)$.

Proof. Fixed $\mathbf{u} \in \mathcal{G}^s(B_1^+)$, by the previous remarks, the last thing left to prove is the validity of the Pohožaev identity (8) for \mathbf{v} , in every ball $B_r(Z_0) \subset B_{X_0,t}^+$, with $Z_0 \in \partial^0 B_{X_0,t}^+$ and

14 NODAL SET OF SEGREGATED CRITICAL CONFIGURATIONS

 $r \in (0, \operatorname{dist}(Z_0, \partial B^+_{X_0,t}))$. We check it by using (18) and by performing the change of variables $(x, y) = (x_0 + tz, tw)$ in the expression of the derivative of the energy $r \mapsto E(Z_0, \mathbf{v}, r)$. More precisely, from (17) let us define $\mathbf{G} = (G_1, \ldots, G_h)$ where

$$G_i(s) = \int_0^s g_i(\tau) d\tau = \frac{t^{1-a}}{\rho^2} \int_0^{\rho s} f_i(\tau) d\tau = \frac{t^{1-a}}{\rho^2} F_i(\rho s),$$

then

$$\begin{aligned} \frac{d}{dr}E(Z_0, \mathbf{v}, r) &= \frac{d}{dr}\frac{1}{\rho^2}E(X_0 + tZ_0, \mathbf{v}, tr) = \frac{t}{\rho^2}\frac{dE}{dr}(X_0 + tZ_0, \mathbf{u}, tr) \\ &= \frac{2t}{\rho^2(tr)^{n-1+a}}\int_{\partial^+ B_{tr}^+(X_0 + tZ_0)} |y|^a |\partial_r \mathbf{u}|^2 \,\mathrm{d}\sigma + \frac{t}{\rho^2}R(X_0 + tZ_0, \mathbf{u}, tr) \\ &= \frac{2}{r^{n-1+a}}\int_{\partial^+ B_r^+(Z_0)} |y|^a |\partial_r \mathbf{v}|^2 \,\mathrm{d}\sigma + R(Z_0, \mathbf{v}, r), \end{aligned}$$

where

$$\begin{aligned} \frac{t}{\rho^2} R(X_0 + tZ_0, \mathbf{u}, tr) &= \frac{t}{\rho^2 (tr)^{n+a}} \int_{\partial^0 B_{tr}^+ (X_0 + tZ_0)} (n+a-1) \langle \mathbf{u}, \mathbf{F}(\mathbf{u}) \rangle - 2n \sum_{i=1}^h F_i(u_i) dx + \\ &+ \frac{t}{\rho^2 (tr)^{n+a-1}} \int_{S_{tr}^{n-1} (X_0 + tZ_0)} 2 \sum_{i=1}^h F_i(u_i) - \langle \mathbf{u}, \mathbf{F}(\mathbf{u}) \rangle dx \\ &= \frac{1}{r^{n+a}} \int_{\partial^0 B_r^+ (Z_0)} (n+a-1) \langle \mathbf{u}, \mathbf{G}(\mathbf{v}) \rangle - 2n \sum_{i=1}^h G_i(v_i) dx + \\ &+ \frac{1}{r^{n+a-1}} \int_{S_r^{n-1} (Z_0)} 2 \sum_{i=1}^h G_i(v_i) - \langle \mathbf{v}, \mathbf{G}(\mathbf{v}) \rangle dx \\ &= R(Z_0, \mathbf{v}, r). \end{aligned}$$

Now, we turn our attention to the convergence of the blow-up sequences, using the previous results about the rescaled functions. Given $\Omega^+ \subset \subset B_1^+$ compactly supported in B_1^+ and $(X_k)_k \in \Omega^+$ and $r_k \searrow 0$, let us consider the following normalized blow-up sequence

$$\mathbf{u}_k(X) = \frac{\mathbf{u}(X_k + r_k X)}{\rho_k} \quad \text{for } X \in B^+_{X_k, r_k}$$
(19)

with

$$\rho_k^2 = \left\| \mathbf{u}(X_k + r_k \cdot) \right\|_{L^{2,a}(\partial^+ B_1^+)}^2 = \frac{1}{r_k^{n+a}} \int_{\partial^+ B_{r_k}^+(X_k)} |y|^a \, \mathbf{u}^2 \, d\sigma = H(X_k, \mathbf{u}, r_k)$$

Hence we have that $\|\mathbf{u}_k\|_{L^{2,a}(\partial^+ B_1^+)} = 1$ and, by Proposition 1.3.1, $\mathbf{u}_k \in \mathcal{G}^s(B_{X_k,r_k}^+)$ since every component solves

$$\begin{cases} -L_a u_{i,k} = 0 & \text{in } B^+_{X_k, r_k} \\ u_{i,k} \left(\partial^a_y u_{i,k} + f_{i,k}(u_{i,k}) \right) = 0 & \text{on } \partial^0 B^+_{X_k, r_k} \end{cases}$$
(20)

where

$$f_{i,k}(s) = \frac{t_k^{1-a}}{\rho_k} f_i(\rho_k s).$$
 (21)

In order to simplify some notations, we introduce the following class of functions which corresponds to the one introduced in [83, 84] in the context of entire segregated profiles.

Definition 1.3.2. For every $s \in (0,1)$ we define with $\mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ the collection of $\mathbf{u} \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1};\mathbb{R}^h)$ with $\mathbf{u} = (u_1, \ldots, u_h)$ continuous, and such that

• $u_i \cdot u_j|_{y=0} \equiv 0$ for every $i \neq j$ and $\mathbf{u} \not\equiv \mathbf{0}$ on Σ . Moreover, for every $i = 1, \ldots, h$ it satisfies

$$\begin{cases} -L_a u_i = 0 & \text{in } \mathbb{R}^{n+1}_+ \\ u_i \partial^a_y u_i = 0 & \text{on } \Sigma; \end{cases}$$
(22)

• for every $X_0 = (x_0, 0) \in \Sigma$ and r > 0, the following Pohožaev type identity holds

$$\frac{1-a-n}{r} \int_{B_{r}^{+}(X_{0})} |y|^{a} |\nabla \mathbf{u}|^{2} dX + \int_{\partial B_{r}^{+}(X_{0})} |y|^{a} |\nabla \mathbf{u}|^{2} d\sigma = 2 \int_{\partial^{+} B_{r}^{+}(X_{0})} |y|^{a} (\partial_{r} \mathbf{u})^{2} d\sigma$$
(23)

In the remaining part of the Section we will prove the following convergence result for blow-up sequences, and present some of its main consequences. First, roughly speaking, we observe that $(B_1^+ - X_k)/r_k$ approaches the whole \mathbb{R}^{n+1} as $k \to +\infty$ since the distance $\operatorname{dist}(X_k, \partial B_1^+) \ge \operatorname{dist}(\Omega^+, \partial B_1^+) > 0$ for every k.

Theorem 1.3.3. Given $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$, let us consider a sequence $(X_k)_k \subset \partial^0 B_1^+$ and $(\mathbf{u}_k)_k$ the associated blow-up sequence defined in (19). Thus, there exists a vector valued function $\overline{\mathbf{u}} \in \mathcal{G}_{loc}^s(\mathbb{R}^{n+1})$ such that, up to a subsequence, $\mathbf{u}_k \to \overline{\mathbf{u}}$ in $C_{loc}^{0,\alpha}(\mathbb{R}^N)$ for every $\alpha \in (0,s)$ and strongly in $H_{loc}^{1,a}(\mathbb{R}^{n+1})$. In particular, the blow-up limit $\overline{\mathbf{u}} = (\overline{u}_1, \dots, \overline{u}_h)$ satisfies

$$\begin{cases} -L_a \overline{u}_i = 0 & \text{in } \mathbb{R}^{n+1}_+ \\ -\overline{u}_i \partial_y^a \overline{u}_i = 0 & \text{on } \Sigma \end{cases} \quad in \, \mathcal{D}'(\mathbb{R}^{n+1}), \tag{24}$$

for every i = 1, ..., h.

The proof will be presented in a series of lemmata, since some results follow directly from the ideas and techniques presented in [83, 84].

Lemma 1.3.4. Let $s \in (0,1)$, $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ and $\Omega^+ \subset B_1^+$ compactly supported in B_1^+ . Then, there exist $\tilde{C} > 0$ and $\tilde{r} > 0$ such that for every $X \in \partial^0 \Omega^+$ and $0 < r \leq \tilde{r}$ we have

$$\frac{1}{r^{n-1+a}} \int_{B_r^+(X_0)} |y|^a \, |\nabla \mathbf{u}|^2 \, \mathrm{d}X + \frac{1}{r^{n+a}} \int_{\partial^+ B_r^+(X_0)} |y|^a \, \mathbf{u}^2 \mathrm{d}X \le C \left(E(X_0, \mathbf{u}, r) + H(X_0, \mathbf{u}, r) \right).$$

The proof of the previous result is a direct consequence of (11) in the proof of Lemma 1.2.2.

Lemma 1.3.5. Under the previous notations, for any given R > 0 we have

$$\|\mathbf{u}_k\|_{H^{1,a}(B_R^+)} \leq C \quad \text{and} \quad \|\mathbf{u}_k\|_{L^\infty(\overline{B_R^+})} \leq C,$$

where C > 0 is a constant independent on k > 0.

Proof. Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ be such that $\mathbf{u} = (u_1, \ldots, u_h)$, with $u_i \in H^{1,a}_{\text{loc}}(B_+)$. First, in order to prove the uniform bound with respect to the $H^{1,a}$ -norm, by the Poincarè inequality (10) we consider the weighted Sobolev space $H^{1,a}(B_R^+)$ endowed with the norm

$$\|v\|_{H^{1,a}(B_R^+)}^2 := \frac{1}{r^{n-1+a}} \int_{B_R^+} |y|^a \, |\nabla v|^2 \, \mathrm{d}X + \frac{1}{r^{n+a}} \int_{\partial^+ B_R^+} |y|^a \, v^2 \mathrm{d}\sigma.$$

By definition of the blow-up sequence $(\mathbf{u}_k)_k$, given C > 0 and $\overline{r} > 0$ be constants such that Proposition 1.2.4, Proposition 1.2.7 and Lemma 1.3.4 hold true. Then, up to taking k so large that $r_k, r_k R \leq \overline{r}$, we get

$$\begin{split} \int_{\partial^+ B_R^+} |y|^a \, \mathbf{u}_k^2 \mathrm{d}\sigma &= \frac{1}{\rho_k^2} \int_{\partial^+ B_R^+} |y|^a \, \mathbf{u}^2 (X_k + r_k X) \mathrm{d}\sigma \\ &= \frac{1}{\rho_k^2 r_k^{n+a}} \int_{\partial^+ B_{Rr_k}^+ (X_k)} |y|^a \, \mathbf{u}^2 \mathrm{d}\sigma \\ &= R^{n+a} \frac{H(X_k, \mathbf{u}, Rr_k)}{H(X_k, \mathbf{u}, r_k)} \\ &\leq R^{n+a} \left(\frac{Rr_k}{r_k}\right)^{2\overline{C}} \end{split}$$
where $\overline{C} > 0$ is defined in (14). Since we proved $\|\mathbf{u}_k\|_{L^{2,a}(\partial^+ B_R^+)}^2 \leq C(R)R^{n+a}$, passing to the second term we conclude

$$\begin{split} \int_{B_R} |y|^a \, |\nabla \mathbf{u}_k|^2 \, \mathrm{d}\sigma &= N(0, \mathbf{u}_k, R) \frac{1}{R} \int_{\partial B_R} |y|^a \, \mathbf{u}_k^2 \mathrm{d}\sigma \\ &\leq C(R) R^{n-1+a} N(X_k, \mathbf{u}, Rr_k) \\ &\leq C(R) R^{n-1+a} e^{C \Psi(\bar{r})} \left(N(X_k, \mathbf{u}, \bar{r}) + 1 \right) \\ &< \overline{C}(R) R^{n-1+a} \end{split}$$

where in the second inequality we used the monotonicity result of Proposition 1.2.4. Since by (14) we obtain $\|\nabla \mathbf{u}_k\|_{L^{2,a}(B_R^+)}^2 \leq C(R)R^{n+a-1}$, it remains to prove the uniform bound with respect to the $L^{\infty}(B_R^+)$ -norm.

Fixed R > 0, let $\mathbf{v}_k \in H^{1,a}(B_R^+)$ be the symmetric extension of \mathbf{u}_k with respect to Σ to the whole B_1 . Since $-\partial_y^a \mathbf{u}_k \leq 0$ on $\partial^0 B_1^+$, the map \mathbf{v}_k is L_a -subharmonic, i.e. $-L_a \mathbf{v}_k \leq 0$, by [88, Lemma A.2] we get

$$\begin{split} \sup_{\substack{B_{R/2}^+ \\ B_{R/2}^-}} \mathbf{u}_k &= \sup_{B_{R/2}} \mathbf{v}_k \le C(n,s) \left(\frac{1}{R^{n+1+a}} \int_{B_R} |y|^a \, \mathbf{v}_k^2 \mathrm{d}X \right)^{1/2} \\ &= 2C(n,s) \left(\frac{1}{R^{n+1+a}} \int_{B_R^+} |y|^a \, \mathbf{u}_k^2 \mathrm{d}X \right)^{1/2} \\ &\le 2C(n,s) \left(\frac{H(0,\mathbf{u}_k,R)}{n+a+1} \right)^{1/2}, \end{split}$$

where in the third inequality we used the monotonicity of $r \mapsto H(0, \mathbf{u}_k, r)$ in (0, R). Finally, the estimate follows directly from the one the $L^{2,a}(\partial^+ B_R^+)$ -norm.

So far we have proved the existence of a nontrivial function $\overline{\mathbf{u}} \in H^{1,a}_{\mathrm{loc}}(\overline{\mathbb{R}^{N+1}_+}; \mathbb{R}^h) \cap L^{\infty}_{\mathrm{loc}}(\mathbb{R}^{n+1}_+)$ such that, up to a subsequence, we have $u_{i,k} \rightharpoonup \overline{u}_i$ weakly in $H^{1,a}_{\mathrm{loc}}(\mathbb{R}^{n+1}_+)$, for every $i = 1, \ldots, h$.

Moreover, since by Definition 1.2.1 and (21) there exists M > 0 such that, for every i = 1, ..., h and k > 0

$$\|f_{i,k}(u_{i,k}))\|_{L^{\infty}(\partial^{0}B_{R}^{+})} \leq Mr_{k}^{1-a} \|u_{i,k}\|_{L^{\infty}(\partial^{0}B_{R}^{+})} \to 0,$$
(25)

since $a \in (-1,1)$ and $r_k \to 0^+$. Hence we deduce

$$\begin{cases} -L_a \overline{u}_i = 0 & \text{in } \mathbb{R}^{n+1}_+ \\ \overline{u}_i \partial_y^a \overline{u}_i = 0 & \text{on } \Sigma \end{cases} \quad \text{in } \mathcal{D}'(\mathbb{R}^{n+1}), \tag{26}$$

for every i = 1, ..., h. The next step is to prove that the convergence of $\mathbf{u}_k \to \overline{\mathbf{u}}$ is indeed strong in $H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1}_+)$ and $C^{0,\alpha}_{\text{loc}}(\overline{\mathbb{R}^{n+1}_+})$ for $\alpha \in (0, s)$.

Lemma 1.3.6. Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ be fixed. Given $(\mathbf{u}_k)_k$ a blow-up sequences of the form (19), then for every R > 0, up to a subsequence, $\mathbf{u}_k \to \overline{\mathbf{u}}$ strongly in $H^{1,a}(B_R^+)$.

Proof. We already know, by compactness, the existence of a blow-up limit $\overline{\mathbf{u}} \in H^{1,a}_{\mathrm{loc}}(\mathbb{R}^n)$, which solves (26) in $\mathcal{D}'(\mathbb{R}^{n+1}_+)$. To prove the strong convergence in $H^{1,a}_{\mathrm{loc}}(\mathbb{R}^{n+1}_+)$ let us consider $\varphi \in C^{\infty}_{c}(B_{2R})$ a cut-off function such that $0 \leq \varphi \leq 1$, $\varphi \equiv 1$ in B_R . First, since it holds

$$-L_a\left(u_{i,k} - \overline{u}_i\right) = 0 \quad \text{in } \mathcal{D}'(B_{2R}^+),$$

testing it with $(u_{i,k} - \overline{u}_i)\varphi$ and integrating by parts, we get

$$\int_{B_{2R}^+} |y|^a \varphi |\nabla(u_{i,k} - \overline{u}_i)|^2 dX + \int_{B_{2R}^+} |y|^a (u_{i,k} - \overline{u}_i) \langle \nabla(u_{i,k} - \overline{u}_i), \nabla\varphi \rangle dX =$$

$$= -\int_{\partial^0 B_{2R}^+} \varphi (u_{i,k} - \overline{u}_k) \partial_y^a (u_{i,k} - \overline{u}_k) dx.$$
(27)

Now, we can conclude just by observing that

$$\begin{split} \left| \int_{B_{2R}^+} |y|^a \left(u_{i,k} - \overline{u}_i \right) \langle \nabla(u_{i,k} - \overline{u}_i), \nabla\varphi \rangle \mathrm{d}X \right| &\leq C \left\| u_{i,k} - \overline{u}_i \right\|_{L^{\infty}(\overline{B_{2R}^+})} \left\| \nabla u_{i,k} \right\|_{L^{2,a}(B_{2R}^+)} \to 0, \\ \left| \int_{\partial^0 B_{2R}^+} \varphi \left(u_{i,k} - \overline{u}_k \right) \partial_y^a \left(u_{i,k} - \overline{u}_k \right) \mathrm{d}x \right| &\leq \left\| u_{i,k} - \overline{u}_i \right\|_{L^{\infty}(\overline{B_{2R}^+})} \int_{\partial^0 B_{2R}^+} \varphi \partial_y^a u_{i,k} \mathrm{d}x + \\ &+ \left\| u_{i,k} - \overline{u}_i \right\|_{L^{\infty}(\overline{B_{2R}^+})} \int_{\partial^0 B_{2R}^+} \varphi \partial_y^a \overline{u}_i \mathrm{d}x \\ &+ C(R) \left\| u_{i,k} - \overline{u}_i \right\|_{L^{\infty}(\overline{B_{2R}^+})} \to 0, \end{split}$$

where in the right hand side of (27) we used [88, Lemma A.2], since $-\partial_y^a u_{i,k} \leq 0$ and $-\partial_y^a u_{i,k} \leq 0$ on Σ .

Similarly, given $\mathbf{v}_k \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ and $\overline{\mathbf{v}} \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ respectively the symmetric extensions of \mathbf{u}_k and $\overline{\mathbf{u}}$ through Σ , one could relate the system (26) to a system of degenerate elliptic equation with a boundary measure data on Σ . More precisely, for every k > 0 there exists a collection of non negative Radon measures $\mu_{i,k} \in \mathcal{M}(B_1^+)$ for $i = 1, \ldots, h$, each one supported on $\partial^0 B_1^+$, such that

$$-L_a v_{i,k} = -\mu_{i,k} \quad \text{in } \mathcal{D}' \left(B_{X_k,r_k}^+ \right),$$

for every i = 1, ..., h. Then, following the strategy in [81, Lemma 3.7] and [81, Lemma 3.11], one could obtain the same strong convergence in $H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ by using the uniform L^{∞} and $H^{1,a}$ estimates in B_R^+ , for every R > 0.

Lemma 1.3.7. Under the previous notations, for every R > 0 there exists C > 0, independent of k, such that

$$[\mathbf{u}_k]_{C^{0,\alpha}(B_R)} = \sup_{X_1, X_2 \in \overline{B}_R} \frac{|\mathbf{u}(X_1) - \mathbf{u}(X_2)|}{|X_1 - X_2|^{\alpha}} \le C$$

for every $\alpha \in (0, \alpha^*)$.

Proof. The proof follows essentially the ideas of the similar results in [83, 84]. Without loss of generality, let R = 1 and suppose by contradiction that up to a subsequence

$$L_{k} = \max_{i=1,\dots,h} \sup_{X_{1},X_{2} \in \overline{B_{1}^{+}}} \frac{|\eta(X_{1})u_{i,k}(X_{1}) - \eta(X_{2})u_{i,k}(X_{2})|}{|X_{1} - X_{2}|^{\alpha}} \to \infty$$

where $\eta \in C_c^{\infty}(B_1)$ is a smooth function such that

$$\begin{cases} \eta(X) = 1, & 0 \le |X| \le 1/2\\ 0 < \eta(X) \le 1, & 1/2 \le |X| \le 1\\ \eta(X) = 0, & |X| = 1. \end{cases}$$

Since we may assume that L_k is achieved by the first component of \mathbf{u}_k and a sequence of points $(X_{1,k}, X_{2,k}) \in \overline{B_1^+} \times \overline{B_1^+}$, given $r_k = |X_{1,k} - X_{2,k}|$ we can prove, as $k \to \infty$, that

•
$$r_k \to 0$$

• $\frac{\operatorname{dist}(X_{1,k}, \partial^+ B_1^+)}{r_k} \to \infty, \frac{\operatorname{dist}(X_{2,k}, \partial^+ B_1^+)}{r_k} \to \infty$

Before to continue, let us fix the notations $X_{1,k} = (x_{1,k}, y_{1,k})$ and $X_{2,k} = (x_{2,k}, y_{2,k})$. Now, since by Lemma 1.3.5 the norm $\|\mathbf{u}_k\|_{L^{\infty}(B_1^+)}$ is uniformly bounded, we have

$$L_{k} \leq \frac{\|\mathbf{u}_{k}\|_{L^{\infty}(B_{1}^{+})}}{r_{k}^{\alpha}} \left(\eta(X_{1,k}) - \eta(X_{2,k})\right),$$
(28)

which immediately implies that $r_k \to 0$. Now, since η is compactly supported in B_1 and it vanishes on $\partial^+ B_1^+$, for every $X \in \overline{B_1^+}$ we have

$$\eta(X) \le \operatorname{dist}(X, \partial^+ B_1^+) \operatorname{Lip}(\eta),$$

where obviously $\text{Lip}(\eta)$ denotes the Lipschitz constant of η . Finally, the inequality (28) becomes

$$\frac{\operatorname{dist}(X_{1,k},\partial^+B_1^+)}{r_k} + \frac{\operatorname{dist}(X_{2,k},\partial^+B_1^+)}{r_k} \ge \frac{L_k r_k^{\alpha-1}}{\operatorname{Lip}(\eta) \|\mathbf{u}_k\|_{L^{\infty}(B_1^+)}} \to \infty$$

and the result follows by recalling that $\alpha < 1$. As in [83, 84], our proof is based on two different blow-up sequences, indeed for every i = 1, ..., h we introduce the auxiliary sequences

$$w_{i,k}(X) = \eta(P_k) \frac{u_{i,k}(P_k + r_k X)}{L_k r_k^{\alpha}} \quad \text{and} \quad \overline{w}_{i,k}(X) = \frac{(\eta u_{i,k})(P_k + r_k X)}{L_k r_k^{\alpha}}$$

for $X \in B^+_{P_k,r_k}$ and $P_k = (p_{x,k}, p_{y,k})$ a suitable sequence of points that will be choose later. On one hand the sequence $(\overline{\mathbf{w}}_k)_k$ has an uniform bound on the α - Hölder seminorm, i.e.

$$\sup_{X_1 \neq X_2 \in \overline{B_{P_k,r_k}^+}} \frac{\left|\overline{w}_{i,k}(X_1) - \overline{w}_{i,k}(X_2)\right|}{\left|X_1 - X_2\right|^{\alpha}} \le \left|\overline{w}_{1,k}\left(\frac{X_1 - P_k}{r_k}\right) - \overline{w}_{1,k}\left(\frac{X_2 - P_k}{r_k}\right)\right| = 1,$$

while on the other hand $(\mathbf{w}_k)_k \in \mathcal{G}^s(B^+_{P_k,r_k})$, where each components satisfy

$$\begin{cases} -L_{a}^{k}w_{i,k} = 0 & \text{in } B_{P_{k},r_{k}}^{+} \\ w_{i,k}\left(\partial_{y}^{a,k}w_{i,k} + g_{i,k}(w_{i,k})\right) = 0 & \text{on } \partial^{0}B_{P_{k},r_{k}}^{+} \end{cases}$$
(29)

with the new operators

$$L_a^k = \operatorname{div}\left(\left(y + \frac{p_{y,k}}{r_k}\right)^a \nabla\right), \quad \partial_y^{a,k} = \lim_{y \to -\frac{p_{y,k}}{r_k}} \left(y + \frac{p_{y,k}}{r_k}\right)^a \partial_y,$$

and

$$g_{i,k}(t) = \eta(P_k) \frac{r_k^{2s-\alpha}}{L_k} f_{i,k} \left(\frac{L_k r_k^{\alpha}}{\eta(P_k)} t\right).$$

By Lemma 1.3.5 and (25), we infer

$$\sup_{\partial^0 B^+_{P_k, r_k}} |g_{i,k}(w_{i,k})| = \eta(P_k) \frac{r_k^{2s-\alpha}}{L_k} \sup_{\partial^0 B^+_1} |f_{i,k}(u_{i,k})| \to 0^+$$

as $k \to +\infty$.

The importance of these two sequences lies in the fact that they have asymptotically equivalent behaviour. Namely, since

$$|w_{i,k}(X) - \overline{w}_{i,k}(X)| \leq \frac{\|\mathbf{u}_k\|_{L^{\infty}(B_1)}}{r_k^{\alpha} L_k} |\eta(P_k + r_k X) - \eta(P_k)|$$

$$\leq \frac{\operatorname{Lip}(\eta) r_k^{1-\alpha}}{L_k} \|\mathbf{u}_k\|_{L^{\infty}(B_1)} |X|$$
(30)

we get, for any compact $K \subset \mathbb{R}^{n+1}$, that

$$\max_{X \in K \cap B^+_{P_k, r_k}} |\mathbf{w}_k(X) - \overline{\mathbf{w}}_k(X)| \longrightarrow 0.$$
(31)

Moreover, since $\mathbf{w}_k(0) = \overline{\mathbf{w}}_k(0)$ we note by (30) that

$$|w_{i,k}(X) - w_{i,k}(0)| \le |w_{i,k}(X) - \overline{w}_{i,k}(X)| + |\overline{w}_{i,k}(X) - \overline{w}_{i,k}(0)|$$
$$\le C\left(\frac{r_k^{1-\alpha}}{L_k}|X| + |X|^{\alpha}\right)$$

and consequently, there exists C = C(K) such that $|\mathbf{w}_k(X) - \mathbf{w}_k(0)| \le C$, for every $X \in K$. Let us prove that it is not restrictive to choose $P_k \in \Sigma$ in the definitions of the sequences $(\mathbf{w}_k)_k$, $(\overline{\mathbf{w}}_k)_k$, showing that $X_{1,k}, X_{2,k}$ must converge to $\partial^0 B_1^+$, i.e. there exists C > 0 such that, for k sufficiently large,

$$\frac{\operatorname{dist}(X_{1,k},\partial^0 B_1^+) + \operatorname{dist}(X_{2,k},\partial^0 B_1^+)}{r_k} \le C.$$

The following proof follows directly the one of [84, Lemma 4.5]) but for the sake of complexness we report some details. Arguing by contradiction, suppose that

$$\frac{\operatorname{dist}(X_{1,k},\partial^0 B_1^+) + \operatorname{dist}(X_{2,k},\partial^0 B_1^+)}{r_k} \longrightarrow \infty$$

and let us choose $P_k = X_{1,k}$ in the definition of $\mathbf{w}_k, \overline{\mathbf{w}}_k$ so that $B_{P_k,r_k}^+ \to \mathbb{R}^{n+1}$ and $p_{y,k}^{-1}r_k \to 0^+$. Given $\mathbf{W}_k = \mathbf{w}_k - \mathbf{w}_k(0)$ and $\overline{\mathbf{W}}_k = \overline{\mathbf{w}}_k - \overline{\mathbf{w}}_k(0)$, by construction $\overline{\mathbf{W}}_k$ is a sequence of functions which share the same bound on the α - Hölder seminorm and they are uniformly bounded in every compact $K \subset \mathbb{R}^{n+1}$ since $\overline{\mathbf{W}}_k(0) = \mathbf{0}$. Thus, by the Ascoli-Arzelá theorem, there exists $\mathbf{W} \in C(K)$ which, up to a subsequence, is the uniform limit of $\overline{\mathbf{W}}_k$. By (??), we also find that $\mathbf{W}_k \to \mathbf{W}$ uniformly con compact sets.

In order to reach a contradiction we can prove that \mathbf{W} is a nonconstant globally Hölder harmonic function with $\alpha \in (0, \alpha^*)$.

Since we already know that $\mathbf{W} \in C^{0,\alpha}(\mathbb{R}^{n+1})$ it reamins to prove the harmonicity of the limit function. To this purpose, let $\varphi \in C_c^{\infty}(\mathbb{R}^{n+1})$ be a compactly supported smooth function and \overline{k} be sufficiently large so that supp $\varphi \subset B_{P_k,r_k}^+$, for all $k \geq \overline{k}$. Fixed $i = 1, \ldots, h$, by testing the first equation in (29) with φ we get

$$\int_{\mathbb{R}^{n+1}} \operatorname{div}\left(\left(1+y\frac{r_k}{p_{y,k}}\right)^a \nabla\varphi\right) w_{i,k} \mathrm{d}X = 0.$$

Passing to the uniform limit and observing that

$$\left(1+y\frac{r_k}{p_{y,k}}\right)^a \to 1 \quad \text{in } C^\infty\left(\operatorname{supp} \varphi\right),$$

we deduce that \mathbf{W} is indeed harmonic. The contradiction follows by the classical Liouville Theorem once we show that \mathbf{W} is globally α - Hölder continuous and not constant. Hence, since $P_k = X_{1,k}$ then, up to a subsequence,

$$\frac{X_{2,k} - P_k}{r_k} = \frac{X_{2,k} - X_{1,k}}{|X_{2,k} - X_{1,k}|} \to X_2 \in \partial B_1$$

Finally, by the equicontinuity and the uniform convergence, we conclude

$$\left|\overline{W}_{1,k}\left(\frac{X_1 - P_k}{r_k}\right) - \overline{W}_{1,k}\left(\frac{X_2 - P_k}{r_k}\right)\right| = 1 \longrightarrow \left|\overline{W}_1(0) - \overline{W}_1(X_2)\right| = 1.$$

At this point, the choice $P_k = (x_{1,k}, 0)$ for every $k \in \mathbb{N}$ guarantees the convergence of the rescaled domains $B^+_{P_k, r_k} \to \mathbb{R}^{n+1}_+$, while for any compact set $K \subset \mathbb{R}^{n+1}$

$$\max_{X \in K \cap B^+_{P_k, r_k}} |\mathbf{w}_k(X) - \overline{\mathbf{w}}_k(X)| \longrightarrow 0$$

Hence, we are left with two possibilities:

- for any compact set $K \subset \Sigma$ we have $w_{1,k}(X) \neq 0$ for every $k \geq k_0$ and $X \in K$;
- there exists a sequence $(X_k)_k \subset \Sigma$ such that $\mathbf{w}_k(X_k) = \mathbf{0}$, for every $k \in \mathbb{N}$.

In the first case, if we define again $\mathbf{W}_k = \mathbf{w}_k - \mathbf{w}_k(0)$ and $\overline{\mathbf{W}}_k = \overline{\mathbf{w}}_k - \overline{\mathbf{w}}_k(0)$ we obtain that the last sequence is uniformly bounded in $C^{0,\alpha}$ and hence $(\mathbf{W}_k)_k$ converges uniformly on compact set to a nonconstant globally α - Hölder continuous L_a -harmonic function \mathbf{W} , with $\partial_y^a W_1 \equiv 0$ and $W_i \equiv 0$ for i > 1, on Σ . Now, extending properly the vector \mathbf{W} to the whole \mathbb{R}^{n+1} , we find a contradiction with the Liouville theorem for entire L_a -harmonic function, since $\alpha < \min\{1, 1-a\}$.

Similarly, in the second case $(\mathbf{w}_k)_k$ itself does converge uniformly on compact sets to a nonconstant globally α - Hölder continuous function \mathbf{w} . In particular, by Lemma 1.3.6, we already know that $(\mathbf{w}_k)_k$ itself converge strongly in $H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1}_+)$ and consequently $\mathbf{w} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+)$. The contradiction follows by the Liouville theorem for entire segregated configurations in $\mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+)$. \Box

We remark that the class of entire segregated profiles $\mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+)$ has been introduced in [84, 84], where the authors proved several properties and monotonicity formulas in order to better understand the asymptotic behaviour of solutions of a competition-diffusion problem with anomalous diffusion and variational competition.

The following is an improved version of a compactness result concerning entire segregated profiles in [84, Proposition 4.7]. Moreover, this result provides a compactness criterion for suitable blow-up sequences, and will be useful in the proof of the gap condition on the possible values of the Almgren frequency formula.

Proposition 1.3.8. Let $(\mathbf{u}_k)_k \subset \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+) \cap C^{0,\alpha}(\overline{B^+_1})$, for some $\alpha \in (0,\nu)$, such that

 $\|\mathbf{u}_k\|_{L^{\infty}(B_1^+)} \le M,$

with M independent on k. Then, for every $\alpha' \in (0, \alpha)$, there exists a constant $C = C(M, \alpha')$ such that

$$\|\mathbf{u}_k\|_{C^{0,\alpha'}(\overline{B_{1/2}^+})} \le C.$$

Furthermore, the subset $(\mathbf{u}_k)_k$ is relatively compact in $H^{1,a}(B^+_{1/2}) \cap C^{0,\alpha'}(\overline{B^+_{1/2}})$, for every $\alpha' \in (0, \alpha)$.

Furthermore, in the context of entire segregated profiles, we can improve Proposition 1.2.4 and Corollary 1.2.6 with the following result.

Proposition 1.3.9. [84, Proposition 2.11] Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s_{loc}(\mathbb{R}^{n+1})$. Then, for every $X_0 \in \Sigma$, the Almgren frequency function

$$N(X_0, \mathbf{u}, r) = \frac{E(X_0, \mathbf{u}, r)}{H(X_0, \mathbf{u}, r)} = \frac{\frac{1}{r^{n+a-1}} \int_{B_r^+(X_0)} |y|^a |\nabla \mathbf{u}|^2 dX}{\frac{1}{r^{n+a}} \int_{\partial^+ B_r^+(X_0)} |y|^a \mathbf{u}^2 d\sigma}$$

well define on $(0, +\infty)$ and monotone non decreasing and it satisfies

$$\frac{d}{dr}\log H(X_0, \mathbf{u}, r) = 2\frac{N(X_0, \mathbf{u}, r)}{r}.$$
(32)

Moreover, if $N(X_0, \mathbf{u}, r) \equiv k$ on an open interval, then $N(X_0, \mathbf{u}, r) \equiv k$ for every r, and $\mathbf{u} = (u_1, \ldots, u_h)$ is k-homogeneous function in \mathbb{R}^{n+1} .

The following is a generalization for $s \in (0, 1)$ of [83, Lemma 3.4], that will be crucial in the study of the structure of the nodal set $\Gamma(\mathbf{u})$.

Corollary 1.3.10. Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}_{loc}^{s}(\mathbb{R}^{n+1})$. Then

- $X \mapsto N(X, \mathbf{u}, 0^+)$ is a non negative upper semi-continuous function on Σ ,
- $X \mapsto N(X, \mathbf{u}, 0^+)$ is constant, even infinite.

Proof. The first part of the Corollary follows because in the case of entire configurations $N(X, \mathbf{u}, 0^+)$ is defined, by monotonicity, as the infimum of continuous functions $X \mapsto N(X, \mathbf{u}, r)$. On the other hand, following the reasoning in Lemma 3.7 in [83], given

$$k = \lim_{r \to +\infty} N(0, \mathbf{u}, r) > 0$$

let us prove the second assertion in the case $k < +\infty$, otherwise it follows with minor changes. By contradiction, suppose there exists $X_0 \in \Sigma$ such that $\sup_{r>0} N(X_0, \mathbf{u}, r) = k - 2\varepsilon$, for some $\varepsilon > 0$. Let moreover $r_0 > 0$ be such that

$$N(0, \mathbf{u}, r_0) \ge k - \varepsilon$$

Up to taking R_1 and R_2 sufficiently large, integrating by parts (32) we get from the previous assumption

$$H(X_0, \mathbf{u}, R_1) \le H(X_0, \mathbf{u}, 1) R_1^{2(k-2\varepsilon)}$$
 and $H(0, \mathbf{u}, R_2) \ge H(0, \mathbf{u}, 1) R_2^{2(k-\varepsilon)}$.

By definition

$$\int_{B_{R_1}^+(X_0)\setminus B_{r_0}^+(X_0)} |y|^a \,\mathbf{u}^2 \mathrm{d}X = \int_{r_0}^{R_1} \rho^{n+a} H(X_0, \mathbf{u}, \rho) \mathrm{d}\rho \le CR_1^{n+a+2(k-2\varepsilon)}$$

and similarly

$$\int_{B_{R_2}^+ \backslash B_{r_0}^+} |y|^a \, \mathbf{u}^2 \mathrm{d}X = \int_{r_0}^{R_2} \rho^{n+a} H(0, \mathbf{u}, \rho) \mathrm{d}\rho \ge CR_2^{n+a+2(k-\varepsilon)}.$$

Now, if we let $|X_0| = R_1 - R_2$, we get

$$CR_{2}^{n+a+2(k-\varepsilon)} \leq \int_{B_{R_{2}}^{+} \setminus B_{r_{0}}^{+}} |y|^{a} \mathbf{u}^{2} dX$$

$$\leq \int_{B_{r_{0}}^{+}(X_{0})} |y|^{a} \mathbf{u}^{2} dX - \int_{B_{r_{0}}^{+}} |y|^{a} \mathbf{u}^{2} dX + \int_{B_{R_{1}}^{+}(X_{0}) \setminus B_{r_{0}}^{+}(X_{0})} |y|^{a} \mathbf{u}^{2} dX$$

$$\leq C + C(R_{2} + |X_{0}|)^{n+a+2(k-2\varepsilon)},$$

and we find a contradiction for R_2 sufficiently large.

Up to now we have dealt with blow-up sequences with arbitrary moving centers $(X_k)_k \subset \Sigma$: the following result emphasizes how some particular choices of X_k provide additional informational on the blow-up limit and how it is correlated to the Almgren frequency formula. More precisely, we have

Proposition 1.3.11. Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$. Fixed a blow-up sequence $(\mathbf{u}_k)_k$ associated to $(X_k)_k \subset \Gamma(\mathbf{u})$, suppose that one of these situations occurs:

• $X_k = X_0$ for every $k \in \mathbb{N}$,

• $X_k \in \Gamma(\mathbf{u})$ and $X_k \to X_0 \in \Gamma(\mathbf{u})$ with $N(X_0, \mathbf{u}, 0^+) = s$.

Then $N(0, \overline{\mathbf{u}}, r) = N(X_0, \mathbf{u}, 0^+) =: \alpha$ for every r > 0 and the blow-up limit $\overline{\mathbf{u}}(r, \theta) = r^{\alpha} \mathbf{g}(\theta)$, where (r, θ) are the generalized polar coordinates centered at the origin in \mathbb{R}^{n+1} .

Proof. First of all we prove that $N(0, \overline{\mathbf{u}}, r)$ is constant for every $r \in (0, +\infty)$. Let us recall that $N(0, \mathbf{u}_k, r) = N(X_k, \mathbf{u}, r_k r)$ and that Theorem 1.3.3 yields that

$$N(0, \overline{\mathbf{u}}, r) = \lim_{k \to \infty} N(0, \mathbf{u}_k, r) = \lim_{k \to \infty} N(X_k, \mathbf{u}, r_k r).$$

If $X_k = X_0$, for some $X_0 \in \Gamma(\mathbf{u})$, then $\lim_k N(X_0, \mathbf{u}, r_k r) = N(X_0, \mathbf{u}, 0^+)$ by Proposition 1.2.4.

In the second case, i.e. $X_k \in \Gamma(\mathbf{u})$ and $X_k \to X_0 \in \Gamma(\mathbf{u})$ with $N(X_0, \mathbf{u}, 0^+) = s$, our purpose is to prove $\lim_k N(X_k, \mathbf{u}, r_k r) = s$.

Denoting with $\overline{r} > 0, C > 0$ the constants associated to Proposition 1.2.4, for any given $\varepsilon > 0$ let us take $0 < \widetilde{r} = \widetilde{r}(\varepsilon) \le \overline{r}$ such that

$$N(X_0, \mathbf{u}, r) \le s + \frac{\varepsilon}{2}$$
 for every $0 < r \le \widetilde{r}$ such that $e^{C\Psi(\widetilde{r})} \le \frac{s+1+2\varepsilon}{s+1+\varepsilon}$.

Furthermore there exists $\delta > 0$ such that

$$N(X, \mathbf{u}, \widetilde{r}) \le s + \varepsilon \quad \text{for } X \in \partial^0 B^+_{\delta}(X_0)$$

Hence, using Proposition 1.2.4 we obtain

$$N(X,\mathbf{u},r) \leq (s+1+\varepsilon)e^{C\Psi(\tilde{r})} - 1 \leq s+2\varepsilon, \quad \text{for } X \in \partial^0 B^+_{\delta}(X_0)$$

and the claim follows by taking into account Corollary 1.2.5.

Finally, let us compute the derivative of $r \mapsto N(0, \overline{\mathbf{u}}, r)$, in order to prove that $\overline{\mathbf{u}}$ is α -homogeneous in $\overline{\mathbb{R}^{n+1}_+}$, i.e. for every $X \in \overline{\mathbb{R}^{n+1}_+} \neq 0$

$$\overline{\mathbf{u}}(X) = |X|^{\alpha} \overline{\mathbf{u}}\left(\frac{X}{|X|}\right).$$

An previously remarked in (9)

$$\frac{d}{dr}H(0,\overline{\mathbf{u}},r) = \frac{2}{r^{n+a}} \int_{\partial^+ B_r^+} |y|^a \left\langle \overline{\mathbf{u}}, \partial_r \overline{\mathbf{u}} \right\rangle dx = \frac{2}{r} E(0,\overline{\mathbf{u}},r)$$

which, together with Theorem 1.3.3, readily implies

$$0 = \frac{1}{2} \frac{d}{dr} N(0, \overline{\mathbf{u}}, r) = \frac{\int_{\partial^+ B_r^+} |y|^a \,\overline{\mathbf{u}}^2 \, d\sigma \int_{\partial^+ B_r^+} |\partial_\nu \overline{\mathbf{u}}|^2 \, d\sigma - \left(\int_{\partial^+ B_r^+} |y|^a \, \langle \overline{\mathbf{u}}, \partial_r \overline{\mathbf{u}} \rangle \, d\sigma\right)^2}{r^{2n+2a-2} H^2(0, \mathbf{u}, r)}$$

for r > 0. This equality yields the existence of C = C(r) > 0 such that $\partial_r \overline{\mathbf{u}} = C(r)\overline{\mathbf{u}}$ for r > 0. Using this fact we get

$$2C(r) = \frac{2\int_{\partial^+ B_r^+} |y|^a \langle \overline{\mathbf{u}}, \partial_r \overline{\mathbf{u}} \rangle \, d\sigma}{\int_{\partial^+ B_r^+} |y|^a \, \overline{\mathbf{u}}^2 \, d\sigma} = \frac{d}{dr} \log H(0, \overline{\mathbf{u}}, r) = \frac{2}{r} N(0, \overline{\mathbf{u}}, r) = \frac{2}{r} \alpha$$

and thus $C(r) = \alpha/r$ and $\overline{\mathbf{u}}(r, \theta) = r^{\alpha} \mathbf{g}(\theta)$ as we claimed.

Moreover, in the case of a blow-up sequences centered we can further improve the convergence result in the following way

Corollary 1.3.12. Given $s \in (0, 1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$, let $(\mathbf{u}_k)_k$ be a blow-up sequence centered in $X_0 \in \Gamma(\mathbf{u})$ and $\Gamma(\mathbf{u}_k)$ the associated nodal sets. Then $\Gamma(\mathbf{u}_k) \to \Gamma(\overline{\mathbf{u}})$ locally with respect to the Hausdorff distance $d_{\mathcal{H}}$ in Σ , i.e. for every R > 0

$$d_{\mathcal{H}}\left(\Gamma(\mathbf{u}_k) \cap \partial^0 B_R^+, \Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_R^+\right) \to 0.$$

In the previous statement we denoted with

$$d_{\mathcal{H}}(A,B) := \max\left\{\sup_{a \in A} \operatorname{dist}(a,B), \sup_{b \in B} \operatorname{dist}(A,b)\right\}, \quad A,B \subseteq \mathbb{R}^{N}$$
(33)

the Hausdorff distance in \mathbb{R}^n . Notice that $d_{\mathcal{H}}(A, B) \leq \varepsilon$ if and only if $A \subseteq N_{\varepsilon}(B)$ and $B \subseteq N_{\varepsilon}(A)$, where $N_{\varepsilon}(\cdot)$ is the closed ε -neighborhood of a set, more precisely

$$N_{\varepsilon}(A) = \{x \in \mathbb{R}^N : \operatorname{dist}(x, A) \le \varepsilon\}, \quad A \subseteq \mathbb{R}^N.$$

Proof of Corollary 1.3.12. It is not restrictive to consider the case R = 1. By the definition of Hausdorff distance, the claimed result is equivalent to prove that for every $\varepsilon > 0$ there exists $\overline{k} > 0$ such that for every $k \ge \overline{k}$

$$\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+ \subseteq N_{\varepsilon} \left(\Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_1^+ \right)$$

$$\Gamma(\overline{\mathbf{u}}) \cap \partial^0 B \subseteq N_{\varepsilon} \left(\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+ \right)$$

Supposing by contradiction that the first inclusion is not true, then there exist $\overline{\varepsilon} > 0$ and a sequence $X_k \in \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+$ such that dist $(X_k, \Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_1^+) > \overline{\varepsilon}$. Moreover, up to a subsequence, $X_k \to \overline{X} \in \Gamma(\overline{\mathbf{u}}) \cap \overline{\partial^0 B_1^+}$ by the L_{loc}^{∞} convergence of $\mathbf{u}_k \to \overline{\mathbf{u}}$. Since by Proposition 1.3.11 the nodal set $\Gamma(\overline{\mathbf{u}})$ is a conical set, i.e. for every $\lambda > 0$ and $\overline{X} \in \Gamma(\overline{\mathbf{u}})$ we have $\lambda \overline{X} \in \Gamma(\overline{\mathbf{u}})$ and $0 \in \Gamma(\overline{\mathbf{u}})$, we deduce that $\text{dist}(\overline{X}, \Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_1^+) = 0$, which provides the contradiction.

Finally, we have to prove that for every $\varepsilon > 0$ there exists $\bar{k} > 0$ such that

$$\Gamma(\overline{\mathbf{u}}) \cap \partial^0 B \subseteq N_{\varepsilon} \left(\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+ \right) \quad \text{for every } k \ge \overline{k}.$$

We start by proving that given $\overline{X} \in \Gamma(\overline{\mathbf{u}}) \in \partial^0 B_1^+$ and $\delta > 0$, the vector valued function \mathbf{u}_k must have a zero in $\partial^0 B_{\delta}^+(\overline{X})$, for k sufficiently large. If not, by recalling that $u_{i,k} \cdot u_{j,k}|_{y=0} \equiv 0$ for every $i \neq j$, we would have that there exists an index 0 < i < h such that

$$\begin{cases} -L_a u_{i,k} = 0 & \text{in } B^+_{\delta}(\overline{X}) \\ -\partial^a_y u_{i,k} = f_i(u_{i,k}) & \text{on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}, \quad u_{i,k} > 0 \text{ on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}$$

and $u_{j,k} \equiv 0$ in $\partial^0 B^+_{\delta}(\overline{X})$, for every $j \neq i$. Passing to the limit, this would imply that

$$\begin{cases} -L_a \overline{u}_i = 0 & \text{in } B^+_{\delta}(\overline{X}) \\ -\partial^a_y \overline{u}_i = 0 & \text{on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}, \quad \overline{u}_i \ge 0 \text{ on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}$$

and $\overline{u}_j \equiv 0$ on $\partial^0 B^+_{\delta}(\overline{X})$ for every $j \neq i$. Since $\overline{X} \in \Gamma(\overline{\mathbf{u}})$ it follows from the Hopf principle (see [43, 13]) that $\overline{\mathbf{u}} \equiv 0$ in $\partial^0 B^+_{\delta}(\overline{X})$, a contradiction with the fact the $\Gamma(\overline{\mathbf{u}})$ has empty interior.

Now, arguing by contradiction, suppose the existence of $\bar{\varepsilon} > 0$ and $(X_k)_k \in \Gamma(\bar{\mathbf{u}}) \cap \partial^0 B_1^+$ such that $X_k \to \overline{X} \in \Gamma(\bar{\mathbf{u}}) \cap \partial^0 B_1^+$ and $\operatorname{dist}(X_k, \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+) > \bar{\varepsilon}$. Since $\Gamma(\bar{\mathbf{u}})$ is a conical set passing through the origin, let us take $\widetilde{X} \in \Gamma(\bar{\mathbf{u}}) \cap \partial^0 B_1^+$ such that $\left| \widetilde{X} - \overline{X} \right| \leq \bar{\varepsilon}/4$. Furthermore, we can take, by using the result proved in the previous paragraph, a sequence $(\overline{X}_k)_k \in \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+$ such that $\left| \overline{X}_k - \widetilde{X} \right| \leq \bar{\varepsilon}/4$ for sufficiently large k. The final contradiction follows noticing that

$$\operatorname{dist}(X_k, \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+) \le \left| X_k - \overline{X}_k \right| \le \left| X_k - \overline{X} \right| + \left| \overline{X} - \widetilde{X} \right| + \left| \widetilde{X} - \overline{X}_k \right| \le \frac{3\overline{\varepsilon}}{4} \le \overline{\varepsilon},$$

for sufficiently large k.

Finally, we define the following class of blow-up which contains all the possible blow-up limit of u centered in a fixed point $X_0 \in \mathbf{u}$.

Definition 1.3.13. Given $s \in (0, 1)$ we define the set $\mathfrak{B}^{s}(\mathbb{R}^{n+1})$ of all possible blow-up limit of $\mathbf{u} \in \mathcal{G}^{s}(B_{1}^{+})$ centered in $X_{0} \in \Gamma(\mathbf{u})$ as the collection of homogenous entire segregated profile in $\mathcal{G}_{\text{loc}}^{s}(\mathbb{R}^{n+1})$.

In particular, given $\mathbf{u} \in \mathfrak{B}^s(\mathbb{R}^{n+1})$ a k-homogenous entire segregated profile, then there exists $\overline{\mathbf{g}} \in H^{1,a}(S^n_+)$ such that

$$\mathbf{u}(X) = |X|^k \mathbf{g}\left(\frac{X}{|X|}\right). \tag{34}$$

We remark that for $a \in (-1, 1)$, given (r, θ) the generalized spherical coordinates in \mathbb{R}^{n+1} with r > 0 and $\theta \in S^n_+$, the weighted Sobolev space $H^{1,a}(S^n_+)$ is defined as the closure of $C^{\infty}_c(S^n_+)$ with respect to the norm

$$||g||_{H^{1,a}(S^n_+)}^2 = \int_{S^n_+} |\sin \theta_n|^a g^2 \mathrm{d}\sigma + \int_{S^n_+} |\sin \theta_n|^a |\nabla_{S^n}g|^2 \,\mathrm{d}\sigma,$$

where θ_n is the spherical coordinate associated to the *y*-direction and ∇_{S^n} the tangential gradient on S^n . Moreover, under the previous notations, we can find a spherical decomposition of the L_a -operator. More precisely,

$$L_a u = \sin^a(\theta_n) \frac{1}{r^n} \partial_r \left(r^{n+a} \partial_r u \right) + \frac{1}{r^a} L_a^{S^n} u$$
(35)

where $y = r \sin(\theta_n)$ and the Laplace-Beltrami type operator is defined as

$$L_a^{S^n} u = \operatorname{div}_{S^n}(\sin^a(\theta_n) \nabla_{S^n} u), \tag{36}$$

with div_{S^n} the tangential divergence on S^n . Inspired by the previous spherical decomposition, we can find a simple characterization of the blow-up limit $\mathbf{u} \in \mathfrak{B}^s(\mathbb{R}^{n+1})$ in term of its trace on the upper sphere S^n_+ .

Proposition 1.3.14. Let $s \in (0,1)$ and $\mathbf{u} \in \mathfrak{B}^{s}(\mathbb{R}^{n+1})$ be a γ -homogeneous blow-up limit, i.e. such that $N(0, \mathbf{u}, 1) = \gamma$. Then, there exists $\mathbf{g} = (g_1, \ldots, g_h) \in H^{1,a}(S^n_+)$ such that, for every $i = 1, \ldots, h$ we get

$$\begin{cases} -L_a^{S^n} g_i = \lambda(\gamma) g_i \sin^a(\theta_n) & \text{in } S^n_+ \\ \partial^a_{\theta_n} g_i = 0 & \text{on } \omega_i \subset S^{n-1} \\ g_i = 0 & \text{on } S^{n-1} \setminus \omega_i, \end{cases}$$
(37)

where $\lambda(\gamma) = \gamma(\gamma + n + a - 1), (\omega_i)_i \subset S^{n-1} \times \{0\}$ and

$$\partial^a_{\theta_n}g(\theta',0) = \lim_{\theta_n \to 0^+} \sin^a(\theta_n) \partial_{\theta_n}g(\theta',\theta_n) \quad \textit{for } \theta' \in S^{n-1}.$$

Proof. Let us consider initially a general case: let $u \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ be a γ -homogeneous L_a -harmonic function in \mathbb{R}^{n+1}_+ such that $u\partial_y^a u = 0$ on Σ . By the homogeneity of u, there exists $\omega \subset S^{n-1}$ such that

$$C_{\omega} = \left\{ X \in \Sigma \colon \frac{X}{|X|} \in \omega \right\} = \left\{ X \in \Sigma \colon \partial_y^a u(X) = 0 \right\},\tag{38}$$

where C_{ω} is the cone in Σ spanned by ω with vertex at zero. Since u is γ -homogeneous, i.e. $u(r, \theta) = r^{\gamma}g(\theta)$ for $r \in (0, +\infty)$ and $\theta \in S^{n-1}$, we get by (35) and (36) that

$$-L_a^{S^n}g = \lambda_1^s(\gamma)g\sin^a(\theta_n) \quad \text{on } S^n_+,$$

with $\lambda_1^s(\gamma) = \gamma(\gamma + n + a - 1)$, and similarly by (38) we get

$$\partial^a_{\theta_n} g(\theta', 0) = 0 \quad \text{on } \omega.$$

Hence, with a slight abuse of notations, for every open region $\omega \subseteq S^{n-1}$, we can define the eigenvalue

$$\lambda_{1}^{s}(\omega) = \inf \left\{ \frac{\int_{S_{+}^{n}} y^{1-2s} |\nabla_{S^{n}} u|^{2} d\sigma}{\int_{S_{+}^{n}} y^{1-2s} u^{2} d\sigma} : \begin{array}{c} u \in H^{1,a}(S_{+}^{n}) \setminus \{0\} \\ u \equiv 0 \text{ in } S^{n-1} \setminus \omega \end{array} \right\}$$
(39)

and similarly the characteristic exponent of the cone C_{ω} spanned by ω as the quantity

$$\gamma_s(C_\omega) = \gamma_s(\lambda_1^s(\omega))$$

where the function $\gamma_s(t)$ is defined by

$$\gamma_s(t) = \sqrt{\left(\frac{n-2s}{2}\right)^2 + t} - \frac{n-2s}{2}.$$

Now, given $\mathbf{u} \in \mathfrak{B}^s(\mathbb{R}^{n+1})$ let $\mathbf{g} \in H^{1,a}(S^n_+; \mathbb{R}^h)$ be its spherical part defined by (34). Since all the components $u_i \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1}_+)$ share the same homogeneity degree γ , we directly get that the eigenvalue $\lambda(g_i)$ is the same for every component of \mathbf{g} .

Using the variational formulation of the eigenvalue problem associated to $\lambda_1^s(\omega)$ defined in (39), for every $\omega \subset S^{n-1}$ we easily get that

$$0 = \lambda_s^1(S^{n-1}) \le \lambda_s^1(\omega) \le \lambda_s^1(\emptyset) = 2s$$

and more generally, for $\omega_1 \subset \omega_2$ it holds $\lambda_s^1(\omega_2) < \lambda_s^1(\omega_1)$ (see [85, 82] for further properties of this eigenvalue problem). We will exhibit several connection between this two formulations of blow-up limits in Chapter 2 finding a different connection with their interpretation on the traces space as *s*-harmonic function on cones. However, by the previous characterization, at this point we can improve the bound on the Almgren frequency for the segregated profile.

Corollary 1.3.15. Let $s \in (0, 1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$. Then, for every $X_0 \in \Gamma(\mathbf{u})$ we get

 $N(X_0, \mathbf{u}, 0^+) < 2s.$

Proof. By the definition of the class of segregated profiles, given $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ we have $\mathbf{u} \neq \mathbf{0}$ on Σ . Given $X_0 \in \mathbf{u}$ and $\overline{\mathbf{u}} \in \mathcal{G}_{loc}^s(\mathbb{R}^{n+1})$ a blow-up limit of \mathbf{u} at X_0 , we get by the uniform convergence that $\overline{\mathbf{u}} \neq 0$ on Σ . By the characterization of Proposition 1.3.14 we get that necessary $\omega \neq \emptyset$ or, in other words, $\lambda_s^1(\omega) < 2s$. The previous inequality implies

$$N(X_0, \mathbf{u}, 0^+) = N(0, \overline{\mathbf{u}}, 1) = \lambda_s^1(\omega) < 2s,$$

as we claimed.

Actually, we remark that combining the previous bound and the monotonicity result Proposition 1.2.4 we can prove that a segregated function in $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ has identically zero trace on $B_1 \cap \Sigma$ if and only if $N(X_0, \mathbf{u}, 0^+) = 2s$ on some point $X_0 \in \Gamma(\mathbf{u})$.

1.4 HAUSDORFF DIMENSION ESTIMATES FOR REGULAR AND SINGULAR SETS

In the same spirit of [19, 18, 36, 81] we prove that there exists a gap in the possible values of the Almgren frequency formula $N(X, \mathbf{u}, 0^+)$ for $X_0 \in \Gamma(\mathbf{u})$. We remark that as in [81], our analysis is not restricted to solutions of minimal energy as in [19].

Proposition 1.4.1. Given $s \in (0,1)$, n = 2 and $u \in \mathcal{G}^s(B_1^+)$, then for $X_0 \in \Gamma(\mathbf{u})$ either

$$N(X_0, \mathbf{u}, 0^+) = s \text{ or } N(X_0, \mathbf{u}, 0^+) \ge s + \delta,$$

for some universal constant $\delta > 0$.

Proof. By contradiction, given $s \in (0, 1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ suppose there exist two sequences $\varepsilon_k \searrow 0^+$ and $(X_k)_k \subset \Gamma(\mathbf{u}) \cap \partial^0 \Omega^+$, for some $\Omega^+ \subset \subset B_1^+$, such that

$$N(X_k, \mathbf{u}, 0^+) \le s + \varepsilon_k.$$

Moreover, it is not restrictive to suppose that $\varepsilon_k \leq s/2$, in order to always have $s + \varepsilon_k < 2s$. Since $\Gamma(\mathbf{u})$ has empty interior in \mathbb{R}^{n+1} , up to a subsequence there exists $X_0 \in \Gamma(\mathbf{u})$ such that $X_k \to X_0$ and, by Corollary 1.2.5 and Corollary 1.2.6 we get $N(X_0, \mathbf{u}, 0^+) = s$.

Therefore, let us construct a sequence of blow-up limit $(\overline{\mathbf{u}}_k)_k \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+)$ in order to translate the absurd hypothesis in the context of entire segregated configurations in $\mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+)$. Hence, for every $k \in \mathbb{N}$ let

$$\mathbf{u}_{X_k,i}(X) = \frac{\mathbf{u}(X_k + r_i X)}{\rho_i}$$

be the blow-up sequence centered in X_k associated to $r_i \searrow 0^+$ and $\rho_i^2 = H(X_k, \mathbf{u}, r_i)$. By Theorem 1.3.3, there exists a family of blow-up limits $(\overline{\mathbf{u}}_k)_k \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ of homogenous function in \mathbb{R}^{n+1} such that $\mathbf{u}_{X_k,i} \to \overline{\mathbf{u}}_k$ and

$$\int_{\partial^+ B_1^+} |y|^a \,\overline{\mathbf{u}}_k^2 \mathrm{d}\sigma = 1 \quad \text{and} \quad N(0, \overline{\mathbf{u}}_k, r) = s + \varepsilon_k \quad \text{for every } r > 0, \tag{40}$$

namely $\overline{\mathbf{u}}_k$ is $(s + \varepsilon_k)$ -homogeneous in \mathbb{R}^{n+1} , i.e.

$$\overline{\mathbf{u}}_k(X) = |X|^{s+\varepsilon_k} \, \mathbf{g}_k\left(\frac{X}{|X|}\right),$$

with $\mathbf{g}_k \in H^{1,a}(S^n_+)$. Following the same idea in the proof of Lemma 1.3.5, since for every R > 0we have $-\partial_y^a \overline{\mathbf{u}}_k \leq 0$ on $\partial^0 B^+_R$, if we consider $\overline{\mathbf{v}}_k \in H^{1,a}(B^+_R)$ the symmetric extension of \mathbf{u}_k with respect to Σ to the whole B_1 we get

$$-L_a \overline{\mathbf{v}}_k \le 0$$
 in \mathbb{R}^{n+1} .

Hence, by [88, Lemma A.2] it follows for every R > 0

$$\begin{split} \sup_{\overline{B_{R/2}^+}} \overline{\mathbf{u}}_k &= \sup_{B_{R/2}} \overline{\mathbf{v}}_k \le C(n,s) \left(\frac{1}{R^{n+1+a}} \int_{B_R} |y|^a \,\overline{\mathbf{u}}_k^2 \mathrm{d}X \right)^{1/2} \\ &= 2C(n,s) \left(\frac{1}{R^{n+1+a}} \int_{B_R^+} |y|^a \,\overline{\mathbf{u}}_k^2 \mathrm{d}X \right)^{1/2} \\ &= 2C(n,s) R^{s+\varepsilon_k} \left(\frac{1}{n+2+2\varepsilon_k} \int_{\partial^+ B_1^+} |y|^a \,\overline{\mathbf{u}}_k^2 \mathrm{d}\sigma \right)^{1/2}, \end{split}$$

where in the last equality we used the $(s + \varepsilon_k)$ -homogeneity of $\overline{\mathbf{u}}_k$. By (40) it follows that for every R > 0 the sequence $(\overline{\mathbf{u}}_k)_k \subset \mathcal{G}_{\text{loc}}^s(\mathbb{R}^{n+1}) \cap C_{\text{loc}}^{0,\alpha}(\overline{\mathbb{R}^{n+1}})$ with $\alpha \in (0, \alpha^*)$ is uniformly bounded in $L^{\infty}(B_R^+)$, for every R > 0, which implies by Theorem 1.3.8 the existence of $\overline{\mathbf{u}} \in \mathcal{G}_{\text{loc}}^s(\mathbb{R}^{n+1})$. Moreover, by the strong convergence in $H_{\text{loc}}^{1,\alpha}(\mathbb{R}^{n+1})$ it follows that $N(0,\overline{\mathbf{u}},r) = s$ for every r > 0, i.e. up to a rotation $\overline{\mathbf{u}} = (\overline{u}_1, \overline{u}_2, 0, \dots, 0)$ where

$$\overline{u}_1(x,y) = C_1 \left(\frac{\sqrt{x_1^2 + y^2} - x_1}{2}\right)^s \quad \text{and} \quad \overline{u}_2(x,y) = C_1 \left(\frac{\sqrt{x_1^2 + y^2} + x_1}{2}\right)^s, \qquad (41)$$

for some positive constant $C_1 > 0$ such that $\|\overline{\mathbf{u}}\|_{L^{2,a}(\partial^+ B_1^+)} = 1$. Up to relabeling the components of $\overline{\mathbf{u}}_k$, let us suppose that the components $\overline{u}_{i,k} \to 0$ strongly in $H^{1,a}_{\text{loc}} \cap C^{0,\alpha}_{\text{loc}}$ for every $i = 2, \ldots, h$, while $(\overline{u}_{1,k}, \overline{u}_{2,k}) \to (\overline{u}_1, \overline{u}_2)$ strongly in $H^{1,a}_{\text{loc}} \cap C^{0,\alpha}_{\text{loc}}$. By (41), since

$$\{\overline{u}_1 = 0\} \cap \Sigma = \{X = (x_1, x_2, y) \in \Sigma \colon x_1 \ge 0\} \{\overline{u}_2 = 0\} \cap \Sigma = \{X = (x_1, x_2, y) \in \Sigma \colon x_1 \le 0\}$$
(42)

32 NODAL SET OF SEGREGATED CRITICAL CONFIGURATIONS

for every k > 0 there exist nonempty $\omega_k \subset \{X \in \Sigma : x_1 > 0\}$ such that ω_k is a connected component in S^{n-1} of $\{g_{1,k} > 0\}$. By contradiction, let us suppose there exists $\overline{k} > 0$ such that $g_{1,\overline{k}} \equiv 0$ on $S^{n-1} \cap \{X \in \Sigma : x_1 \ge 0\}$. Then, necessary we must have $g_{2,\overline{k}} \equiv 0$ on $S^{n-1} \cap \{X \in \Sigma : x_1 \le 0\}$, otherwise by the monotonicity of the eigenvalue (37) we would obtain $\lambda(g_{1,\overline{k}}) > \lambda(g_{2,\overline{k}})$ in contradiction with the definition of $\overline{\mathbf{u}} \in \mathfrak{B}^s(\mathbb{R}^{n+1})$. Hence, since $\overline{\mathbf{u}}_{\overline{k}}$ and $\overline{\mathbf{u}}$ satisfy the same boundary condition, by uniqueness their homogeneities must be equal, in contradiction with the fact that $\overline{\mathbf{u}}_{\overline{k}}$ is $(s + \varepsilon_{\overline{k}})$ -homogeneous in \mathbb{R}^{n+1} . The same contradiction follows from the sequence $(B_k)_k$.

Now, since we are working in dimension n = 2, there exist two sequences $(P_k)_k, (Q_k)_k \subset S^1 \cap \{X \in \Sigma : x_1 > 0\}$ such that $\omega_k = (P_k, Q_k)$ can be seen as an arc of S^1 between the endpoints P_k and Q_k . Since by compactness of S^1 there exist, up to a subsequence, $P, Q \in S^1 \cap \{X \in \Sigma : x_1 \ge 0\}$ respectively limit of $(P_k)_k$ and $(Q_k)_k$, let us consider separately the cases $|P_k - Q_k| \to 0$ and $P \neq Q$.

If $P \neq Q$, let $\omega = (P, Q)$ be the limit of the sequence $(\omega_k)_k$ and C_{ω} be the cone in Σ spanned by ω , i.e.

$$C_{\omega} = \left\{ X \in \Sigma \colon \frac{X}{|X|} \in \omega \subset S^1 \right\}.$$

One one hand, by definition of ω_k , since $u_{1,k}$ and $u_{2,k}$ are segregated on Σ , we get passing $\overline{u}_{2,k} \equiv 0$ on ω_k and then, passing to the limit for $k \to \infty$, $\overline{u}_2 \equiv 0$ on ω . On the other hand, by the L^{∞}_{loc} -convergence of the sequence $(\overline{\mathbf{u}}_k)_k$ we get from (42) that $\|\overline{g}_{1,k}\|_{L^{\infty}(\omega_k)} \to 0$ which implies, passing to its homogeneous extension, that $\overline{u}_1 \equiv 0$ on every compact set $K \subset C_{\omega}$, and similarly $\overline{\mathbf{u}} \equiv 0$ on every compact set in C_{ω} , in contradiction with (41).

Hence in the other case, given $r_k = |P_k - Q_k| \searrow 0$ and $P_k \in \Sigma$ let introduce the blow-up sequence $(\overline{\mathbf{w}}_k)_k \subset \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ centered in $(P_k)_k \in \Gamma(\mathbf{u}_k) \cap S^{n-1}$ and associated to $r_k > 0$, i.e.

$$\overline{\mathbf{w}}_k(X) = \frac{\overline{\mathbf{u}}_k(P_k + r_k X)}{\rho_k} \quad \text{with } \rho_k = \sqrt{H(P_k, \overline{\mathbf{u}}_k, r_k)}.$$

such that $\|\overline{\mathbf{w}}_k\|_{L^{2,a}(\partial^+B_1^+)} = 1$. By construction, for every $k \in \mathbb{N}$ we have

$$\overline{\mathbf{w}}_{k}(0) = \mathbf{0} = \overline{\mathbf{w}}_{k}\left(\frac{Q_{k} - P_{k}}{r_{k}}\right),\tag{43}$$

where $(Q_k - P_k)/r_k \to \nu \in S^{n-1} \times \{0\}$ by compactness in S^{n-1} . Now, following the same ideas in the proof of Lemma 1.3.5, given C > 0 and $\overline{r} > 0$ be such that Proposition 1.2.4, Proposition 1.2.7 and Lemma 1.3.4 hold true then, up to taking k so large that $r_k, r_k R \leq \overline{r}$, we get for every R > 0 that

$$\begin{split} \int_{\partial^+ B_R^+} |y|^a \, \overline{\mathbf{w}}_k^2 \mathrm{d}\sigma &= \frac{1}{\rho_k^2 r_k^{n+a}} \int_{\partial^+ B_{Rr_k}^+(P_k)} |y|^a \, \overline{\mathbf{u}}_k^2 \mathrm{d}\sigma \\ &= R^{n+a} \frac{H(P_k, \overline{\mathbf{u}}_k, Rr_k)}{H(P_k, \overline{\mathbf{u}}_k, r_k)} \\ &\leq R^{n+a} \left(\frac{Rr_k}{r_k}\right)^{2\overline{C}}, \end{split}$$

which implies that

$$\sup_{\overline{B}_{R/2}^+} \overline{\mathbf{w}}_k \le 2C(n,s) \left(\frac{1}{R^{n+1+a}} \int_{B_R^+} |y|^a \, \overline{\mathbf{w}}_k^2 \mathrm{d}X \right)^{1/2} \le 2C(n,s) \left(\frac{H(0,\overline{\mathbf{w}}_k,R)}{n+a+1} \right)^{1/2}$$

is uniformly bounded for k > 0. By Proposition 1.3.8, there exists a blow-up limit $\overline{\mathbf{w}} \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1}) \cap C^{0,\alpha}_{\text{loc}}(\mathbb{R}^{n+1})$ for every $\alpha \in (0,s)$ such that $\overline{\mathbf{w}} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ with $\|\overline{\mathbf{w}}\|_{L^{2,a}(\partial^+B_1^+)} = 1$.

Since the blow-up sequence $(\overline{\mathbf{w}}_k)_k$ is constructed starting from a family of homogeneous entire segregated profiles in $\mathfrak{B}^s_{\text{loc}}(\mathbb{R}^{n+1})$, we can prove that $\overline{\mathbf{w}}$ is constant along the direction parallel to $P \in S^{n-1}$ and that its restriction on the orthogonal half plane belongs to $\mathcal{G}^s_{\text{loc}}(\mathbb{R}^n)$. Hence, let $X \in \mathbb{R}^{n+1}$ and $\lambda \in \mathbb{R}$ be fixed. By the homogeneity of $\overline{\mathbf{u}}_k$ we obtain

$$\overline{\mathbf{w}}_{k}(X+\lambda P_{k}) = \frac{\overline{\mathbf{u}}_{k}(P_{k}+r_{k}(X+\lambda P_{k}))}{\rho_{k}} = \frac{\overline{\mathbf{u}}_{k}((1+r_{k}\lambda)P_{k}+r_{k}X)}{\rho_{k}}$$
$$= \frac{(1+r_{k}\lambda)}{\rho_{k}}^{s+\varepsilon_{k}}\overline{\mathbf{u}}_{k}\left(P_{k}+\frac{r_{k}}{1+r_{k}\lambda}X\right)$$
$$= (1+r_{k}\lambda)^{s+\varepsilon_{k}}\overline{\mathbf{w}}_{k}\left(\frac{r_{k}}{1+r_{k}\lambda}X\right)$$

and then

$$\begin{aligned} \left| \overline{\mathbf{w}}_{k}(X + \lambda P_{k}) - \overline{\mathbf{w}}_{k}(X) \right| &\leq \left| (1 + r_{k}\lambda)^{s + \varepsilon_{k}} \overline{\mathbf{w}}_{k} \left(\frac{r_{k}}{1 + r_{k}\lambda} X \right) - \overline{\mathbf{w}}_{k} \left(\frac{r_{k}}{1 + r_{k}\lambda} X \right) \right| + \\ &+ \left| \overline{\mathbf{w}}_{k} \left(\frac{r_{k}}{1 + r_{k}\lambda} X \right) - \overline{\mathbf{w}}_{k}(X) \right| \\ &\leq \left(\left| (1 + r_{k}\lambda)^{s + \varepsilon_{k}} - 1 \right| + \left| \frac{r_{k}}{1 + r_{k}\lambda} - 1 \right|^{\alpha} |X|^{\alpha} \right) \left\| \overline{\mathbf{w}}_{k} \right\|_{C^{0,\alpha}(\overline{B}_{1}^{+})} \end{aligned}$$

with $\alpha \in (0, s)$. Thus, as $r_k \searrow 0$, by the uniform convergence $(\overline{\mathbf{w}}_k)_k$ on every compact set, we get

$$|\overline{\mathbf{w}}(X + \lambda P) - \overline{\mathbf{w}}(X)| = 0$$
 for every $\lambda \in \mathbb{R}$,

where $P = \lim_{k} P_k \in S^{n-1} \times \{0\}.$

Now, given the section of $\overline{\mathbf{w}}$ with respect to the direction λP , with $\lambda \in \mathbb{R}$, we observe that the equations and the segregation conditions are trivially satisfied and the Pohožaev identities on every ball $B_r(X_0)$ on \mathbb{R}^n_+ follow immediately by the ones for $\overline{\mathbf{w}}$ on the corresponding ball in \mathbb{R}^{n+1}_+ having $B_r(X_0)$ as *n*-dimensional section. Hence, with slight abuse of notation we still denote the *n*-dimensional section by $\overline{\mathbf{w}} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^n)$. By Corollary 1.3.10, since for every k > 0 and $t \in (0, +\infty)$

$$N(P_k, \overline{\mathbf{u}}_k, t) \le N(P_k, \overline{\mathbf{u}}_k, +\infty)$$

= $N(0, \overline{\mathbf{u}}_k, +\infty) = s + \varepsilon_k$

we get from Proposition 1.2.4 and Proposition 1.3.9 that for every R > 0

$$N(0, \overline{\mathbf{w}}, R) = \lim_{r_k \to 0} N(0, \overline{\mathbf{w}}_k, R) = \lim_{r_k \to 0} N(P_k, \overline{\mathbf{u}}_k, r_k R) \le s + \varepsilon_k.$$
(44)

Finally, we reach the contradiction applying a blow-down analysis on the limit function $\overline{\mathbf{w}} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^n)$. With a slight abuse of notations, for $r_k \to +\infty$, consider the blow-down sequence $(\overline{\mathbf{v}}_k)_k$ centered in the origin defined as

$$\overline{\mathbf{v}}_k(X) = rac{\overline{\mathbf{w}}(r_k X)}{
ho_k} \quad ext{ for } X \in B^+_{X_k, r_k}$$

with

$$\rho_k^2 = \|\overline{\mathbf{w}}(r_k \cdot)\|_{L^{2,a}(\partial^+ B_1^+)}^2 = \frac{1}{r_k^{n+a}} \int_{\partial^+ B_{r_k}^+} |y|^a \,\overline{\mathbf{w}}^2 \, d\sigma = H(0, \overline{\mathbf{w}}, r_k).$$

Fixed R > 1, since $\|\overline{\mathbf{w}}\|_{L^{2,a}(\partial^+ B_1^+)} = 1$, we get from integrating (32) between 1 and R that

$$H(0, \overline{\mathbf{w}}, R) \le R^{2(s+\varepsilon_k)},$$

and consequently, up to relabeling the constant, we get

$$\sup_{\overline{B_R^+}} \overline{\mathbf{v}}_k \le 2C(n,s) \left(\frac{H(0,\overline{\mathbf{v}}_k,2R)}{n+a+1}\right)^{1/2} \le C(n,s)R^{s+\varepsilon_k}.$$

Since $(\overline{\mathbf{v}}_k)_k \subset \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$, by the previous uniform bound in $L^{\infty}_{\text{loc}}(\mathbb{R}^{n+1})$ there exists a blowdown limit $\overline{\mathbf{v}} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ such that $\overline{\mathbf{v}}_k \to \overline{\mathbf{v}}$ strongly in $H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ and uniformly in $C_{\text{loc}}^{0,\alpha}(\mathbb{R}^{n+1})$, for $\alpha \in (0,s)$.

Moreover, the uniform convergence and Proposition 1.3.9 yield that

$$N(0, \overline{\mathbf{v}}, R) = \lim_{k \to \infty} N(0, \overline{\mathbf{v}}_k, R) = \lim_{k \to \infty} N(0, \overline{\mathbf{w}}, r_k R) = N(0, \overline{\mathbf{w}}, +\infty),$$

for every R > 0, which implies by Proposition 1.3.9 that $\overline{\mathbf{v}} \in \mathfrak{B}^{s}(\mathbb{R}^{n+1})$ with degree homogeneity $\gamma(\overline{\mathbf{v}}) \leq s + \varepsilon_{k}$. By the gap condition in the lower dimensional case \mathbb{R}^{n} , we get that necessary $\gamma(\overline{\mathbf{v}}) = s$, which it implies, going back to the function $\overline{\mathbf{w}}$, that

$$s \leq N(0, \overline{\mathbf{w}}, r) \leq N(0, \overline{\mathbf{w}}, +\infty) = s.$$

In other words, by the monotonicity result Proposition 1.3.9, the Almgren monotonicity formula satisfies $N(0, \overline{\mathbf{w}}, r) = s$ for every r > 0, i.e. up to a rotation $\overline{\mathbf{w}} = (\overline{w}_1, \overline{w}_2, 0, \dots, 0)$ where

$$\overline{w}_1(x,y) = C_1 \left(\frac{\sqrt{x_1^2 + y^2} - x_1}{2}\right)^s \quad \text{and} \quad \overline{w}_2(x,y) = C_1 \left(\frac{\sqrt{x_1^2 + y^2} + x_1}{2}\right)^s, \quad (45)$$

for some positive constant $C_1 > 0$ such that $\|\overline{\mathbf{w}}\|_{L^{2,a}(\partial^+ B_1^+)} = 1$. The contradiction follows immediately by (43).

We remark that in general, the latter statement is equivalent to the following one:

for every α - homogeneous $\overline{\mathbf{u}} \in \mathfrak{B}^{s}(\mathbb{R}^{n+1})$, either $\alpha = s$ or $\alpha \geq s + \delta_{n}$,

for some universal constant $\delta_n > 0$. Moreover, we can actually generalize the previous result by proving that the possible values of the Almgren frequency formula are a discrete subset of the interval [s, 2s).

Proposition 1.4.2. Given $s \in (0, 1)$, n = 2 and $u \in \mathcal{G}^s(B_1^+)$, then for $X_0 \in \Gamma(\mathbf{u})$ the possible values of the Almgren frequency formula $N(X_0, \mathbf{u}, 0^+)$ are a discrete set in [s, 2s) with 2s as point of accumulation.

Proof. The proof of the first part follows the one of Proposition 1.4.1 since it was based on a contradiction argument due to the a gap in dimension n = 1 for the possible values of the Almgren frequency formula. Hence, let us prove the result by induction on the Almgren frequency. Since the first step is prove in Proposition 1.4.1, let us consider the inductive step. Let $A \in (0, 1)$ be a possible value of the Almgren frequency and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$. By induction we already know that $A = B + \delta$, for some $\delta > 0$ and $B \in [s, 2s)$ lower Almgren frequency. By contradiction, suppose there exist two sequences $\varepsilon_k \searrow 0^+$ and $(X_k)_k \subset \Gamma(\mathbf{u}) \cap \partial^0 \Omega^+$, for some $\Omega^+ \subset B_1^+$, such that

$$N(X_k, \mathbf{u}, 0^+) \le A + \varepsilon_k.$$

with $\varepsilon_k \leq s - A/2$, in order to always have $A + \varepsilon_k < 2s$. Since $\Gamma(\mathbf{u})$ has empty interior in \mathbb{R}^{n+1} , up to a subsequence there exists $X_0 \in \Gamma(\mathbf{u})$ such that $X_k \to X_0$ and, by the induction hypothesis and Corollary 1.2.6 we get $N(X_0, \mathbf{u}, 0^+) = A$.

Therefore, let us construct a sequence of blow-up limit $(\overline{\mathbf{u}}_k)_k \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+)$ in order to translate the absurd hypothesis in the context of entire segregated configurations in $\mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+)$. Hence, following the details in Proposition 1.4.1, we can construct a family of blow-up limits $(\overline{\mathbf{u}}_k)_k \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ of homogenous function in \mathbb{R}^{n+1} such that $\mathbf{u}_{X_k,i} \to \overline{\mathbf{u}}_k$ and

$$\int_{\partial^+ B_1^+} |y|^a \,\overline{\mathbf{u}}_k^2 \mathrm{d}\sigma = 1 \quad \text{and} \quad N(0, \overline{\mathbf{u}}_k, r) = A + \varepsilon_k \quad \text{for every } r > 0, \tag{46}$$

namely $\overline{\mathbf{u}}_k$ is $(A + \varepsilon_k)$ -homogeneous in \mathbb{R}^{n+1} ., i.e.

$$\overline{\mathbf{u}}_k(X) = |X|^{A+\varepsilon_k} \mathbf{g}_k\left(\frac{X}{|X|}\right),$$

with $\mathbf{g}_k \in H^{1,a}(S^n_+)$. By (46) it follows that for every R > 0 the sequence $(\overline{\mathbf{u}}_k)_k \subset \mathcal{G}^s_{\mathrm{loc}}(\mathbb{R}^{n+1}) \cap C^{0,\alpha}_{\mathrm{loc}}(\overline{\mathbb{R}^{n+1}_+})$ with $\alpha \in (0,s)$ is uniformly bounded in $L^{\infty}(B^+_R)$, for every R > 0, which implies by Theorem 1.3.8 the existence of $\overline{\mathbf{u}} \in \mathcal{G}^s_{\mathrm{loc}}(\mathbb{R}^{n+1})$. Moreover, by the strong convergence in $H^{1,a}_{\mathrm{loc}}(\mathbb{R}^{n+1})$ it follows that $N(0, \overline{\mathbf{u}}, r) = A$ for every r > 0.

Since $\overline{\mathbf{u}} \in \mathfrak{B}^{s}(\mathbb{R}^{n+1})$, we get that $\Gamma(\overline{\mathbf{u}})$ is a conic set and its domain of positivity are totally defined by its spherical section $\mathbf{g} \in H^{1,a}(S^{n}_{+})$ (this is a generalisation of the previous case in Proposition 1.4.1, since in this case we do not know the explicit formulation of the blow-up limit $\overline{\mathbf{u}}$).

Now, let us prove the existence of an index i = 1, ..., h such that, given $\overline{\omega} = \{\theta \in S^{n-1} : g_i \equiv 0\}$, then for every k > 0 there exist a nonempty $\omega_k \subset \overline{\omega}$ such that ω_k is a connected component in S^{n-1} of $\{g_{i,k} > 0\}$.

By contradiction, suppose that for every index i = 1, ..., h there exists $\overline{k} > 0$ such that $g_{i,\overline{k}} \equiv 0$ on the zero set of g_i on S^{n-1} . Then, by uniqueness of the eigenvalue problem (37) we must obtain that $\lambda(\mathbf{g}_k) = \lambda(\mathbf{g})$, in contradiction with the definition of $\overline{\mathbf{u}}_k$ and its homogeneity.

Now, since we are working in dimension n = 2, there exist two sequences $(P_k)_k, (Q_k)_k \subset S^1$ such that $\omega_k = (P_k, Q_k)$ can be seen as an arc of S^1 between the endpoints P_k and Q_k . Since by compactness of S^1 there exist, up to a subsequence, $P, Q \in S^1$ respectively limit of $(P_k)_k$ and $(Q_k)_k$, let us consider separately the cases $|P_k - Q_k| \to 0$ and $P \neq Q$.

If $P \neq Q$, let $\omega = (P, Q)$ be the limit of the sequence $(\omega_k)_k$ and C_{ω} be the cone in Σ spanned by ω , i.e.

$$C_{\omega} = \left\{ X \in \Sigma \colon \frac{X}{|X|} \in \omega \subset S^1 \right\}.$$

One one hand, by definition of ω_k , since $u_{i,k} \cdot u_{j,k} = 0$ on Σ , we get $\overline{u}_{j,k} \equiv 0$ on $\omega_k \subseteq \overline{\omega}$ for every $\omega \neq i$ and then, passing to the limit for $k \to \infty$, $\overline{u}_j \equiv 0$ on ω , for every $j \neq i$. The contradiction follows from the fact that since, by definition $g_i \equiv 0$ on $\overline{\omega}$ and necessary must exist a component of **g** non identically zero on $\omega \subset \overline{\omega}$.

Hence in the other case, given $r_k = |P_k - Q_k| \searrow 0$ and $P_k \in \Sigma$ let introduce the blow-up sequence $(\overline{\mathbf{w}}_k)_k \subset \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ centered in $(P_k)_k \in \Gamma(\mathbf{u}_k) \cap S^{n-1}$ and associated to $r_k > 0$, i.e.

$$\overline{\mathbf{w}}_k(X) = \frac{\overline{\mathbf{u}}_k(P_k + r_k X)}{\rho_k} \quad \text{with } \rho_k = \sqrt{H(P_k, \overline{\mathbf{u}}_k, r_k)}$$

such that $\|\overline{\mathbf{w}}_k\|_{L^{2,a}(\partial^+B^+_1)} = 1$. By construction, for every $k \in \mathbb{N}$ we have

$$\overline{\mathbf{w}}_{k}(0) = \mathbf{0} = \overline{\mathbf{w}}_{k}\left(\frac{Q_{k} - P_{k}}{r_{k}}\right),\tag{47}$$

where $(Q_k - P_k)/r_k \to \nu \in S^{n-1} \times \{0\}$ by compactness in S^{n-1} . Now, following the same ideas of the proof of Proposition 1.4.1, by Proposition 1.3.8 there exists a blow-up limit $\overline{\mathbf{w}} \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1}) \cap C^{0,\alpha}_{\text{loc}}(\mathbb{R}^{n+1})$ for every $\alpha \in (0, s)$ such that $\overline{\mathbf{w}} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ with $\|\overline{\mathbf{w}}\|_{L^{2,a}(\partial^+B_1^+)} = 1$.

Since the blow-up sequence $(\overline{\mathbf{w}}_k)_k$ is constructed starting from a family of homogeneous entire segregated profiles in $\mathfrak{B}^s_{\text{loc}}(\mathbb{R}^{n+1})$, we can prove that $\overline{\mathbf{w}}$ is constant along the direction parallel to $P \in S^{n-1}$ and that its restriction on the orthogonal half plane belongs to $\mathcal{G}^s_{\text{loc}}(\mathbb{R}^n)$.

Hence, with slight abuse of notation we still denote the *n*-dimensional section by $\overline{\mathbf{w}} \in \mathcal{G}_{\text{loc}}^s(\mathbb{R}^n)$. By Corollary 1.3.10, since for every k > 0 and $t \in (0, +\infty)$

$$N(P_k, \overline{\mathbf{u}}_k, t) \le N(P_k, \overline{\mathbf{u}}_k, +\infty)$$

= $N(0, \overline{\mathbf{u}}_k, +\infty) = A + \varepsilon_k$

we get from Proposition 1.2.4 and Proposition 1.3.9 that for every R > 0

$$N(0, \overline{\mathbf{w}}, R) = \lim_{r_k \to 0} N(0, \overline{\mathbf{w}}_k, R) = \lim_{r_k \to 0} N(P_k, \overline{\mathbf{u}}_k, r_k R) \le A + \varepsilon_k.$$
(48)

Finally, as in Proposition 1.4.1, we reach the contradiction applying a blow-down analysis on the limit function $\overline{\mathbf{w}} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^n)$. Indeed, applying the same procedure, we get by the gap condition in the lower dimensional case \mathbb{R}^n that

$$s \le N(0, \overline{\mathbf{w}}, r) \le N(0, \overline{\mathbf{w}}, +\infty) = s.$$

In other words, by the monotonicity result Proposition 1.3.9, the Almgren monotonicity formula satisfies $N(0, \overline{\mathbf{w}}, r) = s$ for every r > 0, i.e. up to a rotation $\overline{\mathbf{w}} = (\overline{w}_1, \overline{w}_2, 0, \dots, 0)$ where

$$\overline{w}_1(x,y) = C_1 \left(\frac{\sqrt{x_1^2 + y^2} - x_1}{2}\right)^s \quad \text{and} \quad \overline{w}_2(x,y) = C_1 \left(\frac{\sqrt{x_1^2 + y^2} + x_1}{2}\right)^s, \quad (49)$$

for some positive constant $C_1 > 0$ such that $\|\overline{\mathbf{w}}\|_{L^{2,a}(\partial^+ B_1^+)} = 1$. The contradiction follows immediately by (47). The second part of the proof is a direct consequence of the results in [85]. In this paper, the authors proved in the case n = 2 the existence of some segregated profiles possessing some natural symmetry. Such solutions are constructed as limit of a competitiondiffusion problem and starting from the eigenvalue problem (39) and in particular they proved that they have growth rate at infinity which is arbitrarily close to the critical one, that is, 2s. \Box

1.5 REGULARITY AND FLATNESS OF THE REGULAR SET

In this Section, we will provide an estimate of the Hausdorff dimension of the whole nodal set

$$\Gamma(\mathbf{u}) = \left\{ X \in \partial^0 B_1^+ \colon \mathbf{u}(X) = \mathbf{0} \right\},\,$$

and, regarding its regularity, we will split $\Gamma(\mathbf{u})$ in two parts:

- the singular part $S(\mathbf{u})$, which we will show to be a local finite collection point in $\partial^0 B_1^+$;
- the regular part $\mathcal{R}(\mathbf{u})$, which is relatively open in $\Gamma(\mathbf{u})$ and satisfies a flatness type condition.

Hence, let us start introducing the notion of regular and singular set in the planar case.

Definition 1.5.1. Given $s \in (0, 1)$, n = 2 and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$, we define its regular and singular sets respectively as

$$\mathcal{R}(\mathbf{u}) = \left\{ X_0 \in \Gamma(\mathbf{u}) \colon N(X_0, \mathbf{u}, 0^+) = s \right\} \quad \text{and} \quad \mathcal{S}(\mathbf{u}) = \left\{ X_0 \in \Gamma(\mathbf{u}) \colon N(X_0, \mathbf{u}, 0^+) > s \right\}.$$

Corollary 1.5.2. For $s \in (0,1)$ and n = 2 the set $\mathcal{R}(\mathbf{u})$ is relatively open in $\Gamma(\mathbf{u})$ and $\mathcal{S}(\mathbf{u})$ is closed in $\partial^0 B_1^+$, whenever $\mathbf{u} \in \mathcal{G}^s(B_1^+)$.

Proof. This result is a direct consequence of Proposition 1.4.1 together with the upper semicontinuity of the Almgren frequency function $X \mapsto N(X, \mathbf{u}, 0^+)$ stated in Corollary 1.2.6. \Box

Next we state and prove some estimates regarding the Hausdorff dimensions of two strata previously defined by using the version of the Federer's Reduction principle in [78]. Hence, let us take a class of functions \mathcal{F} invariant under rescaling and translation and \mathcal{S} a map which associate to each function $\Phi \in \mathcal{S}$ a subset of \mathbb{R}^{n+1} . Thus, this principle establishes conditions on \mathcal{F} and \mathcal{S} which imply that to control the Hausdorff dimension of $\mathcal{S}(\Phi)$ for every $\Phi \in \mathcal{S}$, we just need to control the Hausdorff dimension of $\mathcal{S}(\Phi)$ for elements which are homogeneous of some degree. In Chapter 3, we will state the Federer's Reduction principle in its most general version. **Theorem 1.5.3.** Given $s \in (0,1)$ and n = 2, let $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ and $\Gamma(\mathbf{u})$ be its nodal set. Then if $\Gamma(\mathbf{u}) \neq \Sigma$, then $\dim_{\mathcal{H}}(\Gamma(\mathbf{u})) \leq 1$ and

$$\dim_{\mathcal{H}}(\mathcal{R}(\mathbf{u})) = 1 \quad \dim_{\mathcal{H}}(\mathcal{S}(\mathbf{u})) = 0.$$

Moreover for any given compact $K \subset \subset \partial^0 B_1^+$ we that $\mathcal{S}(\mathbf{u}) \cap K$ is a finite set.

Proof. A preliminary remark is that we only need to prove the Hausdorff dimensional estimates for the localization of the sets in $K \subset B_1^+$, since the general statement follows because a countable union of sets with Hausdorff dimension less than or equal to some $n \in \mathbb{R}_0^+$ also has Hausdorff dimension less than or equal to n. Moreover, since the Hausdorff dimension of the nodal set of a function is invariant under rescaling, in order to simplify the notations we claim that

 $\dim_{\mathcal{H}}(\Gamma(\mathbf{u}) \cap \partial^0 B_1^+) \le 1 \quad \dim_{\mathcal{H}}(\mathcal{S}(\mathbf{u}) \cap \partial^0 B_1^+) = 0.$

Let us consider the class of functions ${\mathcal F}$ defined as

$$\mathcal{F} = \left\{ \mathbf{u} \in \left(L^{\infty}_{\text{loc}}(\mathbb{R}^{2+1}_+) \right)^h : \mathbf{u} \in \mathcal{G}^s(B^+_r(X_0)), \text{ for } r \in \mathbb{R}, X_0 \in \Sigma \text{ such that } B^+_r(X_0) \subset B^+_1 \right\}.$$

By the linearity of the L_a operator, we already know that the closure under rescaling, translation and normalization and assumption (F1) are all satisfied.

On the other hand, let $X_0 \in \partial^0 B_1^+$, $r_k \downarrow 0^+$ and $u \in \mathcal{F}$, and choose $\rho_k = \|\mathbf{u}(X_0 + r_k \cdot x)\|_{L^{2,a}(\partial^+ B_1^+)}$. Theorem 1.3.3 and Proposition 1.3.11 yield the existence of a blow-up limit $\overline{\mathbf{u}} \in \mathcal{F}$, i.e. up to a subsequence $\mathbf{u}_k \to \overline{\mathbf{u}}$ in \mathcal{F} and $\overline{\mathbf{u}}$ is a homogeneous entire segregated profile of degree $k = N(X_0, \mathbf{u}, 0+) \ge s$. Hence also (F2) holds.

Next we choose the map \overline{S} in (F3) according to our needs.

1. Dimensional estimate of the nodal set $\Gamma(\mathbf{u})$

First, let us consider \overline{S} : $\mathbf{u} \mapsto \Gamma(\mathbf{u})$. By the continuity of \mathbf{u} , we already know that the set $\Gamma(\mathbf{u}) \cap B_1^+$ is obviously closed in B_1^+ and it is quite straightforward to check the two hypotheses in (F3).

Hence, in order to conclude the analysis, the only thing left to prove is that the integer d in (131) is equal to 1. Suppose by contradiction that d = 2, then this would imply the existence of $\mathbf{v} \in \mathcal{F}$ with $\mathcal{S}(\mathbf{v}) = \mathbb{R}^2$ i.e., $\mathbf{v} \equiv 0$ on Σ , which contradicts the definition of \mathcal{G}^s .

Actually, taking $V = \mathbb{R}^1 \times \{(0,0)\}$ and $\overline{\mathbf{v}} = (\overline{v}_1, \overline{v}_2, 0, \dots, 0)$ where

$$\overline{v}_1(x_1, x_2, y) = \left(\frac{\sqrt{x_1^2 + y^2} - x_1}{2}\right)^s$$
 and $\overline{v}_2(x_1, x_2, y) = \left(\frac{\sqrt{x_1^2 + y^2} + x_1}{2}\right)^s$,

we obtain the claimed estimate on d.

2. Dimensional estimate of the regular set $\mathcal{R}(\mathbf{u})$

Let us consider $\overline{\mathcal{S}}$: $\mathbf{u} \mapsto \mathcal{R}(\mathbf{u})$. By the inclusion in $\Gamma(\mathbf{u})$, we already know that

$$\dim_{\mathcal{H}}(\mathcal{R}(\mathbf{u}) \cap \partial^0 B_1^+) \le 1.$$

Finally, we can apply the Reduction principle since (F3) is completely satisfied. More precisely, for $X_0 \in \partial^0 B_1^+$, $\rho > 0$ and t > 0 if $X \in \mathcal{R}(\rho \mathbf{u}_{X_0,t})$ then obviously $X_0 + tX \in \mathcal{R}(\mathbf{u})$, i.e. $N(X_0 + tX, \mathbf{u}, 0^+) = 1$. Secondly, given $\mathbf{u}_k, \overline{\mathbf{u}} \in \mathcal{F}$ as in (F3), suppose by contradiction that there exists a sequence $X_i \in \partial^0 B_1^+$ and $\overline{\varepsilon} > 0$ such that

$$N(X_k, \mathbf{u}_k, 0^+) = s$$

and dist $(X_k, \overline{S}(\overline{\mathbf{u}})) \geq \overline{\varepsilon}$. Since, up to a subsequence, $X_k \to \overline{X}$, by the upper semi-continuity of the Almgren frequency formula, we already know that $N(\overline{X}, \overline{\mathbf{u}}, 0^+) \geq 1$. Moreover, up to a subsequence, $X_k \to \overline{X} \in \Gamma(\overline{\mathbf{u}}) \cap \Sigma \cap \overline{B_1}$ by the L^{∞}_{loc} convergence of $u_i \to \overline{u}$.

Now, since $\Gamma(\overline{\mathbf{u}})$ is a conical set, i.e. for every $\lambda > 0$ and $\overline{X} \in \Gamma(\overline{\mathbf{u}})$ we have $\lambda \overline{X} \in \Gamma(\overline{\mathbf{u}})$, we deduce that if we can prove $\overline{X} \in \mathcal{R}(\overline{\mathbf{u}}) \cap \overline{\partial^0 B_1^+}$ we provide a contradiction, more precisely we get dist $(\overline{X}, \overline{S}(\overline{\mathbf{u}}) \cap \partial^0 B_1^+) = 0$. Since there exists $\Omega \subset \subset \partial^0 B_1^+$ such that $(X_0 + r_i X_i)_i \subset \Omega$, if we consider

$$R_{1} = \min_{p \in \overline{\partial^{0} \Omega^{+}}} \operatorname{dist}(p, S^{1} \times \{0\}),$$
$$\overline{C} = \sup_{p \in \overline{\partial^{0} \Omega^{+}}} N(p, \mathbf{u}, R_{1}),$$

we easily get from Corollary 1.2.7 that there exists $\overline{C} > 0$ and $\overline{r} > 0$ such that for $p \in \Omega \cap \mathcal{R}(u)$ and $r < \min\{\overline{r}, R_1\}$ we have

$$N(p, \mathbf{u}, r) \le N(p, \mathbf{u}, R_1) \left(\frac{R_1}{r}\right)^{2+a-1+2\overline{C}} \le \overline{C} \frac{1}{r^{2+a-1+2\overline{C}}}$$

In particular, from the previous inequality we get that there exists $\overline{R} = \overline{R}(a) > 0$ sufficiently small, such that for $r < \overline{R}$ we have

$$s \le N(X_k, \mathbf{u}_k, r) \le s + \frac{\delta}{2}.$$

Since $\lim_k N(X_k, \mathbf{u}_k, r) = N(\overline{X}, \overline{\mathbf{u}}, r)$ for sufficiently small r, we directly obtain from Proposition 1.4.1 that $N(\overline{X}, \overline{\mathbf{u}}, 0^+) = s$, as we claimed.

As before, let us suppose now that there exist $\overline{\mathbf{v}} \in \mathcal{F}$ and a *d*-dimensional subspace $V \subset \Sigma$, with $d \leq 1$, and $k \geq 0$ such that

$$\overline{\mathbf{v}}_{Y,r} = r^k \overline{\mathbf{v}} \ \text{ for all } \ Y \in V, r > 0 \ \text{ and } \ \mathcal{R}(\varphi) \cap \partial^0 B_1^+ = V \cap \partial^0 B_1^+$$

Since $\overline{\mathbf{v}} \in \mathfrak{B}^{s}(\mathbb{R}^{2+1})$ is homogenous of degree k with respect to any $Y \in V = \mathcal{R}(\varphi)$, namely $N(Y, \overline{\mathbf{v}}, 0^{+}) = k$, we get that necessary k = s and that $\mathcal{R}(\varphi)$ is d-dimensional. Since the only entire segregated profiles with degree s is, up to rotation, of the form $\overline{\mathbf{v}} = (\overline{v}_1, \overline{v}_2, 0, \dots, 0)$ where

$$\overline{v}_1(x_1, x_2, y) = \left(\frac{\sqrt{x_1^2 + y^2} - x_1}{2}\right)^s$$
 and $\overline{v}_2(x_1, x_2, y) = \left(\frac{\sqrt{x_1^2 + y^2} + x_1}{2}\right)^s$,

we get that $\mathcal{R}(\varphi)$ must be 1-dimensional, and consequently that

$$\dim_{\mathcal{H}}(\mathcal{R}(u) \cap \partial^0 B_1^+) = 1.$$

3. Dimensional estimate of the singular set $\mathcal{S}(\mathbf{u})$

Let us focus on the singular strata, namely given \overline{S} : $\mathbf{u} \mapsto S(\mathbf{u})$, the map satisfies the first part of (F3) thanks to (18), since for $X_0 \in \partial^0 B_1^+$, $\rho > 0$ and t > 0, if $X \in \overline{S}(\rho \mathbf{u}_{X_0,t})$ we get

$$N(X, \rho \mathbf{u}_{X_0, t}, 0^+) > s \longleftrightarrow N(X_0 + tX, \mathbf{u}, 0^+) > s,$$

which is equivalent to $X_0 + tX \in \mathcal{S}(\mathbf{u})$. Now, given $\mathbf{u}_k = \rho_k \mathbf{u}_{X_0, r_k}, \overline{\mathbf{u}} \in \mathcal{F}$ as in (F3), suppose by contradiction that there exists a sequence $X_k \in \partial^0 B_1^+$ and $\overline{\varepsilon} > 0$ such that, up to a subsequence, $X_k \to \overline{X}$ and

$$N(X_k, \mathbf{u}_k, 0^+) \ge s + \delta \tag{50}$$

and dist $(X_k, \overline{\mathcal{S}}(\overline{\mathbf{u}})) \geq \overline{\varepsilon}$. Then, following the same reasoning in Corollary 1.2.6, by Proposition 1.2.4 there exists a constant C > 0 and $\overline{r} > 0$ such that, for $r \in (0, \overline{r})$

$$N(X_k, \mathbf{u}, r) = e^{-C\Psi(r)} e^{C\Psi(r)} \left(N(X_k, \mathbf{u}, r) \right) - 1 > e^{-C\Psi(r)} (s + 1 + \delta) - 1$$

and hence, since for $r \in (0, \overline{r})$ it holds $N(X_k, \mathbf{u}_k, r) \to N(\overline{X}, \overline{\mathbf{u}}, r) \to$, we get $N(\overline{X}, \overline{\mathbf{u}}, 0^+) \ge s + \delta$, which implies a contradiction.

Since $S(\mathbf{u}) \subseteq \Gamma(\mathbf{u})$, we already know that

$$\dim_{\mathcal{H}}(\mathcal{S}(\mathbf{u}) \cap \partial^0 B_1^+) \le 1,\tag{51}$$

which is not the optimal bound for the singular set. Indeed, suppose d = 1, then must exist $\mathbf{v} \in \mathcal{F}$ homogeneous with respect to every point in $\mathbb{R} \times \{(0,0)\}$, i.e. there exists k > 0 such that

$$\mathbf{v}(Y + \lambda X) = \lambda^k \mathbf{v}(X) \text{ for all } Y \in \mathbb{R} \times \{(0,0)\}, X \in \mathbb{R}^{2+1}_+$$

such that $S(\mathbf{v}) = \mathbb{R} \times \{(0,0)\}$. Hence, given $Y_0 \in S(\mathbf{v})$, we get for every $\mu > 0$ that $\mathbf{v}(\mu Y_0 + X) = \mathbf{v}(X)$, which implies that $\mathbf{v} \in \mathfrak{B}^s(\mathbb{R}^{1+1}_+)$ with $N(0, \mathbf{v}, 0^+) = k > s$ and $S(\mathbf{v}) = 0$. The absurd follows from the fact that necessary $N(0, \mathbf{v}, 0^+) = 2s$ and hence $\mathbf{v} \equiv 0$ on Σ , in contradiction with the definition of \mathcal{F} and the upper bound (51).

At this point, combining Corollary 1.3.12 and Corollary 1.5.2 we can state the following results about the flatness of the regular part $\mathcal{R}(\mathbf{u})$ of the nodal set $\Gamma(\mathbf{u})$. More precisely, the following result prove that $\mathcal{R}(\mathbf{u})$ verifies the so called vanishing Reifenberg flat condition, i.e. the (δ, R) -Reifenberg flat condition for every $\delta \in (0, 1)$ and some $R = R(\delta) > 0$.

Proposition 1.5.4. Given $s \in (0,1)$ and n = 2 consider $\mathbf{u} \in \mathcal{G}^s(B_1^+)$. Then, fixed $\Omega^+ \subset B_1^+$, for any given $\delta \in (0,1)$ there exists R > 0 such that $X \in \mathcal{R}(\mathbf{u}) \cap \partial^0 \Omega^+$ and 0 < r < R there exists an hyper-plane $H = H_{X,r}$ passing through X such that

$$d_{\mathcal{H}}(\Gamma(\mathbf{u}) \cap \partial^0 B_r^+(X), H_{X,r} \cap \partial^0 B_r^+(X)) \leq \delta r,$$

where $d_{\mathcal{H}}$ is the Hausdorff distance defined in (33).

The idea of the proof is similar to the one in Corollary 1.3.12 since, roughly speaking, they are both a consequence of Proposition 1.3.11. Indeed, by Theorem 1.3.3 we already know the topology in which the blow-up sequence converges and Proposition 1.3.11 ensure the existence of an homogeneous blow-up limit for a specific choices of X_k , which are the ones considered in Proposition 1.3.12 and in this result.

Proof of Proposition 1.5.4. Arguing by contradiction, let us suppose there exists $\overline{\delta} > 0$ and a sequence $(X_k)_k \subset \mathcal{R}(\mathbf{u}) \cap \partial^0 \Omega^+, r_k \to 0$ such that

$$d_{\mathcal{H}}(\Gamma(\mathbf{u}) \cap \partial^0 B^+_{r_k}(X_k), H \cap \partial^0 B^+_{r_k}(X_k)) > \overline{\delta}r_k.$$

for every hyper-plane H passing through X_k . If we consider the blow-up sequence centered in $(X_k)_k$ associated to r_k , i.e. let

$$\mathbf{u}_k(X) = \frac{\mathbf{u}(X_k + r_k X)}{\sqrt{H(X_k, \mathbf{u}, r_k)}},$$

then "rescaling" the contradiction statement we get

$$d_{\mathcal{H}}(\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+, H \cap \partial^0 B_1^+) > \overline{\delta},$$

whenever *H* is a hyper-plane that passes through the origin. Hence, since up to a subsequence $X_k \to \overline{X} \in \Gamma(\mathbf{u}) \in \partial^0 \Omega^+$, Theorem 1.3.3 together with property Proposition 1.3.11 implies the existence of a blow-up limit $\overline{\mathbf{u}}$ whose nodal set $\Gamma(\overline{\mathbf{u}})$ is a hyper-plane containing the origin. Hence we obtain a contradiction once we are able to prove that

$$d_{\mathcal{H}}(\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+, \Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_1^+) \to 0.$$

Equivalently, the claimed result is to prove that for every $\varepsilon > 0$ there exists $\overline{k} > 0$ such that for every $k \ge \overline{k}$

$$\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+ \subseteq N_{\varepsilon} \left(\Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_1^+ \right)$$

$$\Gamma(\overline{\mathbf{u}}) \cap \partial^0 B \subseteq N_{\varepsilon} \left(\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+ \right).$$

Supposing by contradiction that the first inclusion is not true, then there exist $\overline{\varepsilon} > 0$ and a sequence $X_k \in \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+$ such that dist $(X_k, \Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_1^+) > \overline{\varepsilon}$. Moreover, up to a subsequence, $X_k \to \overline{X} \in \Gamma(\overline{\mathbf{u}}) \cap \overline{\partial^0 B_1^+}$ by the L_{loc}^∞ convergence of $\mathbf{u}_k \to \overline{\mathbf{u}}$. Since by Proposition 1.3.11 the nodal set $\Gamma(\overline{\mathbf{u}})$ is a conical set, i.e. for every $\lambda > 0$ and $\overline{X} \in \Gamma(\overline{\mathbf{u}})$ we have $\lambda \overline{X} \in \Gamma(\overline{\mathbf{u}})$ and $0 \in \Gamma(\overline{\mathbf{u}})$, we deduce that $\text{dist}(\overline{X}, \Gamma(\overline{\mathbf{u}}) \cap \partial^0 B_1^+) = 0$, which provides the contradiction.

Finally, we have to prove that for every $\varepsilon > 0$ there exists $\bar{k} > 0$ such that

$$\Gamma(\overline{\mathbf{u}}) \cap \partial^0 B \subseteq N_{\varepsilon} \left(\Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+ \right) \quad \text{for every } k \ge \overline{k}.$$

We start by proving that given $\overline{X} \in \Gamma(\overline{\mathbf{u}}) \in \partial^0 B_1^+$ and $\delta > 0$, the vector valued function \mathbf{u}_k must have a zero in $\partial^0 B_{\delta}^+(\overline{X})$, for k sufficiently large. If not, by recalling that $u_{i,k} \cdot u_{j,k}|_{y=0} \equiv 0$ for every $i \neq j$, we would have that there exists an index 0 < i < h such that

$$\begin{cases} -L_a u_{i,k} = 0 & \text{in } B^+_{\delta}(\overline{X}) \\ -\partial^a_y u_{i,k} = f_i(u_{i,k}) & \text{on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}, \quad u_{i,k} > 0 \text{ on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}$$

and $u_{j,k} \equiv 0$ in $\partial^0 B^+_{\delta}(\overline{X})$, for every $j \neq i$. Passing to the limit, this would imply that

$$\begin{cases} -L_a \overline{u}_i = 0 & \text{in } B^+_{\delta}(\overline{X}) \\ -\partial^a_y \overline{u}_i = 0 & \text{on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}, \quad \overline{u}_i \ge 0 \text{ on } \partial^0 B^+_{\delta}(\overline{X}) \end{cases}$$

and $\overline{u}_j \equiv 0$ on $\partial^0 B^+_{\delta}(\overline{X})$ for every $j \neq i$. Since $\overline{X} \in \Gamma(\overline{\mathbf{u}})$ it follows from the Hopf principle (see [13, 43]) that $\overline{\mathbf{u}} \equiv 0$ in $\partial^0 B^+_{\delta}(\overline{X})$, a contradiction with the fact the $\Gamma(\overline{\mathbf{u}})$ has empty interior.

Now, arguing by contradiction, suppose the existence of $\bar{\varepsilon} > 0$ and $(X_k)_k \in \Gamma(\bar{\mathbf{u}}) \cap \partial^0 B_1^+$ such that $X_k \to \overline{X} \in \Gamma(\bar{\mathbf{u}}) \cap \partial^0 B_1^+$ and $\operatorname{dist}(X_k, \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+) > \bar{\varepsilon}$. Since $\Gamma(\bar{\mathbf{u}})$ is a conical set passing through the origin, let us take $\widetilde{X} \in \Gamma(\bar{\mathbf{u}}) \cap \partial^0 B_1^+$ such that $|\widetilde{X} - \overline{X}| \leq \bar{\varepsilon}/4$. Furthermore, we can take, by using the result proved in the previous paragraph, a sequence $(\overline{X}_k)_k \in \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+$ such that $\left|\overline{X}_k - \widetilde{X}\right| \leq \overline{\varepsilon}/4$ for sufficiently large k. The final contradiction follows noticing that

$$\operatorname{dist}(X_k, \Gamma(\mathbf{u}_k) \cap \partial^0 B_1^+) \le \left| X_k - \overline{X}_k \right| \le \left| X_k - \overline{X} \right| + \left| \overline{X} - \widetilde{X} \right| + \left| \widetilde{X} - \overline{X}_k \right| \le \frac{3\overline{\varepsilon}}{4} \le \overline{\varepsilon},$$

for sufficiently large k.

With the vanishing Reifenberg property we are able to prove a local separation result. We remark that the following result follows the idea of Proposition 5.4 in [81].

Proposition 1.5.5. Given $s \in (0, 1)$ and n = 2 consider $\mathbf{u} \in \mathcal{G}^s(B_1^+)$. Then, given $X_0 \in \Gamma^*$ there exists a radius $R_0 > 0$ such that $\partial^0 B_{R_0}^+(X_0) \cap \mathcal{R}(\mathbf{u}) = \partial^0 B_{R_0}^+(X_0) \cap \Gamma(\mathbf{u})$ and $\partial^0 B_{R_0}^+(X_0) \setminus \Gamma(\mathbf{u}) = \partial^0 B_{R_0}^+(X_0) \cap \{\mathbf{u} > 0\}$ has exactly two connected components, i.e.

$$\partial^0 B^+_{R_0}(X_0) \cap \mathcal{R}(\mathbf{u}) = \Omega_+ \cup \Omega_-$$

More precisely, there exists $\delta > 0$ such that, given $Y \in \Gamma(\mathbf{u}) \in \partial^0 B^+_R(X_0)$ and $r \in (0, R_0 - |Y - X_0|)$ there exists a hyperplane $H_{Y,r}$ and a vector $\nu_{Y,r} \in S^{n-1}$ orthogonal to $H_{Y,r}$ such that

$$\{X \pm t\nu_{Y,r} \in \partial^0 B_r^+(Y) : X \in H_{Y,r}, t \ge \delta r\} \subset \Omega_{\pm}$$

Proof. Fixed $X_0 \in \mathcal{R}(\mathbf{u})$, since $\mathcal{R}(\mathbf{u})$ is a relatively open set in $\Gamma(\mathbf{u})$, there exists τ such that $\partial^0 B_{2\tau}^+(X_0) \cap \mathcal{R}(\mathbf{u}) = \partial^0 B_{2\tau}^+(X_0) \cap \Gamma(\mathbf{u})$ and fix $\delta < 1/6$. Using the notations of Proposition 1.5.4, for $\Omega^+ = B_{\tau}^+(X_0)$ there exists R > 0 such that $\Gamma(\mathbf{u}) \cap \partial^0 B_{\tau}^+(X_0)$ satisfies the (δ, R) -Reifenberg flatness condition. Let us prove our result with $R_0 := \min\{R, \tau\}$.

By Proposition 1.5.4, there exists a hyperplane H_{X_0,R_0} containing X_0 and such that

$$\mathbf{d}_{\mathcal{H}}(\Gamma(\mathbf{u}) \cap \partial^0 B^+_{R_0}(X_0), H_{X_0, R_0} \cap \partial^0 B^+_{R_0}(X_0)) \le \delta R_0.$$
(52)

Hence the subset $\partial^0 B^+_{R_0}(X_0) \setminus N_{2\delta R_0}(H_{X_0,R_0})$ has exactly two connected components, namely D_1 and D_2 , which do not intersect the nodal set $\Gamma(\mathbf{u})$. Hence, let us define the function

$$\sigma(X) = \begin{cases} 1 & \text{if } X \in D_1, \\ -1 & \text{if } X \in D_2. \end{cases}$$

Take now a point $X_1 \in \Gamma(\mathbf{u}) \cap \partial^0 B_{R_0}^+(X_0) \subseteq N_{\delta R_0}(H_{X_0,R_0}) \cap \partial^0 B_{R_0}^+(X_0)$. As we did before, by using Proposition 1.5.4, considering a ball of radius $R_0/2$ centered at X_1 there exists a hyperplane $H_{X_1,R_0/2}$ such that

$$d_{\mathcal{H}}(\Gamma(\mathbf{u}) \cap \partial^0 B^+_{R_0/2}(X_1), H_{X_1,R_0/2} \cap \partial^0 B^+_{R_0/2}(X_1)) \le \delta \frac{R_0}{2}.$$

This inequality combined with (52) yields that

$$N_{\delta R_0/2}(H_{X_1,R_0/2}) \cap \partial^0 B^+_{R_0/2}(X_1) \cap \partial^0 B^+_{R_0}(X_0) \subseteq N_{2\delta R_0}(H_{X_0,R_0}) \cap \partial^0 B^+_{R_0}(X_0).$$

Hence $\partial^0 B^+_{R_0}(X_0) \cap \partial^0 B^+_{R_0/2}(X_1) \setminus N_{\delta R_0}(H_{X_1,R_0/2})$ has exactly two connected components, each one intersecting D_1 or D_2 and not both. Thus the set

$$\left(\bigcup_{X_1\in\Gamma(\mathbf{u})\cap\partial^0 B_{R_0}^+(X_0)}\partial^0 B_{R_0}^+(X_0)\cap\partial^0 B_{R_0/2}^+(X_1)\setminus N_{\delta R_0}(H_{X_1,R_0/2})\right)\cup D_1\cup D_2$$
(53)

has exactly two connected components which do not interest the nodal set $\Gamma(\mathbf{u})$ and hence we can continuously extend σ to this set.

Iterating this argument to a ball of radius $R_0/2^k$ centered at a point of $\Gamma(\mathbf{u})$ we find two connected and disjoint set Ω_+, Ω_- such that $\partial^0 B^+_{R_0}(X_0) \setminus \Gamma(\mathbf{u}) = \Omega_+ \cup \Omega_-$, with $D_1 \subseteq \Omega_+$ and $D_2 \subseteq \Omega_-$. Furthermore, the map $\sigma : \partial^0 B^+_1 \setminus \Gamma(\mathbf{u}) \to \{+1, -1\}$ such that

$$\sigma \colon X \mapsto \chi_{\Omega_+}(X) - \chi_{\Omega_-}(X)$$

is continuous and thus $\partial^0 B^+_{R_0}(X_0) \setminus \Gamma(\mathbf{u})$ has exactly two connected components.

In order to check the continuity, take $X \in \partial^0 B^+_{R_0}(X_0)$ such that $\operatorname{dist}(X, \Gamma(\mathbf{u}) \cap \partial^0 B^+_{R_0}(X_0)) = \gamma > 0$, with $\overline{X} \in \Gamma(\mathbf{u}) \cap \partial^0 B^+_{R_0}(X_0)$ the point of minimum distance, and k so large that $R_0/2^{k+1} \leq \gamma < R_0/2^k$; then $X \in B_{R_0/2^k}(\overline{X}) \setminus N_{\delta R_0/2^k}(H_{\overline{X}, R_0/2^{k-1}})$, due to the choice of δ , and hence σ is constant in a small neighborhood of X. \Box

1.6 SINGULAR LIMIT OF A COMPETITION-DIFFUSION PROBLEM

In this last Section we consider the class of segregated profiles arising from a competitiondiffusion problem with nonlocal diffusion and variational competition. In particular, we motivate our definition of the class $\mathcal{G}^s(B_1^+)$ and we give some result in the context of self-segregation, comparing them with the ones known for the local case.

As we mentioned in the introduction, in the papers [83, 84] the authors proved that given $\beta > 0$ and $(f_{i,\beta})_{\beta}$ a collection of continuous functions uniformly bounded with respect to β on bounded sets, the sequence $(\mathbf{u}_{\beta})_{\beta} \in H^{1,a}(B_1^+; \mathbb{R}^h)$ of solutions $\mathbf{u}_{\beta} = (u_{1,\beta}, \ldots, u_{h,\beta})$ of the problems

$$\begin{cases} -L_a u_{i,\beta} = 0 & \text{in } B_1^+ \\ -\partial_y^a u_{i,\beta} = f_{i,\beta}(u_{i,\beta}) - \beta u_{i,\beta} \sum_{j \neq i} a_{ij} u_{j,\beta}^2 & \text{on } \partial^0 B_1^+. \end{cases}$$

uniformly bounded in $L^{\infty}(B_1^+)$ with respect to β , does converge uniformly on compact sets and strongly in $H^{1,a}(K \cap B_1^+)$, for every $K \subset B_1$, to a vector valued function $\mathbf{u} = (u_1, \ldots, u_h) \in$ $H^{1,a}_{\text{loc}}(B_1^+; \mathbb{R}^h)$ whose components are all non negative, continuous functions such that

- (1) $\mathbf{u} \in H^{1,a}(K \cap B_1^+) \cap C^{0,\alpha}(\overline{K \cap B_1^+})$, for every compact set $K \subset B$ and every $\alpha \in (0, \alpha^*)$;
- (2) $u_i \cdot u_j|_{y=0} \equiv 0$ for every $i \neq j$ and $\mathbf{u} \neq \mathbf{0}$ on $B_1 \cap \Sigma$. Moreover, for $i = 1, \ldots, h$ it satisfies

$$\begin{cases} -L_a u_i = 0 & \text{in } B_1^+ \\ u_i \left(\partial_y^a u_i + f_i(u_i) \right) = 0 & \text{on } \partial^0 B_1^+ \end{cases}$$

where $f_i: \mathbb{R}^+ \to \mathbb{R}$ are the nonnegative \mathcal{C}^1 limits of $f_{i,\beta}$, such that $f_i(s) = O(s)$ for $s \to 0$;

(3) for every $X_0 = (x_0, 0) \in \partial^0 B_1^+$ and $r \in (0, \text{dist}(X_0, \partial B))$, the following Pohožaev type identity holds

$$(1 - a - n) \int_{B_r^+(X_0)} |y|^a |\nabla \mathbf{u}|^2 dX + r \int_{\partial B_r^+(X_0)} |y|^a |\nabla \mathbf{u}|^2 d\sigma + \\ + 2n \int_{\partial^0 B_r^+(X_0)} \sum_{i=1}^h F_i(u_i) dx - 2r \int_{S_r^{n-1}(X_0)} \sum_{i=1}^h F_i(u_i) dx = 2r \int_{\partial^+ B_r^+(X_0)} |y|^a (\partial_r \mathbf{u})^2 d\sigma$$

where $\mathbf{F}(s) = (F_1(s), \dots, F_h(s))$ with $F_i(s) = \int_0^s f_i(t) dt$ for every $i = 1, \dots, h$.

In particular, they proved that for s = 1/2 the limit profile are $C_{\text{loc}}^{0,1/2}(B_1^+)$ while in general they estimate the solutions in the Hölder spaces by the use of a fractional versions of the Alt-Caffarelli-Friedman and Almgren monotonicity formula. More precisely, let us recall the fractional version of the spectral problem beyond the Alt-Caffarelli-Friedman formula used in [84, 83]. Consider the set of 2-partitions of S^{n-1} as

$$\mathcal{P}^2 := \left\{ (\omega_1, \omega_2) : \omega_i \subseteq S^{n-1} \text{ open, } \omega_1 \cap \omega_2 = \emptyset, \ \overline{\omega_1} \cup \overline{\omega_2} = S^{n-1} \right\},$$

the optimal partition value ν_s^{ACF} is defined as

$$\nu_s^{ACF} := \frac{1}{2} \inf_{(\omega_1, \omega_2) \in \mathcal{P}^2} \sum_{i=1}^2 \gamma_s(\lambda_1^s(\omega_i)).$$
(54)

It is easy to see, by a Schwarz symmetrization argument, that ν_s^{ACF} is achieved by a pair of complementary spherical caps $(\omega_{\theta}, \omega_{\pi-\theta}) \in \mathcal{P}^2$ with aperture 2θ and $\theta \in (0, \pi)$ (for a detailed proof of this kind of symmetrization we refer to [85]), that is:

$$\nu_s^{ACF} = \min_{\theta \in [0,\pi]} \Gamma^s(\theta) = \min_{\theta \in [0,\pi]} \frac{\gamma_s(\theta) + \gamma_s(\pi - \theta)}{2}.$$

Then, for $s \in (0, 1)$ the previous statement for the limiting profile holds true for

$$\alpha^* = \begin{cases} \nu_s^{ACF}, & 0 < s \le \frac{1}{2}, \\ \min\{\nu_s^{ACF}, 2s - 1\}, & \frac{1}{2} < s < 1. \end{cases}$$

As we mentioned, the threshold s = 1/2 is due to the presence of the phenomenon of selfsegregation of nonlocal problem with $s \in (1/2, 1)$, namely centered on the nodal set $\Gamma(\mathbf{u})$ there exists a ball $\widetilde{B}^+ \subset \mathbb{R}^{n+1}_+$ sufficiently small and an index $i = 1, \ldots, h$ such that all the components u_i of **u** with $j \neq i$ are identically zero on the ball make exception of u_i which is not identically zero and such that

$$\partial^0 \widetilde{B}^+ \setminus \Gamma(\mathbf{u}) = \{ X \in \partial^0 \widetilde{B}^+ \colon u_i(X) > 0 \}.$$

The presence of self-segregation in the context of competition-diffusion problem is a phenomenon well known in the literature, even in the local case. Indeed, in [81] the authors dealt with the case of self-segregated profile relabeling the restrictions of the profile on each connected component of the positive set. Since in that case the operator is local, the restriction itself satisfies the assumption of the segregated profile, but unfortunately the nonlocal attitude of our operator does not allow this strategy. Moreover, we mention the work [35], where the authors proved that the self-segregation is a phenomenon that does not appear as singular limit of the local counterpart of our competition-diffusion problem (3).

In the context of the fractional Laplacian $(-\Delta)^s$, as pointed out in [84], the main point is that the fundamental solution turns out to be bounded near 0 and in $H^{1,a}(B_1^+)$ whenever s > 1/2, n = 1. This implies that, when $s \in [1/2, 1)$ and $n \ge 2$, the function

$$u(x,y)=\left(x_1^2+y^2\right)^{\frac{2s-1}{2}}$$

is a positive L_a -harmonic function in \mathbb{R}^{n+1}_+ with non trivial trace on Σ and $\partial_u^a u = 0$ on Σ . In particular, its trace on Σ is self-segregated since it has two disconnected positivity regions.

The following result is a refinement of the Liouville theorem [83, Theorem 7.1] for every $s \in (0, 1)$ and it is based on the division of the nodal set in its segregated and self-segregated part, in such a way we can apply our result obtained in the previous Sections to the segregated part of $\Gamma(\mathbf{u})$.

Definition 1.6.1. Let $s \in (0,1)$ and $u \in \mathcal{G}^s(B_1)$. Then the nodal set $\Gamma(\mathbf{u})$ is said to be either "segregated" in $K \cap \Sigma$ if

$$\Gamma(\mathbf{u}) = \partial^0 K^+ \setminus \bigcup_{i=1}^h \operatorname{int}\left(\overline{\{u_i > 0\}}\right)$$

or "self-segregated" in $K \cap \Sigma$ if

$$\Gamma(\mathbf{u}) = \bigcup_{i=1}^{h} \partial \{u_i > 0\} \setminus \operatorname{int}\left(\overline{\{u_i > 0\}}\right).$$

47



Figure 3: One-dimensional configurations in $\mathcal{G}^s_{\mathrm{loc}}(\mathbb{R})$ respectively segregated and self-segregated

In general, the nodal set is said to be segregated (self-segregated) in Σ if it is segregated (self-segregated) on every compact set $K \cap \Sigma$.

Theorem 1.6.2. *Let* $s \in (0, 1)$ *. If*

- either $\Gamma(\mathbf{u})$ is segregated in Σ and $\mathbf{u} \in \mathcal{G}^s_{\mathrm{loc}}(\mathbb{R}^{n+1}_+) \cap C^{0,\alpha}(\overline{\mathbb{R}^{n+1}_+})$ for $\alpha \in (0,s)$,
- or $\Gamma(\mathbf{u})$ is self-segregated in Σ and $\mathbf{u} \in \mathcal{G}_{loc}^{s}(\mathbb{R}^{n+1}_{+}) \cap C^{0,\alpha}(\overline{\mathbb{R}^{n+1}_{+}})$ for $\alpha \in (0, 2s-1)$,

then **u** is constant.

In the remaining part of the Chapter we prove this result following the procedure in [83]. This result implies a refinement in theory of the regularity of segregated profiles near the nodal set.

Corollary 1.6.3. Let $s \in (0,1)$ and $\mathbf{u} \in \mathcal{G}^s(B_1^+)$ be limit of a sequence $(\mathbf{u}_\beta)_\beta$ of solutions of (3). Then $\mathbf{u} \in C^{0,\alpha}_{\text{loc}}(\overline{B_1^+})$ for every $\alpha \in (0,s)$ if and only if $\Gamma(\mathbf{u})$ is segregated in $B_1 \cap \Sigma$.

The proof of this Corollary is based on our Liouville type theorem and on the techniques developed in the mentioned works [83, 84].

Proof of Theorem 1.6.2. Since by assumption $\mathbf{u} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1}_+) \cap C^{0,\alpha}(\overline{\mathbb{R}^{n+1}_+})$, we easily get by [84, Corollary 2.12] that \mathbf{u} is homogeneous of degree γ with respect to any of its possible zeros. Thus, for $\alpha > 0$ and for every dimension n, as in [83], let us introduce the critical value

$$\nu_s^{\text{Liouv}}(n) = \inf \left\{ \alpha > 0 \colon \mathfrak{B}^s(\mathbb{R}^{n+1}) \cap C^{0,\alpha}_{\text{loc}}(\overline{\mathbb{R}^{n+1}_+}) \text{ is non empty} \right\}.$$

Since for every $s \in (0,1)$ and $n \ge 1$ the function $(y^{2s}, 0, \cdot, 0) \in \mathfrak{B}^s(\mathbb{R}^{n+1}) \cap C^{0,2s}_{\text{loc}}(\overline{\mathbb{R}^{n+1}_+})$, we get $\nu_s^{\text{Liouv}} \le 2s$. However, since we need to take care of the structure of the nodal set we introduce the following critical value

$$\nu_s^1(n) = \inf \left\{ \alpha > 0 \left| \begin{array}{c} \mathfrak{B}^s(\mathbb{R}^{n+1}) \cap C^{0,\alpha}_{\mathrm{loc}}(\overline{\mathbb{R}^{n+1}_+}) \text{ is non empty} \\ \Gamma(\mathbf{u}) \text{ is segregated in } \Sigma \end{array} \right. \right\}$$

such that $\nu_s^1(n) \leq \nu_s^{\text{Liouv}}(n)$. The main idea is to reduce such problem to the ones of estimating $\nu_s^{\text{Liouv}}(1)$ and $\nu_1^1(1)$, which can be computed explicitly: let us prove that for any dimension $n \geq 2$ it holds

$$\nu_s^{\text{Liouv}}(n) \ge \nu_s^{\text{Liouv}}(n-1).$$

Since $\mathbf{u} \in \mathfrak{B}^{s}(\mathbb{R}^{n+1})$, as we previously remarked, we have that $\Gamma(\mathbf{u})$ is a cone with vertex at the origin and $N(0, \mathbf{u}, r) = \alpha$, for every $r \in (0, +\infty)$.

We can easily exclude the case $\Gamma(\mathbf{u}) = \Sigma$, since in that case all the components of \mathbf{u} have trivial trace on Σ . As a consequence, the odd extension of \mathbf{u} through Σ is a nontrivial vector of harmonic functions on \mathbb{R}^{n+1} , forcing $\alpha \geq 2s \geq \nu_s^{\text{Liouv}}(n-1)$.

Similarly, since $n \ge 2$ and for dim $\Gamma(\mathbf{u}) \le n - 2s$ we obtain that $\Gamma(\mathbf{u})$ has null L_a -capacity, we can exclude the case $\Gamma(\mathbf{u}) = \{0\}$.

Now, given $X_0 \in \Gamma(\mathbf{u}) \cap S^{n-1}$, let us introduce the blow-up sequence of \mathbf{u} associated to $r_k > 0$ as

$$\mathbf{u}_k(X) = rac{\mathbf{u}(X_0 + r_k X)}{
ho_k} \quad ext{with }
ho_k^2 = H(X_0, \mathbf{u}, r_k).$$

Now, following the same ideas in the proof of Lemma 1.3.5, given C > 0 and $\overline{r} > 0$ be such that Proposition 1.2.4, Proposition 1.2.7 and Lemma 1.3.4 hold true then, up to taking k so large that $r_k, r_k R \leq \overline{r}$, we get for every R > 0 that

$$\begin{split} \int_{\partial^+ B_R^+} |y|^a \, \mathbf{u}_k^2 \mathrm{d}\sigma &= \frac{1}{\rho_k^2 r_k^{n+a}} \int_{\partial^+ B_{Rr_k}^+(X_0)} |y|^a \, \mathbf{u}_k^2 \mathrm{d}\sigma \\ &= R^{n+a} \frac{H(X_0, \mathbf{u}, Rr_k)}{H(X_0, \mathbf{u}, r_k)} \\ &\leq R^{n+a} \left(\frac{Rr_k}{r_k}\right)^{2\overline{C}}, \end{split}$$

which implies, by the L_a -subharmonicity of the odd extension of \mathbf{u}_k through Σ , that

$$\sup_{B_{R/2}^+} \mathbf{u}_k \le 2C(n,s) \left(\frac{1}{R^{n+1+a}} \int_{B_R^+} |y|^a \, \mathbf{u}_k^2 \mathrm{d}X \right)^{1/2} \le 2C(n,s) \left(\frac{H(0,\mathbf{u}_k,R)}{n+a+1} \right)^{1/2}$$

is uniformly bounded for k > 0. By Proposition 1.3.8, there exists a blow-up limit $\overline{\mathbf{u}} \in \mathcal{G}^s_{\text{loc}}(\mathbb{R}^{n+1})$ with $\|\overline{\mathbf{u}}\|_{L^{2,a}(\partial^+ B_1^+)} = 1$.

Since the blow-up sequence $(\mathbf{u}_k)_k$ is constructed starting from a family of homogeneous entire segregated profiles in $\mathfrak{B}^s_{\mathrm{loc}}(\mathbb{R}^{n+1})$, we can prove that $\overline{\mathbf{u}}$ is constant along the direction parallel to $X_0 \in S^{n-1}$ and that its restriction on the orthogonal half plane belongs to $\mathcal{G}^s_{\mathrm{loc}}(\mathbb{R}^n)$. Moreover, since the blow-up sequence $(\mathbf{u}_k)_k$ is centered in a fixed point X_0 , by Proposition 1.2.4, Corollary 1.2.6 and Proposition 1.3.11 we get that $\overline{\mathbf{u}} \in \mathfrak{B}^s(\mathbb{R}^n) \cap C^{0,\gamma}_{\mathrm{loc}}(\mathbb{R}^{n+1})$, with $N(0,\overline{\mathbf{u}},1) = \gamma \leq \alpha$. Finally, since α is arbitrary choose in $\nu_s^{\mathrm{Liouv}}(n)$, the thesis follows from the bound $\nu_s^{\mathrm{Liouv}}(n-1) \leq \alpha$.

If instead, we consider the critical values $\nu_s^1(n)$, the result of dimensional descent still holds since the uniform convergence on compact sets of the blow-up sequence $(\mathbf{u}_k)_k$ ensures that the nodal set $\Gamma(\overline{\mathbf{u}})$ is still segregated in Σ , and hence

$$\nu_s^1(n) \ge \nu_s^1(n-1),$$

for every $s \in (0,1)$ and $n \ge 2$. Thus, by a complete classification of the elements in $\mathfrak{B}^{s}(\mathbb{R}^{2}_{+})$, see [88, Section 5], we can finally reach the claimed result since

$$\nu_s^{\text{Liouv}}(1) = \begin{cases} s, & \text{if } \left(0, \frac{1}{2}\right) \\ 2s - 1, & \text{if } s \in \left[\frac{1}{2}, 1\right) \end{cases} \text{ and } \nu_s^1(1) = s \text{ for any } s \in (0, 1).$$

2

ON S-HARMONIC FUNCTIONS ON CONES

2.1 INTRODUCTION

Let $n \ge 2$ and C be an open cone in \mathbb{R}^n with vertex in 0, for a given $s \in (0, 1)$, we consider the problem of the classification of nontrivial functions which are *s*-harmonic inside the cone and vanish identically outside, that is:

$$\begin{cases} (-\Delta)^s u_s = 0 & \text{in } C, \\ u_s \ge 0 & \text{in } \mathbb{R}^n \\ u_s \equiv 0 & \text{in } \mathbb{R}^n \setminus C. \end{cases}$$
(55)

By [5, Theorem 3.2], it is known that there exists a homogeneous, nonnegative and nontrivial solution to (55) of the form

$$u_s(x) = |x|^{\gamma_s} u_s\left(\frac{x}{|x|}\right),$$

where $\gamma_s := \gamma_s(C)$ is a definite homogeneity degree (characteristic exponent), which depends on the cone. Moreover, such a solution is continuous in \mathbb{R}^n and unique, up to multiplicative constants. We can normalize it in such a way that $||u_s||_{L^{\infty}(S^{n-1})} = 1$. We consider the case when s approaches 1, wondering whether solutions of the problem do converge to a harmonic function in the same cone and, in case, which are the suitable spaces for convergence.

Our problem (55) can be linked to a specific spectral problem of local nature in the upper half sphere by using the extension technique popularized in [23] by Caffarelli and Silvestre, which characterize the fractional Laplacian in \mathbb{R}^n as the Dirichlet-to-Neumann map for a variable v depending on one more space dimension.

Hence, let us consider an open region $\omega \subseteq S^{n-1} = \partial S^n_+$, with $S^n_+ = S^n \cap \{y > 0\}$, and define the eigenvalue

$$\lambda_1^s(\omega) = \inf\left\{\frac{\int_{S^n_+} y^{1-2s} |\nabla_{S^n} u|^2 \mathrm{d}\sigma}{\int_{S^n_+} y^{1-2s} u^2 \mathrm{d}\sigma} : u \in H^1(S^n_+; y^{1-2s} \mathrm{d}\sigma) \setminus \{0\} \text{ and } u \equiv 0 \text{ in } S^{n-1} \setminus \omega\right\}.$$

Next, define the *characteristic exponent* of the cone C_{ω} spanned by ω (see Definition 2.2.1) as

$$\gamma_s(C_\omega) = \gamma_s(\lambda_1^s(\omega)) , \qquad (56)$$

where the function $\gamma_s(t)$ is defined by

$$\gamma_s(t) := \sqrt{\left(\frac{n-2s}{2}\right)^2 + t} - \frac{n-2s}{2}$$

We recall the existence of a remarkable link between the nonnegative $\lambda_1^s(\omega)$ -eigenfunctions and the $\gamma_s(\lambda_1^s(\omega))$ -homogeneous L_a -harmonic functions. As pointed out in Proposition 1.3.14, given φ_s the first nonnegative eigenfunction to $\lambda_1^s(\omega)$ and v_s its $\gamma_s(\lambda_1^s(\omega))$ -homogeneous extension to \mathbb{R}^{n+1}_+ , i.e.

$$\psi_s(r,\theta) = r^{\gamma_s(\lambda_1^s(\omega))}\varphi_s(\theta),$$

we easily get that v_s is L_a -harmonic in the upper half-space, with $a = 1 - 2s \in (-1, 1)$. Moreover its trace $u_s(x) = v_s(x, 0)$ is s-harmonic in the cone C_ω spanned by ω , vanishing identically outside: in other words u_s is a solution of our problem (55). In a symmetric way, for the standard Laplacian, we consider the problem of γ -homogeneous functions which are harmonic inside the cone spanned by ω and vanish outside:

$$\begin{cases} -\Delta u_1 = 0 & \text{in } C_{\omega}, \\ u_1 \ge 0 & \text{in } \mathbb{R}^n \\ u_1 = 0 & \text{in } \mathbb{R}^n \setminus C_{\omega}. \end{cases}$$
(57)

Is is well known that the associated eigenvalue problem on the sphere is that of the Laplace-Beltrami operator with Dirichlet boundary conditions:

$$\lambda_1(\omega) = \inf\left\{\frac{\int_{S^{n-1}} |\nabla_{S^{n-1}}u|^2 \mathrm{d}\sigma}{\int_{S^{n-1}} u^2 \mathrm{d}\sigma} : u \in H^1(S^{n-1}) \setminus \{0\} \text{ and } u = 0 \text{ in } S^{n-1} \setminus \omega\right\},\$$
and the *characteristic exponent* of the cone C_{ω} is

$$\gamma(C_{\omega}) = \sqrt{\left(\frac{n-2}{2}\right)^2 + \lambda_1(\omega)} - \frac{n-2}{2} = \gamma_{s|s=1}(\lambda_1(\omega)) .$$
(58)

In the classical case, the characteristic exponent enjoys a number of nice properties: it is minimal on spherical caps among sets having a given measure. Moreover for the spherical caps, the eigenvalues enjoy a fundamental convexity property with respect to the colatitude θ (see the results in [3, 47]). We remark that the convexity plays a major role in the proof of the Alt-Caffarelli-Friedman monotonicity formula, a key tool in the Free boundary theory (see [21] for a general excursus on the subject).

Since the standard Laplacian can be viewed as the limiting operator of the family $(-\Delta)^s$ as $s \nearrow 1$, some questions naturally arise:

Problem 2.1.1. Is it true that

- (1) $\lim_{s\to 1} \gamma_s(C) = \gamma(C)$?
- (2) $\lim_{s \to 1} u_s = u_1$ uniformly on compact sets, or better, in Hölder local norms?
- (3) for spherical caps of opening θ is there any convexity of the map $\theta \mapsto \lambda_1^s(\theta)$ at least, for *s* near 1?

We therefore addressed the problem of the asymptotic behavior of the solutions of problem (55) for $s \nearrow 1$, obtaining a rather unexpected result: our analysis shows high sensitivity to the opening solid angle ω of the cone C_{ω} , as evaluated by the value of $\gamma(C)$. In the case of wide cones, when $\gamma(C) < 2$ (that is, $\theta \in (\pi/4, \pi)$ for spherical caps of colatitude θ), our solutions do converge to the harmonic homogeneous function of the cone; instead, in the case of narrow cones, when $\gamma(C) \ge 2$ (that is, $\theta \in (0, \pi/4]$ for spherical caps), then limit of the homogeneity degree will be always two and the limiting profile will be something different, though related through a correction term. Similar transition phenomena have been detected in other contexts for some types of free boundary problems on cones (see [74]). Moreover, we will see that an important quantity which appears in this estimates and plays a fundamental role is

$$\frac{C(n,s)}{2s - \gamma_s(C)}$$

where C(n, s) > 0 is the normalization constant given in (62). It will be therefore very important to bound this quantity uniformly in s. Our main result is the following Theorem.

Theorem 2.1.2. Let C be an open cone with vertex at the origin. There exist finite the following limits:

$$\overline{\gamma}(C) := \lim_{s \to 1^{-}} \gamma_s(C) = \min\{\gamma(C), 2\}$$

and

$$\mu(C) := \lim_{s \to 1^-} \frac{C(n,s)}{2s - \gamma_s(C)} = \begin{cases} 0 & \text{if } \gamma(C) \le 2, \\ \mu_0(C) & \text{if } \gamma(C) \ge 2, \end{cases}$$

where C(n, s) is defined in (62) and

$$\mu_0(C) := \inf \left\{ \frac{\int_{S^{n-1}} |\nabla_{S^{n-1}} u|^2 - 2nu^2 \mathrm{d}\sigma}{\left(\int_{S^{n-1}} |u| \mathrm{d}\sigma\right)^2} : u \in H^1(S^{n-1}) \setminus \{0\} \text{ and } u = 0 \text{ in } S^{n-1} \setminus C \right\}.$$

Let us consider the family (u_s) of nonnegative solutions to (55) such that $||u_s||_{L^{\infty}(S^{n-1})} = 1$. Then, as $s \nearrow 1$, up to a subsequence, we have

- 1. $u_s \to \overline{u}$ in $L^2_{\text{loc}}(\mathbb{R}^n)$ to some $\overline{u} \in H^1_{\text{loc}}(\mathbb{R}^n) \cap L^{\infty}(S^{n-1})$.
- 2. The convergence is uniform on compact subsets of C, \overline{u} is nontrivial with $||\overline{u}||_{L^{\infty}(S^{n-1})} = 1$ and is $\overline{\gamma}(C)$ -homogeneous.
- *3.* The limit \overline{u} solves

$$\begin{cases} -\Delta \overline{u} = \mu(C) \int_{S^{n-1}} \overline{u} d\sigma & \text{in } C, \\ \overline{u} = 0 & \text{in } \mathbb{R}^n \setminus C. \end{cases}$$
(59)

Uniqueness of the limit \overline{u} and therefore existence of the limit of u_s as $s \nearrow 1$ holds in the case of connected cones and, in any case, whenever $\gamma(C) > 2$. We will see in Remark 2.4.2 that under symmetry assumptions on the cone C, the limit function \overline{u} is unique and hence it does not depend on the choice of the subsequence.

A further motivation to our study of (55), as shown in [83, 84] and in the blow-up analysis in Chapter 1, is its deep relation with the exponent of the optimal Hölder regularity of segregated profiles and the geometric analysis of the segregation phenomenon. In [84, 83], estimates in Hölder spaces have been obtained by the use of fractional versions of the Alt-Caffarelli-Friedman and Almgren monotonicity formulas. Let us state here the fractional version of the spectral problem beyond the first monotonicity formula: consider the set of 2-partitions of S^{n-1} as

$$\mathcal{P}^2 := \Big\{ (\omega_1, \omega_2) : \omega_i \subseteq S^{n-1} \text{ open, } \omega_1 \cap \omega_2 = \emptyset, \ \overline{\omega_1} \cup \overline{\omega_2} = S^{n-1} \Big\}$$

and define the optimal partition value as:

$$\nu_s^{ACF} := \frac{1}{2} \inf_{(\omega_1,\omega_2)\in\mathcal{P}^2} \sum_{i=1}^2 \gamma_s(\lambda_1^s(\omega_i)).$$
(60)

It is easy to see, by a Schwarz symmetrization argument, that ν_s^{ACF} is achieved by a pair of complementary spherical caps $(\omega_{\theta}, \omega_{\pi-\theta}) \in \mathcal{P}^2$ with aperture 2θ and $\theta \in (0, \pi)$ (for a detailed proof of this kind of symmetrization we refer to [85]), that is:

$$\nu_s^{ACF} = \min_{\theta \in [0,\pi]} \Gamma^s(\theta) = \min_{\theta \in [0,\pi]} \frac{\gamma_s(\theta) + \gamma_s(\pi - \theta)}{2}$$

This gives a further motivation to our study of (55) for spherical caps. A classical result by Friedland and Hayman, [47], yields $\nu^{ACF} = 1$ for the case s = 1, and the minimal value is achieved for two half spheres; this equality is the core of the proof of the classical Alt-Caffarelli-Friedman monotonicity formula.



Figure 4: Possible values of $\Gamma^{s}(\theta) = \Gamma^{s}(\omega_{\theta}, \omega_{\pi-\theta})$ for s < 1 and s = 1 and n = 2.

It was proved in [84] that ν_s^{ACF} is linked to the threshold for uniform bounds in Hölder norms for competition-diffusion systems, as the interspecific competition rate diverges to infinity, as well as the exponent of the optimal Hölder regularity for their limiting profiles. It was also conjectured that $\nu_s^{ACF} = s$ for every $s \in (0, 1)$. Unfortunately, the exact value of ν_s^{ACF} is still unknown, and we only know that $0 < \nu_s^{ACF} \le s$ (see [84, 83]). Our contribution to this open problem is a byproduct of Theorem 2.1.2. Corollary 2.1.3. In any space dimension we have

$$\lim_{s \to 1} \nu_s^{ACF} = 1 \; .$$

Moreover, exploiting the connection between s-harmonic functions on cones and the traces on \mathbb{R}^n of the blow-up limits in $\mathfrak{B}^s(\mathbb{R}^{n+1})$, discussed in Proposition 1.3.14 in Chapter 1, we can reasonably state the following remark on the asymptotic limit of the segregated configurations in $\mathcal{G}^s_{\text{loc}}(B_1^+)$.

More precisely, given for $s \in (0,1)$ the class $\mathfrak{B}^{s}(\mathbb{R}^{n})$ of the traces of the blow-up limit in $\mathfrak{B}^{s}(\mathbb{R}^{n+1})$, we directly get, from Theorem 2.1.2 that

$$\sup\left\{N(0,\overline{u},1):\overline{u}\in\mathfrak{B}^{s}(\mathbb{R}^{n+1})\right\}\leq\sup_{\theta\in(0,\pi)}\gamma_{s}(\theta)\leq 2s\leq 2$$

for every $s \in (0, 1)$. This simple bound suggests that the possible blow-up limits of the segregated profiles arising from a competition-diffusion problem with nonlocal diffusion and variational competition can not converge to the ones of the segregated solutions studied in [18, 81]. This remark suggests that even for the case of segregated configurations we have to expect, for $s \nearrow 1$, a rather unexpected result.

The Chapter is organized as follows. In Section 2.2 we introduce our setting and we state the relevant known properties of homogeneous *s*-harmonic functions on cones. After this, we will obtain local $C^{0,\alpha}$ -estimates in compact subsets of C and local H^s -estimates in compact subsets of \mathbb{R}^n for solutions of (55). In Section 2.3 we analyze the asymptotic behaviour of $\gamma_s(C)$ as *s* converges to 1, in order to understand the quantities $\overline{\gamma}(C)$ and $\mu(C)$. To do this, we will establish a distributional semigroup property for the fractional Laplacian for functions which grow at infinity. In Section 2.4 we prove Theorem 2.1.2 and Corollary 2.1.3. Eventually, in Section 2.5, we prove a nontrivial improvement of the main Theorem concerns uniform bounds in Hölder spaces holding uniformly for $s \to 1$.

2.2 HOMOGENOUS S-HARMONIC FUNCTIONS ON CONES

In this Section, we focus our attention on the local properties of homogeneous *s*-harmonic functions on *regular cones*. Since in Section 2.3 we will study the behaviour of the characteristic exponent as *s* approaches 1, in this section we recall some known results related to the boundary behaviour of the solution of (55) restricted to the unitary sphere S^{n-1} and some estimates of the Hölder and H^s seminorm.

Definition 2.2.1. Let $\omega \subset S^{n-1}$ be an open set, that may be disconnected. We call *unbounded cone* with vertex in 0, spanned by ω the open set

$$C_{\omega} = \{ rx : r > 0, x \in \omega \}.$$

Moreover we say that $C = C_{\omega}$ is narrow if $\gamma(C) \ge 2$ and wide if $\gamma(C) < 2$. We call C_{ω} regular cone if ω is connected and of class $C^{1,1}$. Let $\theta \in (0, \pi)$ and $\omega_{\theta} \subset S^{n-1}$ be an open spherical cap of colatitude θ . Then we denote by $C_{\theta} = C_{\omega_{\theta}}$ the right circular cone of aperture 2θ .

Hence, let C be a fixed unbounded open cone in \mathbb{R}^n with vertex in 0 and consider

$$\begin{cases} (-\Delta)^s u_s = 0 & \text{in } C, \\ u_s \equiv 0 & \text{in } \mathbb{R}^n \setminus C. \end{cases}$$

with the condition $||u_s||_{L^{\infty}(S^{n-1})} = 1$. By Theorem 3.2 in [5] there exists, up to a multiplicative constant, a unique nonnegative function u_s smooth in C and $\gamma_s(C)$ -homogenous, i.e.

$$u_s(x) = |x|^{\gamma_s(C)} u_s\left(\frac{x}{|x|}\right)$$

where $\gamma_s(C) \in (0, 2s)$. As it is well know (see for example [8, 76]), the fractional Laplacian $(-\Delta)^s$ is a nonlocal operator well defined in the class of integrability $\mathcal{L}_s^1 := \mathcal{L}^1 (dx/(1+|x|)^{n+2s})$, namely the normed space of all Borel functions u satisfying

$$\|u\|_{\mathcal{L}^{1}_{s}} := \int_{\mathbb{R}^{n}} \frac{|u(x)|}{(1+|x|)^{n+2s}} \mathrm{d}x < +\infty.$$
(61)

Hence, for every $u \in \mathcal{L}^1_s, \varepsilon > 0$ and $x \in \mathbb{R}^n$ we define

$$(-\Delta)^{s}_{\varepsilon}u(x) = C(n,s)\int_{\mathbb{R}^{n}\setminus B_{\varepsilon}(x)}\frac{u(x)-u(y)}{|x-y|^{n+2s}}\mathrm{d}y,$$

where

$$C(n,s) = \frac{2^{2s}s\Gamma(\frac{n}{2}+s)}{\pi^{n/2}\Gamma(1-s)} \in \left(0, 4\Gamma\left(\frac{n}{2}+1\right)\right].$$
(62)

and we can consider the fractional Laplacian as the limit

$$(-\Delta)^s u(x) = \lim_{\varepsilon \downarrow 0} (-\Delta)^s_\varepsilon u(x) = C(n,s) \text{ P.V.} \int_{\mathbb{R}^n} \frac{u(x) - u(y)}{|x - y|^{n + 2s}} \mathrm{d}y.$$

We remark that $u \in \mathcal{L}_s^1$ is such that $u \in \mathcal{L}_{s+\delta}^1$ for any $\delta > 0$, which will be an important tool in this part of the manuscript, in order to compute high order fractional Laplacians. Another definition of the fractional Laplacian, which can be constructed by a double change of variables as in [38], is

$$(-\Delta)^{s}u(x) = \frac{C(n,s)}{2} \int_{\mathbb{R}^{n}} \frac{2u(x) - u(x+y) - u(x-y)}{|y|^{n+2s}} \mathrm{d}y$$

which emphasize that given $u \in C^2(D) \cap \mathcal{L}^1_s$, we obtain that $x \mapsto (-\Delta)^s u(x)$ is a continuous and bounded function on D, for some bounded $D \subset \mathbb{R}^n$.

By [67, Lemma 3.3], if we consider a regular unbounded cone C symmetric with respect to a fixed axis, there exists two positive constant $c_1 = c_1(n, s, C)$ and $c_2 = c_2(n, s, C)$ such that

$$c_1 |x|^{\gamma_s - s} \operatorname{dist}(x, \partial C)^s \le u_s(x) \le c_2 |x|^{\gamma_s - s} \operatorname{dist}(x, \partial C)^s \tag{63}$$

for every $x \in C$. We remark that this result can be easily generalized to regular unbounded cones C_{ω} with $\omega \subset S^{n-1}$ which is a finite union of connected $C^{1,1}$ domain ω_i , such that $\overline{\omega}_i \cup \overline{\omega}_j = \emptyset$ for $i \neq j$, since the reasonings in [67] rely on a Boundary Harnack principle and on sharp estimates for the Green function for bounded $C^{1,1}$ domain non necessary connected (for more details [28]).

Through the paper we will call the coefficient of homogeneity γ_s as "*characteristic exponent*", since it is strictly related to an eigenvalue partition problem.

As we already mentioned, our solutions are smooth in the interior of the cone and locally $C^{0,s}$ near the boundary $\partial C \setminus \{0\}$ (see for example [67]), but we need some quantitative estimates in order to better understand the dependence of the Hölder seminorm on the parameter $s \in (0, 1)$.

Before showing the main result of Hölder regularity, we need the following estimates about the fractional Laplacian of smooth compactly supported functions: this result can be found in [8, Lemma 3.5] and [37, Lemma 5.1], but here we compute the formula with a deep attention on the dependence of the constant with respect to $s \in (0, 1)$.

Proposition 2.2.2. Let $s \in (0,1)$ and $\varphi \in C_c^2(\mathbb{R}^n)$. Then

$$|(-\Delta)^{s}\varphi(x)| \le \frac{c}{(1+|x|)^{n+2s}}, \qquad \forall x \in \mathbb{R}^{n},$$
(64)

where the constant c > 0 depends only on n and the choice of φ .

Proof. Let $K \subset \mathbb{R}^n$ be the compact support of φ and $k = \max_{x \in K} |\varphi(x)|$. There exists R > 1 such that $K \subset B_{R/2}(0)$. Let |x| > R.

$$\begin{split} |(-\Delta)^{s}\varphi(x)| &= \left| C(n,s) \int_{\mathbb{R}^{n}} \frac{\varphi(x) - \varphi(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| = \left| C(n,s) \int_{K} \frac{\varphi(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \\ &\leq \left| \frac{C(n,s)k}{|x|^{n + 2s}} \int_{K} \frac{1}{(1 - |\frac{y}{x}|)^{n + 2s}} \mathrm{d}y \leq \frac{C(n,s)k2^{n + 2s}|K|}{|x|^{n + 2s}} \right| \\ &\leq \left| \frac{C(n,s)k2^{2(n + 2s)}|K|}{(1 + |x|)^{n + 2s}} \leq \frac{c}{(1 + |x|)^{n + 2s}}, \end{split}$$

where c > 0 depends only on n and the choice of φ .

Let now $|x| \leq R$. We use the fact that any derivative of φ of first and second order is uniformly continuous in the compact set K and the fact that in $B_R(0)$ the function $(1+|x|)^{n+2s}$ has maximum given by $(1+R)^{n+2s}$. Hence there exist $0 < \delta < 1$ and a constant M > 0, both depending only on n and the choice of φ such that

$$|\varphi(x+z) + \varphi(x-z) - 2\varphi(x)| \le M|z|^2 \qquad \forall z \in B_{\delta}(0).$$

Hence

$$\begin{split} |(-\Delta)^{s}\varphi(x)| &= \left| C(n,s) \int_{\mathbb{R}^{n} \setminus B_{\delta}(x)} \frac{\varphi(x) - \varphi(y)}{|x - y|^{n + 2s}} \mathrm{d}y + C(n,s) \int_{B_{\delta}(x)} \frac{\varphi(x) - \varphi(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \\ &\leq 2kC(n,s) \int_{\mathbb{R}^{n} \setminus B_{\delta}(x)} \frac{1}{|x - y|^{n + 2s}} \mathrm{d}y + \frac{C(n,s)}{2} \int_{B_{\delta}(0)} \frac{|\varphi(x + z) + \varphi(x - z) - 2\varphi(x)|}{|z|^{n + 2s}} \mathrm{d}z \\ &\leq 2kC(n,s)\omega_{n-1} \int_{\delta}^{+\infty} r^{-1 - 2s} \mathrm{d}r + \frac{C(n,s)\omega_{n-1}M}{2} \int_{0}^{\delta} r^{1 - 2s} \mathrm{d}r \\ &= \frac{kC(n,s)\omega_{n-1}}{s\delta^{2s}} + \frac{C(n,s)\omega_{n-1}M\delta^{2 - 2s}}{4(1 - s)} \\ &\leq \frac{c}{\delta^{2}} + c = c \frac{(1 + |x|)^{n + 2s}}{(1 + |x|)^{n + 2s}} \leq \frac{c(1 + R)^{n + 2}}{(1 + |x|)^{n + 2s}} = \frac{c}{(1 + |x|)^{n + 2s}}, \end{split}$$

where c > 0 depends only on n and the choice of φ . This concludes the proof.

By the previous calculations we have also the following result.

Remark 2.2.3. Let $s \in (0,1)$ and $\varphi \in C_c^2(\mathbb{R}^n)$. Then there exists a constant $c = c(n,\varphi) > 0$ and a radius $R = R(\varphi) > 0$ such that

$$|(-\Delta)^{s}\varphi(x)| \le c \frac{C(n,s)}{(1+|x|)^{n+2s}}, \qquad \forall x \in \mathbb{R}^{n} \setminus B_{R}(0).$$
(65)

The following result provides interior estimates for the Hölder norm of our solutions.

Proposition 2.2.4. Let C be a cone and $K \subset C$ be a compact set and $s_0 \in (0, 1)$. Then there exist a constant c > 0 and $\overline{\alpha} \in (0, 1)$, both dependent only on s_0, K, n, C , such that

$$||u_s||_{C^{0,\alpha}(K)} \le c \left(1 + \frac{C(n,s)}{2s - \gamma_s(C)}\right),$$

for any $\alpha \in (0, \overline{\alpha}]$ and any $s \in [s_0, 1)$.

By a standard covering argument, there exists a finite number of balls such that $K \subset \bigcup_{j=1}^{k} B_r(x_j)$, for a given radius r > 0 such that $\bigcup_{j=1}^{k} \overline{B_{2r}(x_j)} \subset C$. Thus, it is enough to prove

Proposition 2.2.5. Let $\overline{B_{2r}(\overline{x})} \subset C$ be a closed ball and $s_0 \in (0,1)$. Then there exist a constant c > 0 and $\overline{\alpha} \in (0,1)$, both dependent only on $s_0, r, \overline{x}, n, C$, such that

$$||u_s||_{C^{0,\alpha}(\overline{B_r(\overline{x})})} \le c\left(1 + \frac{C(n,s)}{2s - \gamma_s(C)}\right),$$

for any $\alpha \in (0, \overline{\alpha}]$ and any $s \in [s_0, 1)$.

In order to achieve the desired result, we need to estimate locally the value of the fractional Laplacian of u_s in a ball compactly contained in the cone C.

Lemma 2.2.6. Let $\eta \in C_c^{\infty}(B_{2r}(\overline{x}))$ be a cut-off function such that $0 \le \eta \le 1$ with $\eta \equiv 1$ in $B_r(\overline{x})$. Under the same assumptions of Proposition 2.2.5,

$$||(-\Delta)^s(u_s\eta)||_{L^{\infty}(B_{2r}(\overline{x}))} \le C_0\left(1 + \frac{C(n,s)}{2s - \gamma_s(C)}\right)$$

for any $s \in [s_0, 1)$, where $C_0 > 0$ depends on $s_0, n, \overline{x}, r, C$, and the choice of the function η .

Proof. Let R > 1 such that $\overline{B_{2r}(\overline{x})} \subset B_{R/2}(0)$. Hence, let fix a point $x \in B_{2r}(\overline{x})$. We can express the fractional Laplacian of $u_s \eta$ in the following way

$$\begin{aligned} (-\Delta)^{s}(u_{s}\eta)(x) &= \eta(x)(-\Delta)^{s}u_{s}(x) + C(n,s)\int_{\mathbb{R}^{n}}u_{s}(y)\frac{\eta(x) - \eta(y)}{|x - y|^{n + 2s}}\mathrm{d}y \\ &= C(n,s)\int_{B_{R}(0)}u_{s}(y)\frac{\eta(x) - \eta(y)}{|x - y|^{n + 2s}}\mathrm{d}y + \\ &+ C(n,s)\int_{\mathbb{R}^{n}\setminus B_{R}(0)}u_{s}(y)\frac{\eta(x) - \eta(y)}{|x - y|^{n + 2s}}\mathrm{d}y. \end{aligned}$$

We recall that $u_s(x) = |x|^{\gamma_s(C)} u_s(x/|x|)$ and that for any $s \in (0,1)$ the functions u_s are normalized such that $||u_s||_{L^{\infty}(S^{n-1})} = 1$. Moreover we remark that $\eta(x) - \eta(y) = \eta(x) \ge 0$ in $B_{2r}(\overline{x}) \times (\mathbb{R}^n \setminus B_R(0))$. Hence, using Proposition 2.2.2 and the fact that $\gamma_s(C) < 2s$, we obtain

$$\begin{aligned} |(-\Delta)^{s}(u_{s}\eta)(x)| &\leq C(n,s) \left| \int_{B_{R}(0)} u_{s}(y) \frac{\eta(x) - \eta(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| + \\ &+ C(n,s) \left| \int_{\mathbb{R}^{n} \setminus B_{R}(0)} u_{s}(y) \frac{\eta(x) - \eta(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \\ &\leq R^{\gamma_{s}(C)} |(-\Delta)^{s} \eta(x)| + C(n,s) 2^{n + 2s} \int_{\mathbb{R}^{n} \setminus B_{R}(0)} \frac{1}{|y|^{n + 2s - \gamma_{s}(C)}} \mathrm{d}y \\ &\leq \frac{cR^{2}}{(1 + |x|)^{n + 2s}} + C(n,s) 2^{n + 2} \omega_{n - 1} \int_{R}^{+\infty} r^{-1 - 2s + \gamma_{s}(C)} \mathrm{d}r \\ &\leq \frac{cR^{2}}{(1 + |x|)^{n + 2s}} + \frac{cC(n,s)}{R^{2s - \gamma_{s}(C)}(2s - \gamma_{s}(C))} \\ &\leq C_{0} \left(1 + \frac{C(n,s)}{2s - \gamma_{s}(C)}\right). \end{aligned}$$

Proof of Proposition 2.2.5. Let as before $\eta \in C_c^{\infty}(B_{2r}(\overline{x}))$ be a cut-off function such that $0 \le \eta \le 1$ with $\eta \equiv 1$ in $B_r(\overline{x})$. First, we remark that there exists a constant $c_0 > 0$ such that for any $s \in (0, 1)$, it holds

$$||u_s\eta||_{L^{\infty}(\mathbb{R}^n)} \le c_0, \tag{66}$$

where c_0 depends only on n, \overline{x}, r . In fact, let R > 0 be such that $\overline{B_{2r}(\overline{x})} \subset B_R(0)$. Then, for any $x \in \mathbb{R}^n$, we have $0 \le u_s \eta(x) \le R^{\gamma_s(C)} \le R^2$. Using the bound (66) and the previous Lemma, we can apply [24, Theorem 12.1] obtaining the existence of $\overline{\alpha} \in (0, 1)$ and C > 0, both depending only on n, s_0 and the choice of $B_r(\overline{x})$ such that

$$\begin{aligned} ||u_s\eta||_{C^{0,\alpha}(\overline{B_r(\overline{x})})} &\leq C(||u_s\eta||_{L^{\infty}(\mathbb{R}^n)} + ||(-\Delta)^s(u_s\eta)||_{L^{\infty}(B_{2r}(\overline{x}))}) \\ &\leq C\left(c_0 + C_0\left(1 + \frac{C(n,s)}{2s - \gamma_s(C)}\right)\right), \end{aligned}$$

for any $s \in [s_0, 1)$ and any $\alpha \in (0, \overline{\alpha}]$. Since $\eta \equiv 1$ in $B_r(\overline{x})$ we obtain the result.

Similarly, now we need to construct some estimate related to the H^s seminorm of the solution u_s , Since the functions do not belong to $H^s(\mathbb{R}^n)$, we need to truncate the solution with some cut off function in order to avoid the problems related to the growth at infinity. In such a way, we can use

$$[v]_{H^{s}(\mathbb{R}^{n})}^{2} = \left\| (-\Delta)^{s/2} v \right\|_{L^{2}(\mathbb{R}^{n})}^{2} = \int_{\mathbb{R}^{n}} v(-\Delta)^{s} v \mathrm{d}x.$$
(67)

which holds for every $v \in H^s(\mathbb{R}^n)$. So, let $\eta \in C_c^{\infty}(B_2)$ be a radial cut off function such that $\eta \equiv 1$ in B_1 and $0 \leq \eta \leq 1$ in B_2 , and consider $\eta_R(x) = \eta(\frac{x-x_0}{R})$ the rescaled cut off function defined in $B_{2R}(x_0)$, for some R > 0 and $x_0 \in \mathbb{R}^n$.

Proposition 2.2.7. Let $s_0 \in (0,1)$ and $\eta_R \in C_c^{\infty}(B_{2R}(x_0))$ previously defined. Then

$$[u_s \eta_R]^2_{H^s(\mathbb{R}^n)} \le c \left(1 + \frac{C(n,s)}{2s - \gamma_s(C)}\right)$$

for any $s \in [s_0, 1)$, where c > 0 is a constant that depends on x_0, R, C, s_0 and η .

Proof. Let $\eta \in C_c^{\infty}(B_2)$ be a radial cut off function such that $\eta \equiv 1$ in B_1 and $0 \leq \eta \leq 1$ in B_2 , and consider the collection of $(\eta_R)_R$ with R > 0 defined by $\eta_R(x) = \eta(\frac{x-x_0}{R})$ with some $x_0 \in \mathbb{R}^n$. By (67), for every R > 0 we obtain

$$[u_s\eta_R]^2_{H^s(\mathbb{R}^n)} = \left\| (-\Delta)^{s/2} (u_s\eta_R) \right\|^2_{L^2(\mathbb{R}^n)} = \int_{\mathbb{R}^n} u_s\eta_R (-\Delta)^s (u_s\eta_R) \mathrm{d}x.$$

By definition of the fractional Laplacian we have

$$\begin{split} \int_{\mathbb{R}^n} u_s \eta_R(-\Delta)^s (u_s \eta_R) \mathrm{d}x &= C(n,s) \int_{\mathbb{R}^n \times \mathbb{R}^n} u_s(x) \eta_R(x) \frac{u_s(x) \eta_R(x) - u_s(y) \eta_R(y)}{|x-y|^{n+2s}} \mathrm{d}y \mathrm{d}x \\ &= \int_{\mathbb{R}^n} \eta_R^2 u_s(-\Delta)^s u_s \mathrm{d}x + C(n,s) \int_{\mathbb{R}^n \times \mathbb{R}^n} \frac{\eta_R(x) - \eta_R(y)}{|x-y|^{n+2s}} u_s(x) u_s(y) \eta_R(x) \mathrm{d}y \mathrm{d}x \\ &= \frac{C(n,s)}{2} \int_{\mathbb{R}^n \times \mathbb{R}^n} \frac{|\eta_R(x) - \eta_R(y)|^2}{|x-y|^{n+2s}} u_s(x) u_s(y) \mathrm{d}y \mathrm{d}x \end{split}$$

where the last equation is obtained by the symmetrization of the previous integral with respect to the variable $(x, y) \in \mathbb{R}^n \times \mathbb{R}^n$. Before splitting the domain of integration into different subset, it is easy to see that

$$\eta_R(x) - \eta_R(y) \equiv 0 \quad \text{in } B_R(x_0) \times B_R(x_0) \cup (\mathbb{R}^n \setminus B_{2R}(x_0)) \times (\mathbb{R}^n \setminus B_{2R}(x_0)) \\ |\eta_R(x) - \eta_R(y)| \equiv 1 \quad \text{in } B_R(x_0) \times (\mathbb{R}^n \setminus B_{2R}(x_0)) \cup (\mathbb{R}^n \setminus B_{2R}(x_0)) \times B_R(x_0).$$

where all the previous balls are centered at the point x_0 . Hence, given the sets $\Omega_1 = B_{3R}(x_0) \times B_{3R}(x_0)$ and $\Omega_2 = B_{2R}(x_0) \times (\mathbb{R}^n \setminus B_{3R}(x_0)) \cup (\mathbb{R}^n \setminus B_{3R}(x_0)) \times B_{2R}(x_0)$ we have

$$\int_{\mathbb{R}^{n} \times \mathbb{R}^{n}} \frac{|\eta_{R}(x) - \eta_{R}(y)|^{2}}{|x - y|^{n + 2s}} u_{s}(x) u_{s}(y) \mathrm{d}y \mathrm{d}x \leq \int_{\Omega_{1}} \frac{|\eta_{R}(x) - \eta_{R}(y)|^{2}}{|x - y|^{n + 2s}} u_{s}(x) u_{s}(y) \mathrm{d}y \mathrm{d}x + \int_{\Omega_{2}} \frac{|\eta_{R}(x) - \eta_{R}(y)|^{2}}{|x - y|^{n + 2s}} u_{s}(x) u_{s}(y) \mathrm{d}y \mathrm{d}x.$$

In particular

$$\begin{split} \int_{\Omega_1} \frac{|\eta_R(x) - \eta_R(y)|^2}{|x - y|^{n + 2s}} u_s(x) u_s(y) \mathrm{d}y \mathrm{d}x &\leq \sup_{B_{3R}(x_0)} u_s^2 \int_{B_{3R}(x_0) \times B_{3R}(x_0)} \frac{\|\nabla \eta_R\|_{L^{\infty}(\mathbb{R}^n)}^2}{|x - y|^{n + 2s - 2}} \mathrm{d}y \mathrm{d}x \\ &\leq \|\nabla \eta_R\|_{L^{\infty}}^2 \sup_{B_{3R}(x_0)} u_s^2 \int_{B_{3R}(0)} \mathrm{d}x \int_{B_{6R}(x)} \frac{1}{|x - y|^{n + 2s - 2}} \mathrm{d}y \\ &\leq \frac{\|\nabla \eta\|_{L^{\infty}}^2}{R^2} \sup_{B_{3R}(x_0)} u_s^2 |B_{3R}| \left|S^{n - 1}\right| \frac{(6R)^{2 - 2s}}{2(1 - s)} \\ &\leq C \|\nabla \eta\|_{L^{\infty}}^2 \frac{R^{n - 2s}}{2(1 - s)} \max\{|x_0|^{2\gamma_s}, (3R)^{2\gamma_s}\} \|u_s\|_{L^{\infty}(S^{n - 1})} \end{split}$$

where in the second inequality we use the changes of variables $x - x_0$ and $y - x_0$ and the fact that $B_{3R}(0) \times B_{3R}(0) \subset B_{3R}(0) \times B_{6R}(x)$ for every $x \in B_{3R}(0)$. Similarly we have

$$\begin{split} \int_{\Omega_2} \frac{\left|\eta_R(x) - \eta_R(y)\right|^2}{\left|x - y\right|^{n+2s}} u_s(x) u_s(y) \mathrm{d}y \mathrm{d}x &\leq 2 \int_{B_{2R}(x_0)} u_s(x) \left(\int_{\mathbb{R}^n \setminus B_{3R}(x_0)} \frac{u_s(y)}{\left|x - y\right|^{n+2s}} \mathrm{d}y\right) \mathrm{d}x \\ &\leq 2 \int_{B_{2R}(0)} u_s(x + x_0) \left(\int_{\mathbb{R}^n \setminus B_{3R}(0)} \frac{u_s(y + x_0)}{\left|y\right|^{n+2s} \left(1 - \frac{|x|}{|y|}\right)^{n+2s}} \mathrm{d}y\right) \mathrm{d}x \\ &\leq 2 \cdot 3^{n+2s} \int_{B_{2R}(0)} u_s(x + x_0) \left(\int_{\mathbb{R}^n \setminus B_{3R}(0)} \frac{C(|y| + |x_0|)^{\gamma_s}}{\left|y\right|^{n+2s}} \mathrm{d}y\right) \mathrm{d}x \\ &\leq C \sup_{B_{2R}(x_0)} u_s \left|B_{2R}\right| \left|S^{n-1}\right| 2^{\gamma_s} G(x_0, R) \end{split}$$

with

$$G(x_0, R) = \begin{cases} \frac{|x_0|^{\gamma_s}}{2s - \gamma_s} (3R)^{-2s} & \text{if } |x_0| \ge 3R\\ \frac{(3R)^{\gamma_s - 2s}}{2s - \gamma_s} & \text{if } |x_0| \le 3R \end{cases} \le \frac{(3R)^{-2s}}{2s - \gamma_s} \max\{|x_0|, 3R\}^{\gamma_s} dx_s \le \frac{(3R)^{-2s}}{2s - \gamma$$

Finally, we obtain the desired bound for the seminorm $[u_s\eta_R]^2_{H^s(\mathbb{R}^n)}$ summing the two terms and recalling that $||u_s||_{L^{\infty}(S^{n-1})} = 1$.

2.3 characteristic exponent $\gamma_s(c)$: properties and asymptotic behaviour

In this Section we start the analysis of the asymptotic behaviour of the homogeneity degree $\gamma_s(C)$ as s converges to 1. The main results are two: first we get a monotonicity result for the map $s \mapsto \gamma_s(C)$, for a fixed regular cone C, which ensures the existence of the limit and, using

some comparison result, a bound on the possible value of the limit exponent. Secondly we study the asymptotic behaviour of the quotient $\frac{C(n,s)}{2s-\gamma_s(C)}$.

In order to prove the first result and compare different order of *s*-harmonic functions for different power of $(-\Delta)^s$, we need to introduce some results which give a natural extension of the classic semigroup property of the fractional Laplacian, for function defined on cones which grow at infinity.

2.3.1 Distributional semigroup property

It is well known that if we deal with smooth functions with compact support, or more generally with functions in the Schwartz space $S(\mathbb{R}^n)$, a semigroup property holds for the fractional Laplacian, i.e. $(-\Delta)^{s_1} \circ (-\Delta)^{s_2} = (-\Delta)^{s_1+s_2}$, where $s_1, s_2 \in (0, 1)$ with $s_1 + s_2 < 1$. Since we have to deal with functions in \mathcal{L}^1_s that grow at infinity, we have to construct a distributional counterpart of the semigroup property, in order to compute high order fractional Laplacians for solutions of the problem given in (55).

First of all, we remark that a solution u_s to (55) for a fixed cone C belongs to \mathcal{L}^1_s since $0 \leq u_s(x) \leq |x|^{\gamma_s(C)}$ in \mathbb{R}^n with $\gamma_s(C) \in (0, 2s)$. Moreover, by the homogeneity one can rewrite the norm (61) in the following way

$$\begin{aligned} \|u_s\|_{\mathcal{L}^1_s} &= \int_{\mathbb{R}^n} \frac{u_s(x)}{(1+|x|)^{n+2s}} \mathrm{d}x = \int_{S^{n-1}} u_s \mathrm{d}\sigma \int_0^\infty \frac{\rho^{n-1+\gamma_s(C)}}{(1+\rho)^{n+2s}} \mathrm{d}\rho \\ &= \frac{\Gamma(n+\gamma_s(C))\Gamma(2s-\gamma_s(C))}{\Gamma(n+2s)} \int_{S^{n-1}} u_s \mathrm{d}\sigma. \end{aligned}$$

In the recent paper [40] the authors introduced a new notion of fractional Laplacian applying to a wider class of functions which grow more than linearly at infinity. This is achieved by defining an equivalence class of functions modulo polynomials of a fixed order. However, it can be hardly exploited to the solutions of (55) as they annihilate on a set of nonempty interior.

As shown in [8, Definition 3.6], if we consider a smooth function with compact support $\varphi \in C_c^{\infty}(\mathbb{R}^n)$ (or $\varphi \in C_c^2(\mathbb{R}^n)$), we can define the distribution k^{2s} by the formula

$$(-\Delta)^s \varphi(0) = (k^{2s}, \varphi).$$

By this definition, it follows that $(-\Delta)^s \varphi(x) = k^{2s} * \varphi(x)$.

Definition 2.3.1. [8, Definition 3.7] For $u \in \mathcal{L}_s^1$ we define the *distributional fractional Laplacian* $(-\widetilde{\Delta})^s u$ by the formula

$$((-\Delta)^s u, \varphi) = (u, (-\Delta)^s \varphi), \quad \forall \varphi \in C_c^{\infty}(\mathbb{R}^n).$$

In particular, since given an open subset $D \subset \mathbb{R}^n$ and $u \in C^2(D) \cap \mathcal{L}^1_s$, the fractional Laplacian exists as a continuous function of $x \in D$ and $(-\tilde{\Delta})^s u = (-\Delta)^s u$ as a distribution in D [8, Lemma 3.8], through the Chapter we will always use $(-\Delta)^s$ both for the classic and the distributional fractional Laplacian. The following is a useful tool to compute the distributional fractional Laplacian.

Lemma 2.3.2. [8, Lemma 3.3] Assume that

$$\iint_{|y-x|>\varepsilon} \frac{|f(x)g(y)|}{|y-x|^{n+2s}} \mathrm{d}x\mathrm{d}y < +\infty \quad and \quad \int_{\mathbb{R}^n} |f(x)g(x)| \,\mathrm{d}x < +\infty, \tag{68}$$

then $((-\Delta)^s_{\varepsilon}f,g) = (f, (-\Delta)^s_{\varepsilon}g)$. Moreover if $f \in \mathcal{L}^1_s$ and $g \in C_c(\mathbb{R}^n)$ the assumptions (68) are satisfied for every $\varepsilon > 0$.

Before proving the semigroup property, we prove the following lemma which ensures the existence of the δ -Laplacian of the *s*-Laplacian, for $0 < \delta < 1$.

Lemma 2.3.3. Let u_s be solution of (55) with C a regular cone. Then we have $(-\Delta)^s u_s \in \mathcal{L}^1_{\delta}$ for any $\delta > 0$, i.e.

$$\int_{\mathbb{R}^n} \frac{|(-\Delta)^s u_s(x)|}{(1+|x|)^{n+2\delta}} \mathrm{d}x < +\infty.$$

Proof. Since the function u_s is *s*-harmonic in *C*, namely $(-\Delta)^s u_s(x) = 0$ for all $x \in C$, we can restrict the domain of integration to $\mathbb{R}^n \setminus C$.

By homogeneity and the results in [8], we have that the function $(-\Delta)^s u_s$ is $(\gamma_s - 2s)$ -homogeneous and in particular $x \mapsto (-\Delta)^s u_s(x)$ is a continuous negative function, for every $x \in D \subset \mathbb{R}^n \setminus C$. In order to compute the previous integral, we focus our attention on the restriction of the fractional Laplacian to the sphere S^{n-1} , in particular, we prove that there exists $\bar{\varepsilon} > 0$ and C > 0 such that

$$|(-\Delta)^{s} u_{s}(x)| \leq \frac{C}{\operatorname{dist}(x, \partial C)^{s}} \quad \forall x \in N_{\bar{\varepsilon}}(\partial C) \cap S^{n-1},$$
(69)

where $N_{\varepsilon}(\partial C) = \{x \in \mathbb{R}^n \setminus C : \operatorname{dist}(x, \partial C) \le \varepsilon\}$ is the tubular neighborhood of ∂C .

Hence, fixed R > 0 small enough, consider initially $\varepsilon < R$ and $x \in S^{n-1} \cap N_{\varepsilon}(\partial C)$: since $u_s(y) \le |y|^{\gamma_s}$ in \mathbb{R}^n and by (63) there exists a constant C > 0 such that for every $y \in C$ we have

$$u_s(y) \le C |y|^{\gamma_s - s} \operatorname{dist}(y, \partial C)^s,$$

it follows, defining $\delta(x) := \operatorname{dist}(x, \partial C) > 0$, that

$$\begin{aligned} |(-\Delta)^{s} u_{s}(x)| &= C(n,s) \int_{C \cap B_{R}(x)} \frac{u_{s}(y)}{|x-y|^{n+2s}} \mathrm{d}y + C(n,s) \int_{C \setminus B_{R}(x)} \frac{u_{s}(y)}{|x-y|^{n+2s}} \mathrm{d}y \\ &\leq C(n,s) \int_{C \cap B_{R}(x)} \frac{C |y|^{\gamma_{s}-s} \operatorname{dist}(y, \partial C)^{s}}{|x-y|^{n+2s}} \mathrm{d}y + C(n,s) \int_{C \setminus B_{R}(x)} \frac{|y|^{\gamma_{s}}}{|x-y|^{n+2s}} \mathrm{d}y. \end{aligned}$$

Since $C \cap B_R(x) \subset B_R(x) \setminus B_{\delta(x)}(x)$, we have

$$\begin{split} |(-\Delta)^s u_s(x)| &\leq C \int_{R \geq |x-y| \geq \delta(x)} \frac{|y|^{\gamma_s - s}}{|x-y|^{n+s}} \mathrm{d}y + \int_{|x-y| \geq R} \frac{(|x-y|+1)^{\gamma_s}}{|x-y|^{n+2s}} \mathrm{d}y \\ &\leq C \int_{R \geq |x-y| \geq \delta(x)} \frac{1}{|x-y|^{n+s}} \mathrm{d}y + \omega_{n-1} \int_R^\infty \frac{(t+1)^{\gamma_s}}{t^{1+2s}} \mathrm{d}t \\ &\leq C \int_{\delta(x)}^R \frac{1}{r^{1+s}} \mathrm{d}r + M \\ &\leq C \frac{1}{\operatorname{dist}(x, \partial C)^s} + M. \end{split}$$

Moreover, again since $s \in (0, 1)$, up to consider a smaller neighborhood $N_{\varepsilon}(\partial C)$, we obtain that there exists a constant $\bar{\varepsilon} > 0$ small enough and C > 0 such that

$$|(-\Delta)^s u_s(x)| \le \frac{C}{\operatorname{dist}(x, \partial C)^s} \quad \text{for every } x \in N_{\bar{\varepsilon}}(\partial C) \cap S^{n-1}.$$

Now, fixed $\delta > 0$ and considered $\bar{\varepsilon} > 0$ of (69), we have

$$\begin{split} \int_{\mathbb{R}^n \setminus C} \frac{|(-\Delta)^s u_s(x)|}{(1+|x|)^{n+2\delta}} \mathrm{d}x &= \int_{\mathbb{R}^n \setminus C} \frac{|x|^{\gamma_s - 2s} \left| (-\Delta)^s u_s\left(\frac{x}{|x|}\right) \right|}{(1+|x|)^{n+2\delta}} \mathrm{d}x \\ &= \int_0^\infty \int_{S^{n-1} \cap (\mathbb{R}^n \setminus C)} \frac{r^{\gamma_s - 2s} \left| (-\Delta)^s u_s(z) \right|}{(1+r)^{n+2\delta}} r^{n-1} \mathrm{d}\sigma(z) \mathrm{d}r \\ &= \int_0^\infty \frac{r^{n-1+\gamma_s - 2s}}{(1+r)^{n+2\delta}} \mathrm{d}r \int_{S^{n-1} \cap (\mathbb{R}^n \setminus C)} |(-\Delta)^s u_s(z)| \, \mathrm{d}\sigma. \end{split}$$

Since $\gamma_s \in (0,2s)$ and $s \in (0,1),$ it follows

$$\begin{split} \int_{\mathbb{R}^n \setminus C} \frac{|(-\Delta)^s u_s(x)|}{(1+|x|)^{n+2\delta}} \mathrm{d}x &\leq C \int_{S^{n-1} \cap N_{\tilde{\varepsilon}}(\partial C)} |(-\Delta)^s u_s(z)| \,\mathrm{d}\sigma + C \int_{((\mathbb{R}^n \setminus C) \setminus N_{\tilde{\varepsilon}}(\partial C)) \cap S^{n-1}} |(-\Delta)^s u_s(z)| \,\mathrm{d}\sigma \\ &\leq C \int_{S^{n-1} \cap N_{\tilde{\varepsilon}}(\partial C)} \frac{1}{\mathrm{dist}(z, \partial C)^s} \mathrm{d}\sigma + M \\ &< +\infty \end{split}$$

where in the second inequality we used that $z \mapsto (-\Delta)^s u_s(z)$ is continuous in every $A \subset \subset S^{n-1} \cap (\mathbb{R}^n \setminus C)$ and in the last one that $\operatorname{dist}(x, \partial C)^{-s} \in L^1(S^{n-1} \cap N_{\bar{\varepsilon}}(\partial C), \mathrm{d}\sigma)$. \Box

Proposition 2.3.4 (Distributional semigroup property). Let u_s be a solution of (55) with C a regular cone and consider $\delta \in (0, 1 - s)$. Then

$$(-\Delta)^{s+\delta}u_s = (-\Delta)^{\delta}[(-\Delta)^s u_s]$$
 in $\mathcal{D}'(C)$

or equivalently

$$((-\Delta)^{s+\delta}u_s,\varphi) = ((-\Delta)^{\delta}[(-\Delta)^s u_s],\varphi), \quad \forall \varphi \in C_c^{\infty}(C).$$

Proof. Since $|u_s(x)| \leq |x|^{\gamma_s}$, with $\gamma_s \in (0, 2s)$, it is easy to see that $u_s \in \mathcal{L}^1_s \cap C^2(C)$. Moreover, as we have already remarked, if $u_s \in \mathcal{L}^1_s$ then $u_s \in \mathcal{L}^1_{s+\delta}$ for every $\delta > 0$. In particular, $(-\Delta)^{s+\delta}u_s$ does exist and it is a continuous function of $x \in C$, for every $\delta \in (0, 1-s)$. By definition of the distributional fractional Laplacian, we obtain

$$((-\Delta)^{s+\delta}u_s,\varphi) = (u_s,(-\Delta)^{s+\delta}\varphi),$$

and since for $\varphi \in C_c^{\infty}(C) \subset S(\mathbb{R}^n)$ in the Schwarz space, the classic semigroup property holds, we obtain that

$$((-\Delta)^{s+\delta}u_s,\varphi) = (u_s,(-\Delta)^s[(-\Delta)^{\delta}\varphi]).$$

On the other hand, since by Lemma 2.3.3 we have $(-\Delta)^s u_s \in \mathcal{L}^1_{\delta}$, it follows

$$((-\Delta)^{\delta}_{\varepsilon}[(-\Delta)^{s}u_{s}],\varphi) = ((-\Delta)^{s}u_{s},(-\Delta)^{\delta}_{\varepsilon}\varphi)$$
(70)

for every $\varepsilon > 0$. Since $(-\Delta)^s u_s \in \mathcal{L}^1_{\delta}$ and $\varphi \in C^{\infty}_c(\mathbb{R}^n)$, the δ -Laplacian of $(-\Delta)^s u_s$ does exists in a distributional sense and hence the left hand side in (70) does converge to $((-\Delta)^{\delta}[(-\Delta)^s u_s], \varphi)$ as $\varepsilon \to 0$. Moreover the right hand side in (70) does converge to $((-\Delta)^s u_s, (-\Delta)^{\delta} \varphi)$ by the dominated convergence theorem, using Proposition 2.2.2 and Lemma 2.3.3 which give

$$\int_{\mathbb{R}^n} (-\Delta)^s u_s(x) (-\Delta)_{\varepsilon}^{\delta} \varphi(x) \mathrm{d}x \le \int_{\mathbb{R}^n} \frac{|(-\Delta)^s u_s(x)|}{(1+|x|)^{n+2\delta}} \mathrm{d}x < +\infty.$$

By the previous remarks,

$$((-\Delta)^{\delta}[(-\Delta)^{s}u_{s}],\varphi) = ((-\Delta)^{s}u_{s},(-\Delta)^{\delta}\varphi).$$

In order to conclude the proof of the distributional semigroup property, we need to show that

$$(u_s, (-\Delta)^s [(-\Delta)^{\delta} \varphi]) = ((-\Delta)^s u_s, (-\Delta)^{\delta} \varphi), \tag{71}$$

which is not a trivial equality, since $(-\Delta)^{\delta}\varphi \in C^{\infty}(\mathbb{R}^n)$ is no more compactly supported.

Let $\eta \in C_c^{\infty}(B_2(0))$ be a radial cutoff function such that $\eta \equiv 1$ in $B_1(0)$ and $0 \leq \eta \leq 1$ in $B_2(0)$, and define $\eta_R(x) = \eta(x/R)$, for R > 0. Obviously, since $u_s \eta_R \in C_c(\mathbb{R}^n)$ and $(-\Delta)^{\delta} \varphi \in \mathcal{L}_s^1$, by Lemma 2.3.2 we have

$$(u_s\eta_R, (-\Delta)^s_{\varepsilon}[(-\Delta)^{\delta}\varphi]) = ((-\Delta)^s_{\varepsilon}(u_s\eta_R), (-\Delta)^{\delta}\varphi)$$
(72)

for every $\varepsilon, R > 0$. First, for R > 0 fixed, we want to pass to the limit for $\varepsilon \to 0$. For the left hand side in (72), we get the convergence to $(u_s \eta_R, (-\Delta)^s [(-\Delta)^{\delta} \varphi])$ since we can apply the dominated convergence theorem. In fact

$$\int_{\mathbb{R}^n} u_s \eta_R(-\Delta)^s_{\varepsilon}[(-\Delta)^{\delta}\varphi] \le c \int_K (-\Delta)^{s+\delta}\varphi < +\infty,$$

where *K* denotes the support of $u_s\eta_R$. For the right hand side in (72) we observe that, for any $x \in \mathbb{R}^n$

$$(-\Delta)^s_{\varepsilon}(u_s\eta_R)(x) = \eta_R(x)(-\Delta)^s_{\varepsilon}u_s(x) + u_s(x)(-\Delta)^s_{\varepsilon}\eta_R(x) - I_{\varepsilon}(u_s,\eta_R)(x),$$

where

$$I_{\varepsilon}(u_s,\eta_R)(x) = C(n,s) \int_{\mathbb{R}^n \setminus B_{\varepsilon}(x)} \frac{(u_s(x) - u_s(y))(\eta_R(x) - \eta_R(y))}{|x - y|^{n+2s}} \mathrm{d}y$$

Obviously the first term $((-\Delta)^s_{\varepsilon}u_s, \eta_R(-\Delta)^{\delta}\varphi) \rightarrow ((-\Delta)^s u_s, \eta_R(-\Delta)^{\delta}\varphi)$ by definition of the distributional *s*-Laplacian, since $u_s \in \mathcal{L}^1_s$ and $\eta_R(-\Delta)^{\delta}\varphi \in C^{\infty}_c(\mathbb{R}^n)$. The second term $(u_s(-\Delta)^s_{\varepsilon}\eta_R, (-\Delta)^{\delta}\varphi) \rightarrow (u_s(-\Delta)^s\eta_R, (-\Delta)^{\delta}\varphi)$ by dominated convergence, since

$$\int_{\mathbb{R}^n} u_s(-\Delta)^s_{\varepsilon} \eta_R(-\Delta)^{\delta} \varphi \mathrm{d}x \le c \int_{\mathbb{R}^n} \frac{u_s(x)}{(1+|x|)^{n+2s}} \mathrm{d}x$$

Finally, the last term $(I_{\varepsilon}(u_s, \eta_R), (-\Delta)^{\delta}\varphi) \to (I(u_s, \eta_R), (-\Delta)^{\delta}\varphi)$ by dominated convergence, since

$$\int_{\mathbb{R}^n} I_{\varepsilon}(u_s, \eta_R) (-\Delta)^{\delta} \varphi \mathrm{d}x \le C \int_{\mathbb{R}^n} \left| (-\Delta)^{\delta} \varphi \right| \mathrm{d}x$$

which is integrable by Proposition 2.2.2. Finally, passing to the limit for $\varepsilon \to 0$, from (72) we get

$$(u_s\eta_R, (-\Delta)^s[(-\Delta)^\delta\varphi]) = ((-\Delta)^s(u_s\eta_R), (-\Delta)^\delta\varphi),$$
(73)

for every R > 0.

Now we want to prove (71), concluding this proof, by passing to the limit in (73) for $R \to +\infty$. Since we know, by dominated convergence, that the left hand side converges to $(u_s, (-\Delta)^s (-\Delta)^{\delta} \varphi)$ for $R \to \infty$, we focus our attention on the other one. At this point, we need to prove that for any $\varphi \in C_c^{\infty}(C)$,

$$\int_{\mathbb{R}^n} (-\Delta)^s (u_s \eta_R) (-\Delta)^\delta \varphi \longrightarrow \int_{\mathbb{R}^n} (-\Delta)^s u_s (-\Delta)^\delta \varphi, \tag{74}$$

as $R \to +\infty$. First of all, we remark that $(-\Delta)^s(u_s\eta_R) \to (-\Delta)^s u_s$ in $L^1_{\text{loc}}(\mathbb{R}^n)$. In fact, let $K \subset \mathbb{R}^n$ be a compact set. There exists $\overline{r} > 0$ such that $K \subset B_{\overline{r}}$. Then, considering any radius $R > \overline{r}$, $\eta_R(x) = 1$ for any $x \in K$. Hence, for any $R > \overline{r}$, using the fact that $u_s(x) = |x|^{\gamma_s} u_s(x/|x|)$, we obtain

$$\begin{split} \int_{K} |(-\Delta)^{s}(u_{s}\eta_{R})(x) - (-\Delta)^{s}u_{s}(x)| \mathrm{d}x &= \int_{K} \mathrm{d}x \left| C(n,s) \mathrm{PN} \int_{\mathbb{R}^{n}} \frac{u_{s}(x)\eta_{R}(x) - u_{s}(y)\eta_{R}(y) + u_{s}(y) - u_{s}(x)}{|x - y|^{n + 2s}} \mathrm{d}x \right| \\ &= C(n,s) \int_{K} \mathrm{d}x \left(\mathrm{PN} \int_{C \setminus B_{R}} \frac{u_{s}(y)[1 - \eta_{R}(y)]}{|x - y|^{n + 2s}} \mathrm{d}y \right) \\ &\leq C(n,s) \int_{K} \mathrm{d}x \left(\mathrm{PN} \int_{C \setminus B_{R}} \frac{|y|^{\gamma_{s}}}{(|y| - \overline{\tau})^{n + 2s}} \mathrm{d}y \right) \\ &\leq C(n,s) \int_{K} \mathrm{d}x \left(\mathrm{PN} \int_{C \setminus B_{R}} \frac{|y|^{\gamma_{s}}}{|y|^{n + 2s}(1 - \frac{\overline{r}}{R})^{n + 2s}} \mathrm{d}y \right) \\ &= C \left(\frac{R}{R - \overline{\tau}} \right)^{n + 2s} \lim_{\rho \to +\infty} \int_{R}^{\rho} \frac{1}{r^{2s - \gamma_{s} + 1}} \mathrm{d}r \\ &= C \left(\frac{R}{R - \overline{\tau}} \right)^{n + 2s} \frac{1}{R^{2s - \gamma_{s}}} \longrightarrow 0, \end{split}$$

as $R \to +\infty$. Hence we obtain also pointwise convergence almost everywhere. Moreover, we can give the following expression

$$(-\Delta)^{s}(u_{s}\eta_{R})(x) = \eta_{R}(x)(-\Delta)^{s}u_{s}(x) + C(n,s)\text{P.V.}\int_{\mathbb{R}^{n}} u_{s}(y)\frac{\eta_{R}(x) - \eta_{R}(y)}{|x-y|^{n+2s}}\mathrm{d}y.$$
 (75)

We remark that $\eta_R(x)(-\Delta)^s u_s(x) \to (-\Delta)^s u_s(x)$ and $\int_{\mathbb{R}^n} u_s(y) \frac{\eta_R(x) - \eta_R(y)}{|x-y|^{n+2s}} dy \to 0$ pointwisely. Moreover we can dominate the first term in the following way

$$\eta_R(x)(-\Delta)^s u_s(x) \le (-\Delta)^s u_s(x),$$

and

$$\int_{\mathbb{R}^n} (-\Delta)^s u_s(x) (-\Delta)^{\delta} \varphi(x) \mathrm{d}x < +\infty$$

since $(-\Delta)^s u_s \in \mathcal{L}^1_{\delta}$ and using Proposition 2.2.2 over $\varphi \in C_c^{\infty}(C)$. In order to prove (74), we want to apply the dominated convergence theorem, and hence we need the following condition for any R > 0

$$I := \left| \int_{\mathbb{R}^n} (-\Delta)^{\delta} \varphi(x) \left(\text{P.V.} \int_{\mathbb{R}^n} u_s(y) \frac{\eta_R(x) - \eta_R(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right) \mathrm{d}x \right| \le c.$$

Therefore, we will obtain a stronger condition; that is, the existence of a value k>0 such that for any R>1

$$I \le \frac{c}{R^k}.$$

We split the region of integration $\mathbb{R}^n \times \mathbb{R}^n$ into five different parts; that is,

$$\Omega_1 := (\mathbb{R}^n \setminus B_{2R}) \times \mathbb{R}^n, \ \Omega_2 := B_{2R} \times B_{2R}, \ \Omega_3 := (B_{2R} \setminus B_R) \times (B_{3R} \setminus B_{2R}),$$
$$\Omega_4 := (B_{2R} \setminus B_R) \times (\mathbb{R}^n \setminus B_{3R}), \ \Omega_5 := B_R \times (\mathbb{R}^n \setminus B_{2R}).$$

First of all, we remark that $(-\Delta)^s \eta_R(x) = R^{-2s}(-\Delta)^s \eta(x/R)$ and also that $||(-\Delta)^s \eta||_{L^{\infty}(\mathbb{R}^n)} < +\infty$. For the first term, using the fact that $\eta_R(x) - \eta_R(y) = 0$ if $(x, y) \in (\mathbb{R}^n \setminus B_{2R}) \times (\mathbb{R}^n \setminus B_{2R})$

$$I_{1} := \int_{\mathbb{R}^{n} \setminus B_{2R}} |(-\Delta)^{\delta} \varphi(x)| \left| \int_{\mathbb{R}^{n}} u_{s}(y) \frac{\eta_{R}(x) - \eta_{R}(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \mathrm{d}x$$

$$\leq \int_{\mathbb{R}^{n} \setminus B_{2R}} \left| (-\Delta)^{\delta} \varphi(x) \right| \left| \int_{B_{2R}} u_{s}(y) \frac{\eta_{R}(x) - \eta_{R}(y)}{|x - y|^{n + 2s}} \right| \mathrm{d}x$$

$$\leq \int_{\mathbb{R}^{n} \setminus B_{2R}} |(-\Delta)^{\delta} \varphi(x)| \left(\sup_{B_{2R}} u_{s} \right) |(-\Delta)^{s} \eta_{R}(x)| \mathrm{d}x$$

$$\leq \frac{c}{R^{2s - \gamma_{s}}} \int_{\mathbb{R}^{n}} \frac{1}{(1 + |x|)^{n + 2\delta}} \mathrm{d}x \leq \frac{c}{R^{2s - \gamma_{s}}}.$$

For the second term, using the fact that $\eta_R(x) - \eta_R(y) \ge 0$ if $(x, y) \in B_{2R} \times (\mathbb{R}^n \setminus B_{2R})$, we obtain as before

$$I_{2} := \int_{B_{2R}} |(-\Delta)^{\delta} \varphi(x)| \left| \int_{B_{2R}} u_{s}(y) \frac{\eta_{R}(x) - \eta_{R}(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \mathrm{d}x$$

$$\leq \int_{B_{2R}} |(-\Delta)^{\delta} \varphi(x)| \left(\sup_{B_{2R}} u_{s} \right) |(-\Delta)^{s} \eta_{R}(x)| \mathrm{d}x$$

$$\leq \frac{c}{R^{2s - \gamma_{s}}} \int_{\mathbb{R}^{n}} \frac{1}{(1 + |x|)^{n + 2\delta}} \mathrm{d}x \leq \frac{c}{R^{2s - \gamma_{s}}}.$$

For the third part

$$I_3 := \int_{B_{2R} \setminus B_R} \left| (-\Delta)^{\delta} \varphi(x) \right| \left| \int_{B_{3R} \setminus B_{2R}} u_s(y) \frac{\eta_R(x) - \eta_R(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \mathrm{d}x,$$

we consider the following change of variables $\xi = x/R \in B_2 \setminus B_1$ and $\zeta = y/R \in B_3 \setminus B_2$. Hence, using the γ_s -homogeneity of u_s and the definition of our cut-off functions, we obtain

$$I_3 \le \frac{R^{2n}}{R^{n+2s-\gamma_s}} \iint_{(B_2 \setminus B_1) \times (B_3 \setminus B_2)} |(-\Delta)^{\delta} \varphi(R\xi)| u_s(\zeta) \frac{\eta(\xi) - \eta(\zeta)}{|\xi - \zeta|^{n+2s}} \mathrm{d}\xi \mathrm{d}\zeta.$$

We use the fact that $u_s \in C^{0,s}(B_3 \setminus B_1)$ (see (63) proved in [67]) and the cut off function $\eta \in \text{Lip}(B_3 \setminus B_1)$; that is, there exists a constant c > 0 such that

$$|u_s(\xi) - u_s(\zeta)| \le c|\xi - \zeta|^s \quad \text{and} \quad |\eta(\xi) - \eta(\zeta)| \le c|\xi - \zeta|, \tag{76}$$

for every $\xi, \zeta \in B_3 \setminus B_1$. Hence,

$$I_{3} \leq \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(B_{3}\setminus B_{2})} |(-\Delta)^{\delta}\varphi(R\xi)| \frac{|u_{s}(\zeta) - u_{s}(\xi)| |\eta(\xi) - \eta(\zeta)|}{|\xi - \zeta|^{n+2s}} \mathrm{d}\xi \mathrm{d}\zeta + \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(B_{3}\setminus B_{2})} |(-\Delta)^{\delta}\varphi(R\xi)| u_{s}(\xi) \frac{|\eta(\xi) - \eta(\zeta)|}{|\xi - \zeta|^{n+2s}} \mathrm{d}\xi \mathrm{d}\zeta = J_{1} + J_{2}.$$

By (76), we obtain

$$J_{1} \leq c \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\backslash B_{1})\times(B_{3}\backslash B_{2})} |(-\Delta)^{\delta}\varphi(R\xi)| \frac{|\xi-\zeta|^{s+1}}{|\xi-\zeta|^{n+2s}} d\xi d\zeta$$

$$\leq c \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\backslash B_{1})\times(B_{3}\backslash B_{2})} \frac{1}{(1+R|\xi|)^{n+2\delta}} \frac{1}{|\xi-\zeta|^{n+s-1}} d\xi d\zeta$$

$$\leq \frac{c}{R^{2s+2\delta-\gamma_{s}}} \iint_{(B_{2}\backslash B_{1})\times(B_{3}\backslash B_{2})} \frac{1}{|\xi-\zeta|^{n+s-1}} d\xi d\zeta \leq \frac{c}{R^{2s+2\delta-\gamma_{s}}}.$$

72 ON *s*-harmonic functions on cones

Moreover, using other two changes of variable $(\xi, \zeta) \mapsto (\xi, \xi + h)$ and $(\xi, \zeta) \mapsto (\xi, \xi - h)$, we obtain

$$J_{2} \leq \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(B_{3}\setminus B_{2})} |(-\Delta)^{\delta}\varphi(R\xi)|u_{s}(\xi)\frac{\eta(\xi)-\eta(\zeta)}{|\xi-\zeta|^{n+2s}} \mathrm{d}\xi \mathrm{d}\zeta$$

$$\leq \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(B_{3}\setminus B_{2})} \frac{1}{(1+R|\xi|)^{n+2\delta}} u_{s}(\xi)\frac{\eta(\xi)-\eta(\zeta)}{|\xi-\zeta|^{n+2s}} \mathrm{d}\xi \mathrm{d}\zeta$$

$$\leq \frac{c}{R^{2s+2\delta-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(B_{3}\setminus B_{2})} \frac{\eta(\xi)-\eta(\zeta)}{|\xi-\zeta|^{n+2s}} \mathrm{d}\xi \mathrm{d}\zeta$$

$$\leq \frac{c}{R^{2s+2\delta-\gamma_{s}}} \left(c + \iint_{(B_{2}\setminus B_{1})\times B_{\varepsilon}} \frac{<\nabla^{2}\eta(\xi)h,h>}{|h|^{n+2s}} \mathrm{d}\xi \mathrm{d}h\right) \leq \frac{c}{R^{2s+2\delta-\gamma_{s}}}.$$

For the fourth part

$$I_4 := \int_{B_{2R} \setminus B_R} \left| (-\Delta)^{\delta} \varphi(x) \right| \left| \int_{\mathbb{R}^n \setminus B_{3R}} u_s(y) \frac{\eta_R(x) - \eta_R(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \mathrm{d}x,$$

we consider, as before, the following change of variables $\xi = x/R \in B_2 \setminus B_1$ and $\zeta = y/R \in \mathbb{R}^n \setminus B_3$. Hence,

$$I_{4} \leq c \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(\mathbb{R}^{n}\setminus B_{3})} |(-\Delta)^{\delta}\varphi(R\xi)| \frac{|\zeta|^{\gamma_{s}}}{|\zeta-\xi|^{n+2s}} d\xi d\zeta$$

$$\leq c \frac{R^{2n}}{R^{n+2s-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(\mathbb{R}^{n}\setminus B_{3})} \frac{1}{(1+R|\xi|)^{n+2\delta}} \frac{|\zeta|^{\gamma_{s}}}{|\zeta-\frac{2\zeta}{|\zeta|}|^{n+2s}} d\xi d\zeta$$

$$\leq \frac{c}{R^{2s+2\delta-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(\mathbb{R}^{n}\setminus B_{3})} \frac{|\zeta|^{\gamma_{s}}}{|\zeta|^{n+2s}(1-\frac{2}{|\zeta|})^{n+2s}} d\xi d\zeta$$

$$\leq \frac{c}{R^{2s+2\delta-\gamma_{s}}} \iint_{(B_{2}\setminus B_{1})\times(\mathbb{R}^{n}\setminus B_{3})} \frac{1}{|\zeta|^{n+2s-\gamma_{s}}} d\xi d\zeta \leq \frac{c}{R^{2s+2\delta-\gamma_{s}}}.$$

Eventually, we consider the last term

$$I_5 := \int_{B_R} \left| (-\Delta)^{\delta} \varphi(x) \right| \left| \int_{\mathbb{R}^n \setminus B_{2R}} u_s(y) \frac{\eta_R(x) - \eta_R(y)}{|x - y|^{n + 2s}} \mathrm{d}y \right| \mathrm{d}x.$$

Hence we obtain

$$\begin{split} I_5 &\leq c \int_{B_R} |(-\Delta)^{\delta} \varphi(x)| \left(\int_{\mathbb{R}^n \setminus B_{2R}} \frac{|y|^{\gamma_s}}{|y-x|^{n+2s}} \mathrm{d}y \right) \mathrm{d}x \\ &\leq c \int_{B_R} |(-\Delta)^{\delta} \varphi(x)| \left(\int_{\mathbb{R}^n \setminus B_{2R}} \frac{|y|^{\gamma_s}}{|y-\frac{Ry}{|y|}|^{n+2s}} \mathrm{d}y \right) \mathrm{d}x \\ &\leq c \int_{B_R} |(-\Delta)^{\delta} \varphi(x)| \left(\int_{\mathbb{R}^n \setminus B_{2R}} \frac{1}{|y|^{n+2s-\gamma_s}} \mathrm{d}y \right) \mathrm{d}x \\ &\leq c \left(\int_{\mathbb{R}^n} \frac{1}{(1+|x|)^{n+2\delta}} \mathrm{d}x \right) \left(\int_{2R}^{+\infty} \frac{1}{r^{1+2s-\gamma_s}} \mathrm{d}r \right) \frac{c}{R^{2s-\gamma_s}}, \end{split}$$

which it implies the desired result.

At this point, fixed $s \in (0, 1)$, by the distributional semigroup property we can compute easily high order fractional Laplacians $(-\Delta)^{s+\delta}$ viewing it as the δ -Laplacian of the *s*-Laplacian.

Corollary 2.3.5. Let C be a regular cone. For every $\delta \in (0, 1 - s)$, the solution u_s of (55) is $(s + \delta)$ -superharmonic in C in the sense of distribution, i.e.

$$((-\Delta)^{s+\delta}u_s,\varphi) \ge 0$$

for every test function $\varphi \in C_c^{\infty}(C)$ nonnegative in C. Moreover, u_s is also superharmonic in C in the sense of distribution, i.e.

$$(-\Delta u_s,\varphi) \ge 0$$

for every test function $\varphi \in C_c^{\infty}(C)$ nonnegative in C.

Proof. As said before, the facts that $u_s \in \mathcal{L}^1_{s+\delta}$ and $u_s \in C^2(A)$ for every $A \subset C$ ensure the existence of the $(-\Delta)^{s+\delta}u_s$ and the continuity of the map $x \mapsto (-\Delta)^{s+\delta}u_s(x)$ for every $x \in A \subset C$. Hence at this point, the only part we need to prove is the positivity of the $(s+\delta)$ -Laplacian in the sense of the distribution, which is a direct consequence of the previous result. Indeed, since u_s is a solution of the problem (55), by Proposition 2.3.4 we know that for every $\varphi \in C_c^{\infty}(C)$ we have

$$((-\Delta)^{s+\delta}u_s,\varphi) = ((-\Delta)^{\delta}[(-\Delta)^s u_s],\varphi)$$

= $\int_C \varphi(x) \text{ P.V.} \int_{\mathbb{R}^n} \frac{(-\Delta)^s u_s(x) - (-\Delta)^s u_s(y)}{|x-y|^{n+2\delta}} \mathrm{d}y \mathrm{d}x.$

74 ON *s*-harmonic functions on cones

where $(-\Delta)^{\delta}[(-\Delta)^{s}u_{s}]$ is well defined since that $(-\Delta)^{s}u_{s} \equiv 0 \in C^{2}(A)$ for every $A \subset C$ and, by Lemma 2.3.3, $(-\Delta)^{s}u_{s} \in \mathcal{L}^{1}_{\delta}$ for every $\delta \in (0, 1-s)$.

Consider now nonnegative test function $\varphi \ge 0$ in C, since $(-\Delta)^s u_s(x) = 0$ for every $x \in C$, we have for every $x \in \mathbb{R}^n \setminus \overline{C}$

$$(-\Delta)^s u_s(x) = -\int_C \frac{u_s(y)}{|x-y|^{n+2s}} \mathrm{d}y \le 0.$$

Similarly,

$$\left((-\Delta)^{\delta}[(-\Delta)^{s}u_{s}],\varphi\right) = \int_{C}\varphi(x)\int_{\mathbb{R}^{n}}\frac{-(-\Delta)^{s}u_{s}(y)}{\left|x-y\right|^{n+2\delta}}\mathrm{d}y\mathrm{d}x \ge 0,$$

since the support of φ is compact in the cone C, and so there exists $\varepsilon > 0$ such that $|x - y| > \varepsilon$ in the above integral. We have obtained that for any $\delta \in (0, 1 - s)$ and any nonnegative $\varphi \in C_c^{\infty}(C)$

$$((-\Delta)^{s+\delta}u_s,\varphi) \ge 0,$$

then, passing to the limit for $\delta \to 1-s$, the function u_s is superharmonic in the distributional sense

$$0 \le \lim_{\delta \to 1-s} ((-\Delta)^{s+\delta} u_s, \varphi) = \lim_{\delta \to 1-s} (u_s, (-\Delta)^{s+\delta} \varphi) = (u_s, -\Delta\varphi) = (-\Delta u_s, \varphi).$$

2.3.2 Monotonicity of $s \mapsto \gamma_s(C)$

The following proposition is a consequence of Corollary 2.3.5 and it follows essentially the proof of Lemma 2 in [9].

Proposition 2.3.6. For any fixed regular cone C with vertex in 0, the map $s \mapsto \gamma_s(C)$ is monotone non decreasing in (0, 1).

Proof. Fixed the cone C, let us denote with γ_s and $\gamma_{s+\delta}$ respectively the homogeneities of u_s and $u_{s+\delta}$. Let us suppose by contradiction that $\gamma_s > \gamma_{s+\delta}$ for a $\delta \in (0, 1-s)$, and let us consider the function

$$h(x) = u_{s+\delta}(x) - u_s(x) \quad \text{in } \mathbb{R}^n,$$

where u_s is the homogeneous solution of (55) and $u_{s+\delta}$ is the unique, up to multiplicative constants, nonnegative nontrivial homogeneous and continuous in \mathbb{R}^n solution for

$$\begin{cases} (-\Delta)^{s+\delta}u = 0, & \text{in } C, \\ u = 0, & \text{in } \mathbb{R}^n \setminus C, \end{cases}$$

of the form

$$u_{s+\delta}(x) = |x|^{\gamma_{s+\delta}} u_{s+\delta}\left(\frac{x}{|x|}\right).$$

The function h is continuous in \mathbb{R}^n and h(x) = 0 in $\mathbb{R}^n \setminus C$. We want to prove that $h(x) \leq 0$ in $\mathbb{R}^n \setminus (C \cap B_1)$. Since h = 0 outside the cone, we can consider only what happens in $C \setminus B_1$. As we already quoted, we have

$$c_1(s)|x|^{\gamma_s-s}\operatorname{dist}(x,\partial C)^s \le u_s(x) \le c_2(s)|x|^{\gamma_s-s}\operatorname{dist}(x,\partial C)^s,\tag{77}$$

for any $x \in \overline{C} \setminus \{0\}$, and there exist two constants $c_1(s + \delta), c_2(s + \delta) > 0$ such that

$$c_1(s+\delta)|x|^{\gamma_{s+\delta}-(s+\delta)}\operatorname{dist}(x,\partial C)^{s+\delta} \le u_{s+\delta}(x) \le c_2(s+\delta)|x|^{\gamma_{s+\delta}-(s+\delta)}\operatorname{dist}(x,\partial C)^{s+\delta}.$$

We can choose u_s and $u_{s+\delta}$ so that $c := c_1(s) = c_2(s+\delta)$ since they are defined up to a multiplicative constant. Then, for any $x \in C \setminus B_1$, since $|x|^{\gamma_{s+\delta}} \leq |x|^{\gamma_s}$, we have

$$h(x) \le c|x|^{\gamma_s} \operatorname{dist}(x, \partial C)^s \left[\frac{\operatorname{dist}(x, \partial C)^{\delta}}{|x|^{\delta}} - 1 \right] \le 0.$$
(78)

In fact, if we take x such that $dist(x, \partial C) \leq 1$, then (78) follows by

$$\frac{\operatorname{dist}(x,\partial C)^{\delta}}{|x|^{\delta}} - 1 \leq \operatorname{dist}(x,\partial C)^{\delta} - 1 \leq 0.$$

Instead, if we consider x so that $dist(x, \partial C) > 1$, then $dist(x, \partial C)^{\delta} < |x|^{\delta}$ and hence (78) follows.

Now we want to show that there exists a point $x_0 \in C \cap B_1$ such that $h(x_0) > 0$. Let us take a point $\overline{x} \in S^{n-1} \cap C$ and let $\alpha := u_{s+\delta}(\overline{x}) > 0$ and $\beta := u_s(\overline{x}) > 0$. Hence, there exists a small r > 0 so that $\alpha r^{\gamma_{s+\delta}} > \beta r^{\gamma_s}$, and so, taking x_0 with $|x_0| = r$ and so that $\frac{x_0}{|x_0|} = \overline{x}$, we obtain $h(x_0) > 0$.

If we consider the restriction of h to $\overline{C \cap B_1}$, which is continuous on a compact set, for the considerations done before and for the Weierstrass Theorem, there exists a maximum point $x_1 \in C \cap B_1$ for the function h which is global in \mathbb{R}^n and is strict at least in a set of positive measure. Hence,

$$(-\Delta)^{s+\delta}h(x_1) = C(n,s) \text{ P.V.} \int_{\mathbb{R}^n} \frac{h(x_1) - h(y)}{|x_1 - y|^{n+2(s+\delta)}} \,\mathrm{d}y > 0,$$

and since $(-\Delta)^{s+\delta}h$ is a continuous function in the open cone, there exists an open set $U(x_1)$ with $\overline{U(x_1)} \subset C$ such that

$$(-\Delta)^{s+\delta}h(x) > 0 \quad \forall x \in U(x_1).$$

But thanks to Corollary 2.3.5 we obtain a contradiction since for any nonnegative $\varphi \in C_c^{\infty}(U(x_1))$

$$((-\Delta)^{s+\delta}h,\varphi) = ((-\Delta)^{s+\delta}u_{s+\delta},\varphi) - ((-\Delta)^{s+\delta}u_s,\varphi) = -((-\Delta)^{s+\delta}u_s,\varphi) \le 0.$$

With the same argument of the previous proof we can show also the following useful upper bound.

Proposition 2.3.7. For any fixed regular cone C with vertex in 0 and any $s \in (0,1)$, $\gamma_s(C) \leq \gamma(C)$.

Proof. Seeking a contradiction, we suppose that there exists $s \in (0, 1)$ such that $\gamma_s > \gamma$. Hence we define the function

$$h(x) = u(x) - u_s(x) \quad \text{in } \mathbb{R}^n,$$

where u_s and u are respectively solutions to (55) and

$$\begin{cases} -\Delta u = 0, & \text{in } C, \\ u = 0, & \text{in } \mathbb{R}^n \setminus C. \end{cases}$$
(79)

We recall that these solutions are unique, up to multiplicative constants, nonnegative nontrivial homogeneous and continuous in \mathbb{R}^n of the form

$$u(x) = |x|^{\gamma} u\left(\frac{x}{|x|}\right), \qquad u_s(x) = |x|^{\gamma_s} u_s\left(\frac{x}{|x|}\right).$$

for some $\gamma_s \in (0, 2s)$ and $\gamma \in (0, +\infty)$. The function h is continuous in \mathbb{R}^n and h(x) = 0 in $\mathbb{R}^n \setminus C$. We want to prove that $h(x) \leq 0$ in $\mathbb{R}^n \setminus (C \cap B_1)$. Since h = 0 outside the cone, we can consider only what happens in $C \setminus B_1$. So, there exist two constants $c_1(s), c_2(s) > 0$ such that, for any $x \in \overline{C} \setminus \{0\}$, it holds (77). Moreover there exist two constants $c_1, c_2 > 0$ such that,

$$c_1|x|^{\gamma-1}\operatorname{dist}(x,\partial C) \le u(x) \le c_2|x|^{\gamma-1}\operatorname{dist}(x,\partial C)$$

We can choose u_s and u so that $c := c_1(s) = c_2$ since they are defined up to a multiplicative constant. Then, for any $x \in C \setminus B_1$, since $|x|^{\gamma} \le |x|^{\gamma_s}$, we have

$$h(x) \le c |x|^{\gamma_s} \operatorname{dist}(x, \partial C)^s \left[\frac{\operatorname{dist}(x, \partial C)^{1-s}}{|x|^{1-s}} - 1 \right] \le 0,$$

with the same arguments of the previous proof.

Now we want to show that there exists a point $x_0 \in C \cap B_1$ such that $h(x_0) > 0$. Let us

take a point $\overline{x} \in S^{n-1} \cap C$ and let $\alpha := u(\overline{x}) > 0$ and $\beta := u_s(\overline{x}) > 0$. Hence, there exists a small r > 0 so that $\alpha r^{\gamma} > \beta r^{\gamma_s}$, and so, taking x_0 with $|x_0| = r$ and so that $\frac{x_0}{|x_0|} = \overline{x}$, we obtain $h(x_0) > 0.$

If we consider the restriction of h to $\overline{C \cap B_1}$, which is continuous on a compact set, for the considerations done before and for the Weierstrass Theorem, there exists at least a maximum point in $C \cap B_1$ for the function h which is global in \mathbb{R}^n . Moreover, since h cannot be constant on $C\cap B_1$ and it is of class C^2 inside the cone, there exists a global maximum $y\in C\cap B_1$ such that, up to a rotation, $\partial_{x_i x_i}^2 h(y) \leq 0$ for any i = 1, ..., n and $\check{\partial_{x_j x_j}^2} h(y) < 0$ for at least a coordinate direction. Hence

$$\Delta h(y) = \sum_{i=1}^{n} \partial_{x_i x_i}^2 h(y) < 0.$$

By the continuity of Δh in the open cone, there exists an open set U(y) with $\overline{U(y)} \subset C$ such that

$$\Delta h(x) < 0 \quad \forall x \in U(y).$$

Since, by Corollary 2.3.5 for any nonnegative $\varphi \in C_c^{\infty}(U(y))$

$$(-\Delta u_s, \varphi) \ge 0,$$

hence

$$(\Delta h, \varphi) = (\Delta u, \varphi) - (\Delta u_s, \varphi) = (-\Delta u_s, \varphi) \ge 0$$

and this is a contradiction.

2.3.3 Asymptotic behavior of $\frac{C(n,s)}{2s-\gamma_s(C)}$

Let us define for any regular cone C the limit

$$\mu(C) = \lim_{s \to 1^{-}} \frac{C(n,s)}{2s - \gamma_s(C)} \in [0, +\infty].$$

Obviously, thanks to the monotonicity of $s \mapsto \gamma_s(C)$ in (0, 1), this limit does exist, but we want to show that $\mu(C)$ can not be infinite. At this point, this situation can happen since $2s - \gamma_s(C)$ can converge to zero and we do not have enough information about this convergence. The study of this limit depends on the cone C itself and so we will consider separately the case of wide cones and narrow cones, which are respectively when $\gamma(C) < 2$ and when $\gamma(C) \geq 2$. In this Section, we prove this result just for regular cones, while in Section 2.4 we will extend the existence of a finite limit $\mu(C)$ to any unbounded cone, without the monotonicity result of Proposition 2.3.6.

Wide cones: $\gamma(C) < 2$

We remark that, fixed a wide cone $C \subset \mathbb{R}^n$, then there exists $\varepsilon > 0$ and $s_0 \in (0, 1)$, both depending on C, such that for any $s \in [s_0, 1)$

$$2s - \gamma_s(C) \ge \varepsilon > 0.$$

In fact we know that $s \mapsto \gamma_s(C)$ is monotone non decreasing in (0,1) and $0 < \gamma_s(C) \le \gamma(C) < 2$. Hence, defining $\overline{\gamma}(C) = \lim_{s \to 1} \gamma_s(C) \in (0,2)$ we can choose

$$s_0 := \frac{\overline{\gamma}(C) - 2}{4} + 1 \in (1/2, 1) \text{ and } \varepsilon := \frac{2 - \overline{\gamma}(C)}{2} > 0,$$

obtaining

$$2s - \gamma_s(C) \ge 2s_0 - \overline{\gamma}(C) = \varepsilon > 0.$$

As a consequence we obtain $\mu(C) = 0$ for any wide cone.

Narrow cones: $\gamma(C) \geq 2$

Before addressing the asymptotic analysis for any regular cone, we focus our attention on the spherical caps ones with "small" aperture. Hence, let us fix $\theta_0 \in (0, \pi/4)$ and for any $\theta \in (0, \theta_0]$, let

$$\lambda_1(\theta) := \lambda_1(\omega_{\theta}) = \min_{\substack{u \in H_0^1(S^{n-1} \cap C_{\theta}) \\ u \neq 0}} \frac{\int_{S^{n-1}} |\nabla_{S^{n-1}} u|^2 \, \mathrm{d}\sigma}{\int_{S^{n-1}} u^2 \mathrm{d}\sigma}.$$

We have that $\lambda_1(\theta) > 2n$, and hence the following problem is well defined

$$\mu_0(\theta) := \min_{\substack{u \in H_0^1(S^{n-1} \cap C_{\theta}) \\ u \neq 0}} \frac{\int_{S^{n-1}} |\nabla_{S^{n-1}} u|^2 - 2nu^2 \mathrm{d}\sigma}{\left(\int_{S^{n-1}} |u| \mathrm{d}\sigma\right)^2}.$$
(8o)

This number $\mu_0(\theta)$ is strictly positive and achieved by a nonnegative $\varphi \in H_0^1(S^{n-1} \cap C_\theta) \setminus \{0\}$ which is strictly positive on $S^{n-1} \cap C_\theta$ and is obviously solution to

$$\begin{cases} -\Delta_{S^{n-1}}\varphi = 2n\varphi + \mu_0(\theta) \int_{S^{n-1}} \varphi d\sigma & \text{in } S^{n-1} \cap C_{\theta}, \\ \varphi = 0 & \text{in } S^{n-1} \setminus C_{\theta}, \end{cases}$$
(81)

where $-\Delta_{S^{n-1}}$ is the Laplace-Beltrami operator on the unitary sphere S^{n-1} . Let now v be the 0-homogeneous extension of φ to the whole of \mathbb{R}^n and r(x) := |x|. Such a function will be solution to

$$\begin{cases} -\Delta v = \frac{2nv}{r^2} + \frac{\mu_0(\theta)}{r^2} \int_{S^{n-1}} v d\sigma & \text{in } C_{\theta}, \\ v = 0 & \text{in } \mathbb{R}^n \setminus C_{\theta}. \end{cases}$$
(82)

Since the spherical cap $C_{\theta} \cap S^{n-1}$ is an analytic submanifold of S^{n-1} and the data $(\partial C_{\theta} \cap S^{n-1}, 0, \partial_{\nu}\varphi)$ are not characteristic, by the classic theorem of Cauchy-Kovalevskaya we can extend the solution φ of (81) to a function $\tilde{\varphi}$, which is defined in a enlarged cone and it satisfies

$$\begin{cases} -\Delta_{S^{n-1}}\tilde{\varphi} = 2n\tilde{\varphi} + \mu_0(\theta) \int_{S^{n-1}} \varphi d\sigma & \text{in} \quad S^{n-1} \cap C_{\theta+\varepsilon} \\ \tilde{\varphi} = \varphi & \text{in} \quad S^{n-1} \cap C_{\theta}, \end{cases}$$

for some $\varepsilon > 0$. As in (82), we can define \tilde{v} as the 0-homogenous extension of $\tilde{\varphi}$. Finally, we introduce the following function

$$v_s(x) := r(x)^{\gamma_s^*(\theta)} v(x),$$

where the choice of the homogeneity exponent $\gamma_s^*(\theta) \in (0, 2s)$ will be suggested by the following important result.

Theorem 2.3.8. Let $\theta \in (0, \theta_0]$, then there exists $s_0 = s_0(\theta) \in (0, 1)$ such that

$$(-\Delta)^s v_s(x) \le 0 \qquad \text{in } C_{\theta},$$

for any $s \in [s_0, 1)$.

Proof. By the $\gamma_s^*(\theta)$ -homogeneity of v_s , it is sufficient to prove that $(-\Delta)^s v_s \leq 0$ on $C_{\theta} \cap S^{n-1}$, since $x \mapsto (-\Delta)^s v_s$ is $(\gamma_s^*(\theta) - 2s)$ -homogenous. In order to ease the notations, through the following computations we will simply use γ instead of $\gamma_s^*(\theta)$ and o(1) for the terms which converge to zero as s goes to 1. Hence, for $x \in S^{n-1} \cap C_{\theta}$, we have

$$(-\Delta)^{s} v_{s}(x) = |x|^{\gamma} (-\Delta)^{s} v(x) + v(x)(-\Delta)^{s} r^{\gamma}(x) - C(n,s) \int_{\mathbb{R}^{n}} \frac{(r^{\gamma}(x) - r^{\gamma}(y))(v(x) - v(y))}{|x - y|^{n+2s}} \mathrm{d}y$$

First for R > 0,

$$\begin{split} (-\Delta)^{s} r^{\gamma}(x) &= C(n,s) \int_{B_{R}(x)} \frac{|x|^{\gamma} - |y|^{\gamma}}{|x - y|^{n+2s}} \mathrm{d}y + C(n,s) \int_{\mathbb{R}^{n} \setminus B_{R}(x)} \frac{|x|^{\gamma} - |y|^{\gamma}}{|x - y|^{n+2s}} \mathrm{d}y \\ &= \frac{C(n,s)}{2} \int_{B_{R}(0)} \frac{2 |x|^{\gamma} - |x + z|^{\gamma} - |x - z|^{\gamma}}{|z|^{n+2s}} \mathrm{d}z + C(n,s) \int_{\mathbb{R}^{n} \setminus B_{R}(x)} \frac{1 - |y|^{\gamma}}{|x - y|^{n+2s}} \mathrm{d}y \\ &= -\frac{C(n,s)}{2} \int_{0}^{R} \frac{\rho^{2} \rho^{n-1}}{\rho^{n+2s}} \mathrm{d}\rho \int_{S^{n-1}} \langle \nabla^{2} |x|^{\gamma} z, z \rangle \mathrm{d}\sigma + o(1) + \\ &+ C(n,s) \left| S^{n-1} \right| \int_{R}^{\infty} \frac{1}{\rho^{1+2s}} \mathrm{d}\rho - C(n,s) \int_{\mathbb{R}^{n} \setminus B_{R}(x)} \frac{|y|^{\gamma}}{|x - y|^{n+2s}} \mathrm{d}y \\ &= -\frac{C(n,s)}{2} \frac{R^{2-2s}}{2-2s} \int_{S^{n-1}} \langle \nabla^{2} |x|^{\gamma} z, z \rangle \mathrm{d}\sigma + \\ &- C(n,s) \int_{R}^{\infty} \frac{\rho^{n-1+\gamma}}{\rho^{n+2s}} \int_{S^{n-1}} \left| \frac{x}{\rho} - \vartheta \right|^{\gamma} \mathrm{d}\omega(\vartheta) \mathrm{d}\rho + o(1). \end{split}$$

Since for every symmetric matrix A we have

$$\int_{S^{n-1}} \langle Az, z \rangle \mathrm{d}\sigma = \frac{\mathrm{tr}A}{n} \omega_{n-1}$$

where ω_{n-1} is the Lebesgue measure of the (n-1)-sphere S^{n-1} , we can simplify the first term since $\operatorname{tr} \nabla^2 |x|^{\gamma} = \Delta(|x|^{\gamma})$ and checking that $\left|\frac{x}{\rho} - \vartheta\right|^{\gamma} = 1 + \gamma \rho^{-1} \langle \vartheta, x \rangle + o(\rho^{-1})$ as $\rho \to \infty$ it follows

$$\begin{split} (-\Delta)^{s} r^{\gamma}(x) &= -\frac{C(n,s)}{2} \frac{R^{2-2s}}{2-2s} \frac{\Delta(|x|^{\gamma})\omega_{n-1}}{n} - C(n,s)\omega_{n-1} \int_{R}^{\infty} \frac{\rho^{n-1+\gamma}}{\rho^{n+2s}} d\rho + o(1) \\ &= -\frac{C(n,s)\omega_{n-1}}{4n(1-s)} \gamma(n-2+\gamma) |x|^{\gamma-2} R^{2-2s} - \frac{C(n,s)}{2s-\gamma} \omega_{n-1} R^{\gamma-2s} + o(1) \\ &= -\frac{C(n,s)\omega_{n-1}}{4n(1-s)} \gamma(n-2+\gamma) R^{2-2s} - \frac{C(n,s)}{2s-\gamma} \omega_{n-1} R^{\gamma-2s} + o(1) \\ &= -\frac{C(n,s)\omega_{n-1}}{4n(1-s)} \gamma(n-2+\gamma) - \frac{C(n,s)}{2s-\gamma} \omega_{n-1} + o(1), \end{split}$$

where in the last equality we choose $\gamma = \gamma_s^*(\theta)$ such that $\gamma_s^*(\theta) - 2s \to 0$ as s goes to 1. Similarly, if \tilde{v} is the 0-homogenous extension of v in an enlarged cone, which is such that $v \ge \tilde{v}$ and $v = \tilde{v}$ on $C_{\theta} \cap S^{n-1}$, it follows

$$\begin{split} (-\Delta)^{s}v(x) &= \frac{C(n,s)}{2} \int_{|z|<1} \frac{2v(x) - v(x+z) - v(x-z)}{|z|^{n+2s}} \mathrm{d}z + C(n,s) \int_{|x-y|>1} \frac{v(x) - v(y)}{|x-y|^{n+2s}} \mathrm{d}y \\ &\leq \frac{C(n,s)}{2} \int_{|z|<1} \frac{2\tilde{v}(x) - \tilde{v}(x+z) - \tilde{v}(x-z)}{|z|^{n+2s}} \mathrm{d}z + C(n,s) \int_{1}^{\infty} \frac{\rho^{n-1}}{\rho^{n+2s}} \int_{S^{n-1}} v(x) - v(y) \mathrm{d}\sigma \mathrm{d}\rho \\ &= -\frac{C(n,s)}{2} \int_{0}^{1} \frac{\rho^{n-1}\rho^{2}}{\rho^{n+2s}} \int_{S^{n-1}} \langle \nabla^{2}\tilde{v}(x)z, z \rangle \mathrm{d}\sigma \mathrm{d}\rho + o(1) \\ &= \frac{C(n,s)\omega_{n-1}}{4n(1-s)} (-\Delta)\tilde{v}(x) + o(1), \end{split}$$

where we can use that \tilde{v} solves

$$-\Delta \tilde{v} = 2n\tilde{v} + \mu_0 \int_{S^{n-1}} v \mathrm{d}\sigma$$

in the enlarged cap $S^{n-1} \cap C_{\theta+\varepsilon}$. Finally,

$$C(n,s) \int_{\mathbb{R}^n} \frac{(|x|^{\gamma} - |y|^{\gamma})(v(x) - v(y))}{|x - y|^{n+2s}} \mathrm{d}y = C(n,s) \left[\int_{|y| < 1} \frac{(1 - |y|^{\gamma})(v(x) - v(y))}{|x - y|^{n+2s}} \mathrm{d}y + \int_{|y| > 1} \frac{(1 - |y|^{\gamma})(v(x) - v(y))}{|x - y|^{n+2s}} \mathrm{d}y \right]$$

where the first term is o(1) since

$$\begin{split} \int_{0}^{1} (1-\rho^{\gamma})\rho^{n-1} \int_{S^{n-1}} \frac{v(x) - v(y)}{|x-\rho y|^{n+2s}} \mathrm{d}\sigma \mathrm{d}\rho &= \int_{0}^{1} (1-\rho^{\gamma})\rho^{n-1} \int_{S^{n-1}} (v(x) - v(y))(1+o(\rho)) \mathrm{d}\sigma \mathrm{d}\rho \\ &+ \int_{0}^{R} (1-\rho^{\gamma})\rho^{n-1} \int_{S^{n-1}} (v(x) - v(y))(n+2s)\rho\langle x, y\rangle \mathrm{d}\sigma \mathrm{d}\rho. \end{split}$$

Hence, we obtain

$$\begin{split} C(n,s) \int_{\mathbb{R}^n} \frac{(|x|^{\gamma} - |y|^{\gamma})(v(x) - v(y))}{|x - y|^{n + 2s}} \mathrm{d}y = C(n,s) \int_{|y| > 1} \frac{(1 - |y|^{\gamma})(v(x) - v(y))}{|x - y|^{n + 2s}} \mathrm{d}y + o(1) \\ = o(1) - C(n,s) \int_{|y| > 1} \frac{|y|^{\gamma} (v(x) - v(y))}{|x - y|^{n + 2s}} \mathrm{d}y + o(1) \\ = o(1) - C(n,s) \int_{1}^{\infty} \rho^{\gamma} \rho^{n - 1} \int_{S^{n - 1}} \frac{v(x) - v(y)}{|x - \rho y|^{n + 2s}} \mathrm{d}\sigma \mathrm{d}\rho \\ = o(1) - C(n,s) \int_{1}^{\infty} \rho^{-1 + \gamma - 2s} \int_{S^{n - 1}} (v(x) - v(y))(1 + o(\rho^{-1})) \mathrm{d}\sigma \\ - C(n,s) \int_{1}^{\infty} \rho^{-1 + \gamma - 2s} \int_{S^{n - 1}} (v(x) - v(y))(n + 2s) \langle y, x \rangle \rho^{-1} \mathrm{d}\sigma \\ = o(1) - \frac{C(n,s)\omega_{n - 1}}{2s - \gamma} v(x) + \frac{C(n,s)}{2s - \gamma} \int_{S^{n - 1}} v(y) \mathrm{d}\sigma. \end{split}$$

Hence, recalling that $\gamma=\gamma^*_s(\theta),$ for $x\in S^{n-1}\cap C_\theta$ we have

$$\begin{aligned} (-\Delta)^s v_s(x) &\leq \left(\mu_0(\theta) \frac{C(n,s)\omega_{n-1}}{4n(1-s)} - \frac{C(n,s)}{2s - \gamma_s^*(\theta)}\right) \int_{S^{n-1}} v_s \mathrm{d}\sigma + \frac{C(n,s)\omega_{n-1}}{4n(1-s)} (n + \gamma_s^*(\theta))(2 - \gamma_s^*(\theta))v_s \\ &\leq \left(\mu_0(\theta) - \frac{C(n,s)}{2s - \gamma_s^*(\theta)}\right) \int_{S^{n-1}} v_s \mathrm{d}\sigma + o(1) \end{aligned}$$

where o(1) is uniform with respect to $\gamma_s^*(\theta)$ as $s \to 1$. In order to obtain a negative right hand side, it is sufficient to choose $\gamma_s^*(\theta) < 2s$ in such a way to make the denominator $2s - \gamma_s^*(\theta)$ small enough and the quotient $\frac{C(n,s)}{2s - \gamma_s^*(\theta)}$ still bounded.

The previous result suggestes the following choice of the homogeneity exponent

$$\gamma_s^*(\theta) := 2s - s \frac{C(n,s)}{\mu_0(\theta)}.$$

We can finally prove the main result of this Section.

Corollary 2.3.9. For any regular cone C we get $\mu(C) < +\infty$.

Proof. We will show that $\mu(\theta) < +\infty$ for any $\theta \in (0, \theta_0]$. Then, fixed an unbounded regular cone C, there exists a spherical cone C_{θ} such that $\theta \in (0, \theta_0]$ and $C_{\theta} \subset C$. Since by inclusion $\gamma_s(C) < \gamma_s(\theta)$, we obtain

$$\mu(C) \le \mu(\theta) < +\infty.$$

We want to show that fixed $\theta \in (0, \theta_0]$, $\gamma_s(\theta) \leq \gamma_s^*(\theta)$ for any $s \in [s_0(\theta), 1)$, where the choice of $s_0(\theta) \in (0, 1)$ is given in Theorem 2.3.8. The proof of this fact is based on considerations done in Proposition 2.3.6. By contradiction, $\gamma_s(\theta) > \gamma_s^*(\theta)$. Let

$$h(x) = v_s(x) - u_s(x).$$

The function h is continuous in \mathbb{R}^n and h(x) = 0 in $\mathbb{R}^n \setminus C_\theta$. We want to prove that $h(x) \leq 0$ in $\mathbb{R}^n \setminus (C_\theta \cap B_1)$. Since h = 0 outside the cone, we can consider only what happens in $C_\theta \setminus B_1$. By (77), there exist two constants $c_1(s), c_2(s) > 0$ such that, for any $x \in \overline{C_\theta} \setminus \{0\}$,

$$c_1(s)|x|^{\gamma_s-s} \operatorname{dist}(x, \partial C_{\theta})^s \le u_s(x) \le c_2(s)|x|^{\gamma_s-s} \operatorname{dist}(x, \partial C_{\theta})^s,$$

and there exist two constants $c_1, c_2 > 0$ such that

$$c_1|x|^{\gamma_s^*-1} \mathrm{dist}(x, \partial C_\theta) \le v_s(x) \le c_2|x|^{\gamma_s^*-1} \mathrm{dist}(x, \partial C_\theta).$$

We can choose v_s so that $c := c_1(s) = c_2$ since it is defined up to a multiplicative constant. Then, for any $x \in C_\theta \setminus B_1$, since $|x|^{\gamma_s^*} \le |x|^{\gamma_s}$, we have

$$h(x) \le c|x|^{\gamma_s} \operatorname{dist}(x, \partial C_{\theta})^s \left[\frac{\operatorname{dist}(x, \partial C_{\theta})^{1-s}}{|x|^{1-s}} - 1 \right] \le 0.$$

Now we want to show that there exists a point $x_0 \in C_{\theta} \cap B_1$ such that $h(x_0) > 0$. Let us consider for example the point $\overline{x} \in S^{n-1} \cap C_{\theta}$ determined by the angle $\vartheta = \theta/2$, and let $\alpha := v_s(\overline{x}) > 0$ and $\beta := u_s(\overline{x}) > 0$. Hence, there exists a small r > 0 so that $\alpha r^{\gamma_s^*} > \beta r^{\gamma_s}$, and so, taking x_0 with angle $\vartheta = \theta/2$ and $|x_0| = r$, we obtain $h(x_0) > 0$.

If we consider the restriction of h to $\overline{C_{\theta} \cap B_1}$, which is continuous on a compact set, for the considerations done before and for the Weierstrass Theorem, there exists a maximum point $x_1 \in C_{\theta} \cap B_1$ for the function h which is global in \mathbb{R}^n and is strict at least in a set of positive measure. Hence,

$$(-\Delta)^{s}h(x_{1}) = C(n,s)$$
 P.V. $\int_{\mathbb{R}^{n}} \frac{h(x_{1}) - h(y)}{|x_{1} - y|^{n+2s}} \, \mathrm{d}y > 0$

and since $(-\Delta)^{s}h$ is a continuous function in the open cone, there exists an open set $U(x_1)$ with $\overline{U(x_1)} \subset C_{\theta}$ such that

 $(-\Delta)^s h(x) > 0 \quad \forall x \in U(x_1).$

But thanks to Theorem 2.3.8 we obtain a contradiction since for any nonnegative $\varphi \in C_c^{\infty}(U(x_1))$

$$((-\Delta)^{s}h,\varphi) = ((-\Delta)^{s}v_{s},\varphi) - ((-\Delta)^{s}u_{s},\varphi) = ((-\Delta)^{s}v_{s},\varphi) \le 0,$$

where the last inequality holds for any $s \in [s_0(\theta), 1)$. Hence, for any $\theta \in (0, \theta_0]$

$$\mu(\theta) = \lim_{s \to 1^-} \frac{C(n,s)}{2s - \gamma_s(\theta)} \le \lim_{s \to 1^-} \frac{C(n,s)}{2s - \gamma_s^*(\theta)} = \mu_0(\theta) < +\infty.$$
(83)

2.4 The limit for $s \nearrow 1$

In this Section we prove the main result, Theorem 2.1.2, emphasizing the difference between wide and narrow cones. Then we improve the asymptotic analysis proving uniqueness of the limit under assumptions on the geometry and the regularity of C.

Let $C \subset \mathbb{R}^n$ be an open cone and consider the minimization problem

$$\lambda_1(C) = \inf\left\{\frac{\int_{S^{n-1}} |\nabla_{S^{n-1}}u|^2 \mathrm{d}\sigma}{\int_{S^{n-1}} u^2 \mathrm{d}\sigma} : u \in H^1(S^{n-1}) \setminus \{0\} \text{ and } u = 0 \text{ in } S^{n-1} \setminus C\right\}, \quad (84)$$

which is strictly related to the homogeneity of the solution of (79) by $\lambda_1(C) = \gamma(C)(\gamma(C) + n-2)$.

Moreover, if $\gamma(C) > 2$, equivalently if $\lambda_1(C) > 2n$, the problem

$$\mu_0(C) := \inf\left\{\frac{\int_{S^{n-1}} |\nabla_{S^{n-1}}u|^2 - 2nu^2 \mathrm{d}\sigma}{\left(\int_{S^{n-1}} |u| \mathrm{d}\sigma\right)^2} : u \in H^1(S^{n-1}) \setminus \{0\} \text{ and } u = 0 \text{ in } S^{n-1} \setminus C\right\}$$
(85)

is well defined and the number $\mu_0(C)$ is strictly positive.

By a standard argument due to the variational characterization of the previous quantities, we already know the existence of a nonnegative eigenfunction $\varphi \in H_0^1(S^{n-1} \cap C) \setminus \{0\}$ associated to the minimization problem (84) and a nonnegative function $\psi \in H_0^1(S^{n-1} \cap C) \setminus \{0\}$ that achieves the minimum (85), since the numerator in (85) is a coercive quadratic form equivalent to the one in (84).

Since the cone C may be disconnected, it is well known that φ is not necessarily unique. Instead, the function ψ is unique up to a multiplicative constant, since it solves

$$\begin{cases} -\Delta_{S^{n-1}}\psi = 2n\psi + \mu_0(C)\int_{S^{n-1}}\psi d\sigma & \text{in } S^{n-1}\cap C,\\ \psi = 0 & \text{in } S^{n-1}\setminus C. \end{cases}$$
(86)

In fact, due to the integral term in the equation, the solution ψ must be strictly positive in every connected component of C and localizing the equation in a generic component we can easily get uniqueness by maximum principle.

A fundamental toll in order to reach as $s \to 1$ the space H^1_{loc} , is the following result

Proposition 2.4.1. [10, Corollary 7] Let $\Omega \subset \mathbb{R}^n$ be a bounded domain. For $1 , let <math>f_s \in W^{s,p}(\Omega)$, and assume that

$$[f_s]_{W^{s,p}(\Omega)} \le C_0.$$

Then, up to a subsequence, (f_s) converges in $L^p(\Omega)$ as $s \to 1$ (and, in fact, in $W^{t,p}(\Omega)$), for all t < 1) to some $f \in W^{1,p}(\Omega)$.

In [10] the authors used a different notation since in our manuscript the normalization constant C(n,s) is incorporate in the seminorm $[\cdot]_{H^s}$, in order to obtain a continuity of the norm $\|\cdot\|_{H^s}$ for $s \in (0,1]$.

Proof of Theorem 2.1.2. Let C be an open cone and C_R be a regular cone with Section on S^{n-1} of class $C^{1,1}$ such that $C_R \subset C$ and $\partial C_R \cap \partial C = \{0\}$.

By monotonicity of the homogeneity degree $\gamma_s(\cdot)$ with respect to the inclusion, we directly obtain $\gamma_s(C) < \gamma_s(C_R)$ and consequently, up to consider a subsequence, we obtain the existence of the following finite limits

$$\overline{\gamma}(C) = \lim_{s \to 1} \gamma_s(C), \quad \mu(C) = \lim_{s \to 1} \frac{C(n,s)}{2s - \gamma_s(C)}.$$
(87)

Since $\gamma_s(C) < 2s$, then $\overline{\gamma}(C) \leq 2$ and similarly $\mu(C) \in [0, +\infty)$.

Let $K \subset \mathbb{R}^n$ be a compact set and consider $x_0 \in K$ and R > 0 such that $K \subset B_R(x_0)$. Given $\eta \in C_c^{\infty}(B_2)$, a radial cut off function such that $\eta \equiv 1$ in B_1 and $0 \leq \eta \leq 1$ in B_2 , consider the rescaled function $\eta_K(x) = \eta(\frac{x-x_0}{R})$ which satisfies $\eta_K \equiv 1$ on K. By Proposition 2.2.7, we have

$$[u_s\eta_K]^2_{H^s(B_{2R}(x_0))} \le [u_s\eta_K]^2_{H^s(\mathbb{R}^n)} \le M(n,K) \left[\frac{C(n,s)}{2(1-s)} + \frac{C(n,s)}{2s-\gamma_s}\right],$$

and similarly

$$\begin{split} \|u_s \eta_K\|_{H^s(B_{2R}(x_0))}^2 &\leq \|u_s \eta_K\|_{L^2(\mathbb{R}^n)}^2 + [u_s \eta_K]_{H^s(\mathbb{R}^n)}^2 \\ &\leq M(n,K) \left[\frac{C(n,s)}{2(1-s)} + \frac{C(n,s)}{2s-\gamma_s} + 1\right] \\ &\leq M(n,K) \left[\frac{2n}{\omega_{n-1}} + c\mu(C) + 1\right]. \end{split}$$

By applying Proposition 2.4.1 with $\Omega = B_{2R}(x_0)$, we obtain that, up to a subsequence, $u_s \eta_K \to \overline{u}\eta_K$ in $L^2(B_{2R}(x_0))$ and

$$\|\overline{u}\eta_K\|^2_{H^1(B_{2R}(x_0))} \le M(n,K)$$

up to relabeling the constant M(n, K).

By construction, since $\eta_K \equiv 1$ on K and $\eta_K \in [0,1]$, we obtain that $u_s \to \overline{u}$ in $L^2(K)$ and similarly

$$\|\overline{u}\|_{H^{1}(K)} \leq \|\overline{u}\eta_{K}\|_{H^{1}(K)} \leq \|\overline{u}\eta_{K}\|_{H^{1}(B_{2R}(x_{0}))} < \infty,$$

which gives us the local integrability in $H^1(\mathbb{R}^n)$.

By Proposition 2.2.4 and Corollary 2.3.9 we obtain, up to pass to a subsequence, uniform in s bound in $C_{\text{loc}}^{0,\alpha}(C)$ for (u_s) . Then, since we obtain uniform convergence on compact subsets of C, the limit must be necessary nontrivial with $||\overline{u}||_{L^{\infty}(S^{n-1})} = 1$, nonnegative and $\overline{\gamma}(C)$ -homogeneous.

Let $\varphi \in C_c^{\infty}(C)$ be a positive smooth function compactly supported such that supp $\varphi \subset B_{\rho}$, for some $\rho > 0$. By definition of the distributional fractional Laplacian

$$0 = \int_{\mathbb{R}^n} \varphi(-\Delta)^s u_s \mathrm{d}x = \int_{\mathbb{R}^n} u_s (-\Delta)^s \varphi \mathrm{d}x = \int_{\mathbb{R}^n \setminus B_\rho} u_s (-\Delta)^s \varphi \mathrm{d}x + \int_{B_\rho} u_s (-\Delta)^s \varphi \mathrm{d}x.$$

Since

$$\frac{1}{|x-y|^{n+2s}} = \frac{1}{|x|^{n+2s}} \left(1 - (n+2s)\frac{y}{|x|} \int_0^1 \frac{\frac{x}{|x|} - t\frac{y}{|x|}}{\left|\frac{x}{|x|} - \frac{ty}{|x|}\right|^{n+2s+2}} \mathrm{d}t \right),$$

by definition of the fractional Laplacian for regular functions, it follows

$$\int_{\mathbb{R}^n \setminus B_{\rho}} u_s(-\Delta)^s \varphi dx = C(n,s) \int_{\mathbb{R}^n \setminus B_{\rho}} u_s(x) \int_{\text{supp } \varphi} \frac{-\varphi(y)}{|y-x|^{n+2s}} dy dx$$
$$= C(n,s) \int_{\mathbb{R}^n \setminus B_{\rho}} \frac{u_s(x)}{|x|^{n+2s}} \int_{\text{supp } \varphi} -\varphi(y) dy dx + C(n,s)(n+2s) \int_{\mathbb{R}^n \setminus B_{\rho}} \frac{u_s(x)}{|x|^{n+2s+1}} \psi(x) dx,$$

for some $\psi \in L^{\infty}$. Moreover, since u_s is $\gamma_s(C)$ -homogeneous with $\gamma_s(C) < 2s$, we have

$$C(n,s)\int_{\mathbb{R}^n\setminus B_{\rho}}\frac{u_s(x)}{|x|^{n+2s}}\mathrm{d}x = \frac{C(n,s)}{2s-\gamma_s(C)}\rho^{\gamma_s(C)-2s}\int_{S^{n-1}}u_s(\theta)\mathrm{d}\sigma$$

and similarly

$$C(n,s) \left| \int_{\mathbb{R}^n \setminus B_\rho} \frac{u_s(x)}{|x|^{n+2s+1}} \psi(x) \mathrm{d}x \right| \le \frac{C(n,s) \|\psi\|_{L^{\infty}}}{2s - \gamma_s(C) + 1} \rho^{\gamma_s(C) - 2s - 1} \int_{S^{n-1}} u_s(\theta) \mathrm{d}\sigma = o(1).$$

Hence, for each $s \in (0, 1)$

$$\begin{split} \int_{B_{\rho}} u_s(-\Delta)^s \varphi \mathrm{d}x &= \int_{\mathbb{R}^n \setminus B_{\rho}} u_s(-\Delta)^s \varphi \mathrm{d}x \\ &= C(n,s) \int_{\mathbb{R}^n \setminus B_{\rho}} u_s(x) \int_{\mathrm{supp }\varphi} \frac{\varphi(y)}{|x-y|^{n+2s}} \mathrm{d}y \mathrm{d}x \\ &= \frac{C(n,s)}{2s - \gamma_s(C)} \int_{\mathrm{supp }\varphi} \varphi(x) \mathrm{d}x \int_{S_{n-1}} u_s \mathrm{d}\sigma + o(1) \end{split}$$

and passing through the limit, up to a subsequence, we obtain

$$\int_{B_{\rho}} \overline{u}(-\Delta)\varphi dx = \mu(C) \int_{S_{n-1}} \overline{u} d\sigma \int_{\text{supp }\varphi} \varphi(x) dx$$
$$= \int_{B_{\rho}} \left(\mu(C) \int_{S_{n-1}} \overline{u} d\sigma \right) \varphi(x) dx,$$

which implies, integrating by parts, that

$$-\Delta \overline{u} = \mu(C) \int_{S^{n-1}} \overline{u} d\sigma \text{ in } \mathcal{D}'(C).$$

Since the function \overline{u} is $\overline{\gamma}(C)$ -homogenous, we get

$$-\Delta_{S^{n-1}}\overline{u} = \overline{\lambda}\overline{u} + \mu(C)\int_{S^{n-1}}\overline{u}\mathrm{d}\sigma \quad \text{on } S^{n-1} \cap C,$$
(88)

where $\overline{\lambda} = \overline{\gamma}(C)(\overline{\gamma}(C) + n - 2)$ is the eigenvalue associated to the critical exponent $\overline{\gamma}(C) \leq 2$.

Consider now a nonnegative $\varphi \in H_0^1(S^{n-1} \cap C) \setminus \{0\}$, strictly positive on $S^{n-1} \cap C$ which achieves (84). Then

$$-\Delta_{S^{n-1}}\varphi = \lambda_1(C)\varphi, \quad \text{in } H^{-1}(S^{n-1} \cap C).$$
(89)

By testing this equation with \overline{u} and integrating by parts, we obtain

$$\left(\lambda_1(C) - \overline{\lambda}\right) \int_{S^{n-1}} \overline{u}\varphi d\sigma = \mu(C) \int_{S^{n-1}} \overline{u} d\sigma \int_{S^{n-1}} \varphi d\sigma \ge 0 \tag{90}$$

which implies that in general $\gamma(C) \geq \overline{\gamma}(C)$ and $\gamma(C) = \overline{\gamma}(C)$ if and only if $\mu(C) = 0$.

Wide cones: $\gamma(C) < 2$

By the previous remark we have $\overline{\gamma}(C) < 2$ and by definition of $\mu(C)$, it follows $\mu(C) = 0$. Since φ is the trace on S^{n-1} of an homogenous harmonic function on C, we obtain that $\overline{\gamma}(C) = \gamma(C)$ and \overline{u} is an homogeneous nonnegative harmonic function on C such that $\|\overline{u}\|_{L^{\infty}(S^{n-1})} = 1$.

Narrow cones: $\gamma(C) \geq 2$

If $\overline{\gamma}(C) < 2$ we have $\mu(C) = 0$ and consequently $\lambda_1(C) = \overline{\lambda}$, which is a contradiction since $\gamma(C) \ge 2 > \overline{\gamma}(C)$. Hence, if C is a narrow cone we get $\overline{\gamma}(C) = 2$. Since $\gamma(C) = 2$ is trivial and it follows directly from the previous computations, consider now $\mu_0(C)$ as the minimum defined in (85), which is well defined and strictly positive since we are focusing on the remaining case $\gamma(C) > 2$. We already remarked that it is achieved by a nonnegative $\psi \in H_0^1(S^{n-1} \cap C) \setminus \{0\}$ which is strictly positive on $S^{n-1} \cap C$ and solution of

$$-\Delta_{S^{n-1}}\psi = 2n\psi + \mu_0(C) \int_{S^{n-1}} \psi d\sigma \quad \text{in } H^{-1}(S^{n-1} \cap C).$$

As we already did in the previous cases, by testing this equation with \overline{u} we obtain $\mu(C) = \mu_0(C)$. By uniqueness of the limits $\overline{\gamma}(C)$ and $\mu(C)$, the result in (87) holds for $s \to 1$ and not just up to a subsequence.

Remark 2.4.2. The possible obstruction to the existence of the limit of u_s as s converge s to one lies in the possible lack of uniqueness of nonnegative solutions to (59) such that $\|\overline{u}\|_{L^{\infty}(S^{n-1})} = 1$. This is the reason why we need to extract subsequences in the asymptotic analysis of Theorem 2.1.2. More precisely, uniqueness of (84) implies uniqueness of the limit \overline{u} in the case $\gamma(C) \leq 2$ and uniqueness of (85) in the case $\gamma(C) > 2$. When C is connected (84) is attained by a unique normalized nonnegative solution via a standard argument based upon the maximum priciple. On the other hand, as we already remarked, when $\gamma(C) > 2$, problem (85) always admits a unique solution. Ultimately, the main obstacle in this analysis is the disconnection of the cone C when $\gamma(C) \leq 2$: in this case we cannot always ensure the uniqueness of the solution of the limit problem and even the positivity of the limit function \overline{u} on every connected components of C.

The following example shows uniqueness of the limit function \overline{u} due to the nonlocal nature of the fractional Laplacian under a symmetry assumption on the cone C.

Proposition 2.4.3. Let $C = C_1 \cup \cdots \cup C_m$ be a union of disconnected cones such that C_1 is connected and there are orthogonal maps $\Phi_2, \ldots, \Phi_m \in O(n)$ (e.g. reflections about hyperplanes) such that $C_i = \Phi_i(C_1)$ and and $\Phi_i(C) = (C)$ for $i = 2, \ldots, m$. Let (u_s) be the family of nonnegative solutions to (55) such that $||u_s||_{L^{\infty}(S^{n-1})} = 1$. Then there exists the limit of u_s as $s \nearrow 1$ in $L^2_{loc}(\mathbb{R}^n)$ and uniformly on compact subsets of C.

Proof. We remark that, for any element of the orthogonal group $\Phi \colon \mathbb{R}^n \to \mathbb{R}^n$,

$$(-\Delta)^{s} (u \circ \Phi) (x) = C(n,s) \text{ P.V.} \int_{\mathbb{R}^{n}} \frac{u(\Phi(x)) - u(y)}{|\Phi(x) - y|^{n+2s}} \mathrm{d}y = (-\Delta)^{s} u(\Phi(x))$$

By the uniqueness result [5, Theorem 3.2] of *s*-harmonic functions on cones, we infer that $u_s \equiv u_s \circ \Phi_i$, for every i = 2, ..., m. Therefore, there holds convergence to \overline{u} , where satisfies $\|\overline{u}\|_{L^{\infty}(S^{n-1})} = 1$, and it is a solution of

$$\begin{cases} -\Delta \overline{u} = \mu(C) \int_{S^{n-1}} \overline{u} d\sigma & \text{in } C, \\ \overline{u} \ge 0 & \text{in } C, \\ \overline{u} = 0 & \text{in } \mathbb{R}^n \setminus C, \end{cases}$$
(91)

such that $\overline{u} \equiv \overline{u} \circ \Phi_i$ for every i = 2, ..., m. Finally, connectedness of C_1 yields uniqueness of such solution also for narrow cones.



Figure 5: Values of the limit $\overline{\Gamma}(\theta) = \lim_{s \to 1} \Gamma^s(\theta)$ and $\Gamma(\theta)$, for n = 2.

Proof of Corollary 2.1.3. Corollary 2.1.3 is an easy application of our main Theorem 2.1.2, since it is a consequence of the Dini's Theorem for a monotone sequence of continuous functions which converges pointwisely to a continuous function on a compact set. In fact, fixed $s \in (0, 1)$,
the function $\theta \mapsto \gamma_s(\theta)$ is continuous in $[0, \pi)$ with $\gamma_s(0) = 2s$ and $\gamma_s(\pi) = 0$. Moreover this function is also monotone decreasing in $[0, \pi]$ and since there exists the limit

$$\lim_{\theta \to \pi^-} \gamma_s(\theta) = \begin{cases} \frac{2s-1}{2} & \text{if } n = 2 \text{ and } s > \frac{1}{2}, \\ \gamma_s(\pi) = 0 & \text{otherwise,} \end{cases}$$

we can extend $\theta \mapsto \gamma_s(\theta)$ to a continuous function in $[0,\pi]$ (see [67]). Nevertheless, the limit $\overline{\gamma}(\theta) = \lim_{s \to 1} \gamma_s(\theta) = \min\{\gamma(\theta), 2\}$ is continuous on $[0, \pi]$ with

$$\overline{\gamma}(\pi) = \begin{cases} \frac{1}{2} & \text{if } n = 2, \\ 0 & \text{otherwise.} \end{cases}$$

Eventually, for any fixed $\theta \in [0, \pi]$, the function $s \mapsto \gamma_s(\theta)$ is monotone nondecreasing in (0, 1). By the Dini's Theorem the convergence is uniform on $[0, \pi]$. This fact obviously implies the uniform convergence

$$\Gamma^{s}(\theta) = \frac{\gamma_{s}(\theta) + \gamma_{s}(\pi - \theta)}{2} \longrightarrow \overline{\Gamma}(\theta) = \frac{\overline{\gamma}(\theta) + \overline{\gamma}(\pi - \theta)}{2}$$

in $[0, \pi]$, and hence

$$\nu_s^{ACF} = \min_{\theta \in [0,\pi]} \Gamma^s(\theta) \longrightarrow \min_{\theta \in [0,\pi]} \overline{\Gamma}(\theta) = \nu^{ACF}.$$

2.5 Uniform estimates in $c^{0, \alpha}$ on annuli

We have already remarked in Section 2.2 that, if you take a cone $C = C_{\omega}$ with $\omega \subset S^{n-1}$ a finite union of connected $C^{1,1}$ domain ω_i , such that $\overline{\omega}_i \cup \overline{\omega}_j = \emptyset$ for $i \neq j$, by [67, Lemma 3.3] we have (63).

Hence solutions u_s to (55) are $C^{0,s}(S^{n-1})$ and for any fixed $\alpha \in (0,1)$, any solution u_s with $s \in (\alpha, 1)$ is $C^{0,\alpha}(S^{n-1})$; that is, there exists $L_s > 0$ such that

$$\sup_{x,y\in S^{n-1}}\frac{|u_s(x)-u_s(y)|}{|x-y|^{\alpha}} = L_s.$$

Let us consider an annulus $A = A_{r_1,r_2} = B_{r_2} \setminus \overline{B_{r_1}}$ with $0 < r_1 < r_2 < +\infty$. We have the following result.

Lemma 2.5.1. Let $\alpha \in (0, 1)$, $s_0 \in (\max\{1/2, \alpha\}, 1)$ and A an annulus centered at zero. Then there exists a constant c > 0 such that any solution u_s to (55) with $s \in [s_0, 1)$ satisfies

$$\sup_{x,y\in A} \frac{|u_s(x) - u_s(y)|}{|x - y|^{\alpha}} \le cL_s.$$

Proof. First of all we remark that

$$\sup_{x,y \in S_r^{n-1}} \frac{|u_s(x) - u_s(y)|}{|x - y|^{\alpha}} \le cL_s,$$
(92)

for any $r \in (r_1, r_2)$. In fact, by the γ_s -homogeneity of our solutions, we have

$$\sup_{x,y\in S_r^{n-1}}\frac{|u_s(x)-u_s(y)|}{|x-y|^{\alpha}} = L_s r^{\gamma_s-\alpha},$$

and since $(2s_0 - 1)/2 \le \gamma_s(C) < 2$ for any $s \in [s_0, 1)$ by the inclusion $C \subset \mathbb{R}^n \setminus \{\text{half} - \text{line from } 0\}$, we obtain (92).

Now we can show what happens considering $x, y \in A$ which are not on the same sphere. We can suppose without loss of generality that $x \in S_R^{n-1}$, $y \in S_r^{n-1}$ with $r_1 < r < R < r_2$. Hence let us take the point z obtained by the intersection between S_r^{n-1} and the half-line connecting 0 and x (z may be y itself). Hence

$$\begin{aligned} |u_s(x) - u_s(y)| &\leq |u_s(x) - u_s(z)| + |u_s(z) - u_s(y)| \\ &\leq u_s(x/|x|) ||x|^{\gamma_s} - |z|^{\gamma_s}| + cL_s|z - y|^{\alpha} \\ &\leq cL_s|x - y|^{\alpha}. \end{aligned}$$

In fact we remark that $||u_s||_{L^{\infty}(S^{n-1})} = 1$. Moreover, since the angle $\beta = \widehat{xzy} \in (\pi/2, \pi]$, obviously $|z - y|^{\alpha} \leq |x - y|^{\alpha}$. Moreover by the α -Hölder continuity of $t \mapsto t^{\gamma_s}$ in (r_1, r_2) and the bounds $(2s_0 - 1)/2 \leq \gamma_s(C) < 2$, one can find a universal constant c > 0 such that

$$||x|^{\gamma_s} - |z|^{\gamma_s}| \le c||x| - |z||^{\alpha} \le c|x-z|^{\alpha} \le c|x-y|^{\alpha}$$

where the last inequality holds since z is the point on S_r^{n-1} which minimizes the distance $\operatorname{dist}(x, S_r^{n-1})$.

A nontrivial improvement of the main Theorem concerns uniform bounds in Hölder spaces holding uniformly for $s \rightarrow 1$.

Theorem 2.5.2. Assume the cone is $C^{1,1}$. Let $\alpha \in (0,1)$, $s_0 \in (\max\{1/2, \alpha\}, 1)$ and A an annulus centered at zero. Then the family of solutions u_s to (55) is uniformly bounded in $C^{0,\alpha}(A)$ for any $s \in [s_0, 1)$.

Proof. Seeking a contradiction,

$$\max_{x,y\in S^{n-1}} \frac{|u_{s_k}(x) - u_{s_k}(y)|}{|x - y|^{\alpha}} = L_{s_k} = L_k \to +\infty, \quad \text{as } s_k \to 1.$$
(93)

We can consider the sequence of points $x_k, y_k \in S^{n-1}$ which realizes L_k at any step. It is easy to see that this couple belongs to $\overline{C} \cap S^{n-1}$. Moreover we can always think x_k as the one closer to the boundary $\partial C \cap S^{n-1}$. Therefore, to have (93), we have $r_k = |x_k - y_k| \to 0$. Hence, without loss of generality, we can assume that x_k, y_k belong defenetively to the same connected component of C and

$$\frac{|u_{s_k}(y_k) - u_{s_k}(x_k)|}{r_k^{\alpha}} = L_k, \qquad \frac{y_k - x_k}{r_k} \to e_1.$$

Let us define

$$u^k(x) = rac{u_{s_k}(x_k+r_kx)-u_{s_k}(x_k)}{r_k^lpha L_k}, \qquad x\in \Omega_k = rac{C-x_k}{r_k}.$$

We remark that $u^k(0) = 0$ and $u^k((y_k - x_k)/r_k) = 1$.

Moreover we can have two different situations.

Case 1: If

$$\frac{r_k}{\operatorname{dist}(x_k,\partial C)} \to 0$$

then the limit of Ω_k is \mathbb{R}^n .

Case 2: If

$$\frac{r_k}{\operatorname{dist}(x_k,\partial C)} \to l \in (0,+\infty],$$

then the limit of Ω_k is an half-space $\mathbb{R}^n \cap \{x_1 > 0\}$.

In any case let us define Ω_{∞} this limit set. Let us consider the annulus $A^* := B_{3/2} \setminus \overline{B_{1/2}}$. By Lemma 2.5.1 and the definition of u^k , we obtain, for any k,

$$\sup_{x,y \in A_k^*} \frac{|u^k(x) - u^k(y)|}{|x - y|^{\alpha}} \le c,$$
(94)

where $A_k^* := \frac{A^* - x_k}{r_k} \to \mathbb{R}^n$ and the constant c > 0 depends only on α and A^* . Let us consider a compact subset K of Ω_{∞} . Since for k large enough $K \subset A_k^*$, functions u^k are $C^{0,\alpha}(K)$ uniformly in k. This is due also to the fact that they are uniformly in $L^{\infty}(K)$, since $|u^k(x) - u^k(0)| \le c|x|^{\alpha}$

on K. Hence $u^k \to \overline{u}$ uniformly on compact subsets of Ω_{∞} . Moreover \overline{u} is globally α -Hölder continuous and it is not constant, since $\overline{u}(e_1) - \overline{u}(0) = 1$. To conclude, we will show that \overline{u} is harmonic in the limit domain Ω_{∞} ; that is, for any $\phi \in C_c^{\infty}(\Omega_{\infty})$

$$\int_{\Omega_{\infty}} \phi(-\Delta) \overline{u} \mathrm{d}x = 0,$$

and this fact will be a contradiction with the global Hölder continuity. In fact we can apply Corollary 2.3 in [72], if $\Omega_{\infty} = \mathbb{R}^n$ directly on the function \overline{u} and if $\Omega_{\infty} = \mathbb{R}^n \cap \{x_1 > 0\}$, since $\overline{u} = 0$ in $\partial \Omega_{\infty}$, we can use the same result over its odd reflection. Hence we want to prove

$$\int_{\Omega_{\infty}} \phi(-\Delta) \overline{u} \mathrm{d}x = \int_{\Omega_{\infty}} \overline{u}(-\Delta) \phi \mathrm{d}x = \lim_{k \to +\infty} \int_{B_R} u^k (-\Delta)^{s_k} \phi \mathrm{d}x = 0,$$

where B_R contains the support of ϕ and the second equality holds by the uniform convergences $u^k \to \overline{u}$ and $(-\Delta)^{s_k} \phi \to (-\Delta)\phi$ on compact subsets of Ω_{∞} , since ϕ is a smooth function compactly supported. Moreover, since u^k is s_k -harmonic on Ω_k , and for k large enough the support of ϕ is contained in this domain, we have

$$\int_{\mathbb{R}^n} u^k (-\Delta)^{s_k} \phi \mathrm{d}x = \int_{\mathbb{R}^n} \phi (-\Delta)^{s_k} u^k \mathrm{d}x = 0.$$

In order to conclude we want

$$\lim_{k \to +\infty} \int_{\mathbb{R}^n \setminus B_R} u^k (-\Delta)^{s_k} \phi \mathrm{d}x = 0.$$

Hence, defining $\eta = x_k + r_k x$ and using Remark 2.2.3, we obtain

$$\left| \int_{\mathbb{R}^n \setminus B_R} u^k (-\Delta)^{s_k} \phi \mathrm{d}x \right| \le \frac{C(n, s_k)}{L_k} r_k^{2s_k - \alpha} \int_{|\eta - x_k| > Rr_k} \frac{|u_{s_k}(\eta) - u_{s_k}(x_k)|}{|\eta - x_k|^{n + 2s_k}} \mathrm{d}\eta.$$

For k large enough, we notice that we can choose $\varepsilon > 0$ such that the set $\{\eta \in \mathbb{R}^n : Rr_k < |\eta - x_k| < \varepsilon\}$ is contained in A^* . So, we can split the integral obtaining

$$\int_{|\eta-x_k|>Rr_k} \frac{|u_{s_k}(\eta) - u_{s_k}(x_k)|}{|\eta - x_k|^{n+2s_k}} \mathrm{d}\eta \leq \int_{Rr_k < |\eta-x_k| < \varepsilon} \frac{|u_{s_k}(\eta) - u_{s_k}(x_k)|}{|\eta - x_k|^{n+2s_k}} \mathrm{d}\eta + \int_{|\eta-x_k| > \varepsilon} \frac{|u_{s_k}(\eta) - u_{s_k}(x_k)|}{|\eta - x_k|^{n+2s_k}} \mathrm{d}\eta$$

where we have

$$\frac{C(n,s_k)r_k^{2s_k-\alpha}}{L_k} \int_{Rr_k < |\eta-x_k| < \varepsilon} \frac{|u_{s_k}(\eta) - u_{s_k}(x_k)|}{|\eta - x_k|^{n+2s_k}} \mathrm{d}\eta \le C(n,s_k)r_k^{2s_k-\alpha}c\omega_{n-1} \int_{Rr_k}^{\varepsilon} t^{-1+\alpha-2s_k}\mathrm{d}t = \frac{C(n,s_k)c\omega_{n-1}}{2s_k-\alpha} \left(R^{\alpha-2s_k} - \frac{r_k^{2s_k-\alpha}}{\varepsilon^{2s_k-\alpha}}\right)$$

and similarly

$$\frac{C(n,s_k)r_k^{2s_k-\alpha}}{L_k} \int_{|\eta-x_k|>\varepsilon} \frac{|u_{s_k}(\eta) - u_{s_k}(x_k)|}{|\eta-x_k|^{n+2s_k}} \mathrm{d}\eta \le \frac{C(n,s_k)r_k^{2s_k-\alpha}c\omega_{n-1}}{L_k} \int_{\varepsilon}^{\infty} \frac{(1+t)^{\gamma_{s_k}}}{t^{1+2s_k}} \mathrm{d}t = \frac{C(n,s_k)r_k^{2s_k-\alpha}c\omega_{n-1}}{L_k} \left(1 + \frac{\varepsilon^{\gamma_{s_k}-2s_k}}{2s_k-\gamma_{s_k}}\right).$$

Finally, recalling that $r_k \to 0$, $C(n, s_k) \to 0$, $L_k \to \infty$ and $2s_k - \alpha > 0$ taking $s_0 > 1/2$, we obtain

$$\left| \int_{\mathbb{R}^n \setminus B_R} u^k (-\Delta)^{s_k} \phi \mathrm{d}x \right| \le \left(C(n, s_k) + \frac{C(n, s_k)}{2s_k - \gamma_{s_k}} \frac{r_n^{2s_k - \alpha}}{L_k} \right) M$$

which converges to zero as we claimed, since

$$\frac{C(n, s_k)}{2s_k - \gamma_{s_k}(C)} \to \mu(C) \in [0, +\infty)$$

in any regular cone $C \subset \mathbb{R}^n$.

94 ON *s*-harmonic functions on cones

NODAL SET OF SOLUTIONS OF DEGENERATE - SINGULAR EQUATIONS

3.1 INTRODUCTION

In literature, the subject of nodal sets, or level sets in general, is an important research topic for solutions of PDEs. Recently in [41, 56, 57, 66] much attention has been paid on the structure of the singular set and on its (n - 2)-Hausdorff measure, and, as pointed out in [48, 49, 66], a starting point of this analysis is the validity of a strong unique continuation principle, in order to ensure the existence of a finite vanishing order.

In this Chapter we consider the nodal set in \mathbb{R}^{n+1} of solution of a peculiar class of degeneratesingular operator, firstly studied in the pioneering works [44, 43]. In the 80s Fabes, Jerison, Kening and Serapioni introduced a general class of degenerate operators $L = \operatorname{div}(A(X)\nabla \cdot)$ whose coefficient $A(X) = (a_{ij}(X))$ are defined starting from a symmetric matrix valued function such that

$$\lambda\omega(X)\,|\xi|^2\leq (A(X)\xi,\xi)\leq\Lambda\omega(X)\,|\xi|^2\,,\quad ext{for some }\lambda,\Lambda>0,$$

where ω may either vanish, or be infinite, or both. In particular, the prototypes of weights considered in their analysis where in the Muckenhoupt A_2 -class, i.e. such that

$$\sup_{B \subset \mathbb{R}^{n+1}} \left(\frac{1}{|B|} \int_B \omega(X) \mathrm{d}X \right) \left(\frac{1}{|B|} \int_B \omega^{-1}(X) \mathrm{d}X \right) < \infty.$$

Given $a \in (-1, 1)$ and $X = (x, y) \in \mathbb{R}^n_x \times \mathbb{R}_y$ we consider the cases of $\omega(X) = |y|^a$, with

$$L_a = \operatorname{div}(|y|^a \,\nabla),$$

where obviously we denote with div and ∇ respectively the divergence and the gradient operator in \mathbb{R}^{n+1} . Our main purpose is to fully understand the local behaviour of L_a -harmonic function near their nodal set and to develop a geometric analysis of its structure and regularity in order to comprehend how the presence of a nontrivial set where the coefficients of an elliptic equation may either vanishes or be infinite can affect the local picture of its solution.

Inspired by this last claim, we introduce the notion of "*characteristic manifold*" Σ associated to the operator L_a , as the set of points where the coefficient either vanishes or blows up, and we studied the properties of the nodal set $\Gamma(u)$ of solutions of

$$-L_a u = 0$$
 in $B_1 \subset \mathbb{R}^{n+1}$.

In particular, since the operator L_a is locally uniformly elliptic on $\mathbb{R}^{n+1} \setminus \Sigma$, we restrict our attention on the structure of the nodal set in a neighbourhood of the manifold Σ , trying to understand the difference between the whole nodal set $\Gamma(u)$ and its restriction on the characteristic manifold Σ .

At first sight, our approach seems to be based upon the validity of an Almgren and Weiss type monotonicity formulas, which guarantee the uniqueness of a non trivial tangent map at every point of the nodal set, and on a complete classification of the possible homogenous configurations appearing at the blow-up limit. Instead, the crucial result of our analysis relies in the decomposition of L_a -harmonic function with respect to the orthogonal direction to the characteristic manifold Σ . More precisely, we prove

Proposition 3.1.1. Given $a \in (-1, 1)$ and u an L_a -harmonic function in B_1 , there exist two unique functions $u_e^a \in H^{1,a}(B_1), u_e^{2-a} \in H^{1,2-a}(B_1)$ symmetric with respect to Σ respectively L_a and L_{2-a} harmonic in B_1 and locally smooth, such that

$$u(X) = u_e^a(X) + u_e^{2-a}(X)y |y|^{-a}$$
 in B_1 .

Heuristically, the presence of a characteristic manifold Σ imposes a quantization of the possible ways in which the nodal set can diffuse across Σ .

With the previous decomposition in mind, we can reduce the classification of the possible blow-up limits to the symmetric ones ad finally recover all the possible cases. In particular, in our analysis we introduce the new notion of "tangent field" Φ^{X_0} of u at a nodal point, which takes care of the different behaviour of both the symmetric and antisymmetric part of an L_a -harmonic function. Namely, by the decomposition and the Definition 3.5.7 of the notion of tangent map, i.e. the unique nonzero map $\varphi^{X_0} \in \mathfrak{B}_k^a(u)$ such that

$$u_{X_0,r}(X) = \frac{u(X_0 + rX)}{r^k} \longrightarrow \varphi^{X_0}(X),$$

with k the vanishing order of u at X_0 , we introduce the following concept.

Definition 3.1.2. Let $a \in (-1,1)$, u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u) \cap \Sigma$, for some $k \ge \min\{1, 1-a\}$. We define as *tangent field* of u at X_0 the unique nontrivial vector field $\Phi^{X_0} \in (H^{1,a}_{loc}(\mathbb{R}^{n+1}))^2$ such that

$$\Phi^{X_0} = (\varphi_e^{X_0}, \varphi_o^{X_0}),$$

where $\varphi_e^{X_0}$ and $\varphi_o^{X_0}$ are respectively the tangent map of the symmetric part u_e of u and of the antisymmetric one u_o .

This new object allows to overcame the obstacle of the degeneracy-singularity of the coefficient and it allows to understand the topology of the nodal set by proving in Proposition 3.5.19 a "*vectorial*" counterpart of the classic result of upper semi-continuity of the vanishing order. Hence, given now the regular $\mathcal{R}(u)$ and singular part $\mathcal{S}(u)$ as

$$\mathcal{R}(u) = \left\{ X \in \Gamma(u) \colon |\nabla_x u(X)|^2 + \left| \partial_y^a u(X) \right|^2 \neq 0 \right\},\$$

$$\mathcal{S}(u) = \left\{ X \in \Gamma(u) \colon |\nabla_x u(X)|^2 + \left| \partial_y^a u(X) \right|^2 = 0 \right\},\$$

we developed a blow-up analysis in order to fully understand the structure of $\Gamma(u)$ in \mathbb{R}^{n+1} and its restriction on Σ . The following is a summary of the main result on the regular and singular set.

Theorem 3.1.3. Let $a \in (-1,1)$, $a \neq 0$ and u be an L_a -harmonic function in B_1 . Then the regular set $\mathcal{R}(u)$ is locally a $C^{k,r}$ hypersurface on \mathbb{R}^{n+1} in the variable $(x, y |y|^{-a})$ with

$$k = \left\lfloor \frac{2}{1-a} \right\rfloor$$
 and $r = \frac{2}{1-a} - \left\lfloor \frac{2}{1-a} \right\rfloor$.

On the other hand, it holds

$$\mathcal{S}(u) \cap \Sigma = \mathcal{S}^*(u) \cup \mathcal{S}^a(u)$$

where $S^*(u)$ is contained in a countable union of (n-2)-dimensional C^1 manifolds and $S^a(u)$ is contained in a countable union of (n-1)-dimensional C^1 manifolds. Moreover

$$\mathcal{S}^*(u) = \bigcup_{j=0}^{n-2} \mathcal{S}^*_j(u) \quad \textit{and} \quad \mathcal{S}^a(u) = \bigcup_{j=0}^{n-1} \mathcal{S}^a_j(u),$$

where both $S_j^*(u)$ and $S_j^a(u)$ are contained in a countable union of j-dimensional C^1 manifolds.

In the last part of Chapter 3 we present an applications of our theory in the context of nonlocal elliptic equations. In particular, inspired by [23, 27, 80], we exploit the local realisation of the fractional Laplacian, and more generally of fractional power of divergence form operator L with Lipschitz leading coefficient, in order to study the structure and the regularity of the nodal set of $(-L)^s$ -harmonic functions, for $s \in (0, 1)$. More precisely, we combine the extension developed in [80] with a geometric reduction introduced in [4] and deeply popularized in [48, 49].

This last Section allows to extend our analysis to fractional powers $(-\Delta_M)^s$ of the Laplace-Beltrami operator on a Riemannian manifold M, also for the case of Lipschitz metric, and moreover to conformal fractional Laplacian on conformally compact Einstein manifolds and asymptotically hyperbolic manifold, thanks to the extension technique developed in [27] and the asymptotic expansion of their geodesic boundary defining function. These examples suggest that our choice of weight collects a wilder class of degenerate-singular elliptic problems.

Moreover, our results show some purely nonlocal feature on the possible local expansion of $(-L)^s$ -harmonic map near their zero set and on the structure of the nodal set itself. One one side we prove that first term of the Taylor expansion of an $(-L)^s$ -harmonic function is either an homogeneous harmonic polynomial or any possible homogeneous polynomial. In particular, it implies

Theorem 3.1.4. Given $s \in (0, 1)$ and L a divergence form operator with Lipschitz leading coefficients, let u be $(-L)^s$ -harmonic in B_1 . Then it holds

$$\mathcal{S}(u) = \mathcal{S}^*(u) \cup \mathcal{S}^s(u)$$

where $S^*(u)$ is contained in a countable union of (n-2)-dimensional C^1 manifolds and $S^s(u)$ is contained in a countable union of (n-1)-dimensional C^1 manifolds. Moreover

$$\mathcal{S}^*(u) = \bigcup_{j=0}^{n-2} \mathcal{S}^*_j(u) \quad \textit{and} \quad \mathcal{S}^s(u) = \bigcup_{j=0}^{n-1} \mathcal{S}^s_j(u),$$

where both $S_i^*(u)$ and $S_i^s(u)$ are contained in a countable union of *j*-dimensional C^1 manifolds.

In the end, we prove what could be seen as the nonlocal counterpart of a conjecture that Lin proposed in [66]. Following his strategy, we give an explicit estimate on the (n-1)-Hausdorff measure of the nodal set $\Gamma(u)$ of *s*-harmonic functions in terms of the Almgren monotonicity formula previously introduced. Finally, we propose an interesting direction of research in order to improve that stated result.

This Chapter is organized as follows. In Section 3.2 we prove some general result about L_a -harmonic function, first of all the decomposition with respect to the direction orthogonal to Σ . After that in Section 3.3 we prove the validity of an Almgren's type monotonicity formula which allows in Section 3.4 to prove the existence of blow-up limit in every point of the nodal set $\Gamma(u)$. Finally, in Section 3.5 we prove a Weiss type monotonicity formula, which allows to introduce the notion of tangent map and tangent field at every point of the nodal set. In Section 3.6 we present some useful result on the stratification of the nodal set and finally in Section 3.7 we prove a general result on the regularity of the whole nodal set $\Gamma(u)$ and on its restriction on the characteristic manifold Σ . In the last two Sections we consider an application of the previous results for solutions of fractional powers of divergence form operator, with Lipschitz leading coefficient. In particular, in Section 3.8 we apply our technique in order to study the nodal set of *s*-harmonic function and, more generally, of solutions of $(-L)^s$ operators, and in Section 3.9 we give a new estimate of the Hausdorff measure of the nodal set of *s*-harmonic functions.

3.2 decomposition of L_a -harmonic functions

In this Section we states some general results on L_a -harmonic function and we introduce some basic additional concept that will be often use through this Chapter in order to better understand the structure of the nodal set $\Gamma(u)$.

In particular, we give a definition of characteristic manifold Σ for a degenerate-singular operator and we prove a crucial decomposition of L_a -harmonic function with respect to the orthogonal direction to Σ . Thanks to this property, we can state a general regularity result on L_a -harmonic functions.

As already remarked, in the pioneering works [44, 43] the authors introduced a class of degeneratesingular operator strictly correlated to some weighted Sobolev spaces with Muckenhoupt A_p weights. In [43, Section 2] they gave six general properties that the weight must satisfy in order to have existence of weak solutions, Sobolev embeddings, Poincaré inequality, Harnack inequality, local solvability in Hölder spaces and estimates on the Green's function and in particular they found a sufficient condition in the definition of the Muckenhoupt A_2 -class. Hence, they introduced for $a \in (-1, 1)$ the weighted Sobolev spaces $H^{1,a}(B_1)$ as the closure of $C^{\infty}(\overline{B_1})$ functions under the norm

$$||u||_{H^{1,a}(B_1)}^2 = \int_{B_1} |y|^a u^2 \mathrm{d}X + \int_{B_1} |y|^a |\nabla u|^2 \,\mathrm{d}X.$$

Anyway, as the authors in [43] pointed out in the study of a special classes of elliptic problem associated to quasi-conformal maps, properties as the Sobolev embeddings, Poincaré inequality, Harnack inequality and local solvability in Hölder spaces still hold for every $a \in (-1, +\infty)$. Thus, the following definition is well defined for every $a \in (-1, +\infty)$.

Definition 3.2.1 ([43]). Given $F = (f_1, \dots, f_n)$ on B_1 such that $|F| \in L^{2,-a}(B_1)$, we say that $u \in H^{1,a}(B_1)$ is a solution of $L_a u = \operatorname{div} F$ if for every $\varphi \in C_c^{\infty}(B_1)$ we have

$$\int_{B_1} |y|^a \langle \nabla u, \nabla \varphi \rangle \mathrm{d}X = \int_{B_1} \langle F, \nabla \varphi \rangle \mathrm{d}X$$

Similarly, a function $u \in H^{1,a}(B_1)$ is said to be L_a -harmonic in B_1 if for every $\varphi \in C_c^{\infty}(B_1)$ we have

$$\int_{B_1} |y|^a \langle \nabla u, \nabla \varphi \rangle \mathrm{d}X = 0.$$

Now, we can finally state the concept of characteristic manifold associated to the operator L_a . We obviously remark that the following definition can be easily generalized to the whole class of Muckenhoupt A_2 -weights, where in general Σ can be any possible non-smooth subset of \mathbb{R}^{n+1} with dimension $0 \le d \le n$.

Definition 3.2.2. Let $a \in (-1,1)$ and L_a the weighted divergence form operator in \mathbb{R}^{n+1} associated to the weight $\omega(X) = |y|^a$. Then we call as "*characteristic manifold*" associated to L_a the collection of points $\Sigma \subset \mathbb{R}^{n+1}$ where the weight takes value zero (degeneracy, a > 0) or infinite (singularity, a < 0).

We remark that in a more general case, on the characteristic manifold the weight could attains both zero and infinite values.

Since the operator L_a is uniformly elliptic on every compact subset of $\mathbb{R}^{n+1} \setminus \Sigma$, the challenging part of our work is the one related to the study of the nodal set near the characteristic manifold Σ associated to L_a . Inspired by this remark, through the Chapter we will focus on the case $X_0 \in \Sigma$ and we will simply compare the result on Σ with the case $\mathbb{R}^{n+1} \setminus \Sigma$, avoiding all the technical details.

In order to better understand the structure of the nodal set and the local behaviour of the L_a -harmonic function, we decided to decompose these functions with respect to the characteristic manifold Σ . More precisely, we construct, starting from an L_a -harmonic function, its parts respectively symmetric and antisymmetric with respect to the orthogonal direction to Σ , since we can imagine that the latter affects the way this functions cross the space of degeneracy-singularity.

Definition 3.2.3. Let $a \in (-1, 1)$ and $u \in H^{1,a}(B_1)$ be an L_a -harmonic function in B_1 . Then, u is said to be *symmetric* with respect to Σ if

$$u(x,-y) = u(x,y)$$
 in \mathbb{R}^{n+1} .

Conversely, the function u is said to be *antisymmetric* with respect to Σ if

$$u(x,-y) = u(x,y)$$
 in \mathbb{R}^{n+1} .

It is easy to see that given an L_a -harmonic function u in B_1 , the functions

$$u_e(x,y) = \frac{u(x,y) + u(x,-y)}{2}$$
 and $u_o(x,y) = \frac{u(x,y) - u(x,-y)}{2}$

are respectively symmetric and antisymmetric with respect to Σ and such that

$$u(X) = u_e(X) + u_o(X).$$

At first sight, the previous decomposition seems to be innocuous and independent on the occurrence of degeneracy of the operator, but with the following Propositions would be clear the complete picture of how the presence of a set where the coefficients take value zero of infinite affect the local behaviour of the solutions.

First, the following result allows us to focus the characterization of the blow-up limits to just the symmetric L_a -harmonic function

Proposition 3.2.4. Let $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 antisymmetric with respect to Σ . Thus, there exists a unique L_{2-a} -harmonic function v symmetric with respect to Σ such that

$$u(x,y) = v(x,y)y |y|^{-a} \quad \text{ in } \mathbb{R}^{n+1}.$$

Proof. Given $v(x, y) = u(x, y) |y|^a y^{-1}$, let us first prove that $v \in H^{1,2-a}(B_1)$, where $2 - a \in (1,3)$. By direct computations we get

$$\int_{B_1} |y|^{2-a} v^2 \mathrm{d}X = \int_{B_1} |y|^a u^2 \mathrm{d}X,$$
(95)

and similarly

$$\begin{split} \int_{B_1} |y|^{2-a} |\nabla v|^2 \, \mathrm{d}X &= \int_{B_1} |y|^a |\nabla u|^2 \, \mathrm{d}X + (a-1)^2 \int_{B_1} |y|^a \frac{u^2}{y^2} \mathrm{d}X \\ &\leq C \left(\int_{B_1} |y|^a \, u^2 \mathrm{d}X + \int_{B_1} |y|^a \, |\nabla u|^2 \, \mathrm{d}X \right), \end{split}$$

where in the last inequality we used the validity of an Hardy type inequality (see [42]). Since that for a.e. $X \in B_1$ we have

$$L_{2-a}v = \operatorname{div}(|y|^{2-a}\nabla v) = (a-1)\partial_y u + \operatorname{div}(y\nabla u) = y |y|^{-a} L_a u,$$
(96)

let us prove v is L_{2-a} -harmonic in B_1 in the sense of Definition 3.2.1. For every $\varphi \in C_c^{\infty}(B_1)$ and $0 < \delta < 1$ let $\eta_{\delta} \in C^{\infty}(B_1)$ be a family of compactly supported cut-off functions such that $0 \le \eta_{\delta} \le 1$ and

$$\eta_{\delta}(x,y) = \begin{cases} 0 & \text{ on } \{(x,y) \in B_1 \colon |y| \le \delta\}, \\ 1 & \text{ on } \{(x,y) \in B_1 \colon |y| \ge 2\delta\}, \end{cases}$$

with $|\nabla \eta_{\delta}| \leq 1/\delta$. Thus, by testing (96) with $\varphi \eta_{\delta}$ we get for every $\delta \in (0, 1)$

$$\int_{B_1} |y|^{2-a} \langle \nabla v, \nabla(\eta_\delta \varphi) \rangle dX = -\int_{B_1} \eta_\delta \varphi L_{2-a} v dX$$
$$= -\int_{B_1} \left(y |y|^{-a} \eta_\delta \varphi \right) L_a u dX = 0,$$

where in the last equality we used that $y |y|^{-a} \eta_{\delta} \varphi \in C_c^{\infty}(B_1)$. Now, by integration by parts

$$\int_{B_1} |y|^{2-a} \langle \nabla v, \nabla(\eta_\delta \varphi) \rangle \mathrm{d}X = \int_{B_1} |y|^{2-a} \eta_\delta \langle \nabla v, \nabla \varphi \rangle \mathrm{d}X + \int_{B_1} |y|^{2-a} \varphi \langle \nabla v, \nabla \eta_\delta \rangle \mathrm{d}X,$$
(97)

where by Dominated convergence we get that

$$\lim_{\delta \to 0^+} \int_{B_1} |y|^{2-a} \eta_{\delta} \langle \nabla v, \nabla \varphi \rangle \mathrm{d}X = \int_{B_1} |y|^{2-a} \langle \nabla v, \nabla \varphi \rangle \mathrm{d}X$$

and by Hölder inequality

$$\begin{split} \int_{B_1} |y|^{2-a} \varphi \langle \nabla v, \nabla \eta_\delta \rangle \mathrm{d}X &\leq \|\varphi\|_{L^{\infty}(B_1)} \left(\int_{B_1} |y|^{2-a} |\nabla v|^2 \,\mathrm{d}X \right)^{1/2} \left(\int_{B_1} |y|^{2-a} |\nabla \eta_\delta|^2 \,\mathrm{d}X \right)^{1/2} \\ &\leq C \, \|\varphi\|_{L^{\infty}(B_1)} \, \|v\|_{H^{1,a}(B_1)} \, \frac{1}{\delta} \left(\int_{\delta}^{2\delta} |y|^{2-a} \,\mathrm{d}y \right)^{1/2} \\ &\leq C \, \|\varphi\|_{L^{\infty}(B_1)} \, \|v\|_{H^{1,a}(B_1)} \left(\frac{2^{3-a}-1}{3-a} \right)^{1/2} \delta^{\frac{1-a}{2}}, \end{split}$$

which imply, passing through $\delta \rightarrow 0$ in (97), that

$$\int_{B_1} |y|^{2-a} \langle \nabla v, \nabla \varphi \rangle \mathrm{d} X = 0 \quad \text{ for } \varphi \in C^\infty_c(B_1),$$

since we are dealing with a < 1.

Hence, for $a \in (-1,1)$ and every L_a -harmonic function $u \in H^{1,a}(B_1)$ there exist $u_e^a \in H^{1,a}(B_1)$ and $u_e^{2-a} \in H^{1,2-a}(B_1)$ two symmetric function with respect to Σ respectively L_a and L_{2-a} harmonic in B_1 such that

$$u(X) = u_e^a(X) + u_e^{2-a}(X)y |y|^{-a} \quad \text{in } B_1.$$
(98)

Thus, through the following Sections we will restrict the classification of the blow-up limit, i.e. the entire homogenous L_a -harmonic functions, to the symmetric with respect to Σ and in the final part of the work we will collect all the result for a generic L_a -harmonic function.

Secondly, the previous decomposition combined with the following result gives a complete picture of the regularity of an L_a -harmonic function.

Proposition 3.2.5 ([87]). Let $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 . Then it holds:

• if u is symmetric with respect to Σ , we get $u \in C^{1,\alpha}_{loc}(B_1)$, for any $\alpha \in (0,1)$;

• if u is antisymmetric with respect to Σ , we get $u \in C^{0,\alpha}_{\text{loc}}(B_1)$, for any $\alpha \in (0, \alpha^*)$ with $\alpha^* = \min\{1, 1-a\}$.

Moreover, if $a \in (-1, +\infty)$ and u is symmetric with respect to Σ , we even get that $u \in C^{\infty}_{loc}(B_1)$.

Proposition 3.2.6 ([87]). Let $a \in (-1, 1)$ and u be L_a -harmonic in B_1 . Then for every i = 1, ..., n we get that $\partial_{x_i} u$ is L_a -harmonic in B_1 and $\partial_y^a u$ is L_{-a} -harmonic in B_1 , where

$$\partial_y^a u = \begin{cases} |y|^a \, \partial_y u & \text{if } X \notin \Sigma \\ \lim_{y \to 0} |y|^a \, \partial_y u(x, y) & \text{if } X \in \Sigma \end{cases}.$$

These results have been recently obtained in [87] using some new approximation technique and Liouville type theorem for a wilder class of degenerate-singular elliptic problems. The main idea is to consider degenerate-singular operator as asymptotic limit of a specific class of uniformly elliptic operator, where the exponent of Hölder regularity can be reached by a blow-up argument combined with some Almgren's type monotonicity formula.

We recall here some general result about L_a -harmonic functions. First we introduce the following Caccioppoli inequality, which enables us to give a priori estimates of the $L^{2,a}$ norm of the derivatives of the solution u in terms of the $L^{2,a}$ -norm of u.

Proposition 3.2.7. Let $a \in (-1, 1)$ and u an L_a -harmonic function in B_1 . Then, for each $X_0 \in B_1 \cap \Sigma$ and $0 < r < R \le 1 - |X_0|$ we have

$$\int_{B_r(X_0)} |y|^a \, |\nabla u|^2 \, \mathrm{d}X \le \frac{C}{(R-r)^2} \int_{B_R(X_0) \setminus B_r(X_0)} |y|^a \, |u-\lambda|^2 \, \mathrm{d}X,\tag{99}$$

for every $\lambda \in \mathbb{R}$.

Proof. Fix $0 < r < R \le 1 - |X_0|$ and consider a smooth cut-off function $\eta \in C_c^{\infty}(B_1)$ such that $0 \le \eta \le 1$ and $\eta \equiv 1$ on $B_r(X_0)$ and $\eta \equiv 0$ on $B_R(X_0) \setminus B_r(X_0)$. Moreover, it is not restrictive to suppose that

$$|
abla \eta| \leq rac{2}{R-r} \quad ext{in } B_R(X_0).$$

Now, by testing the equation $-L_a u = 0$ with the test function $\varphi = (u - \lambda)\eta^2$ and integrating by parts, we get

$$\int_{B_1} |y|^a \eta^2 |\nabla u|^2 dX + 2 \int_{B_1} |y|^a \eta \langle \nabla \eta, \nabla u \rangle (u - \lambda) dX = 0,$$

and using the Hölder inequality

$$\int_{B_R(X_0)} |y|^a \eta^2 |\nabla u|^2 \, \mathrm{d}X \le \left(\int_{B_R(X_0)} |y|^a \eta^2 |\nabla u|^2 \, \mathrm{d}X \right)^{1/2} \left(\int_{B_R(X_0)} 4 \, |y|^a \, |u - \lambda|^2 \, |\nabla \eta|^2 \, \mathrm{d}X \right)^{1/2}$$

Hence, dividing by the first term in the right hand side and taking into account the properties of η we obtain

$$\int_{B_{r}(X_{0})} |y|^{a} |\nabla u|^{2} dX \leq \int_{B_{R}(X_{0})} |y|^{a} \eta^{2} |\nabla u|^{2} dX$$
$$\leq \frac{16}{(R-r)^{2}} \int_{B_{R}(X_{0}) \setminus B_{r}(X_{0})} |y|^{a} |u-\lambda|^{2} dX.$$

Now, for $a \in (-1, 1)$ let us fix

$$|S^n|_a = \int_{\partial B_1} |y|^a \,\mathrm{d}\sigma,$$

which implies

$$|S_{r}^{n}|_{a} = r^{n+a} \left|S^{n}\right|_{a}$$
 and $\left|B_{r}^{n+1}\right|_{a} = \frac{r^{n+a+1}}{n+a+1} \left|S^{n}\right|_{a}$.

Lemma 3.2.8 ([88, Lemma A.1]). Let $a \in (-1, 1)$ ad u be an L_a -harmonic function in B_1 . Then, for each ball $B_r(X_0)$, with $X_0 \in \Sigma$ and $r \in (0, 1 - |X_0|)$, we have

$$u(X_0) = \frac{1}{|S^n|_a r^{n+a}} \int_{\partial B_r(X_0)} |y|^a \, u \mathrm{d}\sigma = \frac{1}{|B^{n+1}|_a r^{n+a+1}} \int_{B_r(X_0)} |y|^a \, u \mathrm{d}X.$$

Proof. Let us consider the case $X_0 = 0$ since the problem is invariant under translation on $B_1 \cap \Sigma$. Set

$$\Phi(r) = \frac{1}{r^{n+a}} \int_{\partial B_r} |y|^a \, u \mathrm{d}\sigma = \int_{\partial B_1} |y|^a \, u(rx) \mathrm{d}\sigma,$$

then

$$\frac{d}{dr}\Phi(r) = \frac{1}{r^{n+a}} \int_{\partial B_r} |y|^a \,\partial_r u \mathrm{d}\sigma.$$

Since $L_a u = 0$ on B_1 , by a Gauss-Green formula we get

$$\int_{\partial B_r} |y|^a \,\partial_r u \mathrm{d}\sigma = \int_{B_r} L_a u \mathrm{d}X = 0$$

 \square

which directly implies that $r \mapsto \Phi(r)$ is constant and consequently

$$\frac{1}{r^{n+a}} \int_{\partial B_r} |y|^a \, u \mathrm{d}\sigma = \lim_{r \to 0} \int_{\partial B_1} |y|^a \lim_{r \to 0} u(rx) \mathrm{d}\sigma$$
$$= u(0) \, |S^n|_a \, .$$

Similarly, by integrating from 0 to r the function $\Phi(r)$, we get

$$\int_0^r \int_{\partial B_t} |y|^a \, u \mathrm{d}\sigma \mathrm{d}t = \int_{B_r} |y|^a \, u \mathrm{d}x$$

and secondly

$$\int_0^r \left(\int_{\partial B_t} |y|^a \, u \mathrm{d}\sigma \right) \mathrm{d}t = u(0) \, |S^n|_a \int_0^r t^{n+a} \mathrm{d}t,$$

from which we get the claimed result.

We remark that in the case of L_a -subharmonic function, i.e. $-L_a u \leq 0$, the previous result holds true in the form of inequality. Finally, by standard Moser's iteration, we also have the following bound

Lemma 3.2.9 ([88, Lemma A.2.]). Let $a \in (-1, 1)$ and u be a L_a -subharmonic function in B_1 . Then, for $X_0 \in B_1 \cap \Sigma$ and $r \in (0, 1 - |X_0|)$ we get

$$||u||_{L^{\infty}(B_{r/2}(X_0))} \le C(n,a) \left(\frac{1}{r^{n+1+a}} \int_{B_r(X_0)} |y|^a u^2 \mathrm{d}X\right)^{1/2},$$

where C(n, a) is a constant depending only on n and a.

3.3 ALMGREN TYPE MONOTONICITY FORMULA

In this Section we introduce the degenerate-singular counterpart of the classical Almgren monotonicity formula for harmonic functions. This computations are more manageable with respect to the ones in Chapter 1 since peculiar phenomena like the problem of the codimension of the nodal set does not manifest in the case of L_a -harmonic functions.

Since we want to understand the structure and regularity of the nodal set of L_a -harmonic function near the characteristic manifold Σ , let us consider $X_0 = (x_0, 0) \in \Sigma$. Hence, for every $r \in (0, R)$, where R > 0 will be defined later, consider

$$E(X_0, u, r) = \frac{1}{r^{n+a-1}} \int_{B_r(X_0)} |y|^a |\nabla u|^2 dX$$
$$H(X_0, u, r) = \frac{1}{r^{n+a}} \int_{\partial B_r(X_0)} |y|^a u^2 d\sigma$$

and the Almgren type monotonicity formula

$$N(X_0, u, r) = \frac{E(X_0, u, r)}{H(X_0, u, r)} = \frac{r \int_{B_r(X_0)} |y|^a \, |\nabla u|^2 \, \mathrm{d}X}{\int_{\partial B_r(X_0)} |y|^a \, u^2 \mathrm{d}\sigma}.$$
 (100)

Since $u \in H^{1,a}_{loc}(B_1)$, both the functional $r \mapsto E(X_0, u, r)$ and $r \mapsto H(X_0, u, r)$ are locally absolutely continuous on $(0, +\infty)$, that is that both them derivatives are in $L^1_{loc}((0, +\infty))$.

Proposition 3.3.1. Let $a \in (-1,1)$ and u be an L_a -harmonic function on B_1 . Then, for every $X_0 \in B_1 \cap \Sigma$ we have that the map $r \mapsto N(X_0, u, r)$ is absolutely continuous and monotone nondecreasing on $(0, 1 - |X_0|)$.

Hence, there always exists finite the limit

$$N(X_0, u, 0^+) = \lim_{r \to 0^+} N(X_0, u, r) = \inf_{r > 0} N(X_0, u, r),$$

which we will call as the Almgren frequency formula.

Proof. Obviously the denominator is nonnegative and at least strictly positive on a nonempty interval (r_1, r_2) , otherwise we get $u \equiv 0$. First, passing to the logarithmic derivatives, the monotonicity of $r \mapsto N(X_0, u, r)$ is a direct consequence of the claim

$$\frac{d}{dr}\log N(X_0, u, r) = \frac{1}{r} + \frac{\frac{d}{dr}\int_{B_r(X_0)} |y|^a |\nabla u|^2 dX}{\int_{B_r(X_0)} |y|^a |\nabla u|^2 dX} - \frac{\frac{d}{dr}\int_{\partial B_r(X_0)} |y|^a u^2 d\sigma}{\int_{\partial B_r(X_0)} |y|^a u^2 d\sigma} \ge 0$$

for $r \in (r_1, r_2)$. Deriving the numerator and using the Pohožaev identity, i.e. for any $X_0 \in \mathbb{R}^{n+1}$ and r > 0

$$\frac{1-n-a}{2} \int_{B_r(X_0)} |y|^a \, |\nabla u|^2 \, \mathrm{d}X + \frac{r}{2} \int_{\partial B_r(X_0)} |y|^a \, |\nabla u|^2 \, \mathrm{d}\sigma = r \int_{\partial B_r(X_0)} |y|^a \, (\partial_r u)^2 \mathrm{d}\sigma + \frac{ay_0}{2} \int_{B_r(X_0)} |y|^a \frac{|\nabla u|^2}{y} \mathrm{d}X$$

we easily get

$$\frac{d}{dr} \int_{B_r(X_0)} |y|^a |\nabla u|^2 dX = \int_{\partial B_r(X_0)} |y|^a |\nabla u|^2 d\sigma$$

$$= 2 \int_{\partial B_r(X_0)} |y|^a (\partial_r u)^2 d\sigma + \frac{n+a-1}{r} \int_{B_r(X_0)} |y|^a |\nabla u|^2 dX$$

$$- \frac{ay_0}{r} \int_{B_r(X_0)} |y|^a \frac{|\nabla u|^2}{y} dX$$
(101)

and similarly

$$\frac{d}{dr} \int_{\partial B_r(X_0)} |y|^a u^2 d\sigma = \frac{d}{dr} \left(r^n \int_{\partial B_1} |y_0 + ry|^a u^2 (X_0 + rX) d\sigma \right)
= 2 \int_{\partial B_r(X_0)} |y|^a u \partial_r u d\sigma - \frac{ay_0}{r} \int_{\partial B_r(X_0)} \frac{|y|^a}{y} u^2 d\sigma + (102)
+ \frac{n+a}{r} \int_{\partial B_r(X_0)} |y|^a u^2 d\sigma.$$

As a consequence, by the Cauchy-Schwarz inequality, if $X_0 \in \Sigma$, i.e. $y_0 = 0$, we get

$$\frac{d}{dr}E(X_0, u, r) = \frac{2}{r^{n+a-1}} \int_{\partial B_r(X_0)} |y|^a (\partial_r u)^2 d\sigma$$
$$\frac{d}{dr}H(X_0, u, r) = \frac{2}{r^{n+a}} \int_{\partial B_r(X_0)} |y|^a u \partial_r u d\sigma$$

and consequently

$$\frac{1}{2}\frac{d}{dr}\log N(X_0, u, r) = \frac{\int_{\partial B_r(X_0)} |y|^a (\partial_r u)^2 \mathrm{d}\sigma}{\int_{\partial B_r(X_0)} |y|^a u \partial_r u \mathrm{d}\sigma} - \frac{\int_{\partial B_r(X_0)} |y|^a u \partial_r u \mathrm{d}\sigma}{\int_{\partial B_r(X_0)} |y|^a u^2 \mathrm{d}\sigma} \ge 0$$

for $r\in(r_1,r_2).$ By the previous differentiation, we have

$$\frac{d}{dr}\log N(X_0, u, r) \ge 0 \quad \text{for } r \in (r_1, r_2)$$

and

$$\frac{d}{dr}\log H(X_0, u, r) = \frac{2}{r}N(X_0, u, r).$$
(103)

Following the same reasoning in [84], it is quite easy to conclude that the maximum interval is the one with $r_1 = 0$.

As a direct consequence of the monotonicity result, we get that the Almgren frequency formula $X \mapsto N(X, u, 0^+)$ on Σ is upper semi-continuous since it is defined as the infimum of continuous function.

A simple consequence of the monotonicity result and (103) is the following comparison property (which, with $r_2 = 2r_1$, is the so called doubling property).

Corollary 3.3.2. Let $a \in (-1,1)$ and u be L_a -harmonic on B_1 . Hence, there given $N = N(X_0, u, 1 - |X_0|)$ such that for every $X_0 \in B_1 \cap \Sigma$,

$$H(X_0, u, r_2) \le H(X_0, u, r_1) \left(\frac{r_2}{r_1}\right)^{2N}$$

for $0 < r_1 < r_2 < 1 - |X_0|$.

Proof. Fixed $R = 1 - |X_0|$ we have that $N(X_0, u, r) \le N(X_0, u, R)$ for every $r \in (0, R)$ and integrating (103) between r_1 and r_2 , with $0 < r_1 < r_2 \le R$, we obtain

$$\frac{H(X_0, u, r_2)}{H(X_0, u, r_1)} \le \left(\frac{r_2}{r_1}\right)^{2N}$$

whit $N = N(X_0, u, R)$.

In other words, for every $X_0 \in B_1 \cap \Sigma$

$$\frac{1}{R^{n+a}} \int_{\partial B_R(X_0)} |y|^a \, u^2 \mathrm{d}\sigma \le \left(\frac{R}{r}\right)^{2N} \frac{1}{r^{n+a}} \int_{\partial B_r(X_0)} |y|^a \, u^2 \mathrm{d}\sigma$$

with $0 < r < R < 1 - |X_0|$ and $N = N(X_0, u, 1 - |X_0|)$, and integrating the previous inequality we get

$$\frac{1}{R^{n+a+1}} \int_{B_R(X_0)} |y|^a \, u^2 \mathrm{d}X \le \left(\frac{R}{r}\right)^{2N-1} \frac{1}{r^{n+a+1}} \int_{B_r(X_0)} |y|^a \, u^2 \mathrm{d}X. \tag{104}$$

In order to justify the analysis of the local behaviour of L_a -harmonic functions, we prove the validity of the strong unique continuation property for the degenerate-singular operator L_a . In general, a function u, is said to vanish of infinite order at a point $X_0 \in \Gamma(u)$ if

$$\int_{|X-X_0| < r} u^2 \mathrm{d}X = O(r^k), \quad \text{for every } k \in \mathbb{N},$$

as $r \to 0$. Given an elliptic operator L, L is said to have the *strong unique continuation property* in B_1 if the only solution of Lu = 0 in $H^1_{loc}(B_1)$ which vanishes of infinite order at a point $X_0 \in \Gamma(u)$ is u = 0. Moreover, L is said to have the *unique continuation property* in B_1 if the solution of Lu = 0 in $H^1_{loc}(B_1)$ which can vanish in an open subset of B_1 is u = 0. (see [48, 49] for more details for the uniformly elliptic case).

Corollary 3.3.3 ([49, Theorem 1.4]). Let $a \in (-1, 1)$ and u be L_a -harmonic in B_1 . Then u cannot vanish of infinite order at $X_0 \in \Gamma(u) \cap B_1$ unless $u \equiv 0$ in B_1 .

In [49] the authors stated the proof for analytic nonnegative weights and pointed out the validity for more general, even degenerate, weighted elliptic equations.

The previous result implies that the nodal set $\Gamma(u)$ has empty interior in \mathbb{R}^{n+1} . As a consequence of our blow-up analysis, we will prove a posteriori unique continuation property for the restriction of $\Gamma(u)$ on Σ .

Corollary 3.3.4. Let $a \in (-1,1)$ and u be an L_a -harmonic function on B_1 . Then, for every $X_0 \in B_1 \cap \Sigma$ given $R = 1 - |X_0|$ we get

$$\frac{1}{n+a+1+2N} \int_{\partial B_R(X_0)} |y|^a u^2 \mathrm{d}\sigma \le \int_{B_R(X_0)} |y|^a u^2 \mathrm{d}X \le \frac{1}{n+a+1} \int_{\partial B_R(X_0)} |y|^a u^2 \mathrm{d}\sigma$$

where $N = N(X_0, u, R)$.

Proof. Let $R = 1 - |X_0|$ and $r \in (0, R)$, we get by (103)

$$H(X_0, u, R) = H(X_0, u, r) \exp\left\{2\int_r^R \frac{N(X_0, u, t)}{t} \mathrm{d}t\right\} \ge H(X_0, u, r)$$

or simply

$$\int_{\partial B_r(X_0)} |y|^a u^2 \mathrm{d}\sigma \le H(X_0, u, R) r^{n+a}.$$

Finally, integrating the previous inequality in (0, R) we obtain

$$\int_{B_R(X_0)} |y|^a u^2 dX = \int_0^R \int_{\partial B_r(X_0)} |y|^a u^2 d\sigma dr$$

$$\leq H(X_0, u, R) \int_0^R r^{n+a} dr = \frac{1}{n+a+1} H(X_0, u, R).$$

On the other hand, for any $r \in (0, R)$, we have

$$H(X_0, u, R) = H(X_0, u, r) \exp\left\{2\int_r^R \frac{N(X_0, u, t)}{t} \mathrm{d}t\right\}$$
$$\geq H(X_0, u, r) \exp\left\{-2N(X_0, u, r) \log r\right\}$$

and consequently

$$H(X_0, u, r) \ge r^{2N(X_0, u, R)} H(X_0, u, R).$$

Thus, as before

$$\int_{B_R(X_0)} |y|^a u^2 \mathrm{d}X = \int_0^R \int_{\partial B_r(X_0)} |y|^a u^2 \mathrm{d}\sigma \ge \frac{1}{n+a+1+2N(X_0, u, R)} H(X_0, u, R).$$

The following result can be viewed as the degenerate-singular counterpart of [55, Theorem 1.6], which gives us a sufficient condition for the presence of the nodal set in the unitary ball.

Proposition 3.3.5. Let $a \in (-1,1)$ and u be an L_a -harmonic function on B_1 . Then, for any $R \in (0,1)$ there exists $N_0 = N_0(R) \ll 1$ such that the following holds:

- 1. if $N(0, u, 1) \leq N_0$, then u does not vanish in B_R ;
- 2. if $N(0, u, 1) > N_0$, then

$$N\left(X_0, u, \frac{1-R}{2}\right) \le CN(0, u, 1)$$
 for any $X_0 \in B_R \cap \Sigma$,

where C is a positive constant depending only on n, a and R.

Moreover, the vanishing order, i.e. the Almgren frequency formula, of u at any point of B_R never exceeds CN(0, u, 1).

Proof. This prof will follows directly the one in [55, 66]. Moreover, the previous result is known to be true if we restrict our study to the set $B_1 \setminus \Sigma$, by the local uniform ellipticity of the operator L_a outside the characteristic manifold. First, the monotonicity of $r \mapsto N(0, u, r)$ implies that the vanishing order of u at 0 never exceeds N(0, u, 1), more precisely

$$\frac{1}{(\lambda R)^{n+1+a}} \int_{B_{\lambda R}} |y|^a \, u^2 \mathrm{d}X \le \lambda^{2N(0,u,1)} \frac{1}{R^{n+1+a}} \int_{B_R} |y|^a \, u^2 \mathrm{d}X$$

for every $R \in (0, 1)$ and $\lambda \in (1, 1/R)$. Through this proof we will use the following notation to identify the average of the integrals

$$\int_{B_r(X_0)} |y|^a \, u^2 \mathrm{d}X = \frac{1}{r^{n+1+a}} \int_{B_r(X_0)} |y|^a \, u^2 \mathrm{d}X,$$

and in order to simplify the notations we will use N = N(0, u, 1) as the frequency of u in B_1 . Under these notations, the previous inequalities become

$$\begin{aligned} & \int_{\partial B_{\lambda R}} |y|^a \, u^2 \mathrm{d}\sigma \leq \lambda^{2N} \int_{\partial B_R} |y|^a \, u^2 \mathrm{d}\sigma \\ & \int_{B_{\lambda R}} |y|^a \, u^2 \mathrm{d}X \leq \lambda^{-1} \lambda^{2N} \int_{B_R} |y|^a \, u^2 \mathrm{d}X \end{aligned}$$

where the second one is a consequence of Corollary 3.3.2.

Now, let us prove the claimed result for the case R = 1/4 since in the general case it follows by scaling. By definition, we have that $B_{3/4}(X_0) \subset B_1$ and $B_{1/4} \subset B_{1/2}(X_0)$ for any $X_0 \in B_{1/4}$. Hence, we have

$$\int_{B_{3/4}(X_0)} |y|^a \, u^2 \mathrm{d}X \le c(n,a) 4^{2N} \int_{B_{1/2}(X_0)} |y|^a \, u^2 \mathrm{d}X$$

for any $X_0 \in B_{1/4}$. Now, let us prove that

$$\int_{\partial B_{5/8}(X_0)} |y|^a \, u^2 \mathrm{d}\sigma \le c(n,a) 4^{2N} \oint_{\partial B_{1/2}(X_0)} |y|^a \, u^2 \mathrm{d}\sigma. \tag{105}$$

Since by (103) the map $r \mapsto H(X_0, u, r)$ is monotone non decreasing on $(0, 1 - |X_0|)$, and hence

$$\int_{B_{3/4}(X_0)} |y|^a u^2 dX \ge \int_{B_{3/4}(X_0) \setminus B_{5/8}(X_0)} |y|^a u^2 dX$$
$$= \int_{5/8}^{3/4} r^{n+a} H(X_0, u, r) dr$$
$$\ge C(n, a) H(X_0, u, 5/8),$$

and similarly

$$\int_{B_{1/2}(X_0)} |y|^a u^2 \mathrm{d}X = \int_0^{1/2} r^{n+a} H(X_0, u, r) \mathrm{d}r \le C(n, a) H\left(X_0, u, \frac{1}{2}\right).$$

Finally, integrating (103) between the previous radii, we obtain

$$\log H(X_0, u, r) \Big|_{1/2}^{5/8} = \int_{1/2}^{5/8} \frac{2N(X_0, u, r)}{r} \mathrm{d}r \ge 2C(n)N(X_0, u, 1/2).$$

Combining the previous inequality with the claimed (105), we get

$$c(n,a)N(X_0, u, 1/2) \le \log\left(c(n,a)4^{2N(0,u,1)}\right)$$

or equivalently $N(X_0, u, 1/2) \le c(n, a)N(0, u, 1) + c(n, a)$. Finally, let us consider the second part of the statement. Hence, given $\varepsilon = \varepsilon(n, a)$, sufficiently small, such that $N(0, u, 1) \le \varepsilon$, let us prove that $u(X_0) \ne 0$ for any $X_0 \in B_{1/4}$. It is not restrictive to assume that H(0, u, 1) = 1, which implies by the definition of the Almgren monotonicity formula that

$$\int_{B_1} |y|^a \, |\nabla u|^2 \, \mathrm{d}X \le \varepsilon.$$

By the L_a -harmonicity of u, for every i = 1, ..., n the derivative $\partial_{x_i} u$ and $\partial_y^a u$ are respectively L_a and L_{-a} -harmonic in B_1 . Hence, by [88, Lemma A.2], we get the following interior estimates

$$\begin{split} \sup_{B_{1/2}} |\partial_{x_i} u| &\leq c(n,a) \left(\frac{1}{r^{n+a+1}} \int_{B_1} |y|^a \, |\partial_{x_i} u|^2 \, \mathrm{d}X \right)^{1/2} \leq c(n,a) \sqrt{\varepsilon} \\ \sup_{B_{1/2}} \left| \partial_y^a u \right| &\leq c(n,-a) \left(\frac{1}{r^{n-a+1}} \int_{B_1} |y|^{-a} \left| \partial_y^a u \right|^2 \, \mathrm{d}X \right)^{1/2} \leq c(n,a) \sqrt{\varepsilon}. \end{split}$$

By the normalization assumption, we have

$$1 = \int_{\partial B_1} |y|^a u^2 \mathrm{d}\sigma \le c_1(n,a) \int_{\partial B_{1/2}} |y|^a u^2 \mathrm{d}\sigma,$$

and consequently the existence of $X_0 \in \partial B_{1/2}$ such that

$$|u(X_0)|^2 \ge \frac{2}{c_1(n,a)2^{n+a}} \int_{\partial B_1} |y|^a \,\mathrm{d}\sigma.$$

Up to relabeling with $c_1(n, a)$ the previous lower bound, we get that for every $X \in B_{1/2}$ that

$$c_1(n,a) \le |u(X_0)| \le |u(X)| + c(n,a)\sqrt{\varepsilon},$$

which yields |u(X)| > 0 of $B_{1/2}$, for $\varepsilon = \varepsilon(n, a)$ sufficiently small.

Corollary 3.3.6. Let u be L_a -harmonic on B_1 , then for every $X_0 \in \Gamma(u) \cap \Sigma$ we have

$$N(X_0, u, 0^+) \ge \min\{1, 1-a\}.$$
(106)

More precisely

- if u is symmetric with respect to Σ , we have $N(X_0, u, 0^+) \ge 1$,
- if u is antisymmetric with respect to Σ we have $N(X_0, u, 0^+) \ge 1 a$.

Proof. This result follows by Proposition 3.2.5. More precisely, et $\alpha^* = \min\{1, 1 - a\}$ be the coefficient of optimal Hölder regularity for L_a -harmonic function, and suppose by contradiction that (106) is not satisfied.

Since the limit $N(X_0, u, 0^+)$ exists, we obtain the existence of R > 0 and $\varepsilon > 0$ such that $N(X_0, u, r) \le \alpha^* - \varepsilon$ for all $0 \le r \le R$. By (103), up to consider a smaller interval of (0, R), we have

$$\frac{d}{dr}\log H(X_0, u, r) = \frac{2}{r}N(X_0, u, r) \le \frac{2}{r}(\alpha^* - \varepsilon).$$

Integrating this inequality between r and R yields

$$\frac{H(X_0, u, R)}{H(X_0, u, r)} \le \left(\frac{R}{r}\right)^{2(\alpha^* - \varepsilon)}$$

which, together with the fact that u is α^* -Hölder continuous and $u(X_0) = 0$, implies

$$C_1 r^{2(\alpha^* - \epsilon)} \le H(X_0, u, r) \le C_2 r^{2\alpha^*}.$$

The contradiction follows for small value of r > 0.

If an addition we suppose that u is symmetric or antisymmetric with respect to Σ , we get respectively that u is Lipschitz continuous or (1 - a)-Hölder continuous, and the lower bound on the Almgren frequency formula follows immediately.

In all the first part of this Section, we had supposed that $X_0 \in B_1 \cap \Sigma$, since the degeneratesingular attitude of the operator L_a is constrained to the characteristic manifold Σ . Instead, if $X_0 = (x_0, y_0) \in B_1 \setminus \Sigma$, since the operator is uniformly elliptic on $B_R(X_0) \subset \mathbb{R}^{n+1}$, with $R = |y_0|$, the structure and the regularity of the nodal set of u is well known.

At this point, we want to remark how the different scaling of the operator on Σ and on $\mathbb{R}^{n+1} \setminus \Sigma$ affects the Almgren monotonicity formula.

Let $X_0 = (x_0, y_0) \in B_1$ and r > 0 and consider u and L_a -harmonic function in B_1 . If we define $u_{X_0,r}(X) = u(X_0 + rX)$ we directly see that

$$\frac{|y_0 + ry|^a}{r^2} \left(\Delta_X u_{X_0, r} + \frac{ar}{y_0 + ry} \partial_y u_{X_0, r} \right) = 0 \quad \text{for } X \in \frac{B_1 - X_0}{r},$$

and

$$\int_{B_r(X_0)} L_a u \mathrm{d}X = \begin{cases} r^{n-1+a} \int_{B_1} \mathrm{div}\left(|y|^a \,\nabla u_{X_0,r}\right) \mathrm{d}X & \text{if } X_0 \in \Sigma \\ r^{n-1} \int_{B_1} \mathrm{div}\left(|y_0 + ry|^a \,\nabla u_{X_0,r}\right) \mathrm{d}X & \text{if } X_0 \notin \Sigma \end{cases}$$

Inspired by the different scalings of L_a operator, for $X_0 \in B_1 \setminus \Sigma$, i.e. $y_0 \neq 0$, and $r \in (0, |y_0|)$, let us introduce the following functionals

$$E(X_0, u, r) = \frac{1}{r^{n-1}} \int_{B_r(X_0)} |y|^a \, |\nabla u|^2 \, \mathrm{d}X,$$
$$H(X_0, u, r) = \frac{1}{r^n} \int_{\partial B_r(X_0)} |y|^a \, u^2 \mathrm{d}\sigma,$$

and consequently the Almgren type monotonicity formula

$$N(X_0, u, r) = \frac{E(X_0, u, r)}{H(X_0, u, r)} = \frac{r \int_{B_r(X_0)} |y|^a |\nabla u|^2 \,\mathrm{d}X}{\int_{\partial B_r(X_0)} |y|^a \,u^2 \mathrm{d}\sigma}.$$
 (107)

As we can see, the expression of the Almgren monotonicity formula is not affected by the position of the point $X_0 \in \mathbb{R}^{n+1}$, i.e if either $X_0 \in B_1 \cap \Sigma$ or $X_0 \in B_1 \setminus \Sigma$. Instead, the rescaling factor in the definitions of $E(X_0, u, r)$ and $H(X_0, u, r)$ are strictly related to the different attitudes of the operator L_a . By [48, 49] we already know the existence of an Almgren type monotonicity formula and the structure/regularity of the nodal set associated to uniformly elliptic operator. For completeness, we give some results on the Almgren type monotonicity result which holds for every $X_0 \in \mathbb{R}^{n+1}$ without using the change of coordinates introduced in [48, 49].

Proposition 3.3.7. Let $a \in (-1,1)$ and u be an L_a -harmonic function on B_1 . Then, for every $X_0 \in B_1 \setminus \Sigma$ there exists C > 0 such that $r \mapsto e^{Cr}N(X_0, u, r)$ is absolutely continuous and monotone nondecreasing on $(0, |y_0|)$.

Hence, there always exists finite the limit

$$N(X_0, u, 0^+) = \lim_{r \to 0^+} N(X_0, u, r),$$

which we will call as the Almgren type frequency formula.

Proof. The strategy of the proof is similar to the one for the case $X_0 \in B_1 \cap \Sigma$. By (101) and (102), we already know that passing to the logarithmic derivatives we get from the Cauchy-Schwarz inequality

$$\frac{d}{dr}\log N(X_0, u, r) = \frac{1}{r} + \frac{\frac{d}{dr}\int_{B_r(X_0)} |y|^a |\nabla u|^2 dX}{\int_{B_r(X_0)} |y|^a |\nabla u|^2 dX} - \frac{\frac{d}{dr}\int_{\partial B_r(X_0)} |y|^a u^2 d\sigma}{\int_{\partial B_r(X_0)} |y|^a u^2 d\sigma}$$
$$\geq \frac{ay_0 \int_{\partial B_r(X_0)} \frac{|y|^a}{y} u^2 d\sigma}{r \int_{\partial B_r(X_0)} |y|^a u^2 d\sigma} - \frac{ay_0 \int_{B_r(X_0)} \frac{|y|^a}{y} |\nabla u|^2 d\sigma}{r \int_{B_r(X_0)} |y|^a |\nabla u|^2 d\sigma}$$

for $r \in (r_1, r_2)$. This remainders come out since the Muckenhoupt A_2 -weight $\omega(X) = |y|^a$ is homogenous with respect to Σ . Obviously if a = 0, by the translation invariance of the Laplacian, we don't need to care anymore about the position of X_0 and also, just substituting a = 0, we obtain the classic Almgren type monotonicity formula of the Laplacian, e.g. [19, 81]. Now, for every $r \in (r_1, r_2)$

$$\frac{d}{dr}\log N(X_0, u, r) \ge \begin{cases} \frac{ay_0}{r} \left(\min_{\partial B_r(X_0)} \frac{1}{y} - \max_{B_r(X_0)} \frac{1}{y}\right) & \text{if } a \cdot y_0 > 0\\ -\frac{ay_0}{r} \left(\min_{B_r(X_0)} \frac{1}{y} - \max_{\partial B_r(X_0)} \frac{1}{y}\right) & \text{if } a \cdot y_0 < 0 \end{cases}$$

which is equivalent to

$$\frac{d}{dr}\log N(X_0, u, r) \ge -\frac{2|ay_0|}{y_0^2 - r^2} \ge -\frac{2|ay_0|}{y_0^2 - r_2^2} \quad \text{for } r \in (r_1, r_2),$$

from which we learn that necessary $r_2 < |y_0|$. Consider now

$$H(X_0, u, r) = \frac{1}{r^{n-1}} \int_{\partial B_r(X_0)} |y|^a \, u^2 \mathrm{d}\sigma$$

such that

$$\frac{d}{dr}\log H(X_0, u, r) = \frac{2}{r}N(X_0, u, r) + \frac{a}{r}\left(1 - y_0\frac{\int_{\partial B_r(X_0)}\frac{|y|^a}{y}u^2\mathrm{d}\sigma}{\int_{\partial B_r(X_0)}|y|^a u^2\mathrm{d}\sigma}\right).$$

Let us prove the existence of the limit of the Almgren frequency formula as $r \to 0^+$, so suppose by contradiction that $r_1 = \inf\{r > 0 : H(X_0, u, r) > 0$ on $(r, |y_0|)\} > 0$ and consider $r \in (r_1, |y_0|)$. By the previous inequality, we have that there exists a positive constant C > 0such that

$$r \mapsto e^{Cr} N(X_0, u, r)$$

is monotone nondecreasing on $(r_1,|y_0|).$ Then, let $r_1 < r < 2r_1 \leq |y_0|,$ since

$$\frac{ay_0 \int_{\partial B_r(X_0)} \frac{|y|^a}{y} u^2 d\sigma}{r \int_{\partial B_r(X_0)} |y|^a u^2 d\sigma} \ge \begin{cases} \frac{1}{r} \frac{ay_0}{y_0 + 2r_1} & \text{if } a \cdot y_0 > 0\\ \\ \frac{1}{r} \frac{ay_0}{y_0 - 2r_1} & \text{if } a \cdot y_0 < 0 \end{cases}$$
(108)

we have

$$\frac{d}{dr}\log H(X_0, u, r) \le \frac{2}{r}e^{2Cr_1}N(X_0, u, 2r_1)$$
(109)

By integrating (109), it follows

$$\frac{H(X_0, u, 2r_1)}{H(X_0, u, r)} \le \left(\frac{2r_1}{r}\right)^{2e^{2Cr_1}N(X_0, u, 2r_1)}$$

and since $r \mapsto H(X_0, u, r)$ is continuous, $H(X_0, u, r_1) > 0$ and we seek the contradiction.

As before, a simple consequence of the monotonicity result and (103) is the following comparison property (which, with $r_2 = 2r_1$, is the so called doubling property).

Corollary 3.3.8. Let u be an L_a -harmonic function in B_1 . For every $X_0 \in B_1 \setminus \Sigma$, there exists C > 0 and R > 0 such that

$$H(X_0, u, r_2) \le H(X_0, u, r_1) \left(\frac{r_2}{r_1}\right)^{2C}$$

for every $0 < r_1 < r_2 < R$.

Proof. Let us consider $R < |y_0|$ and $0 < r_1 < r_2 \le R$. In order to use the monotonicity of $r \mapsto N(X_0, u, r)$ in this case we need to fix C, R > 0 depending on the distance of X_0 from Σ . By (108) we get

$$\frac{d}{dr}\log H(X_0, u, r) \le \frac{2}{r}e^{2CR}N(X_0, u, R)$$

Now, by integrating the previous inequality we get the claimed result.

Moreover, since the operator L_a is uniformly elliptic outside Σ , we can apply the same reasoning using the Lipschitz optimal regularity in $\mathbb{R}^n \setminus \Sigma$ and proving

Corollary 3.3.9. Let u be an L_a -harmonic function in B_1 . For every $X_0 \in \Gamma(u) \setminus \Sigma$ we have $N(X_0, u, 0^+) \ge 1$.

3.4 COMPACTNESS OF BLOW-UP SEQUENCES

All techniques presented in the following Sections involve a local analysis of the solutions, which will be performed via a blow-up procedure. Fix $a \in (-1, 1)$ and u an L_a -harmonic function in B_1 . Consider now $X_0 \in \Gamma(u)$ a point on the nodal set of u, then for any $r_k \downarrow 0^+$ we define as the blow-up sequence the collection

$$u_k(X) = \frac{u(X_0 + r_k X)}{\sqrt{H(X_0, u, r_k)}} \quad \text{for } X \in X \in B_{X_0, r_k} = \frac{B_1 - X_0}{r_k},$$

such that $L_a u_k = 0$ and $||u_k||_{L^{2,a}(\partial B_1)} = 1$. Through this Chapter we will always apply a blow-up analysis centered in point of the nodal set $\Gamma(u)$ on the characteristic manifold Σ , since as we already remarked the local behaviour of L_a -harmonic function is known outside the characteristic manifold.

In this Section we will prove the convergence of the blow-up sequence and the classification of the blow-up limits starting from the following convergence result.

Theorem 3.4.1. Let $a \in (-1, 1)$ and $\alpha^* = \min\{1, 1-a\}$. Given $X_0 \in \Gamma(u) \cap \Sigma$ and a blow-up sequence u_k centered in X_0 and associated to some $r_k \downarrow 0^+$, there exists $p \in H^{1,a}_{loc}(\mathbb{R}^n)$ such that, up to a subsequence, $u_k \to p$ in $C^{0,\alpha}_{loc}(\mathbb{R}^n)$ for every $\alpha \in (0, \alpha^*)$ and strongly in $H^{1,a}_{loc}(\mathbb{R}^n)$. In particular, the blow-up limit is and entire L_a -harmonic function, i.e.

$$L_a p = 0$$
 in \mathbb{R}^{n+1} .

In particular, the previous result can be easily improved in the case of L_a -harmonic function purely symmetric with respect to Σ . More precisely, inspired by Proposition 3.2.5, in the first case the convergence holds in $C_{\text{loc}}^{1,\alpha}$ for every $\alpha \in (0,1)$, and this difference relies on the Liouville type theorems introduced in [87].

As in Chapter 1, the proof will be presented in a series of lemmata.

Lemma 3.4.2. Let $X_0 \in \Gamma(u) \cap \Sigma$. For any given R > 0, we have

$$\|u_k\|_{H^{1,a}(B_R)} \leq C \quad \text{and} \quad \|u_k\|_{L^\infty(\overline{B_R})} \leq C,$$

where C > 0 is a constant independent on k > 0.

Proof. Let us consider $\rho_k^2 = H(X_0, u, r_k)$, then by definition of the blow-up sequence u_k and Corollary 3.3.2 we get

$$\begin{split} \int_{\partial B_R} |y|^a u_k^2 \mathrm{d}\sigma &= \frac{1}{\rho_k^2} \int_{\partial B_R} |y|^a u^2 (X_0 + r_k X) \mathrm{d}\sigma \\ &= \frac{1}{\rho_k^2 r_k^{n+a}} \int_{\partial B_{Rr_k}(X_0)} |y|^a u^2 \mathrm{d}\sigma \\ &= R^{n+a} \frac{H(X_0, u, Rr_k)}{H(X_0, u, r_k)} \\ &\leq R^{n+a} \left(\frac{Rr_k}{r_k}\right)^{2\tilde{C}} \end{split}$$

which gives us $||u_k||^2_{L^{2,a}(\partial B_R)} \leq C(R)R^{n+a}$. Similarly

$$\int_{B_R} |y|^a |\nabla u_k|^2 d\sigma = N(0, u_k, R) \frac{1}{R} \int_{\partial B_R} |y|^a u_k^2 d\sigma$$

$$\leq C(R) R^{n-1+a} N(X_0, u, Rr_k)$$

$$\leq C(R) R^{n-1+a} N(X_0, u, R)$$
(110)

where in the last inequality we used the monotonicity result of Proposition 3.3.1. Since the map u_k is L_a - harmonic, by [88, Lemma A.2] we get

$$\begin{split} \sup_{\overline{B_{R/2}}} u_k &\leq C(n,s) \left(\frac{1}{R^{n+1+a}} \int_{B_R} |y|^a \, u_k^2 \mathrm{d}X \right)^{1/2} \\ &\leq C(n,s) \left(\frac{H(0,u_k,R)}{n+a+1} \right)^{1/2}, \end{split}$$

where in the second inequality we used the monotonicity of $r \mapsto H(0, u_k, r)$ in (0, R). Finally, the estimate follows directly from the one the $L^{2,a}(\partial B_R)$ -norm.

So far we have proved the existence of a nontrivial function $p \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1}) \cap L^{\infty}_{\text{loc}}(\mathbb{R}^{n+1})$ such that, up to a subsequence, we have $u_k \rightharpoonup p$ weakly in $H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ and $L_a p = 0$ in $\mathcal{D}'(\mathbb{R}^{n+1})$.

The next step is to prove that for $X_0 \in \Gamma(u) \cap \Sigma$ the convergence $u_k \to p$ is indeed strong in $H_{\text{loc}}^{1,a}$ and in $C_{\text{loc}}^{0,\alpha}$ for $\alpha \in (0, \alpha^*)$.

Lemma 3.4.3. For every R > 0, up to a subsequence, $u_k \to p$ strongly in $H^{1,a}(B_R)$.

Proof. We already know the existence of a blow-up limit $p \in H^{1,a}_{loc}(\mathbb{R}^n)$, which solves $L_a p = 0$ in $\mathcal{D}'(\mathbb{R}^n)$. Let $\varphi \in C_c^{\infty}(B_{2R})$ be a cut-off function such that $0 \leq \varphi \leq 1$, $\varphi \equiv 1$ in B_R . By the L_a -harmonicity of u, we get

$$\int_{B_{2R}} |y|^a \varphi |\nabla(u_k - p)|^2 dX + \int_{B_{2R}} |y|^a (u_k - p) \langle \nabla(u_k - p), \nabla\varphi \rangle dX = 0$$

and consequently, we can conclude just by observing that

$$\left| \int_{B_{2R}} |y|^a (u_k - p) \langle \nabla(u_k - p), \nabla\varphi \rangle \mathrm{d}X \right| \le C \|u_k - p\|_{L^{\infty}(B_{2R})} \|\nabla u_k\|_{L^{2,a}(B_{2R})} \to 0.$$

Lemma 3.4.4. For every R > 0 there exists C > 0, independent of k, such that

$$[u_k]_{C^{0,\alpha}(B_R)} = \sup_{X_1, X_2 \in \overline{B}_R} \frac{|u(X_1) - u(X_2)|}{|X_1 - X_2|^{\alpha}} \le C$$

for every $\alpha \in (0, \alpha^*)$.

Proof. The proof follows essentially the ideas of the similar results in [83, 84]. Without loss of generality, let R = 1 and suppose by contradiction that up to a subsequence

$$L_{k} = \sup_{X_{1}, X_{2} \in \overline{B}_{1}} \frac{|\eta(X_{1})u_{k}(X_{1}) - \eta(X_{2})u_{k}(X_{2})|}{|X_{1} - X_{2}|^{\alpha}} \to \infty$$

where $\eta \in C_c^{\infty}(B_1)$ is a smooth function such that

$$\begin{cases} \eta(X) = 1, & 0 \le |X| \le 1/2\\ 0 < \eta(X) \le 1, & 1/2 \le |X| \le 1\\ \eta(X) = 0, & |X| = 1. \end{cases}$$

Since we may assume that L_k is achieved by $(X_{1,k}, X_{2,k}) \in \overline{B}_1 \times \overline{B}_1$, given $r_k = |X_{1,k} - X_{2,k}|$ we can prove, as $k \to \infty$, that

•
$$r_k \to 0$$

• $\frac{\operatorname{dist}(X_{1,k}, \partial B_1)}{r_k} \to \infty, \frac{\operatorname{dist}(X_{2,k}, \partial B_1)}{r_k} \to \infty.$

Before to continue, let us fix the notations $X_{1,k} = (x_{1,k}, y_{1,k})$ and $X_{2,k} = (x_{2,k}, y_{2,k})$. Now, since by Lemma 3.4.2 the norm $||u_k||_{L^{\infty}(B^+)}$ is uniformly bounded, we have

$$L_k \le \frac{\|u_k\|_{L^{\infty}(B_1)}}{r_k^{\alpha}} \left(\eta(X_{1,k}) - \eta(X_{2,k})\right), \tag{111}$$

which immediately implies that $r_k \to 0$. Now, since η is compactly supported in B_1 , for every $X \in \overline{B}_1$ we have

$$\eta(X) \leq \operatorname{dist}(X, \partial B_1) \operatorname{Lip}(\eta),$$

where obviously $\text{Lip}(\eta)$ denotes the Lipschitz constant of η . Finally, the inequality (111) becomes

$$\frac{\operatorname{dist}(X_{1,k},\partial B)}{r_k} + \frac{\operatorname{dist}(X_{2,k},\partial B)}{r_k} \geq \frac{L_k r_k^{\alpha-1}}{\operatorname{Lip}(\eta) \|u_k\|_{L^\infty(B_1)}} \to \infty$$

and the result follows by recalling that $\alpha < \alpha^* = \min\{1, 1-a\} \le 1$. As in [83, 84], our proof is based on two different blow-up sequences, indeed we introduce the auxiliary sequences

$$w_k(X) = \eta(P_k) \frac{u_k(P_k + r_k X)}{L_k r_k^{\alpha}} \quad \text{and} \quad \overline{w}_k(X) = \frac{(\eta u_k)(P_k + r_k X)}{L_k r_k^{\alpha}}$$

for $X \in B_{P_k,r_k}$ and $P_k = (p_{x,k}, p_{y,k})$ a suitable sequence of points. On one hand, following the same strategy of the blow-up analysis in Chapter 1, \overline{w}_k has an uniform bound on the α -Hölder seminorm, i.e.

$$\sup_{X_1 \neq X_2 \in \overline{B_{P_k,r_k}}} \frac{\left|\overline{w}_k(X_1) - \overline{w}_k(X_2)\right|}{|X_1 - X_2|^{\alpha}} \le \left|\overline{w}_k\left(\frac{X_1 - P_k}{r_k}\right) - \overline{w}_k\left(\frac{X_2 - P_k}{r_k}\right)\right| = 1,$$

while on the other hand

$$-L_a^k w_{i,k} = 0 \quad \text{in } B_{P_k, r_k}, \quad \text{with } L_a^k = \operatorname{div}\left(\left(y + \frac{p_{y,k}}{r_k}\right)^a \nabla\right). \tag{112}$$

The importance of these two sequences lies in the fact that they have asymptotically equivalent behaviour. Namely, since

$$|w_{k}(X) - \overline{w}_{k}(X)| \leq \frac{\|u_{k}\|_{L^{\infty}(B_{1})}}{r_{k}^{\alpha}L_{k}} |\eta(P_{k} + r_{k}X) - \eta(P_{k})|$$

$$\leq \frac{\operatorname{Lip}(\eta)r_{k}^{1-\alpha}}{L_{k}} \|u_{k}\|_{L^{\infty}(B_{1})} |X|$$
(113)

we get, for any compact $K \subset \mathbb{R}^{n+1},$ that

$$\max_{X \in K \cap B_{P_k, r_k}} |w_k(X) - \overline{w}_k(X)| \longrightarrow 0.$$
(114)

Moreover, since $w_k(0) = \overline{w}_k(0)$ we note by (113) that

$$|w_k(X) - w_k(0)| \le |w_k(X) - \overline{w}_k(X)| + |\overline{w}_k(X) - \overline{w}_k(0)|$$
$$\le C \left(\frac{r_k^{1-\alpha}}{L_k} |X| + |X|^{\alpha}\right)$$

and consequently, there exists C = C(K) such that $|w_k(X) - w_k(0)| \le C$, for every $X \in K$. Let us prove that it is not restrictive to choose $P_k \in \Sigma$ in the definition of w_k, \overline{w}_k , since $X_{1,k}, X_{2,k}$ must converge to $B_1 \cap \Sigma$, i.e. there exists C > 0 such that, for k sufficiently large,

$$\frac{\operatorname{dist}(X_{1,k}, B_1 \cap \Sigma) + \operatorname{dist}(X_{2,k}, B_1 \cap \Sigma)}{r_k} \le C.$$

Arguing by contradiction, suppose

$$\frac{\operatorname{dist}(X_{1,k}, B_1 \cap \Sigma) + \operatorname{dist}(X_{2,k}, B_1 \cap \Sigma)}{r_k} \longrightarrow \infty$$

and let us choose $P_k = X_{1,k}$ in the definition of w_k, \overline{w}_k so that $B_{P_k,r_k} \to \mathbb{R}^{n+1}$ and $p_{y,k}^{-1}r_k \to 0^+$. By definition, since \overline{w}_k is a sequence of functions which share the same α -Hölder seminorm and uniformly bounded in every compact $K \subset \mathbb{R}^{n+1}$, by the Ascoli-ArzelÃă theorem, there exists a limit $w \in C(K)$ which, up to a subsequence, is the uniform limit of w_k . By (114), we also find that $w_k \to w$ uniformly con compact sets.

In order to reach a contradiction we can prove that w is a nonconstant globally Hölder harmonic function To this purpose, let $\varphi \in C_c^{\infty}(\mathbb{R}^{n+1})$ be a compactly supported smooth function and \overline{k} be sufficiently large so that supp $\varphi \subset B_{P_k,r_k}$, for all $k \geq \overline{k}$. Fixed $i = 1, \ldots, h$, by testing the first equation in (112) with φ we get

$$\int_{\mathbb{R}^{n+1}} \operatorname{div}\left(\left(1+y\frac{r_k}{p_{y,k}}\right)^a \nabla\varphi\right) w_k \mathrm{d}X = 0.$$

Passing to the uniform limit and observing that

$$\left(1+y\frac{r_k}{p_{y,k}}\right)^a \to 1 \quad \text{in } C^\infty\left(\operatorname{supp} \varphi\right),$$

we deduce that w is actually harmonic and the contradiction follows naturally by the classical Liouville Theorem, once we have shown that w is globally α - Hölder continuous and not constant. since $P_k = X_{1,k}$ then, up to a subsequence,

$$\frac{X_{2,k} - P_k}{r_k} = \frac{X_{2,k} - X_{1,k}}{|X_{2,k} - X_{1,k}|} \to X_2 \in \partial B_1.$$

Finally, by the equicontinuity and the uniform convergence, we conclude

$$\left|\overline{w}_k\left(\frac{X_1 - P_k}{r_k}\right) - \overline{w}_k\left(\frac{X_2 - P_k}{r_k}\right)\right| = 1 \longrightarrow |\overline{w}(0) - \overline{w}(X_2)| = 1.$$

At this point, the choice $P_k = X_{1,k}$ for every $k \in \mathbb{N}$ guarantees the convergence of the domains $B_{P_k,r_k} \to \mathbb{R}^{n+1}$, while for any compact set $K \subset \mathbb{R}^{n+1}$

$$\max_{X \in K \cap B_{P_k, r_k}} |w_k(X) - \overline{w}_k(X)| \longrightarrow 0.$$

Hence, we are left with two possibilities:

- for any compact set $K \subset \Sigma$ we have $w_k(X) \neq 0$ for every $k \geq k_0$ and $X \in K$;
- there exists a sequence $(X_k)_k \subset \Sigma$ such that $w_k(X_k) = 0$ for every $k \in \mathbb{N}$.

In the first case, if we define $W_k = w_k - w_k(0)$ and $\overline{W}_k = \overline{w}_k - \overline{w}_k(0)$ we obtain that the last sequence is uniformly bounded in $C^{0,\alpha}$ and hence $(W_k)_k$ converges uniformly on compact set to a nonconstant globally α -Hölder continuous L_a -harmonic function W and similarly in the second case the sequence $(w_k)_k$ does converge to a nonconstant globally α -Hölder continuous L_a -harmonic function.

In both cases, the contradiction follows from the Liouville theorem for L_a -harmonic functions since $\alpha \in (0, \alpha^*)$, with $\alpha^* = \min\{1, 1 - a\}$. Now, since $\alpha \in (0, \alpha^*)$ with $\alpha^* = \min\{1, 1 - a\}$, the contradiction follows immediately from the Liouville theorem for L_a -harmonic functions. \Box

If instead we consider the general case $X_0 \in \Gamma(u)$ we can prove the following general result

Theorem 3.4.5. Let u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma(u)$ a point on its nodal set. Given the blow-up sequence u_k centered in X_0 and associated to $r_k \downarrow 0^+$ we have this two cases:

1. if $X_0 \in \Sigma$, there exists $p \in H^{1,a}_{loc}(\mathbb{R}^n)$ such that $u_k \to p$ in $C^{0,\alpha}_{loc}(\mathbb{R}^n)$ for every $\alpha \in (0, \alpha^*)$ and strongly in $H^{1,a}_{loc}(\mathbb{R}^n)$. In particular the blow-up limit solves

$$-L_a p = 0$$
 in \mathbb{R}^n .

2. if $X_0 \notin \Sigma$, there exists $p \in H^1_{loc}(\mathbb{R}^n)$ such that $u_k \to p$ in $C^{0,\alpha}_{loc}(\mathbb{R}^n)$ for every $\alpha \in (0,1)$ and strongly in $H^1_{loc}(\mathbb{R}^n)$. In particular the blow-up limit solves

$$-\Delta p = 0$$
 in \mathbb{R}^n .

Now we will mention some counterpart of the previous results for the case $X_0 \in \Gamma(u) \setminus \Sigma$.

Lemma 3.4.6. Let $X_0 \in \Gamma(u) \setminus \Sigma$. For any given R > 0, we have $||u_k||_{H^1(B_R)} \leq C$ where C is independent on k > 0.

Proof. Since for $X_0 \in \Gamma(u) \setminus \Sigma$ we have

$$\rho_k^2 = H(X_0, u, r_k) = \frac{1}{r_k^{n-1}} \int_{\partial B_{r_k}(X_0)} |y|^a \, u^2 \mathrm{d}\sigma = \int_{\partial B_1} |y_0 + r_k y|^a \, u_k^2 \mathrm{d}\sigma,$$

we get by Corollary 3.3.8 that

$$\int_{\partial B_R} |y_0 + r_k y|^a u_k^2 d\sigma = \frac{1}{\rho_k^2} \int_{\partial B_R} |y_0 + r_k y|^a u^2 (X_0 + r_k X) d\sigma$$
$$= \frac{1}{\rho_k^2 r_k^{n-1}} \int_{\partial B_{Rr_k}(X_0)} |y|^a u^2 d\sigma$$
$$= R^{n-1} \frac{H(X_0, u, Rr_k)}{H(X_0, u, r_k)}$$
$$\leq R^{n-1} \left(\frac{Rr_k}{r_k}\right)^{2\tilde{C}}.$$

Consequently, by the Almgren monotonicity formula, for k so large that $r_k, Rr_k < |y_0|$, we have

$$\int_{B_R} |y_0 + r_k y|^a \, |\nabla u_k|^2 \, \mathrm{d}X \le C(R) R^{n-2} N(X_0, u, R).$$

Since $X_0 \in \mathbb{R}^n \setminus \Sigma$ and $r_k \downarrow 0^+$ we have

$$\inf_{B_R} |y_0 + r_k y|^a = \inf_{\partial B_R} |y_0 + r_k y|^a \ge |y_0|^a \min\{|1 + r_0|^a, |1 - r_0|^a\}$$

and finally

$$\frac{1}{R^{n-1}} \int_{\partial B_R} u_k^2 d\sigma + \frac{1}{R^{n-2}} \int_{B_R} |\nabla u_k|^2 \, dX \le \frac{C'(R)}{|y_0|^a}$$

which gives the claimed bound.

For completeness we just remarked the uniform bound in H^1 for the blow-up sequence centered outside Σ , but the convergence result for the blow-up sequence centered in $X_0 \notin \Sigma$ is a direct consequence of [81, Theorem 3.3]. Indeed, for every k > 0 we have that $\operatorname{div}(|y_0 + r_k y|^a \nabla u_k) = 0$ or equivalently

$$-\Delta u_k = a \frac{r_k}{|y_0 + r_k y|} \partial_y u_k,$$

which implies the existence of a nontrivial function $p \in H^1_{\text{loc}}(\mathbb{R}^{n+1}) \cap L^{\infty}_{\text{loc}}(\mathbb{R}^{n+1})$ such that, up to a subsequence, we have $u_k \rightharpoonup p$ weakly in $H^1_{\text{loc}}(\mathbb{R}^{n+1})$ and $-\Delta p = 0$ in $\mathcal{D}'(\mathbb{R}^{n+1})$.

Proposition 3.4.7. Let $X_0 \in \Gamma(u) \setminus \Sigma$ and p be the blow-up limit of u centered in X_0 , as previously defined. Then the following Almgren monotonicty formula

$$N(Z_0, p, r) = \frac{\frac{1}{r^{n-2}} \int_{B_r(Z_0)} |\nabla p|^2 \, \mathrm{d}X}{\frac{1}{r^{n-1}} \int_{\partial B_r(Z_0)} p^2 \, \mathrm{d}\sigma}$$

is well defined for $Z_0 \in \mathbb{R}^{n+1}$ and r > 0. In particular the map $r \mapsto N(Z_0, p, r)$ is monotone nondecreasing for every $Z_0 \in \mathbb{R}^{n+1}$ and $r \in (0, +\infty)$.

Moreover, $N(0, p, r) = N(X_0, u, 0^+) =: k$ for every r > 0, namely p is k-homogeneous

$$p(X) = |X|^{\gamma} p\left(\frac{X}{|X|}\right)$$
 for every $X \in \mathbb{R}^{n+1}$,

with $k \in \mathbb{N}, k \geq 1$.

Now we focus our attention on the blow-up limit itself in the challenging case $X_0 \in \Gamma(u) \cap \Sigma$ and the relationship between the value of the Almgren frequency formula and its local behaviour. More precisely, we have **Proposition 3.4.8.** Let $X_0 \in \Gamma(u) \cap \Sigma$ and p be a blow-up limit of u centered in X_0 , as previously defined. Then $N(0, p, r) = N(X_0, u, 0^+) =: k$ for every r > 0 and p is k-homogeneous, i.e.

$$p(X) = |X|^k p\left(\frac{X}{|X|}\right)$$
 for every $X \in \mathbb{R}^{n+1}$

Proof. First of all we prove that $r \mapsto N(0, p, r)$ is constant. Let us observe that $N(0, u_k, r) = N(X_0, u, rr_k)$ and that Theorem 3.4.1 yields that $N(0, p, r) = \lim_k N(0, u_k, r)$. Similarly, for the right hand side we get $\lim_k N(X_0, u, rr_k) = N(X_0, u, 0^+)$ by Proposition 3.3.1.

We now compute the derivative of $r \mapsto N(0, p, r)$, in order to prove that p is k-homogeneous, where obviously $k = N(X_0, u, 0^+)$ is the Almgren frequency formula. As in the proof of Proposition 3.3.1, we know that

$$\frac{d}{dr}H(0,p,r) = \frac{2}{r^{n+a-1}} \int_{\partial B_r} |y|^a p \partial_r p \mathrm{d}\sigma$$

and by integration by parts that

$$\frac{d}{dr}E(0,p,r) = \frac{1}{r^{n+a-1}} \int_{\partial B_r} |y|^a \left(\partial_r p\right)^2 \mathrm{d}\sigma.$$

Hence, this two equalities imply

$$0 = \frac{d}{dr}N(0,p,r) = \frac{2}{r^{2n+2a-2}}\frac{1}{H^2(0,p,r)} \left[\int_{\partial B_r} |y|^a p^2 \mathrm{d}\sigma \int_{\partial B_r} |y|^a |\partial_r p|^2 \,\mathrm{d}\sigma - \left(\int_{\partial B_r} |y|^a p \partial_r p \mathrm{d}\sigma \right)^2 \right]$$

for r > 0. This equality yields the existence of C = C(r) > 0 such that $\partial_r p = C(r)p$ for every r > 0. Using this fact in (103) we get

$$2C(r) = \frac{\int_{\partial B_r} |y|^a p \partial_r p \mathrm{d}\sigma}{\int_{\partial B_r} |y|^a p^2 \mathrm{d}\sigma} = \frac{d}{dr} \log H(0, p, r) = \frac{2}{r} N(0, p, r) = \frac{2}{r} k$$

and thus C(r) = k/r and p is k-homogenous as we claimed.

In the final part of this Section we classify the possible values of the Almgren frequency formula on the restriction $\Gamma(u) \cap \Sigma$ and consequently the possible blow-up limits, in order to better understand the structure and the stratification of the nodal set of u.

A crucial Corollary of this analysis is that a blow-up limit of u in a point of the nodal set on Σ is either symmetric or antisymmetric with respect Σ : this attitude is due to the fact that
near the characteristic manifold the local behaviour of L_a -harmonic function symmetric with respect to Σ is different to the one of antisymmetric L_a -harmonic function, which it is a feature of the degenerate-singular case for $a \in (-1, 1)$ and $a \neq 0$.

At this point, we already know that given an L_a -harmonic function u on B_1 , for every $X_0 \in \Gamma(u) \cap \Sigma$ and $r_k \downarrow 0^+$ we have, up to a subsequence, that

$$u_k(X) = \frac{u(X_0 + r_k X)}{\sqrt{H(X_0, u, r_k)}} \to p(X),$$

where $p \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ is an nonconstant entire L_a -harmonic function homogenous of order $k \in \mathbb{R}$ with $\|p\|_{L^{2,a}(\partial B_1)} = 1$. In particular, by Proposition 3.4.8 we already know that $k = N(X_0, u, 0^+)$.

Inspired by Proposition 3.2.4, let us consider separately the case when u is symmetric with respect to Σ and the antisymmetric one.

Lemma 3.4.9. Let $a \in (-1,1)$ and u be an L_a -harmonic function symmetric with respect to Σ . Then, for every $X_0 \in \Gamma(u) \cap \Sigma$, we have

$$N(X_0, u, 0^+) \in 1 + \mathbb{N}.$$

Proof. Let $X_0 \in \Gamma(u) \cap \Sigma$ and $k = N(X_0, u, 0^+)$ be the Almgren frequency formula in X_0 . For every $r_k \to 0^+$ we already know that, up to a subsequence, by Theorem 3.4.1 and Proposition 3.4.8 that

$$u_k(X) = \frac{u(X_0 + r_k X)}{\sqrt{H(X_0, u, r_k)}} \to p(X),$$

where p is an L_a -harmonic k-homogenous function symmetric with respect to Σ . Since, by Corollary 3.3.6 we already know that $k \ge 1$, let us suppose by contradiction that there exists an homogenous L_a -harmonic function of order k > 1 such that $k \notin \mathbb{N}$. Since for every $i = 1, \ldots, n$, we have

$$L_a(\partial_{x_i}p) = \partial_{x_i}L_ap = 0,$$

fixed $k = \lfloor k \rfloor$, by Euler's homogeneous function Theorem, we already know that any k-order partial derivative of p with respect to the variables x_1, \ldots, x_n must be an homogenous L_a -harmonic of order $\alpha = k - \lfloor k \rfloor \in (0, 1)$. The contradiction follows from Proposition 3.3.6, since the homogeneity of an homogenous function is equal to the Almgren frequency formula evaluated in the origin, hence in the symmetric case it must be greater or equal to 1.

Lemma 3.4.10. Let $a \in (-1, 1)$ and u be an L_a -harmonic function antisymmetric with respect to Σ . Then, for every $X_0 \in \Gamma(u) \cap \Sigma$, we have

$$N(X_0, u, 0^+) \in 1 - a + \mathbb{N}.$$

Proof. As in the previous Lemma, let $X_0 \in \Gamma(u) \cap \Sigma$ and $k = N(X_0, u, 0^+)$ be the Almgren frequency formula in X_0 . For every $r_k \to 0^+$ we already know that, up to a subsequence, we have by Theorem 3.4.1 and Proposition 3.4.8 that $u_k \to p$ where p is an L_a -harmonic k-homogenous function antisymmetric with respect to Σ .

By Proposition 3.2.4, there exists $q \in H^{1,2-a}_{loc}(\mathbb{R}^{n+1})$ and L_{2-a} -harmonic function symmetric with respect to Σ , such that $p = qy |y|^{-a}$. Since p is k-homogenous, we already know that q must be (k-1+a)-homogenous, i.e.

$$q(X) = p(X)y^{-1}|y|^a = |X|^{k-1+a} p\left(\frac{X}{|X|}\right) \frac{y^{-1}|y|^a}{|X|^{-1+a}} = |X|^{k-1+a} q\left(\frac{X}{|X|}\right)$$

for every $X \in \mathbb{R}^{n+1}$.

Obviously if $q(0) \neq 0$, then k = 1 - a and q is zero-homogenous, i.e. $q \equiv q(0)$ on \mathbb{R}^{n+1} , instead, if q(0) = 0, by Lemma 3.4.9 we know that $N(0, q, 0^+) \in 1 + \mathbb{N}$ and consequently $k \in 2 - a + \mathbb{N}$. Similarly, since these two cases correspond to $N(0, q, 0^+) = 0$ and $N(0, q, 0^+) \in 1 + \mathbb{N}$, the final result on k can be formulated as $N(X_0, u, 0^+) \in 1 - a + \mathbb{N}$.

Proposition 3.4.11. Let $a \in (-1, 1)$ and u be an L_a -harmonic function. Given $X_0 \in \Gamma(u) \cap \Sigma$ and a blow-up sequence u_k centered in X_0 and associated to some $r_k \downarrow 0^+$. Then the blow-up limit $p \in H^{1,a}_{loc}(\mathbb{R}^{n+1})$ is either symmetric or antisymmetric with respect to Σ and

$$N(X_0, u, 0^+) \in \begin{cases} 1 + \mathbb{N}, & \text{if } p \text{ is symmetric,} \\ 1 - a + \mathbb{N}, & \text{if } p \text{ is antisymmetric.} \end{cases}$$

Proof. The proof is a direct consequence of the previous Lemmas. Indeed, let $X_0 \in \Gamma(u) \cap \Sigma$ and $k = N(X_0, u, 0^+)$ be the Almgren frequency formula in X_0 . For every $r_k \to 0^+$ we already know that, up to a subsequence, we have by Theorem 3.4.1 and Proposition 3.4.8 that

$$u_k(X) = \frac{u(X_0 + r_k X)}{\sqrt{H(X_0, u, r_k)}} \to p(X),$$

where p is an L_a -harmonic k-homogenous function. By Proposition 3.2.4 and (98), there exist a unique L_a -harmonic function $p_e \in H^{1,a}_{loc}(\mathbb{R}^{n+1})$ and an L_{2-a} -harmonic function $q_e \in H^{1,2-a}_{loc}(\mathbb{R}^{n+1})$ both symmetric with respect to Σ , such that

$$p(x,y) = p_e(x,y) + q_e(x,y)y |y|^{-a}$$
, for every $(x,y) \in \mathbb{R}^{n+1}$

Since p is k-homogeneous, we already know by the Definition 3.2.3 that also p_e and $q_e y |y|^{-a}$ are k-homogeneous, i.e.

$$N(0, p_e, 0^+) = k = N(0, q_e, 0^+) + 1 - a.$$
(115)

If p is purely symmetric or antisymmetric with respect to Σ , the result follows respectively by Lemma 3.4.9 and Lemma 3.4.10. Instead, suppose by contradiction that $p_e \neq 0$ and $q_e \neq 0$, then by (115) the two homogeneity of p_e and q_e can not be simultaneously in $\mathbb{N} + 1$, in contradiction with the previous Lemmas.

In order to understand the local behaviour of the L_a -harmonic function, we need to construct explicitly the homogenous L_a -harmonic function. As before, we start by classifying the homogeneous solution symmetric with respect to Σ in order to classify all the possible ones.

Lemma 3.4.12. Let $p \in H^{1,a}_{loc}(\mathbb{R}^{n+1})$ be a nonconstant homogeneous L_a -harmonic function, symmetric with respect to Σ . Then p does not depend on the variable y if and only if it is harmonic in the variable x_1, \ldots, x_n .

The proof is trivial and the main consequence is that for every $k \in 1 + \mathbb{N}$ an homogenous harmonic function in the variable x_1, \ldots, x_n of order k is an admissible blow-up limit. For this reason, let us concentrate our attention on the case of blow-up limits that depend on the variable y.

Lemma 3.4.13. [22, Lemma 2.7] Let $p \in H^{1,a}_{loc}(\mathbb{R}^{n+1})$ be an entire L_a -harmonic function symmetric with respect to Σ , such that

$$|p(X)| \le C\left(1 + |X|^k\right) \quad \text{in } \mathbb{R}^{n+1},$$

for some $k \in \mathbb{N}$. Then p is a polynomial.

In order to give an explicit formulation of the blow-up limits, at least for n + 1 = 2, we remark that if p is a k-homogenous L_a -harmonic function, then for every i = 1, ..., n the functions $\partial_{x_i} u$ are (k - 1)-homogeneous L_a -harmonic function and $\partial_{yy}^2 u + ay^{-1} \partial_y u$ is a (k - 2)-homogenous L_a -harmonic function. More precisely,

Lemma 3.4.14. Let $p \in H^{1,a}_{loc}(\mathbb{R}^{n+1})$ be an L_a -harmonic homogenous polynomial of degree $k \ge 2$ symmetric with respect to Σ . Then p is L_a -harmonic if and only if $\partial_{x_i}p$ and $\partial_{yy}^2p + ay^{-1}\partial_yp$ are L_a -harmonic, for every i = 1, ..., n.

Proof. Since the derivatives commute, we get

$$|y|^{-a} L_a \partial_{x_i} p = \partial_{x_i} \left(|y|^{-a} L_a p \right), \ \forall i = 1, \dots, n$$

$$L_a \left(\partial_{yy}^2 p + a y^{-1} \partial_y p \right) = \partial_{yy}^2 L_a p + a y^{-1} \partial_y L_a p.$$
 (116)

The first implication is obvious since replacing $L_a p = 0$, we get the conditions on the derivatives. Now let us suppose that (116) holds true, since p is an homogenous polynomial of degree k, the function $(x, y) \mapsto |y|^{-a} (L_a p)(x, y)$ is an homogenous polynomial of degree k - 2 symmetric with respect to Σ . Moreover, since $\partial_{x_i} p$ is L_a -harmonic, for every $i = 1, \ldots, n$, from the first conditions in (116) we have $L_a p = q(y)$, with q(y) = q(-y) and

$$\partial_{yy}^2 q + ay^{-1} \partial_y q = 0, \tag{117}$$

from the last condition on the derivatives with respect to y.

Since the general solution of (117) is $q(y) = c_1 + c_2 |y|^{-a}$, we immediately get, from the constrained of symmetry, that $q(y) \equiv 0$ and hence that $L_a p = 0$ on \mathbb{R}^{n+1} .

The following Proposition gives a complete picture of the possible entire configurations in \mathbb{R}^2 . This profiles will be useful in the stratification result of Section 3.7.

Proposition 3.4.15. Let $p \in H^{1,a}_{loc}(\mathbb{R}^2)$ be a nonconstant entire L_a -harmonic function symmetric with respect to Σ such that N(0, p, r) = k for every r > 0. Suppose that p depends on the variable y, then if $k \in 2\mathbb{N}$ we have

$$p(x,y) = \frac{(-1)^{\frac{k}{2}}\Gamma\left(\frac{1}{2} + \frac{a}{2}\right)}{2^{k}\Gamma\left(1 + \frac{k}{2}\right)\Gamma\left(\frac{1}{2} + \frac{a}{2} + \frac{k}{2}\right)} {}_{2}F_{1}\left(-\frac{k}{2}, -\frac{k}{2} - \frac{a}{2} + \frac{1}{2}, \frac{1}{2}, -\frac{x^{2}}{y^{2}}\right)y^{k},$$
(118)

and if $k \in 2\mathbb{N} + 1$ we get

$$p(x,y) = -\frac{(-1)^{\frac{k}{2} + \frac{1}{2}}\Gamma\left(\frac{1}{2} + \frac{a}{2}\right)}{2^{k-1}\Gamma\left(\frac{1}{2} + \frac{k}{2}\right)\Gamma\left(\frac{a}{2} + \frac{k}{2}\right)} \, {}_{2}F_{1}\left(\frac{1}{2} - \frac{k}{2}, 1 - \frac{k}{2} - \frac{a}{2}, \frac{3}{2}, -\frac{x^{2}}{y^{2}}\right) xy^{k-1}, \quad (119)$$

where $_2F_1$ is the hypergeometric function.

Proof. The proof is by induction and based on the properties related to the derivatives of homogenous L_a -harmonic functions. By Lemma 3.4.13, we already know that every homogenous L_a -harmonic function symmetric with respect to Σ is a polynomial p(x, y) such that, for every $x \in \Sigma$ the map $y \mapsto p(x, y)$ is a polynomial of even degree. Fix k = 2m with $m \in \mathbb{N}$, consider

$$c(m,a,t) = \frac{(-1)^{m-t}}{2t!} \frac{1}{2^{m-t}(m-t)!} \prod_{i=1}^{m-t} \frac{1}{2i+a-1} = \frac{(-1)^{m-t}\Gamma\left(\frac{1}{2} + \frac{a}{2}\right)}{2t!(m-t)!2^{2m-2t}\Gamma\left(m-t+\frac{1}{2} + \frac{a}{2}\right)}$$
(120)

and consequently

$$p(x,y) = \frac{x^{2m}}{2m!} + \sum_{t=0}^{m-1} c(m,a,t) x^{2t} y^{2m-2t}.$$

which is equivalent to (118). By a direct computation, it is easy to see that $L_a p(x, y) = 0$ for every $(x, y) \in \mathbb{R}^2$. Now, let us prove by induction on $k \ge 2$ that every homogenous L_a -harmonic function is of the form (118). Since the case k = 0 is trivial, let us take k = 2. Since p must be of degree 2 and even in the variable y, the polynomial must be like $p(x, y) = a_1 x^2 + a_2 y^2$ and consequently

$$L_a p = 0 \quad \longleftrightarrow \quad a_2 = -\frac{1}{1+a}a_1,$$

and for $a_1 = 1/2$ we obtain the formula in (120).

Suppose (120) are true for $k \in 2\mathbb{N}$, and consider a L_a -harmonic polynomial p of degree k + 2, i.e.

$$p(x,y) = a_{m+1}x^{2m+2} + \sum_{t=0}^{m} a_t x^{2t} y^{2m-2t}.$$

Since $\partial_x^2 p$ is a L_a -harmonic polynomial of degree k, we must have by the inductive hypothesis

$$a_{m+1}(2m+2)(2m+1) = \frac{1}{2m!}, \quad 2t(2t-1)a_t = c(m,a,t-1) \text{ for } t = 1, \dots, m$$

which imply, by definition (120), that

$$a_{m+1} = \frac{1}{(2m+2)!}, \quad a_t = \frac{c(m,a,t-1)}{2t(2t-1)} = c(m+1,a,t)$$

for t = 1, ..., m. Finally, let $w = -\partial_{yy}^2 p - ay^{-1}\partial_y p$ be a polynomial of degree k. By Lemma 3.4.14 w is L_a -harmonic and, by the inductive hypothesi, we get by linearity that

$$-\partial_{yy}^2(a_0y^{2m+2}) - ay^{-1}\partial_y(a_0y^{2m+2}) = c(m,a,0)y^{2m},$$

or in other words that $-(2m+2)(2m+1+a)a_0 = c(m, a, 0)$, which implies that

$$a_0 = \frac{c(m, a, 0)}{2(m+1)(2m+1+a)} = c(m+1, a, 0).$$

We have already proved the formula for the case $k \in 2\mathbb{N}$, while the other one is obtained via an integration respect to the variable x.

Before to consider the general case $n \ge 3$, we complete the Section with some concrete examples of blow-up profiles in 2-dimensional case. This example, and more generally the class of homogeneous function described by the previous Proposition, will summarize all the possible behaviour of the (n-2)-dimensional singular set, as we will see in Section 3.7. For $n \ge 3$, we can not give an explicit formula for the blow- up limits which depend on the variable y, but we can prove that every polynomial in \mathbb{R}^n admits a unique L_a -harmonic extension symmetric with respect to Σ . Since we want to classify the possible blow-up limit of s-harmonic functions on the nodal set, this result suggests that s-harmonic functions can vanish like any polynomial. We will discuss in the following Sections the implication of this classification.

Lemma 3.4.16. [51, Lemma 5.2] Let p(x) be an homogeneous polynomial of degree d in \mathbb{R}^n . Then, there exists a unique polynomial q(X) = q(x, y) of degree d in \mathbb{R}^{n+1} such that

$$\begin{cases} L_a q = 0 & \text{ in } \mathbb{R}^{n+1} \\ q(x,y) = q(x,-y) & \text{ in } \mathbb{R}^{n+1} \\ q(x,0) = p(x) & \text{ on } \mathbb{R}^n. \end{cases}$$

In particular, it can proved that this extension is obtained by

$$q(x,y) = \sum_{k\geq 0}^{d/2} (-1)^k c_{2k} \Delta^k \frac{x^{\alpha}}{\alpha!} \frac{y^{2k}}{(2k!)}, \quad c_{2k} = \prod_{i=1}^k \frac{2i-1}{2i-2s},$$

where $\alpha = (\alpha_1, \ldots, \alpha_n) \in \mathbb{N}^d$, $x^{\alpha} = x_1^{\alpha_1} \cdots x_n^{\alpha_n}$ and $\alpha! = \alpha_1! \cdots \alpha_n!$.

Inspired by the previous results, let us introduce the following classes of blow-up limit.

Definition 3.4.17. Given $a \in (-1, 1)$ and $k \in \mathbb{R}$, we define the set of all possible blow-up limit of order k, i.e. the set of all L_a -harmonic symmetric polynomials of degree k, as

$$\mathfrak{B}_k^a(\mathbb{R}^{n+1}) = \left\{ p \in H^{1,a}_{\mathrm{loc}}(\mathbb{R}^{n+1}) \middle| \begin{array}{c} L_a p = 0 \text{ in } \mathbb{R}^{n+1} \\ p(X) = |X|^k \, p\left(\frac{X}{|X|}\right) \text{ in } \mathbb{R}^{n+1} \right\}.$$

Similarly, the set of blow-up limit of order k respectively symmetric or antisymmetric with respect to Σ are defined as

$$\mathfrak{sB}_k^a(\mathbb{R}^{n+1}) = \left\{ p \in \mathfrak{B}_k^a(\mathbb{R}^{n+1}) \, | \, p \text{ symmetric with respect to } \Sigma \right\},$$

$$\mathfrak{aB}_k^a(\mathbb{R}^{n+1}) = \left\{ p \in \mathfrak{B}_k^a(\mathbb{R}^{n+1}) \, | \, p \text{ antisymmetric with respect to } \Sigma \right\}.$$

By Proposition 3.2.4 and Lemma 3.4.12 we can classify even more the structure of the previous classes emphasizing two subclasses of blow-up limit.

Definition 3.4.18. Given $a \in (-1, 1)$ and $k \in \mathbb{R}$, let us define $\mathfrak{sB}_k^*(\mathbb{R}^{n+1}) = \mathfrak{B}_k^0(\mathbb{R}_x^n)$ the set of functions $p \in \mathfrak{B}_k^a(\mathbb{R}^{n+1})$ such that $\Delta_x p = 0$, namely p(x, y) = p(x) in $\mathbb{R}^{n+1} = \mathbb{R}_x^n \times \mathbb{R}_y$.

By the previous Section, we already know that for $a \in (-1, 1)$ we have $\mathfrak{B}_1^a(\mathbb{R}^{n+1}) = \mathfrak{B}_1^*(\mathbb{R}^{n+1})$ and for $k \ge 2$ we have $\mathfrak{sB}_k^a(\mathbb{R}^{n+1}) \setminus \mathfrak{B}_k^*(\mathbb{R}^{n+1}) \ne \emptyset$ and it consists of all blow-up limit which depends on the variable y. Finally

Corollary 3.4.19. For $a \in (-1, 1)$, let u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u)$, for some $k \in 1 + \mathbb{N}$ or $k \in 1 - a + \mathbb{N}$. Then, every blow-up limit p centered in $X_0 \in \Gamma_k(u)$ is either in $\mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ or in $\mathfrak{aB}_k^a(\mathbb{R}^{n+1})$. Moreover, for every $a \in (-1, 1)$ we have

$$\mathfrak{aB}_k^a(\mathbb{R}^{n+1}) = \mathfrak{sB}_{k+a-1}^{2-a}(\mathbb{R}^{n+1})y |y|^{-a}$$

3.5 UNIQUENESS AND CONTINUITY OF TANGENT MAPS AND TANGENT FIELDS

In this Section we start introducing a Weiss type monotonicity formula, which is a fundamental tool well suited for the blow-up analysis at the nodal points $X_0 \in \Gamma(u)$ where $N(X_0, u, 0^+) = k$. Starting from this result we will improve our knowledge of the blow-up convergence by proving the existence of a unique no-zero blow-up limit at every point of the nodal set $\Gamma(u)$, which will be called the tangent map φ^{X_0} of u at X_0 .

In particular, inspired by the decomposition in (98), we introduce the notion of tangent "*field*" at the nodal point Φ^{X_0} , which take the main role in our blow-up analysis.

Definition 3.5.1. Given u an L_a -harmonic function in B_1 , for $k \ge \min\{1, 1-a\}$, we define

$$\Gamma_k(u) \coloneqq \{X_0 \in \Gamma(u) \colon N(X_0, u, 0^+) = k\}.$$

One has to point out that the sets $\Gamma_k(u)$ may be nonempty only for k in a certain set of values. Indeed, by Proposition 3.4.11, we already know that $\Gamma_k(u) \cap \Sigma$ is nonempty if and only if $k \in 1 + \mathbb{N}$ or $k \in 1 - a + \mathbb{N}$. We remark that all the following results are well known for the case $X_0 \in \Gamma_k(u) \setminus \Sigma$ since the L_a operator is uniformly elliptic outside Σ .

Proposition 3.5.2. Let u be a nontrivial L_a -harmonic function in B_1 . For $X_0 \in \Gamma_k(u) \cap \Sigma$, we introduce the k-Weiss type formula

$$W_k(X_0, u, r) = \frac{1}{r^{n+a-1+2k}} \int_{B_r(X_0)} |y|^a |\nabla u|^2 \, \mathrm{d}X - \frac{k}{r^{n+a+2k}} \int_{\partial B_r(X_0)} |y|^a |u|^2 \, \mathrm{d}\sigma.$$

For $r \in (0, 1 - |X_0|)$ *we have*

$$\frac{d}{dr}W_k(X_0, u, r) = \frac{2}{r^{n+a+1+2k}} \int_{\partial B_r(X_0)} |y|^a \left(\langle \nabla u, X - X_0 \rangle - ku\right)^2 \mathrm{d}\sigma.$$
(121)

which implies that $r \mapsto W_k(X_0, u, r)$ is monotone nondecreasing in $(0, 1 - |X_0|)$. Furthermore, the map $r \mapsto W_k(X_0, u, r)$ is constant if and only if u is homogeneous of degree k.

Proof. By the definition of the Almgren monotonicity formula, we have

$$W_k(X_0, u, r) = \frac{H(X_0, u, r)}{r^{2k}} \left(N(X_0, u, r) - k \right)$$
(122)

which directly implies that

$$\begin{aligned} \frac{d}{dr}W_k(X_0, u, r) &= \frac{-2k}{r^{2k+1}} \left(E(X_0, u, r) - kH(X_0, u, r) \right) + \frac{1}{r^{2k}} \left(\frac{d}{dr} E(X_0, u, r) - k\frac{d}{dr} H(X_0, u, r) \right) \\ &= \frac{-4k}{r^{2k+1}} E(X_0, u, r) + \frac{2k^2}{r^{2k+1}} H(X_0, u, r) + \frac{1}{r^{2k}} \frac{d}{dr} E(X_0, u, r) \\ &= \frac{2}{r^{n+a+1+2k}} \int_{\partial B_r(X_0)} |y|^a \left(\langle \nabla u, X - X_0 \rangle - ku \right)^2 \mathrm{d}\sigma \end{aligned}$$

as we previously claimed.

By a integration by parts, we can rewrite the k-Weiss monotonicity formula as

$$W_k(X_0, u, r) = \frac{1}{r^{n+a+2k}} \int_{\partial B_r(X_0)} |y|^a u \left(\langle \nabla u, X - X_0 \rangle - u \right) \mathrm{d}\sigma.$$

Proposition 3.5.3. Let $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u) \cap \Sigma$. For every homogenous L_a -harmonic polynomial $p \in \mathfrak{B}_k^a(\mathbb{R}^{n+1})$, the map

$$r \mapsto \frac{H(X_0, u - p_{X_0}, r)}{r^{2k}} = \frac{1}{r^{n+a+2k}} \int_{\partial B_r(X_0)} |y|^a \left(u - p_{X_0}\right)^2 \mathrm{d}\sigma$$

is monotone non decreasing in $(0, 1 - |X_0|)$, where $p_{X_0}(X) = p(X - X_0)$.

Through the following Section, we will use the notation $r \mapsto M(X_0, u, p_{X_0}, r)$ for the previous map.

Proof. Since $X_0 \in \Gamma_k(u) \cap \Sigma$ and p is a k-homogenous L_a -harmonic function, we already know that $W_k(X_0, u, r) \ge 0$ and $W_k(X_0, p_{X_0}, r) = 0$ for every $r \in (0, 1 - |X_0|)$. Let $w = u - p_{X_0}$, then

$$\begin{split} W_k(X_0, u, r) &= W_k(X_0, u, r) + W_k(X_0, p_{X_0}, r) \\ &= \frac{1}{r^{n+a-1+2k}} \left(\int_{B_r(X_0)} |y|^a \, |\nabla w|^2 + 2 \, |y|^a \, \langle \nabla w, \nabla p \rangle \mathrm{d}X - \frac{k}{r} \int_{\partial B_r(X_0)} |y|^a \, w^2 + 2 \, |y|^a \, w \mathrm{pd}\sigma \right) \\ &= W_k(X_0, w, r) + \frac{2}{r^{n+a+2k}} \int_{\partial B_r(X_0)} |y|^a \, w(\langle \nabla p_{X_0}, X - X_0 \rangle - kp) \mathrm{d}\sigma \\ &= W_k(X_0, u - p_{X_0}, r). \end{split}$$

Hence , by (122), we finally get

$$\frac{d}{dr}\frac{H(X_0, u - p_{X_0}, r)}{r^{2k}} = 2\frac{H(X_0, u - p_{X_0}, r)}{r^{2k+1}} \left(N(X_0, u - p_{X_0}, r) - k\right)$$
$$= \frac{2}{r}W_k(X_0, u - p_{X_0}, r) \ge 0.$$

Now, we apply the previous monotonicity formulas to study the growth rate of the L_a -harmonic function at the points of the nodal set. In particular, we prove a nondegeneracy and uniqueness result of the blow-up limit, for every points of the nodal set.

Lemma 3.5.4. Let $a \in (-1,1)$ and u be an L_a -harmonic function in B_1 . Then, for every $X_0 \in \Gamma_k(u) \cap \Sigma$, there exists C > 0 such that

$$|u(X)| \le C |X - X_0|^k$$
 in $B_{R/2}(X_0)$.

where $R = 1 - dist(X_0, \partial B_1)$.

Proof. Since whenever $X_0 \in \Gamma_k(u)$ we have $N(X_0, u, r) \ge N(X_0, u, 0^+) = k$, then for every $r \in (0, R)$

$$\frac{d}{dr}\log H(X_0, u, r) \ge \frac{2}{r}N(X_0, u, r) \ge \frac{2k}{r}$$

and similarly

$$\log \frac{H(X_0, u, R)}{H(X_0, u, r)} \ge 2k \log \frac{1}{r},$$

which implies $H(X_0, u, r) \leq H(X_0, u, R)r^{2k}$. Now, by [88, Lemma A.2.] and the previous estimate, we get for every $r \in (0, R)$

$$\begin{split} \sup_{\overline{B_{r/2}}} u &\leq C(n,a) \left(\frac{1}{r^{n+1+a}} \int_{B_r} |y|^a \, u^2 \mathrm{d}X \right)^{1/2} \\ &\leq C(n,a) \left(\frac{H(0,u,R)}{n+a+1} \right)^{1/2}, \end{split}$$

where in the second inequality we used the monotonicity of $r \mapsto H(0, u_k, r)$ in (0, R).

Lemma 3.5.5 (Nondegeneracy). Let $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 . Then, for every $X_0 \in \Gamma_k(u) \cap \Sigma$ there exists C > 0 such that

$$\sup_{\partial B_r(X_0)} |u(X)| \ge Cr^k \quad \text{for } 0 < r < R$$

where $R = 1 - dist(X_0, \partial B_1)$.

Proof. Fix $X_0 \in \Gamma_k(u)$ and suppose by contradiction, given a decreasing sequence $r_j \downarrow 0$, that

$$\lim_{j \to \infty} \frac{H(X_0, u, r_j)^{1/2}}{r_j^k} = \lim_{j \to \infty} \left(\frac{1}{r_j^{n+a+2k}} \int_{\partial B_{r_j}(X_0)} |y|^a \, u^2 \, d\sigma \right)^{1/2} = 0.$$

Consider now the blow-up sequence

$$u_j(X) = \frac{u(X_0 + r_j X)}{\rho_j}$$
 where $\rho_j = H(X_0, u, r_j)^{1/2}$

constructed starting from r_j and centered in $X_0 \in \Gamma_k(u)$. By Theorem 3.4.1, up to a subsequence $u_j \to p$ uniformly, where p is a nontrivial L_a -harmonic homogenous polynomial of degree k such that H(0, p, 1) = 1.

Let us focus our attention on the functional $M(X_0, u, p_{X_0}, r)$ with p_{X_0} as above. By the assumption on the growth of u it follows

$$M(X_0, u, p_{X_0}, 0^+) = \lim_{r \to 0} \frac{1}{r^{n+a+2k}} \int_{\partial B_r(X_0)} |y|^a (u - p_{X_0})^2 d\sigma$$

= $\lim_{r \to 0} \int_{\partial B_1} |y|^a \left(\frac{u(X_0 + rX)}{r^k} - p(X)\right)^2 d\sigma$
= $\int_{\partial B_1} |y|^a p^2 d\sigma$
= $\frac{1}{r^{n+a+2k}} \int_{\partial B_r(X_0)} |y|^a p_{X_0}^2 d\sigma.$

By the monotonicity result of Proposition 3.5.3 on the map $r \mapsto M(X_0, u, p_{X_0}, r)$, we obtain

$$\frac{1}{r^{n+a-1+2k}} \int_{\partial B_r(X_0)} |y|^a \left(u - p_{X_0}\right)^2 d\sigma \ge \frac{1}{r^{n+a-1+2k}} \int_{\partial B_r(X_0)} |y|^a p_{X_0}^2 d\sigma$$

and similarly

$$\int_{\partial B_r(X_0)} |y|^a \left(u^2 - 2up_{X_0} \right) d\sigma \ge 0.$$

On the other hand, rescaling the previous inequality and using the blow-up sequence u_j defined as above, we get

$$\int_{\partial B_1} |y|^a \left(H(X_0, u, r_j) u_j^2 - 2H(X_0, u, r_j)^{1/2} r_j^k u_j p \right) \, d\sigma \ge 0$$

and

$$\int_{\partial B_1} |y|^a \left(\frac{H(X_0, u, r_j)^{1/2}}{r_j^k} u_j^2 - 2u_j p \right) \, d\sigma \ge 0.$$

The absurd follows passing to the limit for $j \to \infty$, indeed by the previous inequality we get

$$\int_{\partial B_1} |y|^a \, p^2 \, d\sigma \le 0$$

in contradiction with $p \not\equiv 0$.

Theorem 3.5.6 (Uniqueness of the blow-up limit). Given $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 , let us consider $X_0 \in \Gamma_k(u) \cap \Sigma$, i.e. $N(X_0, u, 0^+) = k$. Then there exists a unique nonzero $p \in \mathfrak{B}_k^a(\mathbb{R}^{n+1})$ blow-up limit such that

$$u_{X_0,r}(X) = \frac{u(X_0 + rX)}{r^k} \longrightarrow p(X).$$
(123)

Proof. Up to a subsequence $r_j \to 0^+$, we have that $u_{X_0,r_j} \to p$ in $\mathcal{C}_{\text{loc}}^{0,\alpha}$. The existence of such limit follows directly from the previous growth estimate $|u(X)| \leq C |X|^k$ and by Lemma 3.5.5 we have p is not identically zero. Now, for any r > 0 we have

$$W_k(0, p, r) = \lim_{j \to \infty} W_k(0, u_{X_0, r_j}, r) = \lim_{j \to \infty} W_k(X_0, u, rr_j) = W_k(X_0, u, 0^+) = 0.$$

In particular, Proposition 3.5.2 implies that the L_a -harmonic function p is k-homogeneous and consequently $p \in \mathfrak{B}_k^a(\mathbb{R}^{n+1})$. By Proposition 3.5.3 the limit $M(X_0, u, p_{X_0}, 0^+)$ exists and can be computed by

$$M(X_0, u, p_{X_0}, 0^+) = \lim_{j \to \infty} M(X_0, u, p_{X_0}, r_j)$$

= $\lim_{j \to \infty} M(0, u_{X_0, r_j}, p, 1)$
= $\lim_{j \to \infty} \int_{\partial B_1} |y|^a (u_{X_0, r_j} - p)^2 d\sigma = 0.$

Moreover, let us suppose by contradiction that for any other sequence $r_i \to 0^+$ we have that the associated sequence converges to another blow-up limit, i.e. $u_{X_0,r_i} \to q \in \mathfrak{B}_k^a(\mathbb{R}^{n+1}), q \neq p$, then

$$0 = M(X_0, u, p_{X_0}, 0^+) = \lim_{i \to \infty} M(X_0, u, p_{X_0}, r_i)$$
$$= \lim_{i \to \infty} \int_{\partial B_1} |y|^a (u_{r_i} - p)^2 d\sigma$$
$$= \int_{\partial B_1} |y|^a (q - p)^2 d\sigma.$$

As we claim, since q and p are both homogenous of degree k they must coincide in \mathbb{R}^n .

Inspired by the previous uniqueness and nondegeneracy results, we introduce the notion of tangent map at every point on the nodal set $\Gamma(u)$.

Definition 3.5.7. Given $a \in (-1, 1)$, let u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u) \cap \Sigma$, for $k \geq \min\{1, 1-a\}$. We define as *tangent map* of u at X_0 the unique nonzero map $\varphi^{X_0} \in \mathfrak{B}_k^a(u)$ such that

$$u_{X_0,r}(X) = \frac{u(X_0 + rX)}{r^k} \longrightarrow \varphi^{X_0}(X).$$

Moreover, we define as *normalized* tangent map of u at X_0 , the unique nonzero map $p^{X_0} \in \mathfrak{B}_k^a(u)$ normalized with respect to the $L^{2,a}(\partial B_1)$ norm, i.e. the map obtained as

$$u_{X_0,r}(X) = \frac{u(X_0 + rX)}{\sqrt{H(X_0, u, r)}} \longrightarrow p^{X_0}$$

Exploiting the deep connection between the existence and uniqueness of the tangent map and the Taylor expansion of an L_a -harmonic function, we can find another characterization of the sets $\Gamma_k(u)$.

Corollary 3.5.8. For $a \in (-1, 1)$, let u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u) \cap \Sigma$, with $k \ge \min\{2, 2-a\}$. Then

- if $k \in 2 + \mathbb{N}$, we have $D^{\nu}u(X_0) = 0$ for every $|\nu| \le k 1$ and there exists $|\nu_0| = k$ such that $D^{\nu_0}u(X_0) \neq 0$;
- if $k \in 2 a + \mathbb{N}$, we have $D^{\nu}(uy |y|^{-a})(X_0) = 0$ for every $|\nu| \le k 1$ and there exists $|\nu_0| = k$ such that $D^{\nu_0}(uy |y|^{-a})(X_0) \ne 0$.

Finally, we can prove the validity of the weak unique continuation principle for the restriction of $\Gamma(u)$ on Σ . This result will improve the study of the nodal set of u by showing that its restriction on the characteristic manifold Σ is either with empty interior in Σ or is Σ itself. While in [73] the author proved a similar weak unique continuation property using a boot strap argument based on some regularity estimates for the L_a -operator, in our case we want to emphasize how our blow-up analysis and the classification of the tangent maps allow to study several local property of L_a -harmonic function.

Proposition 3.5.9. Let $a \in (-1,1)$ and u be an L_a -harmonic function in B_1 . If there exists $X_0 \in B_1 \cap \Sigma$ and $R < 1 - |X_0|$ such that u = 0 on $B_R(X_0) \cap \Sigma$, then $u \equiv 0$ on $B_1 \cap \Sigma$.

Proof. Let $X_0 \in \Gamma(u) \cap \Sigma$ and $R < 1 - |X_0|$. Since we are focusing the attention on the restriction of the nodal set on Σ , by definition of the symmetric part of u with respect to Σ , we can assume

that $u = u_e$ is purely symmetric with respect to Σ .

The idea of the proof is to prove that u is identically zero in the whole ball $B_R(X_0)$ in order to apply the Strong Unique continuation property Corollary 3.3.3, which is actually a stronger result since it does not only concern the trace of u on Σ .

Suppose by contradiction that $u \not\equiv 0$ on $B_R(X_0)$, then

$$H(X_0, u, r) = \frac{1}{r^{n+a}} \int_{\partial B_r(X_0)} |y|^a \, u^2 \mathrm{d}X > 0$$

for all $r \in (0, R)$. Now, since $X_0 \in \Gamma(u)$, there exists by Theorem 3.5.6 a unique nontrivial tangent map $\varphi^{X_0} \in \mathfrak{B}_k^a(\mathbb{R}^{n+1})$ of u at X_0 , where $k = N(X_0, u, 0^+)$. Since u is symmetric with respect to Σ , by Corollary 3.4.9 we know that $\varphi^{X_0} \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$, with $k \in 1 + \mathbb{N}$.

Let us see the points in $B_R(X_0) \cap \Sigma$ as the collection of point $X_0 + r\nu$ for r < R and $\nu \in S^n \cap \Sigma$. By the L^{∞}_{loc} convergence of the blow-up sequence we get that $\varphi^{X_0}(\nu) = 0$ for all $\nu \in S^n \cap \Sigma$, i.e. $\varphi^{X_0} \equiv 0$ on Σ . Let us prove now that $\varphi^{X_0} \equiv 0$ on \mathbb{R}^{n+1} by induction on the homogeneity $k = N(0, \varphi^{X_0}, 0^+)$.

Let k = 1, then up to a rotation $\varphi^{X_0}(x, y) = C\langle X, e_1 \rangle = Cx_1$, where $x = (x_1, \dots, x_n)$ and consequently C = 0. Now let us suppose that every k-homogenous L_a -harmonic polynomial symmetric with respect to Σ which is zero on Σ is actually identically zero in \mathbb{R}^{n+1} and consider the case k + 1. Given $v_i = \partial_{x_i} \varphi^{X_0} \in H^{1,a}(B_1)$ we have that

$$\begin{cases} L_a v_i = 0 & \text{ in } \mathbb{R}^{n+1} \\ v_i = 0 & \text{ on } \Sigma, \\ N(0, v_i, 0^+) \le k. \end{cases}$$

By the induction hypothesi we have that for every $i = 1, \dots, n v_i \equiv 0$ on \mathbb{R}^{n+1} , i.e. $\partial_{x_i} \varphi^{X_0} \equiv 0$ and consequently φ^{X_0} does not depend on $x \in \Sigma$. The absurd follows immediately since the only L_a -harmonic polynomial in the *y*-variable is purely antisymmetric and equal, up to a multiplicative constant, to $f(y) = y |y|^{-a}$.

Inspired by the doubling estimate in [73], we get

Proposition 3.5.10. Let $a \in (-1, 1)$ and u a L_a -harmonic function in B_1 . Then $\Gamma(u)$ has empty interior in \mathbb{R}^{n+1} and its restrictions $\Gamma(u) \cap \Sigma$ is either equal to Σ or it has empty interior in Σ itself. More generally,

$$\Gamma(u) \cap \Sigma = \Gamma(u_e) \cap \Sigma.$$

Proof. Assume by contradiction that there exists $X_0 \in \Gamma(u)$ such that $d = \text{dist}(X_0, \partial \Gamma(u)) < R$, where $R = 1 - |X_0|$. By definition of d, we have $H(X_0, u, r) > 0$ for $r \in (d, d + \varepsilon)$, for some $\varepsilon > 0$. By (103), the map $r \mapsto H(X_0, u, r)$ solves the Cauchy problem

$$\begin{cases} H'(r) = a(r)H(r), & \text{for } r \in (d, d + \varepsilon) \\ H(d) = 0, \end{cases}$$
(124)

where $a(r) = 2N(X_0, u, r)/r$, which is continuous at d by the monotonicity result of $r \mapsto N(X_0, u, r)$, i.e. Proposition 3.3.1. Then by uniqueness, $H(r) \equiv 0$ for r > d, which contradicts the definition of d and the assumption that u is not identically zero in B_1 . Now, let us consider $\Gamma(u) \cap \Sigma$. By definition of u_e, u_o we easily get

$$\Gamma(u) \cap \Sigma = \Gamma(u_e) \cap \Sigma.$$

Hence, let us suppose that $u \not\equiv u_o$, i.e. $\Gamma(u) \cap \Sigma \subsetneq \Sigma$, and assume as before that there exists $X_0 \in \Gamma(u) \cap \Sigma = \Gamma(u_e) \cap \Sigma$ such that $d = \text{dist}(X_0, \partial \Gamma(u_e) \cap \Sigma) < R$, where $R = 1 - |X_0|$. In other words, the symmetric part u_e of u solves for every r < d,

$$\begin{cases} L_a u_e = 0 & \text{on } B_r(X_0) \\ u_e = 0 & \text{on } B_r(X_0) \cap \Sigma \\ \partial_u^a u_e = 0 & \text{on } B_r(X_0) \cap \Sigma, \end{cases}$$

which implies that $u_e \equiv 0$ in $B_d(X_0)$, i.e. $H(X_0, u_e, d) = 0$. As before, by the uniqueness of the Cauchy problem (124), we get that u_e is identically zero in B_1 , in contradiction with the assumption $\Gamma(u) \cap \Sigma \subsetneq \Sigma$.

Looking again to the blow-up sequence, we can establish an auxiliary result concerning the convergence with respect to the Hausdorff distance $d_{\mathcal{H}}$. In particular, we will prove that given the blow-up sequence $(u_{X_0,r})_r$ of u at X_0 , then the nodal sets $\Gamma(u_{X_0,r})$ converge to $\Gamma(\varphi^{X_0})$ with respect to the Hausdorff distance. More precisely, given two sets A, B, the Hausdorff distance $d_{\mathcal{H}}(A, B)$ is defined as

$$d_{\mathcal{H}}(A,B) := \max\left\{\sup_{a \in A} \operatorname{dist}(a,B), \sup_{b \in B} \operatorname{dist}(A,b)\right\}.$$

Notice that $d_{\mathcal{H}}(A, B) \leq \varepsilon$ if and only if $A \subseteq N_{\varepsilon}(B)$ and $B \subseteq N_{\varepsilon}(A)$, where $N_{\varepsilon}(\cdot)$ is the closed ε -neighborhood of a set, i.e.

 $N_{\varepsilon}(A) = \left\{ X \in \mathbb{R}^{n+1} : \operatorname{dist}(X, A) \leq \varepsilon \right\}.$

Proposition 3.5.11. Let u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u) \cap \Sigma$. Given, $u_{X_0,r}$ the blow-up sequence at X_0 , i.e.

$$u_{X_0,r}(X) = \frac{u(X_0 + rX)}{r^k} \to \varphi^{X_0}(X).$$

Then $\Gamma(u_{X_0,r}) \cap \Sigma \to \Gamma(\varphi^{X_0}) \cap \Sigma$ with respect to the Hausdorff distance $d_{\mathcal{H}}$ in B_1 . More precisely, for every $k \ge \min\{1, 1-a\}$ we have that

$$\Gamma_k(u_{X_0,r}) \cap \Sigma \to \Gamma_k(\varphi^{X_0}) \cap \Sigma$$

with respect to the Hausdorff distance $d_{\mathcal{H}}$ in B_1

Proof. Let $r_i \to 0^+$ and $u_i = u_{X_0,r_i}$ be the blow-up sequence of u at X_0 associated to r_i and $\Gamma_k(u_i)$ be the sequence of nodal sets associated to the blow-up sequence. Through the proof, we will omit the fact that we are just focusing on the restriction of the nodal sets on Σ and we will call $\Gamma_k(\varphi^{X_0})$ as the tangent cone of $\Gamma_k(u)$ at X_0 . By Theorem 3.5.6 we already know that φ^{X_0} and $\Gamma(\varphi^{X_0})$ do not depend on the choice of the sequence r_k . By the definition of Hausdorff distance, the claimed result

$$d_{\mathcal{H}}\left(\Gamma_k(u_i) \cap B_1, \Gamma_k(\varphi^{X_0}) \cap B_1\right) \to 0$$

is equivalent to prove that for every $\varepsilon > 0$ there exists $\overline{i} > 0$ such that for every $i \ge \overline{i}$

$$\Gamma_k(u_i) \cap B_1 \subseteq N_{\varepsilon} \left(\Gamma_k(\varphi^{X_0}) \cap B_1 \right)$$

$$\Gamma_k(\varphi^{X_0}) \cap B_1 \subseteq N_{\varepsilon} \left(\Gamma_k(u_i) \cap B_1 \right).$$

Supposing by contradiction that the first inclusion is not true, then there exist $\overline{\varepsilon} > 0$ and a sequence $X_i \in \Gamma_k(u_i) \cap B_1$ such that dist $(X_i, \Gamma_k(\varphi^{X_0}) \cap B_1) > \overline{\varepsilon}$. Up to a subsequence, $X_i \to \overline{X} \in \Gamma(\varphi^{X_0}) \cap \overline{B_1}$ by the L^{∞}_{loc} convergence of $u_i \to \varphi^{X_0}$. Since $X_i \in \Gamma_k(u_i)$ is equivalent to $X_0 + r_i X_i \in \Gamma_k(u)$, given $\Omega \subset B_1$ such that $(X_0 + r_i X_i)_i \subset \Omega$, let us consider

$$R_{1} = \min_{p \in \overline{\Omega}} \operatorname{dist}(p, \partial B_{1}) < 1,$$
$$\widetilde{C} = \sup_{p \in \overline{\Omega}} N(p, u, R_{1}).$$

Hence, by the monotonicity result Proposition 3.3.1 and Corollary 3.3.2, for $p \in \Omega \cap \Gamma_k(u)$ and $r < R_1$ we get that $N(p, u, r) \ge k$ and

$$N(p, u, r) \le N(p, u, R_1) \left(\frac{R_1}{r}\right)^{n+a-1+2\tilde{C}} \le \tilde{C} \frac{1}{r^{n+a-1+2\tilde{C}}}.$$

In particular, from the second inequality we can easily state that for every $\varepsilon > 0$ there exists $\overline{R} = \overline{R}(n, a, \Omega, \varepsilon) > 0$ such that

$$N(p, u, r) \le k + \varepsilon_s$$

for every $p \in \Omega \cap \Gamma_k(u)$ and $r < \overline{R}$.

Now, since for i > 0 sufficiently large $N(X_i, u_i, r) \le N(X_0 + r_i X_i, u, r)$, if we take $p = X_0 + r_i X_i$ in the previous inequality, we get that there exists $\overline{R} = \overline{R}(n, a, X_0) > 0$ sufficiently small, such that for $r < \overline{R}$ we have

$$k \le N(X_i, u_i, r) \le k + \min\left(\frac{1}{2}, \frac{1-a}{2}, \frac{|a|}{2}\right)$$

Since $\lim_i N(X_i, u_i, r) = N(\overline{X}, \varphi^{X_0}, r)$ for sufficiently small r, we directly obtain from Proposition 3.4.11 that $N(\overline{X}, \varphi^{X_0}, 0^+) = k$, i.e. $\overline{X} \in \Gamma_k(\varphi^{X_0}) \cap \overline{B_1}$. Finally, the absurd follows immediately since $\Gamma_k(\varphi^{X_0}) \cup \{0\}$ is an homogeneous cone passing through the origin and hence it implies that $\operatorname{dist}(\overline{X}, \Gamma_k(\varphi^{X_0}) \cap B_1) = 0$.

Now let us consider the second inclusion, i.e. for every $\varepsilon > 0$ there exists $\overline{i} > 0$ such that for every $i \ge \overline{i}$

$$\Gamma_k(\varphi^{X_0}) \cap B_1 \subseteq N_{\varepsilon} (\Gamma_k(u_i) \cap B_1).$$

Let us start by proving that given $\overline{X} \in \Gamma_k(\varphi^{X_0})$ and $\delta > 0$ such that $B_{\delta}(\overline{X}) \cap \Gamma(\varphi^{X_0}) = B_{\delta}(\overline{X}) \cap \Gamma_k(\varphi^{X_0})$ there exiss $\overline{i} > 0$ such that for every $i \ge \overline{i}$ the function u_i must admit a zero of order k in $B_{\delta}(\overline{X})$, $\Gamma_k(u_i) \cap B_{\delta}(\overline{X})$. Suppose it is not true, we would have two possibilities: first that $u_i > 0$ in $B_{\delta}(\overline{X})$ for every k > 0 or secondly that every zeros of u_i is not of order k. In the first case, the positivity implies that φ^{X_0} must be an homogeneous L_a -harmonic function nonnegative in $B_{\delta}(\overline{X})$ with $\varphi^{X_0}(\overline{X}) = 0$, and therefore $\varphi^{X_0} \equiv 0$ in \mathbb{R}^{n+1} . In this case the contradiction follows by Lemma 3.5.5 and Theorem 3.5.6.

Secondly, since up to a subsequence there exists a sequence $X_i \in \Gamma_h(u_i) \cap B_{\delta}(\overline{X})$ for $h \neq k$, by arguing as in the proof of the other inclusion, we can prove that $X_i \to \widetilde{X} \in \Gamma_h(\varphi^{X_0}) \cap B_{\delta}(\overline{X})$, in contradiction with the definition of $\delta > 0$.

Finally, suppose the existence of $\overline{\varepsilon} > 0$ and $X_i \in \Gamma_k(\varphi^{X_0}) \cap B_1, X_i \to X \in \overline{\Gamma_k(\varphi^{X_0})} \cap \overline{B_1}$, such that $\operatorname{dist}(X_i, \Gamma_k(\varphi^{X_0}) \cap B_1) > \overline{\varepsilon}$. Since $\overline{X} = \{0\}$ is a trivial case, let us focus on the case $\overline{X} \in \Gamma_k(\varphi^{X_0}) \cap \overline{B_1}$. By definition, $\Gamma_k(\varphi^{X_0}) \cup \{0\}$ is an homogenous cone passing through the origin and hence we can take $\overline{X} \in \Gamma_k(\varphi^{X_0}) \cap B_1$ such that $\left|X - \overline{X}\right| \leq \overline{\varepsilon}/4$. Moreover, by the previous paragraph, there exist a sequence $\overline{X}_i \in \Gamma(u_i) \cap B_1$ and $\overline{i} > 0$, such that for $i \geq \overline{i}$ we have $\left|\overline{X}_i - \overline{X}\right| \leq \min\{\delta, \overline{\varepsilon}\}/4$ Hence, we get

$$\operatorname{dist}(X_i, \Gamma_k(\varphi^{X_0}) \cap B_1) \leq \left| X_i - \overline{X}_i \right| \leq \left| X_i - X \right| + \left| X - \overline{X} \right| + \left| \overline{X} - \overline{X}_i \right| < \overline{\varepsilon},$$

which leads a contradiction for large i > 0.

The following result will be a fundamental tool in the study of $\Gamma(u) \cap \Sigma$. Indeed, by using the continuation of the tangent map with respect to the $L^{2,a}(\partial B_1)$, we will prove a separation property for the set $\Gamma_k(u) \cap \Sigma$, for $k \ge \min\{2, 2-a\}$.

Theorem 3.5.12 (Continuation of the tangent map on $\Gamma_k(u)$). Let $X_0 \in \Gamma_k(u) \cap \Sigma$ and φ^{X_0} be the tangent map of u at X_0 , such that

$$u(X) = \varphi^{X_0}(X - X_0) + o(|X - X_0|^k).$$
(125)

Thus, the map $X_0 \mapsto \varphi^{X_0}$ from $\Gamma_k(u)$ to $\mathfrak{B}^a_k(\mathbb{R}^{n+1})$ is continuous. Moreover, for any compact set $K \subset \Gamma_k(u) \cap B_1$ there exists a modulus of continuity σ_K such that $\sigma_K(0) = 0$ and

$$|u(X) - \varphi^{X_0}(X - X_0)| \le \sigma_K (|X - X_0|) |X - X_0|^k,$$

for any $X_0 \in K$.

Proof. Since $\mathfrak{B}_k^a(\mathbb{R}^{n+1})$ is a convex subset of a finite-dimensional vector space, namely the space of all k-homogeneous polynomials in \mathbb{R}^{n+1} , all the norms on such space are equivalent and hence we can then endow $\mathfrak{B}_k^a(\mathbb{R}^{n+1})$ with the norm of $L^{2,a}(\partial B_1)$.

Fixed $X_0 \in \Gamma(u) \cap \Sigma$, by Theorem 3.5.6 we have the following expansion

$$u(X) = \varphi^{X_0}(X - X_0) + o(|X - X_0|^k).$$

where φ^{X_0} is the unique blow-up limit of u in X_0 . Given $\varepsilon > 0$, consider $r_{\varepsilon} = r_{\varepsilon}(X_0)$ such that

$$M(X_0, u, \varphi^{X_0}, r_{\varepsilon}) = \frac{1}{r_{\varepsilon}^{n+a+2k}} \int_{\partial B_{r_{\varepsilon}}} |y|^a \left(u(X_0 + X) - \varphi^{X_0}(X) \right)^2 d\sigma < \varepsilon.$$

There exists also $\delta_{\varepsilon} = \delta_{\varepsilon}(X_0)$ such that if $X_1 \in \Gamma_k(u) \cap \Sigma$ and $|X_1 - X_0| < \delta_{\varepsilon}$ then

$$\frac{1}{r_{\varepsilon}^{n+a+2k}} \int_{\partial B_{r_{\varepsilon}}} |y|^a \left(u(X_1+X) - \varphi^{X_0}(X) \right)^2 d\sigma < 2\varepsilon$$

or similarly

$$\int_{\partial B_1} |y|^a \left(\frac{u(X_1 + r_{\varepsilon}X)}{r_{\varepsilon}^k} - \varphi^{X_0}(X) \right)^2 d\sigma < 2\varepsilon$$

From Proposition 3.5.3, we have that $M(X_1, u, \varphi^{X_0}, r) < 2\varepsilon$ for $r \in (0, r_{\varepsilon})$, which implies

$$M(X_1, u, \varphi^{X_0}, 0^+) = \lim_{r \to 0} M(X_1, u, \varphi^{X_0}, r)$$

=
$$\lim_{r \to 0} \int_{\partial B_1} |y|^a \left(\frac{u(X_1 + rX)}{r^k} - \varphi^{X_0}(X) \right)^2 d\sigma$$

=
$$\int_{\partial B_1} |y|^a \left(\varphi^{X_1} - \varphi^{X_0} \right)^2 d\sigma \le 2\varepsilon.$$

Now, by the previous computations, for $|X_1 - X_0| < \delta_{\varepsilon}, 0 < r < r_{\varepsilon}$ we get

$$\left\| u_{X_{1},r} - \varphi^{X_{1}} \right\|_{L^{2,a}(\partial B_{1})} \leq \left\| u_{X_{1},r} - \varphi^{X_{0}} \right\|_{L^{2,a}(\partial B_{1})} + \left\| \varphi^{X_{0}} - \varphi^{X_{1}} \right\|_{L^{2,a}(\partial B_{1})} \leq 2\sqrt{2\varepsilon},$$

where $u_{X_1,r}$ and $u_{X_0,r}$ are the blow-up sequences defined in (123) centered respectively in X_1 and X_0 . Now, covering the compact set $K \subset \Gamma_k(u) \cap B_1$ with finitely many balls $B_{\delta_{\varepsilon}(X_0^i)}(X_0^i)$, for some points $X_0^i \in K$, i = 1, ..., N, we obtain that the previous inequality is satisfied for all $X_1 \in K$ with $r < r_{\varepsilon}^K = \min\{r_{\varepsilon}(X_0^i) : i = 1, ..., N\}$.

Now, since $u_{X_1,r} - \varphi^{X_1}$ is an L_a -harmonic function in B_1 , by [88, Lemma A.2] and (103), we get

$$\sup_{B_{1/2}} \left| u_{X_1,r} - \varphi^{X_1} \right| \le C(n,a) \left(\int_{B_1} \left| y \right|^a \left(u_{X_1,r} - \varphi^{X_1} \right)^2 \mathrm{d}X \right)^{1/2}$$
$$\le 2C(n,a) \sqrt{\frac{2\varepsilon}{n+a+1}}$$

for all $X_1 \in K, 0 < r < r_{\varepsilon}^K$, which immediately implies the second part of the Theorem. \Box

The following definition allows us to study the structure of the restriction $\Gamma(u) \cap \Sigma$. Inspired by Proposition 3.5.10, since $\Gamma(u) \cap \Sigma = \Gamma(u_e) \cap \Sigma$, where u_e is the symmetric part of u with respect to Σ , we characterize the sets $\Gamma_k(u)$ starting from the unique tangent map of u_e . Moreover, since we are dealing with a purely symmetric function, we will see that the structure of the nodal set on Σ is completely defined starting from the blow-up classes $\mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ and $\mathfrak{B}_k^*(\mathbb{R}^{n+1})$.

Definition 3.5.13. Given u an L_a -harmonic function on B_1 , for $k \ge \min\{1, 1-a\}$ we define on Σ

$$\Gamma_k^*(u) = \left\{ X_0 \in \Gamma_k(u) \cap \Sigma \colon \varphi_e^{X_0} \in \mathfrak{sB}_k^*(\mathbb{R}^{n+1}) \right\} \text{ and } \Gamma_k^a(u) = \Gamma_k(u) \setminus \Gamma_k^*(u),$$

where $\varphi_e^{X_0} \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ is the unique tangent map of u_e at X_0 .

In particular $\Gamma_1(u) = \Gamma_1^*(u)$ and for $k \ge 2$ the points in $\Gamma_k^a(u)$ are the ones whose tangent map depends on the variable y.

Corollary 3.5.14. For every $k \ge 2$ we have that $\overline{\Gamma_k^*(u)} \cap \Gamma_k^a(u) = \emptyset = \Gamma_k^*(u) \cap \overline{\Gamma_k^a(u)}$.

Proof. The proof of this result is based on the continuation of the tangent map of u on $\Gamma_k(u) \cap \Sigma$ with respect to the norm $L^{2,a}(\partial B_1)$.

First, suppose by contradiction that there exists a sequence $(X_i)_i \subset \Gamma_k^*(u)$ such that $X_i \to X_0 \in \Gamma_k^a(u)$. Let $\varphi^{X_i} = \varphi_e^{X_i}$ and $\varphi^{X_0} = \varphi_e^{X_0}$ be respectively the tangent map of u_e at X_i and X_0 , then

by Theorem 3.5.12 we get that $\varphi^{X_i} \to \varphi^{X_0}$ strongly in $L^{2,a}(\partial B_1)$, i.e. for every $\varepsilon > 0$ there exists $N = N(\varepsilon) > 0$ such that if i > N, then

$$\int_{\partial B_1} |y|^a \left(\varphi^{X_i} - \varphi^{X_0}\right)^2 \mathrm{d}\sigma \le \varepsilon$$

Hence, fixed $w_i = \varphi^{X_i} - \varphi^{X_0}$ we get that $L_a w_i = 0$ in B_1 and $||w_i||_{L^{2,a}(\partial B_1)} \to 0$. Since w_i is homogenous of degree k, we have

$$\int_{B_1} |y|^a w_i^2 \mathrm{d}X = \frac{1}{n+a+2k+1} \int_{\partial B_1} |y|^a w_i^2 \mathrm{d}\sigma$$
$$\int_{B_1} |y|^a |\nabla w_i|^2 \mathrm{d}X = k \int_{\partial B_1} |y|^a w_i^2 \mathrm{d}\sigma$$

which implies that $w_i \to 0$ strongly in $H^{1,a}(B_1)$. In particular, for every $\phi \in H^{1,a}(B_1)$ we have

$$\int_{B_1} |y|^a \langle \nabla \varphi^{X_0}, \nabla \phi \rangle \mathrm{d}X = \lim_{i \to \infty} \int_{B_1} |y|^a \langle \nabla \varphi^{X_i}, \nabla \phi \rangle \mathrm{d}X$$

If $\phi = \phi(y) \in C_c^{\infty}((-1,1))$ we get $\nabla \phi = \partial_y \phi e_y$ and consequently, since $\varphi^{X_i} \in \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$, that

$$\int_{B_1} |y|^a \,\partial_y \varphi^{X_0} \partial_y \phi \mathrm{d}X = \lim_{i \to \infty} \int_{B_1} |y|^a \,\partial_y \varphi^{X_i} \partial_y \phi \mathrm{d}X = 0,$$

in contradiction with the fact that $\varphi^{X_0} \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$. Similarly, suppose now there exists a sequence $(X_i)_i \subset \Gamma_k^a(u)$ such that $X_i \to X_0 \in \Gamma_k^*(u)$. As before, let φ^{X_i} and φ^{X_0} be respectively the tangent map of u_e at X_i and X_0 , fixed $w_i = \varphi^{X_i} - \varphi^{X_0}$ we get

$$\int_{B_1} |y|^a \phi \Delta_x w_i \mathrm{d}X + \int_{B_1} |y|^a \left(-\partial_{yy}^2 \varphi^{X_i} - \frac{a}{y} \partial_y \varphi^{X_i} \right) \phi \mathrm{d}X = k \int_{\partial B_1} |y|^a w_i \phi \mathrm{d}\sigma$$

for every $\phi \in H^{1,a}(B_1)$. The idea now is to reach the contradiction by induction on k, proving that it is impossible that the sequence of L_a -harmonic polynomials in $\mathfrak{sB}_k^a(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$ converges strongly in the $L^{2,a}(\partial B_1)$ -topology to a function in $\mathfrak{sB}_k^*(\mathbb{R}^{n+1})$. First, for $\phi \in H_0^{1,a}(B_1)$, we have

$$\left| \int_{B_1} |y|^a \left(\partial_{yy}^2 \varphi^{X_i} + \frac{a}{y} \partial_y \varphi^{X_i} \right) \phi \mathrm{d}X \right| \le C(n,k,a) \left(\|\nabla w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} \le C(n,k,a) \left(\|\nabla w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} \le C(n,k,a) \left(\|\nabla w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} \le C(n,k,a) \left(\|\nabla w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} \le C(n,k,a) \left(\|\nabla w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} \le C(n,k,a) \left(\|\nabla w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} \le C(n,k,a) \left(\|\nabla w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} \right) \|\phi\|_{H^{1,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)} + \|w_i\|_{L^{2,a}(B_1)}$$

which gives us that

$$\psi_i = -\partial_{yy}^2 \varphi^{X_i} - \frac{a}{y} \partial_y \varphi^{X_i} \rightharpoonup 0 \text{ in } L^{2,a}(B_1), \tag{126}$$

where ψ_i is a sequence of homogeneous L_a -harmonic polynomial of degree $k-2 \ge 0$. Since $X_i \mapsto \psi_i$ is continuous, by Lebesgue's dominated convergence theorem, we get $\|\psi_i\|_{L^{2,a}(B_1)} \to 0$, i.e. $\psi_i \to 0$ strongly in $L^{2,a}(B_1)$.

Hence, let k = 2 and $\varphi^{X_i} \in \mathfrak{sB}_2^a(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_2^*(\mathbb{R}^{n+1})$ be the sequence that converges to some $\varphi^{X_0} \in \mathfrak{sB}_2^*(\mathbb{R}^{n+1})$. As in (126), let us consider the associate sequence ψ_i of L_a -harmonic polynomial of degree k - 2 = 0, i.e. a sequence of nonzero constants. Since $\varphi^{X_i} \in \mathfrak{sB}_2^a(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_2^*(\mathbb{R}^{n+1})$, by the reasoning in Section 3.4, there exists, up to a multiplicative constant, a unique homogeneous polynomial $u^{X_i} = u^{X_i}(x)$ of degree 2, such that

$$\varphi^{X_i}(x,y) = u^{X_i}(x) - y^2 \text{ in } \mathbb{R}^{n+1},$$

where $\Delta_x u^{X_i} = 2(1+a)$ in \mathbb{R}^n . In particular, by (126) we get $\psi_i \equiv 2(1+a)$, and the contradiction follows immediately since $a \in (-1, 1)$.

Suppose now that we have proved the statement for every $k \leq K$ and let us consider the case K + 1. By contradiction, let us suppose that $\mathfrak{sB}_{K+1}^{a}(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_{K+1}^{*}(\mathbb{R}^{n+1})$ is not closed in the $L^{2,a}(\partial B_1)$ topology and $\varphi^{X_i} \to \varphi^{X_0}$ strongly in $L^{2,a}(\partial B_1)$, with $\varphi^{X_0} \in \mathfrak{sB}_{K+1}^{*}(\mathbb{R}^{n+1})$. Thus, we already know that the sequence ψ_i defined by (126) strongly converges to the zero function with respect to the $L^{2,a}(B_1)$ topology. Now, since $(\psi_i)_i$ are (K-1)-homogenous, we have that the $L^{2,a}(B_1)$ and $L^{2,a}(\partial B_1)$ topologies are equivalent. Finally, given that $0 < K - 1 \leq K$, we have constructed a sequence of (K-1)-homogenous L_a -harmonic polynomials ψ_i that converges to the zero function $0 \in \mathfrak{sB}_{K-1}^*(\mathbb{R}^{n+1})$, which contradicts the inductive hypothesis. \Box

In the uniformly elliptic case, the Almgren and Weiss monotonicity formulas allow to prove the uniqueness and non degeneracy of the tangent map and also to construct the generalized Taylor expansion of u at X_0 .

In this degenerate-singular setting, since as we already pointed out that the symmetric and antisymmetric cases are complementary, we introduce the notion of *tangential field* of u on its nodal set in order to take care of both this aspect of the solution u.

Definition 3.5.15. Let $a \in (-1, 1)$, u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u) \cap \Sigma$, for some $k \ge \min\{1, 1-a\}$. We define as *tangent field* of u at X_0 the unique nontrivial vector field $\Phi^{X_0} \in (H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1}))^2$ such that

$$\Phi^{X_0} = (\varphi_e^{X_0}, \varphi_o^{X_0}),$$

where $\varphi_e^{X_0}$ and $\varphi_o^{X_0}$ are respectively the tangent map of the symmetric part u_e of u and of the antisymmetric one u_o .

The notion of tangent field will allows us to better understand the regularity of the nodal set $\Gamma(u)$. Indeed, the main weakness of the concept of tangent map in this context is that it take care either of the symmetric part of u or of the even one since they do not share the same optimal regularity and even the same possible vanishing orders. More precisely, by Definition 3.5.7, for every $X_0 \in \Gamma_k(u)$

$$u_{X_0,r}(X) = \frac{u_e(X_0 + rX)}{r^k} + \frac{u_o(X_0 + rX)}{r^k}$$
$$= \frac{u_e^a(X_0 + rX)}{r^k} + \frac{u_e^{2-a}(X_0 + rX)}{r^{k-1+a}} y |y|^{-a}$$

where both u_e^a and u_e^{2-a} are symmetric with respect to Σ . By Proposition 3.4.11 we already know that the tangent map of u at X_0 is either the tangent map of u_e or the one of u_o .

Definition 3.5.16. Let $a \in (-1, 1)$, u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma_k(u)$, for some $k \ge \min\{1, 1 - a\}$. We define as Almgren monotonicity formula associated to the tangent field Φ^{X_0} of u at X_0 as the vector

$$N(X, \Phi^{X_0}, r) = \left(N(X, \varphi_e^{X_0}, r), N(X, \varphi_o^{X_0}, r) \right).$$

Obviously, the "*vectorial*" notion of the Almgren frequency formula can be naturally extended to the L_a -harmonic function u as

$$N(X_0, u, r) = (N(X_0, u_e, r), N(X_0, u_o, r)).$$

for every $X_0 \in \Sigma$, but we will avoid this ambiguity on this notion. However, if the function u is symmetric or antisymmetric with respect to Σ , the Almgren monotonicity formula associated to Φ is equal to the one of the tangent map φ^{X_0} of u at X_0 and it does not contain further information on the local behaviour of u at X_0 . In general, proving uniqueness result on both the symmetric and antisymmetric part of u with respect to Σ gives the following generalized Taylor expansion

Corollary 3.5.17. Given $a \in (-1, 1)$, let u be an L_a -harmonic function in B_1 and $X_0 \in \Gamma(u) \cap \Sigma$. Then

$$u(X) = \varphi_e^{X_0}(X - X_0) + \varphi_o^{X_0}(X - X_0) + o(|X - X_0|^k)$$

where $\varphi_e^{X_0} \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ and $\varphi_o^{X_0} \in \mathfrak{aB}_k^a(\mathbb{R}^{n+1})$ are respectively the tangent maps of u_e and u_o at X_0 and $k = \max\{N(0, \varphi_e^{X_0}, 0^+), N(0, \varphi_o^{X_0}, 0^+)\}$.

Lemma 3.5.18. Let u be an L_a -harmonic function in B_1 and $\mathcal{R}(u)$ the set

$$\mathcal{R}(u) = \{X_0 \in \Gamma(u) \colon N(X_0, u_e, 0^+) = 1 \quad \text{or} \quad N(X_0, u_o, 0^+) = 1 - a\}$$
$$= \{X_0 \in \Gamma(u) \colon N(X_0, \Phi^{X_0}, 0^+) = (1, 1 - a)\}.$$

Then $\mathcal{R}(u) \cap \Sigma$ is relatively open in $\Gamma(u) \cap \Sigma$, while for $k \ge 2$ the set $\Gamma_k(u)$ is F_{σ} , i.e. it is a union of countably many closed sets.

Proof. The first part of the Lemma is a direct consequence of the upper semi-continuity of $X \mapsto N(X, u, 0^+)$ on Σ . More precisely, since $\Gamma(u) \cap \Sigma = \Gamma(u_e) \cap \Sigma$, we can restrict our attention on functions symmetric with respect to Σ and hence, we have

$$\mathcal{R}(u) \cap \Sigma = \{X_0 \in \Gamma(u) \cap \Sigma \colon N(X_0, u_e, 0^+) = 1\}$$

Now, by Lemma 3.4.9 we get

$$\{X_0 \in \Gamma(u) \colon N(X_0, u_e, 0^+) = 1\} = \{X_0 \in \Gamma(u) \colon N(X_0, u_e, 0^+) \le \frac{3}{2}\}.$$

Hence, let us focus our attention on the case $\Gamma_k(u_e) \cap \Sigma$, with $k \ge 2$. For $j \in \mathbb{N}$, let us define with E_j the set of points of Σ such that

$$E_{j} = \left\{ X_{0} \in \Gamma_{k}(u_{e}) \cap \Sigma \cap \overline{B_{1-1/j}}: \frac{1}{j}\rho^{k} \leq \sup_{|X-X_{0}|=\rho} |u_{e}(X)| < j\rho^{k}, \ 0 < \rho < 1 - |X_{0}| \right\}.$$

By Lemma 3.5.4 and Lemma 3.5.5 we have that

$$\Gamma_k(u) \cap \Sigma = \bigcup_{j=1}^{\infty} E_j$$

The result follows immediately once we prove that E_j is a collection of closed sets. Given $X_0 \in \overline{E_j}$, since it satisfies

$$\frac{1}{j}\rho^k \le \sup_{|X-X_0|=\rho} |u_e(X)| < j\rho^k,$$
(127)

we need only to show that $X_0 \in \Gamma_k(u) \cap \Sigma$, i.e. $N(X_0, u_e, 0^+) = k$. Since $X \mapsto N(X, u_e, 0^+)$ is upper semi-continuous on Σ , we readily have $N(X_0, u_e, 0^+) \ge k$. On the other hand, if $N(X_0, u_e, 0^+) = k' > k$, we would have

$$|u_e(X)| \le C |X - X_0|^{k'}$$
 in $B_{1-|X_0|}(X_0) \cap \Sigma$

which contradicts Lemma 3.5.4 and implies that $X_0 \in E_j$.

An other important consequence of our analysis of the tangent field of u at some nodal point $X_0 \in \Gamma(u) \cap \Sigma$ is the following a posteriori result about the "quasi" upper semi-continuity of the Almgren frequency $X \to N(X, u, 0^+)$ in the whole \mathbb{R}^{n+1} .

Obviously, the restriction of this map on the characteristic manifold Σ and the one on its complementary are both upper semi-continuous, but in general in the whole space \mathbb{R}^{n+1} the upper

semi-continuity is not a immediate consequence of the Almgren monotonicity formula.

topology of the nodal set near Σ .

This result is based on the decomposition (98) of L_a -harmonic functions and on the regularity result of Proposition 3.2.5 for L_a -harmonic function symmetric with respect to Σ . Moreover, the following result can be seen as the "*vectorial*" counterpart of the classic one, since it will establish the validity of an upper semi-continuity property for the Almgren frequency in the vectorial sense of Definition 3.5.16. In particular, it allows to relate the notion of vanishing order on Σ to the one on $\mathbb{R}^{n+1} \setminus \Sigma$, which is a fundamental step in order to comprehend the complete

Proposition 3.5.19. Let u be an L_a -harmonic function in B_1 . Given $(X_i)_i \in \Gamma_k(u) \setminus \Sigma$, with $k \in 1 + \mathbb{N}$ such that $X_i \to X_0 \in \Gamma(u) \cap \Sigma$, then

$$N(X_i, u, 0^+) \le \begin{cases} N(X_0, u_e, 0^+), \\ N(X_0, u_o, 0^+) + a. \end{cases}$$

Proof. By Definition 3.2.3 and Proposition 3.2.4, we already know that there exist $f \in H^{1,a}(B_1), g \in H^{1,2-a}(B_1)$ symmetric with respect to Σ and respectively L_a and L_{2-a} -harmonic in B_1 , such that

$$u(x,y) = f(x,y) + g(x,y)y|y|^{-a} \quad \text{in } B_1,$$
(128)

where respectively the first term is the symmetric part u_e of u with respect to Σ and the second one the antisymmetric part u_o .

Through this proof, let us suppose that up to a subsequence $y_i > 0$. Since $(X_i)_i \in \Gamma_k(u) \setminus \Sigma$ and the operator L_a is locally uniformly elliptic on $\mathbb{R}^{n+1} \setminus \Sigma$ we know that $D^{\nu}u(X_i) = 0$, for any $|\nu| < k$ and there exists $|\nu_0| = k$ such that $D^{\nu_0}u(X_i) \neq 0$. Let us prove the main result by induction on $k \ge 2$. If $k = N(X_i, u, 0^+) = 2$, then for every $j = 1, \dots, n$ we get from (128) that

$$\partial_{x_j} f(x_i, y_i) = -\partial_{x_j} g(x_i, y_i) y_i^{1-a},$$

$$-y_i^a \partial_y f(x_i, y_i) = (1-a)g(x_i, y_i) + y_i \partial_y g(x_i, y_i),$$

where the maps $X \mapsto \partial_{x_j} f(X), X \mapsto \partial_y f(X), X \mapsto \partial_{x_j} g(X)$ and $X \mapsto \partial_y g(X)$ are all smooth in B_1 thanks to Proposition 3.2.5. Passing through the limit as $i \to \infty$ we get

$$\partial_{x_i} f(X_0) = 0$$
 and $-\partial_u^a f(X_0) = (1-a)g(X_0).$

First, since $\partial_y^a f$ is antisymmetric with respect to Σ we get that $g(X_0) = 0$ and consequently $N(X_0, u_o, 0^+) = N(X_0, g, 0^+) + 1 - a \ge 2 - a$. Similarly, if $\varphi^{X_0}(f)$ is the tangent map of f at X_0 , we get $\left| \nabla_X(\varphi^{X_0}(f))(X_0) \right| = 0$ and consequently that $N(X_0, f, 0^+) = N(0, \varphi^{X_0}, 0^+) \ge 2$,

as required.

Now, let us prove the inductive step $k - 1 \mapsto k$. Let us consider for $j = 1, \ldots, n$ the collection of symmetric L_a -harmonic functions $v_j = \partial_{x_j} u$ and the antisymmetric L_{-a} -harmonic function $w = \partial_y^a u$. Since $N(X_i, u, 0^+) = k$, we get

$$N(X_i, v_j, 0^+) = k - 1$$
 for $i = 1, ..., n$ and $N(X_i, w, 0^+) = k - 1$,

where we remark that since $X_i \notin \Sigma$ it is the same to consider the order of vanishing of $\partial_y u$ or of the covariant derivative $w = \partial_y^a u$.

By the inductive hypothesi, passing through the limit as $i \to \infty$ we get

$$\begin{cases} N(X_0, v_{j,e}, 0^+) \ge k - 1\\ N(X_0, v_{j,o}, 0^+) + a \ge k - 1 \end{cases} \quad \text{for } j = 1, \dots, n \quad \text{and} \quad \begin{cases} N(X_0, w_e, 0^+) \ge k - 1\\ N(X_0, w_o, 0^+) - a \ge k - 1 \end{cases}$$

Hence, comparing this result with the notations in (128), since $v_j = \partial_{x_j} f$ and w = (1 - a)g on Σ , we get

$$\begin{cases} N(X_0, \partial_{x_j} f, 0^+) \ge k - 1\\ N(X_0, \partial_{x_j} g, 0^+) + a \ge k - 1 \end{cases} \quad \text{for } j = 1, \dots, n \quad \text{and} \quad \begin{cases} N(X_0, g, 0^+) \ge k - 1\\ N(X_0, \partial_{x_j}^a f, 0^+) - a \ge k - 1 \end{cases},$$

which directly imply that $N(X_0, u_e, 0^+) \ge k$ and $N(X_0, u_o, 0^+) \ge k - a$, as required.

Lemma 3.5.20. Let u be an L_a -harmonic function in B_1 and $\mathcal{R}(u)$ the set

$$\begin{aligned} \mathcal{R}(u) &= \left\{ X_0 \in \Gamma(u) \middle| \begin{array}{cc} N(X_0, u, 0^+) = 1 & \text{if } X_0 \notin \Sigma \\ N(X_0, u_e, 0^+) = 1 \text{ or } N(X_0, u_o, 0^+) = 1 - a & \text{if } X_0 \in \Sigma \end{array} \right\} \\ &= \left\{ X_0 \in \Gamma(u) \middle| \begin{array}{c} N(X_0, u, 0^+) = 1 & \text{if } X_0 \notin \Sigma \\ N(0, \Phi^{X_0}, 0^+) = (1, 1 - a) & \text{if } X_0 \in \Sigma \end{array} \right\}, \end{aligned}$$

is relatively open in $\Gamma(u)$, while for $k \ge \min\{2, 2-a\}$ the set $\Gamma_k(u)$ is F_{σ} , i.e. it is a union of countably many closed sets.

Proof. The first part of the Lemma is a direct consequence of the upper semi-continuity of $X \mapsto N(X, u, r)$ restricted to Σ and to $\mathbb{R}^{n+1} \setminus \Sigma$ and of the Proposition 3.5.19. Hence, let us focus our attention on the case $\Gamma_k(u)$, with $k \ge \min\{2, 2-a\}$. For $j \in \mathbb{N}$, let us define with E_j the set of points of Σ such that

$$E_{j} = \left\{ X_{0} \in \Gamma_{k}(u) \cap \Sigma \cap \overline{B_{1-1/j}}: \ \frac{1}{j}\rho^{k} \leq \sup_{|X-X_{0}|=\rho} |u(X)| < j\rho^{k}, \ 0 < \rho < 1 - |X_{0}| \right\}.$$

By Lemma 3.5.4 and Lemma 3.5.5 we have that

$$\Gamma_k(u) \cap \Sigma = \bigcup_{j=1}^{\infty} E_j.$$

The result follows immediately once we prove that E_j is a collection of closed sets. Given $X_0 \in \overline{E_j}$, since it satisfies

$$\frac{1}{j}\rho^k \le \sup_{|X-X_0|=\rho} |u(X)| < j\rho^k,$$

we need only to show that $X_0 \in \Gamma_k(u) \cap \Sigma$, i.e. $N(X_0, u, 0^+) = k$. Since $X \mapsto N(X, u, 0^+)$ is upper semi-continuous on Σ , we readily have $N(X_0, u, 0^+) \ge k$. On the other hand, if $N(X_0, u, 0^+) = k' > k$, we would have

$$|u(X)| \le C |X - X_0|^{k'}$$
 in $B_{1-|X_0|}(X_0)$,

which contradicts Lemma 3.5.4 and implies that $X_0 \in E_j$.

3.6 HAUSDORFF DIMENSION ESTIMATES FOR THE NODAL SET

In this Section we prove different estimates on the Hausdorff dimension of the sets $\Gamma(u)$ and $\Gamma(u) \cap \Sigma$. In the latter, we improve our analysis taking care of the regular and singular part of the restricted nodal set $\Gamma(u) \cap \Sigma$.

Hence, given $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 , let us split the nodal set $\Gamma(u)$ in its regular part

$$\mathcal{R}(u) = \left\{ X_0 \in \Gamma(u) \middle| \begin{array}{cc} N(X_0, u, 0^+) = 1 & \text{if } X_0 \notin \Sigma \\ N(X_0, u_e, 0^+) = 1 \text{ or } N(X_0, u_o, 0^+) = 1 - a & \text{if } X_0 \in \Sigma \end{array} \right\},$$
(129)

and its singular part

$$\mathcal{S}(u) = \left\{ X_0 \in \Gamma(u) \middle| \begin{array}{cc} N(X_0, u, 0^+) \ge 2 & \text{if } X_0 \notin \Sigma \\ N(X_0, u_e, 0^+) \ge 2 \text{ and } N(X_0, u_o, 0^+) \ge 2 - a & \text{if } X_0 \in \Sigma \end{array} \right\}.$$
(130)

The main idea is to apply a version of the Federer's Reduction Principle due to [78, Appendix A]. More precisely, given a class \mathcal{F} of functions invariant under rescaling and translation and a map \mathcal{S} which associates to each function a subset of \mathbb{R}^n , by the Reduction principle we can establishes conditions on F and S which imply that to control the Hausdorff dimension of $\mathcal{S}(u)$ for every $u \in \mathcal{F}$, we just need to control the Hausdorff dimension of $\mathcal{S}(u)$ for elements which are homogeneous of some degree.

Theorem 3.6.1 (Federer's Reduction Principle). Let $\mathcal{F} \subseteq L^{\infty}_{\text{loc}}(\mathbb{R}^{n+1})$ and define, for any given $u \in \mathcal{F}, X_0 \in \mathbb{R}^{n+1}$ and r > 0, the rescaled and translated function

$$u_{X_0,r} := u(X_0 + r \cdot).$$

We say that $u_n \to u$ in \mathcal{F} if and only if $u_n \to u$ uniformly on every compact set of \mathbb{R}^{n+1} . Moreover, let us assume that \mathcal{F} satisfies the following conditions:

- (F1) (Closure under rescaling, translation and normalization) Given any $|X_0| \le 1 r, 0 < r, \rho > 0$ and $u \in \mathcal{F}$, we have that $\rho u_{X_0,r} \in \mathcal{F}$.
- (F2) (Existence of a homogeneous blow- up) Given $|X_0| < 1, r_k \searrow 0$ and $u \in \mathcal{F}$, there exists a sequence $\rho_k \in (0, \infty)$, a real number $\alpha \ge 0$ and a function $\overline{u} \in \mathcal{F} \alpha$ -homogenous such that, if we define $u_k(x) = u(X_0 + r_k x) / \rho_k$ then ,up to a subsequence, we have

$$u_k \to \overline{u}$$
 in \mathcal{F} .

(F3) (Singular Set hypotheses) *There exists a map* \overline{S} : $\mathcal{F} \to \mathcal{C}$, where

$$\mathcal{C} := \{ A \subset \mathbb{R}^{n+1} : A \cap B_1(0) \text{ is relatively closed in } B_1(0) \}$$

such that

(1) Given $|X_0| \le 1 - r, 0 < r < 1$ and $\rho > 0$, it holds

$$\overline{\mathcal{S}}(\rho \, u_{X_0,r}) = \left(\overline{\mathcal{S}}(u)\right)_{X_0,r} := \frac{\overline{\mathcal{S}}(u) - X_0}{r}$$

(2) Given $|X_0| < 1, r_k \searrow 0$ and $u, \overline{u} \in \mathcal{F}$ such that there exists $\rho_k > 0$ satisfying $u_k := \rho_k u_{X_0, r_k} \to \overline{u}$ in \mathcal{F} , the following property holds:

$$\forall \varepsilon > 0, \exists k = k(\varepsilon) > 0 \text{ such that for every } k \le k(\varepsilon) \\ \overline{\mathcal{S}}(u_k) \cap B_1(0) \subseteq \{ x \in \mathbb{R}^{n+1} : dist(x, \overline{\mathcal{S}}(\overline{u})) < \varepsilon \}.$$

(131)

Then, if we define

$$d := \max \{ \dim V : V \text{ is a vector subspace of } \mathbb{R}^{n+1} \text{ and there exists } u \in \mathcal{F} \text{ and } \alpha \geq 0 \}$$

such that $\overline{\mathcal{S}}(u) \neq \emptyset$ and $u_{y,r} = r^{\alpha}u, \forall y \in V, r > 0 \Big\},\$

either $\overline{S}(u) \cap B_1(0) = \emptyset$ for every $u \in \mathcal{F}$ or else $\dim_{\mathcal{H}} (\overline{S}(u) \cap B_1(0)) \leq d$ for every $u \in \mathcal{F}$. Furthermore in the latter case there exists a function $\varphi \in \mathcal{F}$, a d-dimensional subspace $V \subseteq \mathbb{R}^{n+1}$ and a real number $\alpha \geq 0$ such that

$$\varphi_{Y,r} = r^{\alpha} \varphi$$
 for all $Y \in V, r > 0$ and $\overline{\mathcal{S}}(\varphi) \cap B_1(0) = V \cap B_1(0)$

At last if d = 0 then $\overline{S}(u) \cap B_{\rho}(0)$ is a fine set for each $u \in \mathcal{F}$ and $0 < \rho < 1$.

We will apply this general result due to Federer in order to construct some estimates on the Hausdorff dimension of the nodal set $\Gamma(u)$ and on its restriction $\Gamma(u) \cap \Sigma$. In the second case, we improve our analysis introducing its regular and singular part on Σ .

Theorem 3.6.2. Let $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 . Then $\dim_{\mathcal{H}}(\Gamma(u)) \leq n$.

Proof. A preliminary remark is that we only need to prove the Hausdorff dimensional estimates for the localization of the sets in $K \subset B_1$, since the general statement follows because a countable union of sets with Hausdorff dimension less than or equal to some $n \in \mathbb{R}^+_0$ also has Hausdorff dimension less than or equal to some \mathcal{F} defined as

$$\mathcal{F} = \left\{ u \in L^{\infty}_{\text{loc}}(\mathbb{R}^{n+1}) \setminus \{0\} \colon L_a u = 0 \text{ in } B_r(X_0), \text{ for some } r \in \mathbb{R}, X_0 \in \mathbb{R}^{n+1} \text{ with } B_r(X_0) \subset B_1 \right\}.$$

By the linearity of the L_a operator, we already know that the closure under rescaling, translation and normalization and assumption (F1) are all satisfied.

On the other hand, let $|X_0| < 1$, $r_k \downarrow 0^+$ and $u \in \mathcal{F}$, and choose $\rho_k = ||u(X_0 + r_k \cdot x)||_{L^{2,a}(\partial B_1)}$. Theorem 3.4.1 and Proposition 3.4.8 yield the existence of a blow-up limit $\varphi^{X_0} \in \mathcal{F}$, i.e. normalized tangent map of u at X_0 , such that, up to a subsequence, $u_k \to \varphi^{X_0}$ in \mathcal{F} and φ^{X_0} is a homogeneous function of degree $k = N(X_0, u, 0+) \ge \min\{1, 1-a\}$. Hence also (F2) holds.

Now, let us consider $\overline{S}: u \mapsto \Gamma(u)$. By the continuity of u, we already know that the set $\Gamma(u) \cap B_1$ is obviously closed in B_1 and it is quite straightforward to check that the two hypotheses in (F3) are satisfied.

Hence, in order to conclude the analysis, the only thing left to prove is that the integer d in (131) is equal to n. Suppose by contradiction that d = n + 1, then this would imply the existence of $\varphi \in \mathcal{F}$ with $\overline{\mathcal{S}}(\varphi) = \mathbb{R}^{n+1}$ i.e., $\varphi \equiv 0$ on \mathbb{R}^{n+1} , which contradicts the fact the fact the $\Gamma(\varphi)$ has empty interior.

Actually, taking $V = \mathbb{R}^{n-1} \times \{0\} \times \mathbb{R}$ and $\varphi(X) = \langle X, e_n \rangle$, we obtain the claimed estimate on d.

Now, we prove a different stratification result for the set $\Gamma(u) \cap \Sigma$, in order to emphasize the different structure of the nodal set with respect of the one of the elliptic case. In particular, with this analysis we want to point out how the different classes of blow-up influence the stratification on the characteristic manifold Σ . Obviously, by Proposition 3.5.10 we already know that

$$\Gamma(u) \cap \Sigma = \Gamma(u_e) \cap \Sigma,$$

and it is either equal to Σ or with empty interior in Σ . Inspired by this fact, since we are dealing with the restriction of the nodal set on the characteristic manifold Σ , we will concentrate our attention on the trace of u on Σ , which is actually equal to the trace of u_e itself.

Theorem 3.6.3. Let $a \in (-1,1)$ and u be an L_a -harmonic function in B_1 . If $\Gamma(u) \cap \Sigma \neq \Sigma$, then, under the previous notations, we have $\dim_{\mathcal{H}}(\Gamma(u) \cap \Sigma) \leq n-1$ and more precisely

 $\dim_{\mathcal{H}}(\mathcal{R}(u) \cap \Sigma) = n - 1$ and $\dim_{\mathcal{H}}(\mathcal{S}(u) \cap \Sigma) \leq n - 1$.

Proof. Let us consider the class of functions \mathcal{F} defined as

$$\mathcal{F} = \left\{ u \in L^{\infty}_{\text{loc}}(\mathbb{R}^{n+1}) \setminus \{0\} \middle| \begin{array}{l} L_a u = 0 \text{ in } B_r(X_0), \text{ for some } r \in \mathbb{R}, X_0 \in \mathbb{R}^{n+1} \\ u \text{ symmetric with respect to } \Sigma \end{array} \right\}.$$

Since the functions in \mathcal{F} are symmetric with respect to Σ and nontrivial, the condition $\Gamma(u) \cap B_r(X_0) \cap \Sigma \neq \Sigma$ is always satisfied.

As before, we already know that the closure under rescaling, translation and normalization and assumption (F1) and (F2) are all satisfied. Moreover, by (129) and (130) we get

$$\mathcal{R}(u) \cap \Sigma = \{X_0 \in \Gamma(u) \cap \Sigma \colon N(X_0, u, 0^+) = 1\},\$$

$$\mathcal{S}(u) \cap \Sigma = \bigcup_{k \ge 2} \Gamma_k(u) \cap \Sigma = \bigcup_{k \ge 2} \{X_0 \in \Gamma(u) \cap \Sigma \colon N(X_0, u, 0^+) = k\}$$

since we are dealing with functions in the class \mathcal{F} . Now, we choose the map $\overline{\mathcal{S}}$ in (F3) according to our needs.

1. Dimensional estimate of $\Gamma(\mathbf{u}) \cap \Sigma$

First, let us consider $\overline{S}: u \mapsto \Gamma(u) \cap \Sigma$. By the continuity of u, we already know that the set $\Gamma(u) \cap \Sigma \cap B_1$ is obviously closed in B_1 and it is quite straightforward to check the two hypothesis in (F3). Therefore, in order to conclude the analysis of $\Gamma(u) \cap \Sigma$, the only thing left to prove is that the integer d in (131) is equal to n-1.

Suppose by contradiction that d = n, this would implies the existence of $\varphi \in \mathcal{F}$ such that $\overline{\mathcal{S}}(\varphi) = \mathbb{R}^n$ i.e., $\varphi \equiv 0$ on Σ . Since φ solves

$$\begin{cases} L_a \varphi = 0 & \text{in } \mathbb{R}^{n+1} \\ \varphi = 0 & \text{on } \Sigma \\ \partial_u^a \varphi = 0 & \text{on } \Sigma, \end{cases}$$

it implies that $\varphi \equiv 0$ on the whole \mathbb{R}^{n+1} , which contradicts the fact the $0 \notin \mathcal{F}$. Actually, by taking $V = \mathbb{R}^{n-1} \times (0,0) \subset \Sigma$ and $\varphi(X) = \langle X, e_n \rangle$, we obtain the claimed estimate on d.

2. Dimensional estimate of $\mathcal{R}(\mathbf{u}) \cap \boldsymbol{\Sigma}$

Let us consider $\overline{S}: u \mapsto \mathcal{R}(u) \cap \Sigma$. Since we are dealing just with symmetric function with

respect to Σ , by Lemma 3.4.9 we get that necessary $N(X_0, u, 0^+) = 1$ for every $X_0 \in \mathcal{R}(u)$. By the inclusion, we already know that

$$\dim_{\mathcal{H}}(\mathcal{R}(u) \cap \Sigma \cap B_1) \le n - 1.$$

Finally, we can apply the Reduction principle since (F3) is completely satisfied. More precisely, for $X_0 \in \Sigma \cap B_1$, $\rho > 0$ and t > 0 if $X \in \mathcal{R}(\rho u_{X_0,t}) \cap \Sigma$ then obviously $X_0 + tX \in \mathcal{R}(u) \cap \Sigma$, i.e. $N(X_0 + tX, u, 0^+) = 1$. Secondly, given $u_i, \overline{u} \in \mathcal{F}$ as in (F3), suppose by contradiction that there exists a sequence $X_i \in \Sigma \cap B_1$ and $\overline{\varepsilon} > 0$ such that

$$N(X_i, u_i, 0^+) = 1$$

and dist $(X_i, \overline{S}(\overline{u})) \geq \overline{\varepsilon}$. Since, up to a subsequence, $X_i \to \overline{X}$, by the upper semi-continuity of the Almgren frequency formula, we already know that $N(\overline{X}, \overline{u}, 0^+) \geq 1$. Moreover, up to a subsequence, $X_i \to \overline{X} \in \Gamma(\overline{u}) \cap \Sigma \cap \overline{B_1}$ by the L^{∞}_{loc} convergence of $u_i \to \overline{u}$. The contradiction follows from the same argument of the proof of the second case of Theorem 3.6.2.

More precisely, since $\Gamma(\overline{u}) \cap \Sigma$ is a conical set, i.e. for every $\lambda > 0$ and $\overline{X} \in \Gamma(\overline{u}) \in \Sigma$ we have $\lambda \overline{X} \in \Gamma(\overline{u}) \cap \Sigma$, we deduce that if we can prove $\overline{X} \in \mathcal{R}(\overline{u}) \cap \overline{B_1} \cap \Sigma$ we provide a contradiction, more precisely we get $\operatorname{dist}(\overline{X}, \overline{\mathcal{S}}(\overline{u}) \cap B_1) = 0$. Since there exists $\Omega \subset B_1 \setminus \Sigma$ such that $(X_0 + r_i X_i)_i \subset \Omega$, if we consider

$$R_1 = \min_{p \in \overline{\Omega}} \operatorname{dist}(p, \partial B_1),$$
$$\overline{C} = \sup_{p \in \overline{\Omega}} N(p, u, R_1),$$

we easily get from Corollary 3.3.2 that for $p \in \Omega \cap \mathcal{R}(u)$ and $r < R_1$ we have

$$N(p, u, r) \le N(p, u, R_1) \left(\frac{R_1}{r}\right)^{n+a-1+2\overline{C}} \le \overline{C} \frac{1}{r^{n+a-1+2\overline{C}}}.$$

In particular, from the previous inequality we get that there exists $\overline{R} = \overline{R}(n, a, X_0, \varepsilon) > 0$ sufficiently small, such that for $r < \overline{R}$ we have

$$1 \le N(X_i, u_i, r) \le 1 + \frac{1}{4}.$$

Since $\lim_i N(X_i, u_i, r) = N(\overline{X}, \overline{u}, r)$ for sufficiently small r, we directly obtain from Lemma 3.4.9 that $N(\overline{X}, \overline{u}, 0^+) = 1$, as we claimed.

As before, let us suppose now that there exist $\varphi \in \mathcal{F}$ and a *d*-dimensional subspace $V \subset \mathbb{R}^{n+1}$, with $d \leq n-1$, and $k \geq 0$ such that

$$\varphi_{Y,r} = r^k \varphi$$
 for all $Y \in V, r > 0$ and $\mathcal{R}(\varphi) \cap \Sigma \cap B_1 = V \cap B_1$

Since $\varphi \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ is homogenous of degree k with respect to any $Y \in V = \mathcal{R}(\varphi) \cap \Sigma$, namely $N(Y, \varphi, 0^+) = k$, we get that necessary k = 1 and that $\mathcal{R}(\varphi) \cap \Sigma$ is *d*-dimensional. Since every homogenous L_a -harmonic function of order k = 1 is one dimensional, i.e. there exists $\nu \in S^{n-1}$ and C > 0 such that either

$$\varphi(X) = C\langle X, (\nu, 0) \rangle$$
, for every $X = (x, y) \in \mathbb{R}^{n+1}$

we get that $\mathcal{R}(\varphi) \cap \Sigma$ must be (n-1)-dimensional, and consequently that

$$\dim_{\mathcal{H}}(\mathcal{R}(u) \cap \Sigma \cap B_1) = n - 1.$$

3. Dimensional estimate of $\mathcal{S}(\mathbf{u}) \cap \boldsymbol{\Sigma}$

Let us focus on the singular strata

$$\mathcal{S}(u) \cap \Sigma = \bigcup_{k \ge 2} \left\{ X_0 \in \Gamma(u) \cap \Sigma \colon N(X_0, u, 0^+) = k \right\}$$

Hence, given \overline{S} : $u \mapsto S(u)$, the map satisfies (F3), since for $X_0 \in \Sigma \cap B_1$, $\rho > 0$ and t > 0, if $X \in \overline{S}(\rho u_{X_0,t})$ we get

$$N(X,\rho u_{X_0,t},0^+) = k \longleftrightarrow N(X_0 + tX, u, 0^+) = k$$

which is equivalent to $X_0 + tX \in \Gamma_k(u) \subset S(u)$. Now, given $u_i = \rho_i u_{X_0,r_i}, \overline{u} \in \mathcal{F}$ as in (F3), suppose by contradiction that there exists a sequence $X_i \in B_1$ and $\overline{\varepsilon} > 0$ such that, up to a subsequence, $X_i \to \overline{X}$ and

$$N(X_i, u_i, 0^+) = k (132)$$

and dist $(X_i, \overline{S}(\overline{u})) \ge \overline{\varepsilon}$. By the upper semi-continuity of the Almgren frequency formula, we already know that $N(\overline{X}, \overline{u}, 0^+) \ge k$. Since $X_i \in \Gamma_k(u_i)$, there exists $\Omega \subset B_1 \setminus \Sigma$ such that $(X_0 + r_i X_i)_i \subset \Omega$, if we consider

$$R_{1} = \min_{p \in \overline{\Omega}} \operatorname{dist}(p, \partial B_{1}),$$

$$\overline{C} = \sup_{p \in \overline{\Omega}} N(p, u, R_{1}),$$

we easily get from Corollary 3.3.2 that for $p \in \Omega \cap \Gamma_k(u)$ and $r < R_1$ we have

$$N(p, u, r) \le N(p, u, R_1) \left(\frac{R_1}{r}\right)^{n+a-1+2\overline{C}} \le \overline{C} \frac{1}{r^{n+a-1+2\overline{C}}}.$$

In particular, from the previous inequality we get that there exists $\overline{R} = \overline{R}(n, a, X_0, \varepsilon) > 0$ sufficiently small, such that for $r < \overline{R}$ we have

$$k \le N(X_i, u_i, r) \le k + \frac{1}{4}.$$

Since $\lim_i N(X_i, u_i, r) = N(\overline{X}, \overline{u}, r)$ for sufficiently small r, we directly obtain from Lemma 3.4.9 that $N(\overline{X}, \overline{u}, 0^+) = k$, as we claimed.

Since $\mathcal{S}(u) \cap \Sigma \subseteq \Gamma(u) \cap \Sigma$, we already know that

$$\dim_{\mathcal{H}}(\mathcal{S}(u) \cap \Sigma \cap B_1) \le n-1,$$

which is actually the optimal bound even for the singular set. Indeed, since there exists $\varphi \in \mathcal{F}$, a (n-1)-dimensional subspace $V \subset \Sigma$ and $k \geq 0$ such that

$$\varphi_{Y,r} = r^k \varphi$$
 for all $Y \in V, r > 0$ and $\mathcal{S}(\varphi) \cap \Sigma \cap B_1 = V \cap B_1$.

In particular, for every $k \ge 2$, $n \ge 1$ it can be seen by taking $V = \mathbb{R}^{n-1} \times \{0, 0\}$ and

$$\varphi(X) = \frac{(-1)^{\frac{k}{2}} \Gamma\left(\frac{1}{2} + \frac{a}{2}\right)}{2^{k} \Gamma\left(1 + \frac{k}{2}\right) \Gamma\left(\frac{1}{2} + \frac{a}{2} + \frac{k}{2}\right)} \, {}_{2}F_{1}\left(-\frac{k}{2}, -\frac{k}{2} - \frac{a}{2} + \frac{1}{2}, \frac{1}{2}, -\frac{\langle X, e_{n} \rangle^{2}}{\langle X, e_{y} \rangle^{2}}\right) \langle X, e_{y} \rangle^{k},$$

as it was previously proved in Section 3.4.

3.7 REGULARITY OF THE REGULAR AND SINGULAR STRATA

In this Section we show some results about the regularity of the regular and singular strata of the nodal set $\Gamma(u)$. As in Section 3.6, we will consider first the stratification in \mathbb{R}^{n+1} of the whole nodal set $\Gamma(u)$, while in the second case we will focus the attention on the restriction $\Gamma(u) \cap \Sigma$ of the nodal set on the characteristic manifold.

The main idea of this stratification is to classify and then to stratify the nodal set by the spines of the normalized tangent maps, i.e. the largest vector space that leaves the tangent map invariant.

Indeed, we will introduce the subset $\Gamma_k^j(u)$ as the set of points at which every tangent map has at most j independent directions of translation invariance in order to correlate the nodal set of u with the dimension of the set where the tangent map φ^{X_0} vanishes with the same order of u. Moreover, by Theorem 3.6.2 we already know that $\Gamma_k^j(u)$ is well defined for $j \leq n-1$.

More precisely, if $k \ge \min\{2, 2-a\}$ given

$$\Gamma_k(u) = \{ X_0 \in \Gamma(u) \colon N(X_0, u, 0^+) = k \}$$

for each $j = 0, \ldots, n-1$ let us define

$$\Gamma_k^j(u) = \left\{ X_0 \in \Gamma_k(u) \colon \dim \Gamma_k(\varphi^{X_0}) = j \right\},\$$

where φ^{X_0} is the unique normalized tangent limit of u at X_0 . Obviously, since the uniformly elliptic case is well studied, we focus on the structure of the nodal set $\Gamma(u)$ near Σ .

Before to continue our analysis, let us prove that the concept of dimension in well defined.

Lemma 3.7.1. Given $a \in (-1, 1)$, for every $\varphi \in \mathfrak{B}_k^a(\mathbb{R}^{n+1})$, the singular set $\Gamma_k(\varphi)$ of order $k \geq \min\{2, 2-a\}$ is the largest vector subspace on Σ which leaves φ and $N(\cdot, \varphi, 0^+)$ invariant, i.e.

$$\Gamma_k(\varphi) = \left\{ Z \in \mathbb{R}^{n+1} \colon \varphi(X+Z) = \varphi(X) \text{ for every } X \in \mathbb{R}^{n+1} \right\}.$$

Proof. We can restrict our proof to the case $\varphi \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ for $k \ge 2$, since by Corollary 3.4.19 we can easily extend the analysis to the antisymmetric case. Thus, we already know by Corollary 3.5.8 that since $\varphi \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ we have

$$\Gamma_k(\varphi) = \left\{ X \in \mathbb{R}^{n+1} \colon D^{\nu}\varphi(X) = 0 \text{ for any } |\nu| \le k-1 \right\}.$$

Obviously $0 \in \Gamma_k(\varphi)$ by the homogeneity of φ and we claim that for every $Z \in \Gamma_k(\varphi)$

$$\varphi(X) = \varphi(X+Z), \quad \text{for all } X \in \mathbb{R}^{n+1},$$

in other words $\Gamma_k(\varphi)$ leaves the map φ invariant. Hence, let $Z \in \Gamma_k(\varphi)$, i.e.

$$D^{\nu}\varphi(Z) = 0 \quad \text{for any } |\nu| \le k - 1 \tag{133}$$

and write the homogenous polynomial $\varphi\in C^\infty$ as

$$\varphi(X) = \sum_{|\nu|=k} a_{\nu} X^{\nu},$$

where $X^{\nu} = x_1^{\nu_1} \cdot x_2^{\nu_2} \cdots y^{\nu_{n+1}}$ and $a_{\nu} \in \mathbb{R}$. By (133) we directly get that

$$\varphi(X) = \sum_{|\nu|=k} a_{\nu} (X - Z)^{\nu},$$

which implies the claimed invariance. Since φ is k-homogenous, for every $\lambda > 0$ and $X \in \mathbb{R}^{n+1}$

$$\begin{split} \varphi(X) &= \varphi(X - Z) \\ &= (\lambda + 1)^k \varphi\left(\frac{X - Z}{\lambda + 1}\right) \\ &= (\lambda + 1)^k \varphi\left(Z + \frac{X - Z}{\lambda + 1}\right) \\ &= \varphi(X + \lambda Z), \end{split}$$

therefore, we obtain $D^{\nu}\varphi(\lambda Z) = 0$ for any $|\nu| \le k - 1$, i.e. $\lambda Z \in \Gamma_k(\varphi)$. Similarly, noticing that for any $Z, W \in \Gamma_k(\varphi)$ we have $\varphi(Z + W + X) = \varphi(W + X) = \varphi(X)$ for any $X \in \mathbb{R}^{n+1}$, we get $Z + W \in \Gamma_k(\varphi)$.

Definition 3.7.2. Let $a \in (-1,1)$ and u be an L_a -harmonic function in B_1 . We call d^{X_0} the dimension of $\Gamma_k(u)$ at $X_0 \in \Gamma_k(u)$ as

$$d^{X_0} = \dim \Gamma_k(\varphi^{X_0})$$

= dim $\left\{ \xi \in \mathbb{R}^{n+1} \colon \langle \xi, \nabla_X \varphi^{X_0}(X) \rangle = 0 \text{ for all } X \in \mathbb{R}^{n+1} \right\}$

Following the previous notations we get $\Gamma_k^j(u) = \{X_0 \in \Gamma_k(u) : d^{X_0} = j\}.$

Hence, given $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 . Then, let us split the nodal set $\Gamma(u)$ in its regular part

$$\mathcal{R}(u) = \left\{ X_0 \in \Gamma(u) \middle| \begin{array}{cc} N(X_0, u, 0^+) = 1 & \text{if } X_0 \notin \Sigma \\ N(X_0, u_e, 0^+) = 1 \text{ or } N(X_0, u_o, 0^+) = 1 - a & \text{if } X_0 \in \Sigma \end{array} \right\},\$$

and its singular part

$$\mathcal{S}(u) = \left\{ X_0 \in \Gamma(u) \mid \begin{array}{cc} N(X_0, u, 0^+) \ge 2 & \text{if } X_0 \notin \Sigma \\ N(X_0, u_e, 0^+) \ge 2 \text{ and } N(X_0, u_o, 0^+) \ge 2 - a & \text{if } X_0 \in \Sigma \end{array} \right\}.$$

As we previously remarked the definition of the regular set $\mathcal{R}(u)$ is well defined in such a way, for every $X_0 \in \Gamma(u) \cap \Sigma$ such that $N(X_0, u_e, 0^+) = 1$ or $N(X_0, u_o, 0^+) = 1 - a$ must exist a sequence of point $(X_i)_i \in \Gamma(u) \setminus \Sigma$ such that $N(X_i, u, 0^+) = 1$ and $X_i \to X_0$. The following result gives a generalization in the context of degenerate-singular operator of the concept of regular hypersurface as the set of points where the function vanishes away from its critical set. **Theorem 3.7.3.** Let $a \in (-1,1)$, $a \neq 0$ and u be an L_a -harmonic function in B_1 . Then the regular set $\mathcal{R}(u)$ is locally a $C^{k,r}$ hypersurface on \mathbb{R}^{n+1} in the variable $(x, y |y|^{-a})$ with

$$k = \left\lfloor \frac{2}{1-a} \right\rfloor$$
 and $r = \frac{2}{1-a} - \left\lfloor \frac{2}{1-a} \right\rfloor$.

Moreover, we have that

$$\mathcal{R}(u) = \left\{ X \in \Gamma(u) \colon |\nabla_x u(X)|^2 + \left| \partial_y^a u(X) \right|^2 \neq 0 \right\}.$$
(134)

Proof. Let us start by proving the characterization of the regular set in terms of the derivatives of the L_a -harmonic function u. By (98), there exist $u_e^a \in H^{1,a}(B_1)$, $u_e^{2-a} \in H^{1,2-a}(B_1)$ respectively L_a and L_{2-a} -harmonic function in B_1 , symmetric with respect to Σ , such that

$$u(X) = u_e^a(X) + u_e^{2-a}(X)y |y|^{-a}$$
 in B_1 .

For every i = 1, ..., h, differentiating the previous equality, we get

$$\partial_{x_i} u(X) = \partial_{x_i} u_e^a(X) + \left(\partial_{x_i} u_e^{2-a}(X)\right) y |y|^{-a}$$
(135)

$$\partial_y^a u(X) = \left((1-a)u_e^{2-a}(X) + y\partial_y u_e^{2-a} \right) + \partial_y^a u_e^a, \tag{136}$$

where we split the two functions as sum of their symmetric and antisymmetric part. If $X_0 \in \mathcal{R}(u) \setminus \Sigma$ the condition in (134) is obviously satisfied by the local uniformly elliptic regularity outside Σ . Instead, if $X_0 \in \mathcal{R}(u) \cap \Sigma$, if $N(X_0, u_e, 0^+) = 1$ it follows

$$u_e(X) = \varphi_e^{X_0}(\nu^{X_0}) \langle X - X_0, \nu^{X_0} \rangle + o(|X - X_0|),$$

for some $\nu^{X_0} \in S^{n-1} = S^n \cap \Sigma$, and by Theorem 3.5.6 and (135) we get

$$\partial_{x_i} u(X_0) = \partial_{x_i} u_e(X_0) = \varphi_e^{X_0}(\nu^{X_0}) \langle e_i, \nu^{X_0} \rangle$$

and by the nondegeneracy of the blow-up limit $|\nabla u(X_0)| = \varphi_e^{X_0}(\nu^{X_0}) \neq 0$ (for further details, we remaind to the proof of Theorem 3.7.6). Similarly, taking care of the antisymmetric part, if $N(X_0, u_o, 0^+) = 1 - a$ we get

$$u_o(X) = \varphi_o^{X_0}(e_y) y |y|^{-a} + o(|X - X_0|^{1-a}),$$

and consequently $\partial_y^a u(X_0) = \partial_y^a u_o(X_0) = (1-a)\varphi_o^{X_0}(e_y) \neq 0$, as we claimed.

Now, let us consider the other part of the Theorem and let us study the regularity of the regular part $\mathcal{R}(u)$. Since the implicit function theorem implies that the nodal set of a smooth function is a smooth hypersurface away from the critical nodal set, we decide to introduce a suitable change

of variable.

More precisely, let us introduce the change of variable $\Phi \colon \mathbb{R}^{n+1} \to \mathbb{R}^{n+1}$ such that

$$\Phi \colon (x,z) \mapsto \left(x, (1-a)z \, |z|^{\frac{a}{1-a}}\right),$$
$$\Phi^{-1} \colon (x,y) \mapsto \left(x, \frac{y \, |y|^{-a}}{(1-a)^{1-a}}\right),$$

with Jacobian $|J_{\Phi^{-1}}(x,y)| = (1-a)^a |y|^{-a}$ and $\Phi(X_0) = X_0$, for every $X_0 \in \Sigma$. By (98) and This change of variable is well known in the literature since it allows to correlate our class of degenerate-singular operator with the class of Baouendi-Grushin Operators (see also [52]). In particular, since $a \in (-1, 1)$, we get by simple computations that $\Phi \in C^{k', r'}(\mathbb{R}^{n+1}, \mathbb{R}^{n+1})$, with

$$k' = \left\lfloor \frac{1}{1-a} \right\rfloor$$
 and $r' = \frac{1}{1-a} - \left\lfloor \frac{1}{1-a} \right\rfloor$

The previous quantity are well defined since $(1-a)^{-1} > 2$, for every $a \in (-1, 1)$ and it blows up as a approaches 1^- .

Now, given $v(x,z) = u(\Phi(x,z))$, we get $\Gamma(u) = \Phi(\Gamma(v))$ and by (98), (135) and (136)

$$v(x,z) = u_e^a(\Phi(x,z)) + u_e^{2-a}(\Phi(x,z))z$$

$$\partial_{x_i}v(x,z) = (\partial_{x_i}u)(\Phi(x,z)), \text{ for every } i = 1, \dots, h$$

$$\partial_z v(x,z) = (\partial_y^a u)(\Phi(x,z)),$$

so in particular $|\nabla v(x,z)|^2 = (|\nabla_x u|^2 + |\partial_y^a u|^2)(\Phi(x,z))$. By (98) and Proposition 3.2.5 we get that given and L_a -harmonic function u in B_1 , since $u_e^a(\Phi(x,z)), u_e^{2-a}(\Phi(x,z)) \in C^{k',r'}(B_{1/2})$ we obtain that $v \in C^{k',r'}(B_{1/2})$. Moreover, as we remarked in Section 3.4, since for every $\varphi^{X_0} \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$ we have that and our change of variables Φ acts only in the y-direction, we get from Proposition 3.2.5 and Theorem 3.5.6 that actually $v \in C^{k,r}(B_{1/2})$ with

$$k = \left\lfloor \frac{2}{1-a} \right\rfloor \ge 1$$
 and $r = \frac{2}{1-a} - \left\lfloor \frac{2}{1-a} \right\rfloor$.

Now, by the first part of the statement, since $X_0 \in \mathcal{R}(u) \cap \Sigma$ we get by Corollary 3.5.17

$$\left|\nabla v(X_0)\right|^2 = \left|\nabla_x u(X_0)\right|^2 + \left|\partial_y^a u(X_0)\right|^2 = \varphi_e^{X_0} (\nu^{X_0})^2 + (1-a)^2 \varphi_o^{X_0} (e_y)^2 \neq 0,$$

where $\varphi_e^{X_0}$ and $\varphi_o^{X_0}$ are respectively the tangent map of the symmetric and the antisymmetric part of u with respect to Σ . Since the conclusion follows after an application of the implicit function theorem on the function v and the relation $\Gamma(u) = \Phi(\Gamma(v))$, let us consider three different cases: (1) $N(X_0, u_e, 0^+) = 1$ and $N(X_0, u_o, 0^+) > 1 - a$, which implies that $\partial_z v(X_0) = 0$ and $\nabla_x v(X_0) = \varphi^{X_0}(\nu^{X_0})\nu^{X_0}$. In this case, up to relabeling the *x*-variables, by the implicit function theorem we get that there exists $\rho > 0$ and $g \in C^{k,r}(B_\rho(X_0))$ such that $x_1 = g(x) = g(x_2, \ldots, x_n, z)$ for every $(x, z) \in \Gamma(v) \cap B_\rho(X_0)$. Going back to the (x, y) variables, we get

$$x_1 = g(x_2, ..., x_n, y |y|^{-a})$$
 for every $X \in \Gamma(u) \cap B_{\rho/2}(X_0)$;

(2) $N(X_0, u_e, 0^+) > 1$ and $N(X_0, u_o, 0^+) = 1 - a$, in this case since $\partial_{x_i} v(X_0) = 0$ for all i = 1, ..., n and $\partial_z v(X_0) \neq 0$ we get that there exists $\rho > 0$ and $g \in C^{k,r}(B_\rho(X_0))$ such that $z = g(x) = g(x_1, ..., x_n)$ for every $(x, z) \in \Gamma(v) \cap B_\rho(X_0)$. Going back to the (x, y) variables, we get

$$|y|^{-a} = g(x)$$
 for every $X \in \Gamma(u) \cap B_{\rho/2}(X_0)$;

(3) $N(X_0, u_e, 0^+) = 1$ and $N(X_0, u_o, 0^+) = 1 - a$, we get that if a < 0, by applying the implicit function theorem with respect to the *x*-variables as in case (1), we get, up to a rotation on Σ , that

$$x_1 = g(x_2, ..., x_n, y |y|^{-a})$$
 for every $X \in \Gamma(u) \cap B_{\rho/2}(X_0)$;

where in this case $y |y|^{-a} \in C^{1,-a}_{loc}(B_1)$. Otherwise, if a > 0 by applying the implicit function theorem on the z-variable as in (2), we get

$$y |y|^{-a} = g(x)$$
 for every $X \in \Gamma(u) \cap B_{\rho/2}(X_0)$,

where in the both cases $g \in C^{k,r}(B_{\rho}(X_0))$.

We remark that the previous records can be changed considering the cases when the minimum between the Almgren frequency of the symmetric and the antisymmetric part of u is achieved by the first or the second one.

Thus, up to consider a smaller radius on the previous cases, the results on $\mathcal{R}(u)$ is a direct consequence of the local one on $\Gamma(u)$ near X_0 , since the regular set is relatively open in $\Gamma(u)$ and hence there exists $\rho > 0$ such that $\Gamma(u) \cap B_{\rho}(X_0) = \mathcal{R}(u) \cap B_{\rho}(X_0)$.

The previous result explains why the tangent map at a point of the restriction of the nodal set $\Gamma(u) \cap \Sigma$ does not allow to fully understand the geometric picture of the nodal set itself, since we need to take care of both the symmetric and antisymmetric part of u.

Furthermore, we can describe the local behaviour of the regular set $\mathcal{R}(u)$ near the characteristic manifold by using the tangent field Φ^{X_0} , which contains all the geometric information of the regular set. More precisely, as a direct consequence of the previous reports we get
Corollary 3.7.4. Let $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 . Then the regular part $\mathcal{R}(u)$ of the nodal set intersects the characteristic manifold Σ either orthogonally or tangentially. More precisely, given $X_0 \in \mathcal{R}(u) \cap \Sigma$

- if $N(X_0, u, 0^+) = 1$ the direction is orthogonal,
- if $N(X_0, u, 0^+) = 1 a$ the direction is tangential.

Moreover, independently on $a \in (-1, 1)$ and on the value of $N(0, \Phi^{X_0}, 0^+)$, the restriction on Σ of $\mathcal{R}(u)$ is completely described by $\varphi_e^{X_0}$.

Instead, since the structure of the singular set is well known outside of the characteristic manifold Σ , we decided to postpone our analysis and to concentrate our attention to the intersection of the nodal set on Σ .

Hence, in this last part of the Section, we extend the previous analysis focusing on the restriction of the regular and singular set on the characteristic manifold. First, since Lemma 3.7.1 relies on the homogeneity and the regularity of the homogenous polynomial $\varphi \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1})$, we can reasonably introduce the concept of *dimension restricted to* Σ .

Definition 3.7.5. Given $a \in (-1, 1)$, let u be an L_a -harmonic function in B_1 . We call $d_{\Sigma}^{X_0}$ the dimension of $\Gamma_k(u) \cap \Sigma$ at $X_0 \in \Gamma_k(u) \cap \Sigma$ as

$$\begin{split} d_{\Sigma}^{X_0} &= \dim \Gamma_k(\varphi^{X_0}) \cap \Sigma \\ &= \dim \left\{ \xi \in \Sigma \colon \langle \xi, \nabla_x \varphi^{X_0}(x, 0) \rangle = 0 \text{ for all } x \in \Sigma \right\}. \end{split}$$

Following the previous notations, we define $\Gamma_k^j(u) \cap \Sigma = \{X_0 \in \Gamma_k(u) \cap \Sigma : d_{\Sigma}^{X_0} = j\}$. In the previous Section, we split the restriction on the nodal set on Σ into its regular part

$$\mathcal{R}(u) \cap \Sigma = \{ X \in \Gamma(u) \cap \Sigma \colon N(X, u_e, 0^+) = 1 \},\$$

and its singular part

$$\mathcal{S}(u) \cap \Sigma = \{ X \in \Gamma(u) \cap \Sigma \colon N(X, u_e, 0^+) \ge 2 \} = \bigcup_{k \ge 2} \Gamma_k(u) \cap \Sigma.$$

Theorem 3.7.6. Let $a \in (-1, 1)$ and u be an L_a -harmonic function in B_1 . Then the regular set $\mathcal{R}(u)$ on Σ is locally a smooth hypersurface on Σ and

$$\mathcal{R}(u) \cap \Sigma = \{ X \in \Gamma(u) \cap \Sigma \colon |\nabla_x u_e(X)| \neq 0 \}.$$

Proof. By Proposition 3.5.10 we already know that

$$\Gamma(u) \cap \Sigma = \Gamma(u_e) \cap \Sigma$$

and it is either equal to Σ or with empty interior in Σ . Inspired by this fact, we will concentrate our attention on the trace of u on Σ , which is actually equal to the trace of u_e itself. In order to simplify we will just write u instead of u_e assuming the symmetry with respect to Σ .

Suppose that $\Gamma(u_e) \neq \Sigma$, by Theorem 3.5.6 and our blow-up classification, for every $X_0 \in \mathcal{R}(u) \cap \Sigma$ there exists a linear map $\varphi^{X_0} \in \mathfrak{sB}_1^a(\mathbb{R}^{n+1})$ such that

$$u(X) = \varphi^{X_0}(X - X_0) + o(|X - X_0|) = \varphi^{X_0}(\nu^{X_0})\langle X - X_0, \nu^{X_0} \rangle + o(|X - X_0|)$$

for some $\nu^{X_0} \in S^{n-1} = S^n \cap \Sigma$.

Moreover, by Theorem 3.5.12 we know that the map $X_0 \mapsto \varphi^{X_0}(\nu^{X_0})\nu^{X_0}$ is continuous. Passing through its trace on Σ , since $\nu \in \Sigma$ we get

$$u(x,0) = \varphi^{X_0}(\nu) \langle x - x_0, \nu \rangle + o(|x - x_0|)$$

Since by Proposition 3.2.5 the function $u \in C^{\infty}(B_{1/2})$, we can use the tangent map in order to compute the directional derivative of u, which will implies the nondegeneracy of the gradient on Σ of u at X_0 . More precisely, for every $\xi \in S^{n-1}$

$$\langle \nabla_x u(X_0), \xi \rangle = \left. \frac{d}{dt} u(X_0 + t\xi) \right|_{t=0} = \lim_{t \to 0} \frac{u(X_0 + t\xi)}{t} = \varphi^{X_0}(\nu^{X_0}) \langle \xi, \nu^{X_0} \rangle,$$

and hence $\nabla_x u(X_0) = \varphi^{X_0}(\nu^{X_0})\nu^{X_0}$ which is nonzero by Theorem 3.5.5. Finally, by the implicit function theorem we get the claimed result.

As we already mentioned, since for $k \ge 2$ we have $\mathfrak{sB}_k^a(\mathbb{R}^{n+1}) \setminus \mathfrak{B}_k^*(\mathbb{R}^{n+1}) \neq \emptyset$, we decide to introduce the following singular sets

$$\mathcal{S}^*(u) = \bigcup_{k \geq 2} \Gamma^*_k(u) \quad \text{and} \quad \mathcal{S}^a(u) = \bigcup_{k \geq 2} \Gamma^a_k(u),$$

where

$$\Gamma_k^*(u) = \left\{ X_0 \in \Gamma_k(u) \cap \Sigma \colon \varphi_e^{X_0} \in \mathfrak{sB}_k^*(\mathbb{R}^{n+1}) \right\} \text{ and } \Gamma_k^a(u) = \left(\Gamma_k(u) \cap \Sigma\right) \setminus \Gamma_k^*(u).$$

The idea is to stratify the singular set taking care of both the dimension $d_{\Sigma}^{X_0}$ and the different classes of tangent map associated to the sets $\Gamma_k^*(u)$ and $\Gamma_k^a(u)$.

Theorem 3.7.7. Given $a \in (-1, 1)$, let u be an L_a -harmonic function in B_1 . Then for $k \in 2 + \mathbb{N}$ and $j = 0, \dots, n-1$ the sets $\Gamma_k^j(u) \cap \Sigma$ is contained in a countable union of j-dimensional C^1 manifolds. *Proof.* The proof of this result follows the strategy of [50, Theorem 1.3.8]. Since φ^{X_0} is a polynomial of degree k on Σ , we can write the following

$$\varphi^{X_0}(x,0) = \sum_{|\alpha|=k} \frac{a_\alpha(x_0,0)}{\alpha!} x^\alpha,$$

where the coefficients $X \mapsto a_{\alpha}(X)$ are continuous on $\Gamma_k(u) \cap \Sigma$ and, since u(X) = 0 on $\Gamma_k(u)$ it holds

$$\left|\varphi^{X_0}(X-X_0)\right| \le \sigma\left(|X-X_0|\right)|X-X_0|^k \quad \text{for every } X, X_0 \in K.$$

For any multi-index $|\alpha| \leq k$, let us introduce for any $X \in \Gamma_k(u)$ the collection

$$f_{\alpha}(X) = \begin{cases} a_{\alpha}(X) & \text{if } |\alpha| = k \\ 0 & \text{if } |\alpha| < k \end{cases}.$$

Let us prove that the compatibility conditions for the Whitney's extension theorem are fully satisfied in order to guarantee the existence of a function $F \in C^k(\mathbb{R}^{n+1})$ such that

$$\partial^{\alpha} F = f_{\alpha} \quad \text{on } E_j,$$

for every $\alpha \leq k$. More precisely, following [92] our claim is that for any $X_0, X \in K$ it holds

$$f_{\alpha}(X) = \sum_{|\beta| \le k - |\alpha|} \frac{f_{\alpha+\beta}(X_0)}{\beta!} (X - X_0)^{\beta} + R_{\alpha}(X, X_0),$$

with

$$|R_{\alpha}(X, X_{0})| \leq \sigma_{\alpha} \left(|X - X_{0}|\right) |X - X_{0}|^{k - |\alpha|}$$
(137)

where $\sigma_{\alpha} = \sigma_{\alpha}^{K}$ is a certain modulus of continuity. If $|\alpha| = k$, since $R_{\alpha}(X, X_{0}) = a_{\alpha}(X) - a_{\alpha}(X_{0})$, we infer from the continuity of $X \mapsto \varphi^{X}$ on K that $|R_{\alpha}(X, X_{0})| \leq \sigma_{\alpha} (|X - X_{0}|)$. Instead, for $0 \leq |\alpha| < k$ we have

$$R_{\alpha}(X, X_{0}) = -\sum_{\substack{\gamma > \alpha \\ |\gamma| = k}} \frac{a_{\gamma}(X_{0})}{(\gamma - \alpha)!} (X - X_{0})^{\gamma - \alpha} = -\partial^{a} \varphi^{X_{0}} (X - X_{0}).$$
(138)

By contradiction, suppose that there is no modulus of continuity σ_{α} such that (137) is satisfied for $X, X_0 \in K$. Then, must exist $\delta > 0$ and two sequences $X^i, X_0^i \in K$ with $\rho_i = |X^i - X_0^i| \searrow 0$ such that

$$\left|\sum_{\substack{\gamma > \alpha \\ |\gamma| = k}} \frac{a_{\gamma}(X_0)}{(\gamma - \alpha)!} (X - X_0)^{\gamma - \alpha}\right| \ge \delta \left|X^i - X_0^i\right|^{k - |\alpha|}.$$

Thus, consider the blow-up sequence associated to the sequences $(X_0^i)_i$ and $(\rho_i)_i$ given by

$$u_i(X) = \frac{u(X_0^i + \rho_i X)}{\rho_i^k}, \quad \xi^i = \frac{X^i - X_0^i}{\rho_i}$$

where it is not restrictive to assume that $X_0^i \to X_0 \in K$ and $\xi^i \to \xi_0 \in \partial B_1$. By Theorem 3.5.12 we get $u^i \to \varphi^{X_0} \in \mathfrak{B}_k^a(\mathbb{R}^{n+1})$ uniformly on compact set and there exist a modulus of continuity such that

$$\left|u_{i}(X) - \varphi^{X_{0}^{i}}(X)\right| \leq \sigma\left(\rho_{i} |X|\right) |X|^{k}.$$

In particular, since $X_0^i, X^i \in K = E_j$, the inequalities (127) holds true for u^i at 0 and ξ^i . Thus, passing to the limit, we obtain that

$$\frac{1}{j}\rho^k \le \sup_{|X-\xi_0|=\rho} \left|\varphi^{X_0}(X)\right| < j\rho^k,$$

for $0 < \rho < +\infty$, which implies that $\xi_0 \in \Gamma_k(\varphi^{X_0})$. Finally, since $\partial^{\alpha} \varphi^{X_0}(\xi_0) = 0$ for $|\alpha| < k$, dividing both the left and the right hand side of (138) by $\rho_i^{k-|\alpha|}$ and passing to the limit, we reach a contradiction since we get

$$\left|\partial^a \varphi^{X_0}(\xi_0)\right| = \left|\sum_{\substack{\gamma > \alpha \\ |\gamma| = k}} \frac{a_{\gamma}(X_0)}{(\gamma - \alpha)!} (X - X_0)^{\gamma - \alpha}\right| \ge \delta.$$

Finally, under the previous notations, let us consider $X_0 = (x_0, 0) \in \Gamma_k^j(u) \cap E_i$, where E_i is defined in Lemma 3.5.20. Hence, by definition of $d_{\Sigma}^{X_0}$, there exists $n - d_{\Sigma}^{X_0}$ linearly independent unit vectors $(\nu_i)_i \subset S^n$, such that

$$\langle
u_i,
abla_X arphi^{X_0}
angle
eq 0 ext{ on } \Sigma_i$$

where $d_{\Sigma}^{X_0} = j$. Hence, there exist multi-indices α_i or order $|\alpha_i| = k - 1$ such that

$$\partial_{\nu_i} D^{\alpha_i} \varphi^{X_0}(0,0) \neq 0.$$

Since φ^{X_0} is a polynomial of degree k on Σ , we can write the following

$$\varphi^{X_0}(x,0) = \sum_{|\alpha|=k} \frac{a_\alpha(x_0,0)}{\alpha!} x^\alpha,$$

where the coefficients $X \mapsto a_{\alpha}(X)$ are continuous on $\Gamma_k(u) \cap \Sigma$. Thus, the nondegeneracy condition on φ^{X_0} implies

$$\partial_{\nu_i} D^{\alpha_i} F(x_0, 0) \neq 0, \quad i = 1, \cdots, n - d_{\Sigma}^{X_0}.$$
 (139)

Finally, since

$$\Gamma_k^j(u) \cap \Sigma \cap E_i \subset \bigcap_{i=1}^{n-j} \{ D^{\alpha_i} F = 0 \} \cap \Sigma,$$

in view of the implicit function Theorem, the condition (139) implies that $\Gamma_k^j(u) \cap \Sigma \cap E_i$ is contained in a *j*-dimensional manifold in a neighborhood of X_0 . The results follows immediately from Lemma 3.5.20

We remark that in this particular case of L_a -harmonic function symmetric with respect to Σ , since by the definition of tangent map at a point of the nodal set we have

$$u(x,0) = \sum_{|\alpha|=k} \frac{a_{\alpha}(x_0,0)}{\alpha!} x^{\alpha} + o(|x-x_0|^k)$$

and $u \in C^{\infty}(B_{1/2})$ thanks to Proposition 3.2.5, we get that $D^{\alpha}u(x_0, 0) = 0$ for $|\alpha| = k - 1$ and $D^{\alpha}u(x_0, 0) = a_{\alpha}(x_0, 0)$ for $|\alpha| = k$. Thus, the nondegeneracy condition on φ^{X_0} implies

$$\partial_{\nu_i} D^{\alpha_i} u(x_0, 0) \neq 0, \quad i = 1, \cdots, n - d_{\Sigma}^{X_0}.$$
 (140)

Hence, we can obtain conclusion just looking at the strata $\{D^{\alpha_i}u = 0\}$, with i = 1, ..., n - j. Instead, the previous proof is more general and it will be applied to a more general class of degenerate-singular operators in Section 3.8.

The following is the main Theorem of this stratification analysis, in particular it allows to emphasize the degenerate-singular attitude of the operator L_a near the characteristic manifold Σ by showing the presence of a (n-1)-dimensional singular stratum for $a \in (-1, 1)$ with $a \neq 0$.

Theorem 3.7.8. Given $a \in (-1, 1)$, let u be L_a -harmonic in B_1 . Then there holds

$$\mathcal{S}(u) \cap \Sigma = \mathcal{S}^*(u) \cup \mathcal{S}^a(u)$$

where $S^*(u)$ is contained in a countable union of (n-2)-dimensional C^1 manifolds and $S^a(u)$ is contained in a countable union of (n-1)-dimensional C^1 manifolds. Moreover

$$\mathcal{S}^*(u) = \bigcup_{j=0}^{n-2} \mathcal{S}^*_j(u) \quad \textit{and} \quad \mathcal{S}^a(u) = \bigcup_{j=0}^{n-1} \mathcal{S}^a_j(u),$$

where both $S_j^*(u)$ and $S_j^a(u)$ are contained in a countable union of j-dimensional C^1 manifolds.

Proof. The proof can be seen as an improvement of Proposition 3.7.7 since it consists on applying the previous strategy for the dimension and the regularity of the set $\Gamma_k^j(u)$ taking care on the

case when the tangent map belongs to $\mathfrak{sB}_k^*(\mathbb{R}^{n+1})$ or not. Indeed, this two cases influence the upper bound on the dimension $d_{\Sigma}^{X_0}$ and consequently the dimension of the singular strata. Hence, let us set

$$\mathcal{S}^{*}(u) = \bigcup_{j=0}^{n-2} \mathcal{S}_{j}^{*}(u) = \bigcup_{j=0}^{n-2} \bigcup_{k\geq 2} \{X \in \Gamma_{k}^{*}(u) \colon d_{\Sigma}^{X_{0}} = j\},\$$
$$\mathcal{S}^{a}(u) = \bigcup_{j=0}^{n-1} \mathcal{S}_{j}^{a}(u) = \bigcup_{j=0}^{n-1} \bigcup_{k\geq 2} \{X \in \Gamma_{k}^{a}(u) \colon d_{\Sigma}^{X_{0}} = j\}.$$

Since for every $k \ge 2$ the functions $\varphi \in \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$ are homogeneous polynomial harmonic in Σ , we have that dim $(\mathcal{S}(\varphi) \cap \Sigma) \le n-2$, and consequently $d_{\Sigma}^{X_0} \le n-2$ for every $X_0 \in \Gamma_k^*(u)$. Similarly, following Proposition 3.4.15 and the remarks in the proof of Theorem 3.6.3, since for every $k \ge 2$ there exists $\varphi \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$ such that dim $(\mathcal{S}(u) \cap \Sigma) = n-1$ we get that for $X_0 \in \Gamma_k^a(u)$ it holds $d_{\Sigma}^{X_0} \le n-1$.

Now, by applying the same argument in the proof of Proposition 3.7.7, if we set

$$\begin{split} \mathcal{S}_j^*(u) &= \bigcup_{k \ge 2} \left\{ X \in \Gamma_k^*(u) \colon d_{\Sigma}^{X_0} = j \right\} \quad \text{for } j = 0, \cdots, n-2 \\ \mathcal{S}_j^a(u) &= \bigcup_{k \ge 2} \left\{ X \in \Gamma_k^a(u) \colon d_{\Sigma}^{X_0} = j \right\} \quad \text{for } j = 0, \cdots, n-1, \end{split}$$

we get that $S_i^*(u)$ and $S_i^a(u)$ are contained in *j*-dimensional C^1 manifold.

Furthermore, by Proposition 3.4.15 we get that for any $X_0 \in S_{n-1}^a(u)$ the leading polynomial of u at X_0 , i.e. the first term of the Taylor expansion of u at X_0 , is an homogenous polynomial of two variables of the form (118) or (119), up to a rotation on Σ .

3.8 FRACTIONAL POWER OF ELLIPTIC OPERATOR IN DIVERGENCE FORM

In this Section, we find an application to the previous analysis relating, via the extension technique, the study of the restriction of the nodal set on the characteristic manifold Σ to the local properties of solutions of fractional power of elliptic differential equations in divergence form. Initially we start focusing the attention on the case of the fractional Laplacians $(-\Delta)^s$ and then we discuss the monotonicity formula and its consequences for solutions of fractional elliptic differential equations of the second order with Lipschitz leading coefficients.

Let $s \in (0,1)$ and $u: B_1 \subset \mathbb{R}^n \to \mathbb{R}$ be a nontrivial *s*-harmonic function in B_1 , that is

$$(-\Delta)^s u(x) = 0$$
 in B_1 . (141)

Here we define the *s*-Laplacian

$$(-\Delta)^s u(x) = C(n,s) \text{ P.V.} \int_{\mathbb{R}^n} \frac{u(x) - u(y)}{|x-y|^{n+2s}} \mathrm{d}y ,$$

where

$$C(n,s) = \frac{2^{2s}s\Gamma(\frac{n}{2}+s)}{\pi^{n/2}\Gamma(1-s)} \in \left(0, 4\Gamma\left(\frac{n}{2}+1\right)\right] \,. \tag{142}$$

In general, the *s*-Laplacian can be defined in various ways, which we review now. First, in order to better understand these definitions, we introduce the spaces

$$\tilde{H}^{s}(\mathbb{R}^{n}) = \left\{ u \in L^{2}(\mathbb{R}^{n}) \colon \left| \xi \right|^{s} (\mathcal{F}u)(\xi) \in L^{2}(\mathbb{R}^{n}) \right\},\$$

where $s \in (0, 1)$ and \mathcal{F} denotes the Fourier transform. In the literature, the spaces $\tilde{H}^s(\mathbb{R}^n)$ are called Bessel spaces and in particular they can be equivalently defined as a Sobolev-Slobodeckij spaces. More precisely, fixed $\Omega \subseteq \mathbb{R}^n$ an open set, for every fractional exponent $s \in (0, 1)$ we define $H^s(\Omega)$ as the set of all functions u defined on Ω with a finite norm

$$\|u\|_{H^{s}(\Omega)} = \left(\int_{\Omega} |u|^{2} dx + \frac{C(n,s)}{2} \int_{\Omega} \int_{\Omega} \frac{|u(x) - u(z)|^{2}}{|x - z|^{n+2s}} dx dz\right)^{1/2},$$

where the term

$$[u]_{H^{s}(\Omega)} = \left(\frac{C(n,s)}{2} \int_{\Omega} \int_{\Omega} \frac{|u(x) - u(z)|^{2}}{|x - z|^{n + 2s}} \mathrm{d}x \mathrm{d}z\right)^{1/2}$$
(143)

is the so-called Gagliardo seminorm of u in $H^s(\Omega)$. It can be proved that $\tilde{H}^s(\mathbb{R}^n) = H^s(\mathbb{R}^n)$ and in particular, for every $u \in H^s(\mathbb{R}^n)$ we get

$$[u]_{H^{s}(\mathbb{R}^{n})}^{2} = \int_{\mathbb{R}^{n}} |\xi|^{2s} |\mathcal{F}u(\xi)|^{2} d\xi = \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})}^{2}.$$

Note that one can also define the fractional Laplacian acting on spaces of functions with weaker regularity.

More precisely, following [77], let S be the Schwartz space of rapidly decreasing smooth functions in \mathbb{R}^n and $S^s(\mathbb{R}^n)$ be the space of smooth function u such that $(1 + |x|^{n+2s})D^k f(x)$ is bounded in \mathbb{R}^n , for every $k \ge 0$, endowed with the topology given by the family of seminorms

$$[f]_k = \sup_{x \in \mathbb{R}^n} \left(1 + |x|^{n+2s} \right) D^k f(x).$$

Under this notations, the fractiona Laplacian of $f \in S$ is well defined in $(-\Delta)^s f \in S_s$ and, by duality, this allows to define the fractional Laplacian for functions in the space

$$\mathcal{L}_{s}^{1}(\mathbb{R}^{n}) = \left\{ u \in L_{loc}^{1}(\mathbb{R}^{n}) \colon \int_{\mathbb{R}^{n}} \frac{|u(x)|}{(1+|x|)^{n+2s}} \mathrm{d}x < +\infty \right\}$$
$$= L_{\mathrm{loc}}^{1}(\mathbb{R}^{n}) \cap \mathcal{S}_{s}'(\mathbb{R}^{n}),$$

where $S'_{s}(\mathbb{R}^{n})$ stands for the dual of $S_{s}(\mathbb{R}^{n})$. We remark that necessary a function in $\mathcal{L}^{1}_{s}(\mathbb{R}^{n})$ needs to keep an algebraic growth of power strictly smaller than 2*s*, in order to make the above expression meaningful, as we pointed out in Chapter 1 and Chapter 2.

In order to study the local behaviour of u, let us look at the extension technique popularized by Caffarelli and Silvestre (see [23]), characterizing the fractional Laplacian in \mathbb{R}^n as the Dirichletto-Neumann map for a variable v depending on one more space dimension. Namely for every $u \in H^s(\mathbb{R}^n)$, let us consider $v \in H^{1,a}(\mathbb{R}^{n+1}_+)$ satisfying

$$\begin{cases} \operatorname{div}(y^a \nabla v) = 0 & \text{in } \mathbb{R}^{n+1}_+, \\ v(x,0) = u(x) & \text{in } \Sigma \end{cases}.$$
(144)

with $a = 1 - 2s \in (-1, 1)$. Such an extension exists unique and is given by the formula

$$v(x,y) = \gamma(n,s) \int_{\mathbb{R}^n} \frac{y^{2s} u(x)}{(|x-\eta|^2 + y^2)^{n/2+s}} \mathrm{d}\eta \qquad \text{where } \gamma(n,s)^{-1} =: \int_{\mathbb{R}^n} \frac{1}{(|\eta|^2 + 1)^{n/2+s}} \mathrm{d}\eta \,,$$

where the nonlocal operator $(-\Delta)^s$ translates into the Dirichlet-to-Neumann opeartor type

$$(-\Delta)^s \colon H^s(\mathbb{R}^n) \to H^{-s}(\mathbb{R}^n), \quad u \longmapsto -\frac{C(n,s)}{\gamma(n,s)} \lim_{y \to 0^+} y^{1-2s} \partial_y v(x,y),$$

with C(n, s) the normalization constant deeply studied in Chapter 2. By [70], it is known that the space $H^s(\mathbb{R}^n)$ coincides with the trace on $\partial \mathbb{R}^{n+1}$ of the weighted Sobolev space $H^{1,a}(\mathbb{R}^{n+1}_+)$ and in general

$$[u]_{H^{s}(\mathbb{R}^{n})}^{2} = \frac{C(n,s)}{\gamma(n,s)} \int_{\mathbb{R}^{n+1}} |y|^{1-2s} |\nabla v|^{2} \,\mathrm{d}X.$$

where v is the L_a -harmonic extension of u defined by (144). Since in the context of the extension problem the equation (141) translates in the homogeneous Neumann condition

$$\partial_y^a v(x,0) = -\frac{C(n,s)}{\gamma(n,s)} \lim_{y \to 0^+} y^{1-2s} \partial_y v(x,y) = 0 \quad \text{on } B_1 \subset \Sigma,$$

by applying the even reflection through Σ , we can study the structure of the nodal set of *s*-harmonic function in \mathbb{R}^n as the restriction of the nodal set $\Gamma(v)$ on the characteristic manifold Σ of the solution

$$\begin{cases} L_a v = 0 & \text{in } B_1^+ \\ v(x, -y) = v(x, y) & \text{in } B_1^+ \\ v(x, 0) = u(x) & \text{in } B_1 \end{cases}$$
(145)

where $a = 1 - 2s \in (-1, 1)$ and B_1^+ is the unitary (n + 1)-dimensional ball in \mathbb{R}^{n+1} . Moreover, by [70] is it well known that the class of trace on $B_1^+ \cap \Sigma = B_1$ of function L_a -harmonic in B_1^+ is equal to the space $H^s(B_1)$.

Through this Section we will always identify as v the L_a -harmonic extension of u in \mathbb{R}^{n+1} symmetric with respect to Σ and with $B_r(x_0)^+$ the ball in \mathbb{R}^{n+1}_+ of radius r > 0 and centered in the point $X_0 = (x_0, 0) \in \Sigma$ in the characteristic manifold associated to Σ .

The following results are a direct consequence of the ones obtained for purely symmetric L_a -harmonic function. For this reason the proof of the majority of them is skipped when the result is obtained just passing through the L_a -harmonic extension.

Proposition 3.8.1. Given $s \in (0, 1)$, let u be s-harmonic in B_1 . Then, there for every $x_0 \in B_1$,

$$\frac{1}{R^n} \int_{B_R(x_0)} u^2 \mathrm{d}x \le C(n,s) \left(\frac{R}{r}\right)^{2N-1} \frac{1}{r^n} \int_{B_r(x_0)} u^2 \mathrm{d}x,\tag{146}$$

for $0 < r < R < 1 - |x_0|$ and $N = N(X_0, v, 1 - |X_0|)$, with v the L_a -harmonic extension of u.

Proof. Let $v \in H^{1,a}(B_1)$ be the L_a -harmonic extension of u in \mathbb{R}^{n+1} , symmetric with respect to Σ . The idea of this proof is to "move" the doubling condition on \mathbb{R}^{n+1} to the characteristic manifold Σ . In [73] the author used a similar strategy to prove a so called "bulk doubling property". In our case we improve the proof by using our blow up analysis developed in Section 3.4 and applying the correct factor of scaling in order to pass from a doubling condition in the dimension n + a + 1 to the one on Σ .

Let $X_0 \in B_1 \cap \Sigma$, and v the L_a -harmonic extension symmetric with respect to Σ . Integrating the inequality in Corollary 3.3.2, see (104), we obtain that

$$\int_{B_{R}^{+}(X_{0})} |y|^{a} v^{2} \mathrm{d}X \leq \left(\frac{R}{r}\right)^{2C+n+a} \int_{B_{r}^{+}(X_{0})} |y|^{a} v^{2} \mathrm{d}X$$

for every $0 < r < R < 1 - |X_0|$, with $N = N(X_0, v, 1 - |X_0|)$. By the interpolation estimate in [73], we get

$$\begin{split} \frac{1}{R^n} \int_{B_R(X_0)} u^2 \mathrm{d}x &\leq C(n, a) \left(\frac{1}{R^{n+a+1}} \int_{B_R^+(X_0)} |y|^a \, v^2 \mathrm{d}X + \frac{1}{R^{n+a-1}} \int_{B_R^+(X_0)} |y|^a \, |\nabla v|^2 \, \mathrm{d}X \right) \\ &\leq C(n, a) \left(\frac{1}{R^{n+a+1}} \int_{B_R^+(X_0)} |y|^a \, v^2 \mathrm{d}X + \frac{1}{R^{n+a+1}} \int_{B_{2R}^+(X_0)} |y|^a \, v^2 \mathrm{d}X \right) \\ &\leq C(n, a) 2^{2N+n+a+1} \frac{1}{R^{n+a}} \int_{B_R^+(X_0)} |y|^a \, v^2 \mathrm{d}X \end{split}$$

where in the second inequality we used the Caccioppoli estimate (99) and in the last one the doubling condition. Since it yields the desired lower bound for the left hand side of the doubling condition on Σ , we left to prove the upper bound. Let us prove by contradiction the existence of C > 0 and a radius $0 < \overline{r} < R$ such that

$$\int_{\partial B_r^+(X_0)} |y|^a v^2 \mathrm{d}X \le C(n,a) r^{a+1} \int_{\partial B_r(X_0)} u^2 \mathrm{d}x \quad \text{for all } 0 < r \le \overline{r}, \tag{147}$$

which will finally implies (146) after a simple integration.

Hence, suppose there exists a sequence $r_k \searrow 0^+$ such that

$$\|v\|_{L^{2,a}(\partial B^+_{r_k}(X_0))} \ge k r_k^{\beta/2} \|u\|_{L^2(\partial B_{r_k}(X_0))},$$
(148)

with $\beta = a + 1$. Then let us consider the blow-up sequence of u centered at X_0 associated to $(r_k)_k$

$$v_k(X) = \frac{v(X_0 + r_k X)}{\rho_k} \quad \text{with } \rho_k^2 = \frac{1}{r_k^{n+a}} \int_{\partial B_{r_k}^+(X_0)} |y|^a v^2 \mathrm{d}X = H(X_0, v, r_k).$$

By definition we have $||v_k||_{L^{2,a}(\partial B_1^+)} = 1$, and by Lemma 3.4.2 the sequence $(v_k)_k$ is uniformly bounded in $H^{1,a}(B_R^+)$ and $L^{\infty}(\overline{B_R})$, for every R > 0. In particular, by (148), we get

$$\|u_k\|_{L^2(\partial B_1)} = \|v_k\|_{L^2(\partial B_1)} = \frac{r_k^{-\frac{n-1}{2}} \|v\|_{L^2(\partial B_{r_k}(X_0))}}{r_k^{-\frac{n+a}{2}} \|v\|_{L^{2,a}(\partial B_{r_k}^+(X_0))}} \le k^{-1} r_k^{\frac{a+1-\beta}{2}} = k^{-1}$$

Thus, up to a subsequence, by Theorem 3.4.1 the blow-up sequence $(v_k)_k$ strongly converge in $H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ and in $C^{0,\alpha}_{\text{loc}}(\mathbb{R}^n)$, for every $\alpha \in (0,1)$ to some homogeneous blow-up limit $\overline{v} \in H^{1,a}_{\text{loc}}(\mathbb{R}^{n+1})$ such that $\overline{v} = 0$ on B_1 , $\|\overline{v}\|_{L^{2,a}(\partial B_1^+)} = 1$ and it satisfies

$$\begin{cases} L_a \overline{v} = 0 & \text{in } \mathbb{R}^{n+1} \\ \partial_y^a \overline{v} = 0 & \text{in } \Sigma. \end{cases}$$

Hence, by Proposition 3.5.10 we get that $\overline{v} \equiv 0$ in contradiction with $\|\overline{v}\|_{L^{2,a}(B_1^+)} = 1$.

In order to justify the analysis of the local behaviour of *s*-harmonic functions, it is necessary to ensure the validity of the strong unique continuation property. It is known by [45] that an *s*-harmonic function in B_1 enjoys the *strong unique continuation property* in B_1 , i.e. the only solutions which vanishes of infinite order at a point $X_0 \in \Gamma(u)$ is $u \equiv 0$. Similarly, an *s*-harmonic function in B_1 is said to satisfies the *unique continuation property* in B_1 if the only solution of $(-\Delta)^s u = 0$ in $H^s_{loc}(B_1)$ which can vanish in an open subset of B_1 is $u \equiv 0$. Indeed, as a direct consequence of Proposition 3.5.10 we prove

Corollary 3.8.2. Let $s \in (0,1)$ and u be s-harmonic in B_1 . Then the nodal set $\Gamma(u)$ has either empty interior in B_1 or $u \equiv 0$.

Hence, it is reasonable to define the notion of vanishing order of u at $x_0 \in \Gamma(u)$. More precisely, the strong unique continuation property guarantees the existence of $k \in \mathbb{R}$ such that

$$\limsup_{r \to 0^+} \frac{1}{r^{n+2k}} \int_{B_r(x_0)} u^2 \mathrm{d}x > 0$$

In order to correlate the notion of vanishing order of s-harmonic functions with the one for their L_a -harmonic extension, let us introduce the following common definition.

Definition 3.8.3. Given $s \in (0, 1)$, let u be an s-harmonic function in B_1 and $x_0 \in \Gamma(u)$. The vanishing order of u in x_0 is defined as the number $\mathcal{O}(u, x_0) \in \mathbb{R}$ such that

$$\limsup_{r \to 0^+} \frac{1}{r^{n-1+2k}} \int_{\partial B_r(x_0)} u^2 \mathrm{d}x = \begin{cases} 0 & \text{if } k < \mathcal{O}(u, x_0) \\ +\infty & \text{if } k > \mathcal{O}(u, x_0). \end{cases}$$

In particular, from Lemma 3.4.9 and Proposition 3.8.1 we get

Corollary 3.8.4. Let $s \in (0,1)$ and $a = 1 - 2s \in (-1,1)$. Given u an s-harmonic function in B_1 , then the vanishing order $\mathcal{O}(u, x_0)$ of u in $x_0 \in \Gamma(u)$ satisfy

$$\mathcal{O}(u, x_0) = N(X_0, v, 0^+) = \lim_{r \to 0^+} \frac{r \int_{B_r^+(X_0)} |y|^a |\nabla v|^2 \, \mathrm{d}X}{\int_{\partial B_r^+(X_0)} |y|^a v^2 \mathrm{d}\sigma},$$

where v is the unique L_a -harmonic extension of u symmetric with respect to Σ and $X_0 = (x_0, 0)$.

Hence, for $k \in 1 + \mathbb{N}$, we define the subsets

$$\Gamma_k(u) \coloneqq \{ x_0 \in \Gamma(u) \colon \mathcal{O}(u, x_0) = k \},\$$

which is coherent with the Definition for the L_a -harmonic case. Indeed, inspired by the results in Section 3.4, we can prove a convergence result for the blow-up sequence associated to $x_0 \in \Gamma(u)$ to some blow-up limit $\varphi \in H^s_{\text{loc}}(\mathbb{R}^n)$.

Before to prove the main convergence result, let us introduce two different classes of tangent maps strictly related to the ones introduced in Definition 3.4.17 and Definition 3.4.18. In particular, we will see that the structure of the nodal set is completely defined starting from these blow-up classes.

Definition 3.8.5. Given $s \in (0, 1)$ and $k \in 1 + \mathbb{N}$, we define the set of all possible blow-up limit of order k, i.e. the set of the traces of all L_a -harmonic polynomial of degree k symmetric with respect to Σ , as

$$\mathfrak{B}^s_k(\mathbb{R}^n) = \left\{ \varphi \in H^s_{\mathrm{loc}}(\mathbb{R}^n): \text{ the } L_a\text{-extension of } \varphi \in \mathfrak{sB}^a_k(\mathbb{R}^{n+1}) \right\}.$$

Moreover, by Lemma 3.4.12 and Lemma 3.4.16, the space $\mathfrak{B}_k^s(\mathbb{R}^n)$ is the set of all possible homogenous polynomial of order k in \mathbb{R}^n , which is, by the results in [70], the space of traces on Σ of $\mathfrak{sB}_k^a(\mathbb{R}^n)$. Similarly, if we define with $\mathfrak{B}_k^*(\mathbb{R}^n)$ the set of function $\varphi \in \mathfrak{B}_k^s(\mathbb{R}^n)$ such that $\Delta \varphi = 0$ in \mathbb{R}^n , namely the collection of homogeneous harmonic polynomial of order k, it holds that $\mathfrak{B}_k^*(\mathbb{R}^n)$ coincides with the set of traces of blow-up limits in $\mathfrak{sB}_a^*(\mathbb{R}^{n+1})$.

The following result is a direct application of Theorem 3.4.1, Lemma 3.5.5, Theorem 3.5.6 and Theorem 3.5.12 on the L_a -harmonic extension of u symmetric with respect to Σ and it ensure the existence of a unique non trivial tangent map at every point of the nodal set of u.

Proposition 3.8.6. Given $s \in (0,1)$, let u be an s-harmonic function in B_1 and $x_0 \in \Gamma_k(u)$. Then there exists a unique k-homogenous polynomial $\varphi^{x_0} \in \mathfrak{B}_k^s(\mathbb{R}^n)$ such that

$$u_{x_0,r}(x) = \frac{u(x_0 + r_k x)}{r^k} \longrightarrow \varphi^{x_0}(x),$$

where the blow-up sequence $(u_{x_0,r})_r$ converges strongly in $H^s_{loc}(\mathbb{R}^n)$ and in $C^{1,\alpha}_{loc}(B_1)$, for every $\alpha \in (0,1)$. Moreover, the unique tangent map φ^{x_0} is nontrivial and it satisfies the following generalized Taylor expansion

$$u(x) = \varphi^{x_0}(x - x_0) + o(|x - x_0|^k),$$

where the map $x_0 \mapsto \varphi^{x_0}$ from $\Gamma_k(u)$ to the space $\mathfrak{B}_k^s(\mathbb{R}^n)$ is continuous.

Thus, let

$$\mathcal{R}(u) = \{ x_0 \in \Gamma(u) \colon \mathcal{O}(u, x_0) = 1 \},\$$

$$\mathcal{S}(u) = \bigcup_{k \ge 2} \Gamma_k(u) = \bigcup_{k \ge 2} \{ x_0 \in \Gamma(u) \colon \mathcal{O}(u, x_0) = k \},\$$

be respectively the *regular* and *singular* part of $\Gamma(u)$. Moreover, by Corollary 3.5.8 we can find a different characterization of the singular strata $\Gamma_k(u)$ for $k \ge 2$, i.e.

$$\Gamma_k(u) = \left\{ x_0 \in \Gamma_k(u) \middle| \begin{array}{c} D^{\nu}u(x_0) = 0 & \text{ for every } |\nu| \le k-1 \\ D^{\nu_0}u(x_0) \neq 0 & \text{ for some } |\nu_0| = k \end{array} \right\}.$$

The following are the main theorems related to the regularity and the geometric structure of the nodal set: while in the first result we focus the attention on the regular part of the nodal set, proving a result similar to its local counterpart (see [56, 66]), in the ones related to the singular strata we highlight the presence of a singular subset $S^s(u)$ strictly related to the nonlocal attitude of the fractional Laplacian.

Theorem 3.8.7. Given $s \in (0, 1)$, let u be s-harmonic in B_1 . Then the regular set $\mathcal{R}(u)$ is relatively open in $\Gamma(u)$ and is locally a smooth hypersurface on \mathbb{R}^n . Moreover

$$\mathcal{R}(u) = \{x \in \Gamma(u) \colon |\nabla u(x)| \neq 0\}.$$

Proof. By Corollary 3.8.2, let us suppose that $u \neq 0$ in B_1 and hence $\Gamma(u)$ has empty interior. Given v the unique L_a -harmonic extension of u symmetric with respect to Σ , well defined in $H^{1,a}(B_1^+)$ by (145), it is obvious to infer that

$$\mathcal{R}(v) \cap \Sigma = \mathcal{R}(u).$$

Moreover, by Lemma 3.5.20 we already know that $\mathcal{R}(u)$ is relatively open in $\Gamma(u)$ and by the application of the Federer reduction principle in Theorem 3.6.3 we get

$$\dim_{\mathcal{H}}(\mathcal{R}(u)) = n - 1$$

Now, by Theorem 3.8.6 and our blow-up classification, for every $x_0 \in \mathcal{R}(u) \cap \Sigma$ there exists a linear map $\varphi^{x_0} \in \mathfrak{B}_1^s(\mathbb{R}^n)$ such that

$$u(x) = \varphi^{x_0}(x - x_0) + o(|x - x_0|) = \varphi^{x_0}(\nu^{x_0}) \langle x - x_0, \nu^{x_0} \rangle + o(|x - x_0|)$$

for some $\nu^{x_0} \in S^{n-1}$.

Moreover, still by Theorem 3.8.6 we know that the map $x_0 \mapsto \varphi^{x_0}(\nu^{x_0})\nu^{x_0}$ is continuous. By Proposition 3.2.5, since $u \in C^{\infty}(B_{1/2})$ we can use the tangent map in order to compute the directional derivative of u, which will ensures the nondegeneracy of the gradient of u at x_0 . More precisely, for every $\xi \in S^{n-1}$

$$\left\langle \nabla u(x_0), \xi \right\rangle = \left. \frac{d}{dt} u(x_0 + t\xi) \right|_{t=0} = \lim_{t \to 0} \frac{u(x_0 + t\xi)}{t} = \varphi^{x_0}(\nu^{x_0}) \langle \xi, \nu^{x_0} \rangle,$$

and hence $\nabla u(x_0) = \varphi^{x_0}(\nu^{x_0})\nu^{x_0}$ which is nonzero by the nondegeneracy of the tangent map. Finally, by the implicit function theorem we get the claimed result. As in Section 3.7, initially we will prove a stratification result for the singular set S(u). The main idea of this stratification is to stratify the nodal set by the spines of the normalized tangent maps. Indeed, we will introduce the subset $\Gamma_k^j(u)$ as the set of points at which every tangent map has at most j independent directions of translation invariance in order to correlate the nodal set of u with the dimension of the set where the tangent map φ^{X_0} vanishes with the same order of u.

We remark that these result are a direct consequence of Theorem 3.7.7 and Theorem 3.7.8, nevertheless, for the sake of completeness, we present some technical details.

From Definition 3.7.5, given $s \in (0, 1)$ we call d^{x_0} the dimension of $\Gamma_k(u)$ at $x_0 \in \Gamma_k(u)$ as

$$d^{x_0} = \dim \left\{ \xi \in \mathbb{R}^n \colon \langle \xi, \nabla \varphi^{x_0}(x) \rangle = 0 \text{ for all } x \in \mathbb{R}^n \right\}.$$

Now, fixed $k \ge 2$, for each $j = 0, \ldots, n-1$ let us define

$$\Gamma_k^j(u) = \{x_0 \in \Gamma_k(u) \colon \dim \Gamma_k(\varphi^{x_0}) = j\},\$$

where φ^{x_0} is the unique tangent limit of u at x_0 . As we already mentioned, since for $k \ge 2$ we have $\mathfrak{B}_k^s(\mathbb{R}^n) \setminus \mathfrak{B}_k^*(\mathbb{R}^n) \neq \emptyset$, we decide to introduce the following singular sets

$$\mathcal{S}^*(u) = \bigcup_{k \geq 2} \Gamma^*_k(u) \quad \text{and} \quad \mathcal{S}^s(u) = \bigcup_{k \geq 2} \Gamma^s_k(u),$$

where

$$\Gamma_k^*(u) = \{x_0 \in \Gamma_k(u) \colon \varphi^{x_0} \in \mathfrak{B}_k^*(\mathbb{R}^n)\} \text{ and } \Gamma_k^s(u) = \Gamma_k(u) \setminus \Gamma_k^*(u).$$

The idea is to stratify the singular set taking care of both the dimension d^{x_0} and the different classes of tangent map associated to the sets $\Gamma_k^*(u)$ and $\Gamma_k^s(u)$.

Theorem 3.8.8. Given $s \in (0, 1)$ let u be s-harmonic in B_1 . Then it holds

$$\mathcal{S}(u) = \mathcal{S}^*(u) \cup \mathcal{S}^s(u)$$

where $S^*(u)$ is contained in a countable union of (n-2)-dimensional C^1 manifolds and $S^s(u)$ is contained in a countable union of (n-1)-dimensional C^1 manifolds. Moreover

$$\mathcal{S}^*(u) = \bigcup_{j=0}^{n-2} \mathcal{S}^*_j(u) \quad \textit{and} \quad \mathcal{S}^s(u) = \bigcup_{j=0}^{n-1} \mathcal{S}^s_j(u),$$

where both $S_i^*(u)$ and $S_i^s(u)$ are contained in a countable union of *j*-dimensional C^1 manifolds.

Proof. The proof is based on a combination of Theorem 3.7.7 Theorem 3.7.8. Since for every $k \geq 2$ the functions $\varphi \in \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$ are homogeneous polynomial harmonic in Σ , we have that $\dim (\mathcal{S}(\varphi) \cap \Sigma) \leq n-2$, and consequently $d_{\Sigma}^{X_0} \leq n-2$ for every $X_0 \in \Gamma_k^*(u)$.

Similarly, following Proposition 3.4.15 and the remarks in the proof of Theorem 3.6.3, since for every $k \geq 2$ there exists $\varphi \in \mathfrak{sB}_k^a(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$ such that dim $(\mathcal{S}(u) \cap \Sigma) = n - 1$ we get that for $X_0 \in \Gamma_k^a(u)$ it holds $d_{\Sigma}^{X_0} \leq n - 1$.

Now, by applying the same argument in the proof of Theorem 3.7.8, if we set

$$S_{j}^{*}(u) = \bigcup_{k \ge 2} \{ x \in \Gamma_{k}^{*}(u) \colon d^{x_{0}} = j \} \text{ for } j = 0, \cdots, n-2$$
$$S_{j}^{s}(u) = \bigcup_{k \ge 2} \{ x \in \Gamma_{k}^{s}(u) \colon d^{x_{0}} = j \} \text{ for } j = 0, \cdots, n-1,$$

we get the claimed result.

Furthermore, by Proposition 3.4.15 we get that for any $x_0 \in S_{n-1}^s(u)$ the leading polynomial of u at x_0 is a monomial of degree k with $k \in 2 + \mathbb{N}$ depending only on one variable of \mathbb{R}^n .

In order to show the optimality of the result, we will now present an explicit example of *s*-harmonic function in $B_1 = (-1, 1) \subset \mathbb{R}$ with vanishing order $k \geq 2$. More precisely, the following construction allows to exhibit an *s*-harmonic in $B_1 \subset \mathbb{R}^n$ with $\Gamma(u) = S_{n-1}^s(u)$. Fixed $s \in (0, 1)$, let $B_1 = (-1, 1) \subset \mathbb{R}$ be the unitary ball in the real line and $f \in \mathcal{L}_s^1(\mathbb{R}) \cap C(\mathbb{R})$ an admissible function. By the classical potential theory is it known that the unique solution of

$$\begin{cases} (-\Delta)^s u = 0 & \text{in } B_1 \\ u = f & \text{in } \mathbb{R} \setminus B_1 \end{cases}$$

an be computed explicitly as

$$u(x) = \int_{\mathbb{R}\setminus B_1} P(x,y)f(y)dy = \frac{\Gamma(1/2)\sin\pi s}{\pi^{3/2}} \left(1 - |x|^2\right)^s \int_{\mathbb{R}\setminus B_1} \frac{1}{(|y|^2 - 1)^s} \frac{f(y)}{|x - y|} dy.$$

We remark that several results and reference about the Poisson kernel can be found in the classical book of Landkof [63].

Now, given $f \in \mathcal{L}^1_s(\mathbb{R}) \cap C(\mathbb{R})$, let us consider $f_e, f_o \in \mathcal{L}^1_s(\mathbb{R}) \cap C(\mathbb{R})$ respectively the even and odd part of f uniquely defined as

$$f_e(x) = \frac{f(x) + f(-x)}{2}$$
 and $f_o(x) = \frac{f(x) - f(-x)}{2}$.

Under this notations, we get for $x \in (-1, 1)$

$$u(x) = \frac{2\Gamma(1/2)\sin\pi s}{\pi^{3/2}} \left(1 - |x|^2\right)^s \left[\int_1^{+\infty} \frac{f_e(y)y}{(|y|^2 - 1)^s(y^2 - x^2)} \mathrm{d}y + x \int_1^{+\infty} \frac{f_o(y)}{(|y|^2 - 1)^s(y^2 - x^2)} \mathrm{d}y\right]$$

Since for every $y \in \mathbb{R} \setminus B_1$ we have |y| > |x|, using the series expression

$$\frac{1}{y^2 - x^2} = \frac{1}{y^2} \sum_{n=0}^{\infty} \frac{x^{2n}}{y^{2n}},$$

we obtain

$$u(x) = \frac{2\Gamma(1/2)\sin\pi s}{\pi^{3/2}} \left(1 - |x|^2\right)^s \left[\sum_{n=0}^{\infty} A_{2n}(f)x^{2n} + \sum_{n=0}^{\infty} A_{2n+1}(f)x^{2n+1}\right],$$

where for every $n \in \mathbb{N}$

$$A_{2n}(f) = \int_{1}^{+\infty} \frac{f_e(y)}{y(|y|^2 - 1)^s y^{2n}} \mathrm{d}y \quad \text{and} \quad A_{2n+1}(f) = \int_{1}^{+\infty} \frac{f_o(y)}{y(|y|^2 - 1)^s y^{2n+1}} \mathrm{d}y.$$

In particular, if we consider $f(x) = (|x|^2 - 1)^s g(x^{-1})$ we get by a simple change of variables

$$A_{2n}(f) = \int_0^1 \frac{g_e(y)}{y} y^{2n} dy$$
 and $A_{2n+1}(f) = \int_0^1 g_o(y) y^{2n} dy.$

Hence, for every fixed order of vanishing $k \in 2 + \mathbb{N}$, there exists a polynomial function g(x) such that $A_i(f) = 0$, for every $i \le k - 1$. We remark that all these coefficients can be computed explicitly. Moreover, this construction it implies that for every vanishing order $k \in 2 + \mathbb{N}$ there exists an *s*-harmonic function in (-1, 1) which vanishes in zero with order k, which shows the purely nonlocal attitude of the singular set of *s*-harmonic functions.

In this part we will generalize the previous result to a more general class of fractional power of divergence form operator following the change of variables first introduced in [4] and deeply popularized in the works [48, 49]. Inspired by works, we consider solutions of homogeneous linear elliptic differential equations of the second order with Lipschitz leading coefficients and no lower order terms. We remark that in general the regularity assumption on the coefficient is optimal thanks to the counterexample of [68].

Let $A(x) = (a_{ij}(x))$ be a symmetric $n \times n$ matrix-valued function in \overline{B}_1 satisfying the following assumptions:

1. there exists $\lambda \in (0, 1)$ such that

$$|\lambda|\xi|^2 \le a_{ij}(x)\xi_i\xi_j \le \lambda^{-1}|\xi|^2$$
 for any $x \in \overline{B}_1$ and $\xi \in \mathbb{R}^n$;

2. there exists $\Gamma > 0$ such that for any $1 \le i, j \le n$

$$|a_{ij}(x) - a_{ij}(z)| \le \gamma |x - z|$$
 for any $x, z \in B_1$

Hence, consider now the uniformly elliptic operator

$$Lu = \operatorname{div}\left(A(x)\nabla u(x)\right) = \frac{\partial}{\partial x_i}\left(a_{ij}(x)\frac{\partial}{\partial x_j}u\right) = 0 \quad \text{in } B_1.$$
(149)

By [80], we already know the existence of characterization for the fractional powers of second order partial differential operators in some suitable class.

Proposition 3.8.9. Let $s \in (0,1)$ and $u \in \mathcal{L}^1_s(\mathbb{R}^n)$. Given $a = 1 - 2s \in (-1,1)$, a solution of the extension problem

$$\begin{cases} Lv + \frac{a}{y}\partial_y v + \partial_{yy}^2 v = 0 & \text{in } \mathbb{R}^{n+1}_+ \\ v(x,0) = u(x) & \text{in } \mathbb{R}^n; \end{cases}$$
(150)

is given by

$$v(x,y) = \frac{1}{\Gamma(s)} \int_0^\infty \left(e^{tL} (-L)^s f \right)(x) e^{-\frac{y^2}{4t}} \frac{\mathrm{d}t}{t^{1-s}},$$

and

$$(-L)^{s}u(x) = -\frac{\Gamma(s)}{2^{1-2s}\Gamma(1-s)} \lim_{y \to 0^{+}} y^{a} \partial_{y}u(x,y).$$

A similar extension can be constructed in the context of fractional powers $(-\Delta_M)^s$ of the Laplace-Beltrami operator on a Riemannian manifold M and to conformal fractional Laplacian on conformally compact Einstein manifolds and asymptotically hyperbolic manifold, thanks to the extension technique developed in [27] and the asymptotic expansion of their geodesic boundary defining function.

In this Section, we just consider the case of divergence form operator L in order to show how to deal with the limit case of Lipschitz coefficients. Therefore, this analysis will extend the results also to the case of Laplace-Beltrami with Lipschitz metric.

As we did for the fractional Laplacian, in order to study the local behaviour of solution of fractional elliptic equation associated to operator L in divergence form, let $s \in (0, 1)$ and u be a solution of the extended problem (150) associated to L, even with respect to the y-direction, i.e. such that

$$\begin{cases} \operatorname{div}_{x,y}(|y|^a \,\overline{A}(x) \nabla_{x,y} u) = 0, & \operatorname{in} \mathbb{R}^{n+1} \\ u(x,y) = u(x,-y), & \operatorname{in} \mathbb{R}^{n+1}. \end{cases}$$
(151)

where $\overline{A}(x)$ is a symmetric $(n+1) \times (n+1)$ matrix-valued function in B_1 such that

$$\overline{A}(x) = \begin{pmatrix} A(x) & 0\\ \hline 0 & 1 \end{pmatrix}.$$
 (152)

Inspired by Definition 3.2.1 we define the natural generalization of notion of L_a -harmonicity in the context of divergence form operator L with Lipschitz leading coefficient.

Definition 3.8.10. Let $a \in (-1, 1)$, we say $u \in H^{1,a}(B_1)$ is L_a^A -harmonic in B_1 if for every $\varphi \in C_c^{\infty}(B_1)$ we have

$$\int_{B_1} |y|^a \, \langle \overline{A}(x) \nabla u, \nabla \varphi \rangle \mathrm{d}X = 0,$$

where $\overline{A}(x)$ is the symmetric $(n+1) \times (n+1)$ matrix-valued function defined in (152).

Through this Section we will state all the result in the contxt of L_a^A -harmonic function in B_1 symmetric with respect to Σ , since the nodal set of the fractional powers $(-L)^s$ is completely defined as the restriction of the nodal set of L_a^A -harmonic function symmetric with respect to Σ , as we did in the previous part of the Section.

Obviously, in order to better understand the behaviour of general degenerate operator with Lipschitz leading coefficient, one could consider general L_a^A -harmonic solution and apply the ideas and the decomposition of the previous Sections.

In order to develop a blow-up analysis, let us construct a monotonicity formula base on a geometrical reduction introduce in [4] and deeply used in the local case [48, 49]. Hence, for $n \ge 3$, define a Lipschitz metric $\overline{g} = \overline{g}_{ij}(x, y) dx_i \otimes dx_j + \overline{g}_{yy}(x, y) dy \otimes dy$ on B_1 by setting

$$\overline{g}_{ij} = \overline{a}^{ij} \left(\det \overline{A} \right)^{\frac{1}{n-1}} = \begin{cases} a^{ij} |A|^{\frac{1}{n-1}}, & \text{if } 1 \le i, j \le n \\ |A|^{\frac{1}{n-1}}, & \text{otherwise} \end{cases}$$
(153)

where $\overline{a}^{i,j}$ and $a^{i,j}$ denote respectively the entries of \overline{A}^{-1} and A^{-1} . Letting similarly \overline{g}^{ij} be the entries of the inverse metric of \overline{g} , consider

$$r(x,y)^{2} = \overline{g}_{ij}(0)x_{i}x_{j} + \overline{g}_{yy}(0)y^{2}$$
$$= |A|^{\frac{1}{n-1}} \left(a^{ij}(0)x_{i}x_{j} + y^{2}\right)$$

and

$$\eta(x,y) = \frac{1}{r^2(x,y)} \left(\overline{g}^{kl}(x) \overline{g}_{ik}(0) \overline{g}_{jl}(0) x_i x_j + \overline{g}^{yy}(x) \overline{g}_{yy}(0) \overline{g}_{yy}(0) y^2 \right)$$
$$= \frac{a_{kl}(x) a^{ik}(0) a^{jl}(0) x_i x_j + y^2}{a^{ij}(0) x_i x_j + y^2}.$$

We can easily verify that η is a positive Lipschitz function in B_1 , whose Lipschitz constant depends on n, λ, Γ but not on $a \in (-1, 1)$.

Next, we introduce a new metric tensor $g = g_{ij}(x, y) dx_i \otimes dx_j + g_{yy}(x, y) dy \otimes dy$ in B_1 by

defining $g = \eta(x, y)\overline{g}$. In the intrinsic geodesic polar coordinates with pole at zero of the Riemannian manifold (B_1, g_{ij}) , the metric tensor takes the form

$$g = \mathrm{d}r \otimes \mathrm{d}r + r^2 b_{ij}(r,\theta) \mathrm{d}\theta_i \otimes \mathrm{d}\theta_j,$$

where

$$b_{ij}(0,0) = \delta_{ij}, \quad |\partial_r b_{ij}(r,\theta)| \le \Lambda(n,\lambda,\Gamma), \text{ for } 1 \le i,j \le n.$$
 (154)

Moreover, if we denote $|g| = |\det g|$ we get

$$\sqrt{|g|} = \eta^{\frac{n+1}{2}} \sqrt{|\overline{g}|} = \eta^{\frac{n+1}{2}} |A|^{\frac{1}{n-1}}.$$
(155)

Here we denote by $\nabla_g u$ and $\operatorname{div}_g X$ respectively the intrinsic gradient of a function u and the intrinsic divergence of a vector field X on B_1 in the metric g, i.e.

$$\nabla_g u = g^{ij} \frac{\partial u}{\partial x_i} \frac{\partial}{\partial x_j}, \quad \operatorname{div}_g X = \frac{1}{\sqrt{|g|}} \left(\sum_{i=1}^n \frac{\partial}{\partial x_i} (\sqrt{|g|} X_i) + \frac{\partial}{\partial y} (\sqrt{|g|} X_y) \right).$$

Finally, in this new metric we rewrite the divergence form equation in (151) as

$$\operatorname{div}_{g}\left(\left|y\right|^{a}\mu\nabla_{g}u\right) = \frac{1}{\sqrt{\left|g\right|}}\frac{\partial}{\partial y}\left[\left(1-\sqrt{\left|g\right|}g^{yy}\mu\right)u\frac{\partial}{\partial y}\left|y\right|^{a}\right]$$

where $\mu = \mu(x, y)$ is a positive Lipschitz function given by

$$\mu(x,y)=\eta(x,y)^{-\frac{n-1}{2}}$$

bounded in \overline{B}_1 and such that, in polar coordinates, it satisfies

$$\mu(0,0) = 1, \quad \left|\frac{\partial}{\partial r}\mu(r,\theta)\right| \le \Lambda(n,\lambda,\Gamma).$$
(156)

By (153), (155) and the definition of μ , for every $(x, y) \in B_1$

$$\sqrt{|g|}g^{yy}\mu = \eta^{\frac{n+1}{2}} |A|^{\frac{1}{n-1}} |A|^{-\frac{1}{n-1}} \eta^{-1}\eta^{-\frac{n-1}{2}} = 1.$$

To proceed, given $u \in H^{1,a}(B_1, \mathrm{d} V_g)$ a solution of

$$\operatorname{div}_g(|y|^a \mu \nabla_g u) = 0 \quad \text{in } B_1 \tag{157}$$

symmetric with respect to Σ , let us define for any $r \in (0, 1)$

$$E_g(u,r) = \frac{1}{r^{n+a-1}} \int_{B_g(r)} |y|^a \, \mu \, |\nabla_g u|^2 \, \mathrm{d}V_g$$
$$H_g(u,r) = \frac{1}{r^{n+a}} \int_{\partial B_g(r)} |y|^a \, \mu u^2 \mathrm{d}V_{\partial B_r}$$

where here $B_g(r)$ represents the geodesic ball in the metric g of radius r centered at the origin. We remark that by the polar decomposition of g, $B_q(r)$ coincides with the usual Euclidian ball.

In [52] the authors introduced a new monotonicity formula for a class of generalized Baouendi-Grushin operators. Since it is well known the existence of a connection between this two families of degenerate elliptic operator, their result gives an analogue counterpart in the context of our weighted degenerate operator, firstly introduced in the pioneering papers [43, 44]. More precisely, let us introduce the change of variable $\Phi \colon \mathbb{R}^{n+1}_+ \to \mathbb{R}^{n+1}_+$ such that

$$(x,z) = \Phi(x,y) = \left(x, \frac{y^{1-a}}{(1-a)^{1-a}}\right),$$

with inverse $\Phi^{-1}(x,z) = \left(x, (1-a)z^{\frac{1}{1-a}}\right)$. Now, given a function u(x,z) defined for $(x,z) \in \mathbb{R}^{n+1}_+$, we define a function $\tilde{u}(x,y)$ with $(x,y) \in \mathbb{R}^{n+1}_+$ as $\tilde{u}(x,y) = u(\Phi(x,y))$. A simple computations gives

$$L\tilde{u}(x,y) + \partial_{yy}\tilde{u}(x,y) + \frac{a}{y}\partial_y\tilde{u}(x,y) = z^{-\frac{2a}{1-a}} \left[\partial_{zz}u(x,z) + z^{\frac{2a}{1-a}}Lu(x,z)\right]$$

As we can see, the operator within square brackets in the right-hand side of the previous equation is a special case of the family of operators in $\mathbb{R}^n_x \times \mathbb{R}^1_z$ known as generalized Baouendi-Grushin operator.

Nevertheless, our problem does not satify the hyphotesis of the remarkable result obtained in [52] and consequently we need to construct a new monotonicity formula, which "does" extend the class of generalized Baouendi-Grushin operator for which a unique continuation principle holds true.

Under the previous notations, for $r \in (0,1)$, we define the Almgren type monotonicity formula as

$$N_g(u,r) = \frac{E_g(u,r)}{H_g(u,r)} = \frac{r \int_{B_g(r)} |y|^a \, \mu \, |\nabla_g u|^2 \, \mathrm{d}V_g}{\int_{\partial B_g(r)} |y|^a \, \mu u^2 \mathrm{d}V_{\partial B_r}}.$$

Theorem 3.8.11. Let $a \in (-1, 1)$ and u be a solution of (157) symmetric with respect to Σ . Then there exist a constant C > 0 such that the map $r \mapsto e^{Cr} N_g(u, r)$ is absolutely continuous and monotone nondecreasing on (0, 1). Hence, there always exists finite the limit

$$N_g(u, 0^+) = \lim_{r \to 0^+} N_g(u, r),$$

which we will call as the Almgren frequency formula.

Proof. By assumption, both $r \mapsto E_g(u, r)$ and $r \mapsto H_g(u, r)$ are locally absolutely continuous function on (0, 1), that is both their derivative are $L^1_{loc}(0, 1)$. First, passing to the logarithmic derivatives, the monotonicity of $r \mapsto N_g(u, r)$ is a direct consequence of the claim

$$\frac{d}{dr}\log N(X_0, u, r) = \frac{1}{r} + \frac{\frac{d}{dr}\int_{B_g(r)} |y|^a \,\mu \,|\nabla u|^2 \,\mathrm{d}V_g}{\int_{B_g(r)} |y|^a \,\mu \,|\nabla u|^2 \,\mathrm{d}V_g} - \frac{\frac{d}{dr}\int_{\partial B_g(r)} |y|^a \,\mu u^2 \mathrm{d}V_{\partial B_r}}{\int_{\partial B_g(r)} |y|^a \,\mu u^2 \mathrm{d}V_{\partial B_r}} \ge 0$$

for $r \in (0,1)$. First, by setting $b(r,\theta) = |\det b_{ij}(r,\theta)|$, we get $\sqrt{g(r,\theta)} = r^n \sqrt{b(r,\theta)}$ and we can rewrite the denominator $H_g(u,r)$ of the Almgren monotonicity formula as

$$H_g(u,r) = \int_{\partial B_g(1)} |\theta_n|^a \,\mu(r,\theta) u^2(r,\theta) \sqrt{b(r,\theta)} \mathrm{d}\theta,$$

where θ_n is the spherical coordinate associated to the *y*-direction. By differentiating respect to $r \in (0, 1, \text{we obtain})$

$$\frac{d}{dr}H_g(u,r) = \frac{2}{r^{n+a}} \int_{\partial B_g(r)} |y|^a \,\mu u \partial_\rho u \mathrm{d}V_{\partial B_r} + \frac{1}{r^{n+a}} \int_{\partial B_g(r)} \frac{|y|^a}{\sqrt{b}} \frac{\partial}{\partial\rho} \left(\mu\sqrt{b}\right) u^2 \mathrm{d}V_{\partial B_r}$$

where $\partial_{\rho} u$ denotes the radial differentiation $\partial_{\rho} u = \langle \nabla_g u, X/\rho \rangle$ for $X \in \mathbb{R}^{n+1}$. Finally, by (154) and (156) we get

$$\frac{d}{dr}H_g(u,r) = \frac{2}{r^{n+a}} \int_{\partial B_g(r)} |y|^a \,\mu u \partial_\rho u \mathrm{d}V_{\partial B_r} + O(1)H_g(u,r),\tag{158}$$

with O(1) a function bounded in absolute value by a constant $C = C(n, \Lambda)$. On the other hand, the divergence theorem gives

$$\int_{B_g(r)} |y|^a \,\mu \,|\nabla u|^2 \,\mathrm{d}V_g = -\int_{B_g(r)} u \mathrm{div}_g \left(|y|^a \,\mu \nabla u\right) \mathrm{d}V_g + \int_{\partial B_g(r)} |y|^a \,\mu u \partial_\rho u \mathrm{d}V_{\partial B_r}$$

and hence we can rewrite (158) as

$$\frac{d}{dr}H_g(u,r) = \frac{2}{r}E_g(u,r) + O(1)H_g(u,r).$$
(159)

We now focus on the derivative of $r \mapsto E_g(u, r)$, following the idea of the radial deformation in [48, 49]: for $0 < r, \Delta r < 1/2$ fixed, we define $w_t \colon \mathbb{R}^+ \to \mathbb{R}^+$ by

$$w_t(\rho) = \begin{cases} t, & \text{if } \rho \leq r \\ 1, & \text{if } \rho \geq r + \Delta r \\ t \frac{r + \Delta r - \rho}{\Delta r} + \frac{\rho - r}{\Delta r}, & \text{if } r \leq \rho \leq r + \Delta r. \end{cases}$$

Now, for $0 < t < 1 + \Delta r / (r + \Delta r)$, we define the bi-Lipschitz map $l_t \colon \mathbb{R}^{n+1} \to \mathbb{R}^{n+1}$ as

$$l_t(X) = w_t(\rho(X))X_t$$

with $\rho(X) = \text{dist}_q(0, X)$, and consequently the radial deformation u_t of u as

$$u^{t}(X) = u(l_{t}^{-1}(X)) \in H^{1,a}(B_{1}, \mathrm{d}V_{g}).$$

By definition we have $u^t(Z) = u(X)$, with $Z = l_t(X)$. Since u is a solution of (157), given the functional $I(t) = E_g(u^t, 1)$ we have

$$\left. \frac{d}{dt} I(t) \right|_{t=1} = 0. \tag{160}$$

In order to ease the notations, through the following computations we will simply use B_r instead of $B_q(r)$. Inspired by the definition of w(t), let us set

$$I(t) = \int_{B_{rt}} |y|^{a} \mu \left| \nabla u^{t} \right|^{2} dV_{B_{r}} + \int_{B_{r+\Delta r} \setminus B_{rt}} |y|^{a} \mu \left| \nabla u^{t} \right|^{2} dV_{B_{r}} + \int_{B_{1} \setminus B_{r+\Delta r}} |y|^{a} \mu \left| \nabla u^{t} \right|^{2} dV_{B_{r}}$$

= $I_{1}(t) + I_{2}(t) + I_{3}(t).$

It is easy to see that

$$I_3(t) = \int_{B_1 \setminus B_{r+\Delta r}} |y|^a \, \mu \left| \nabla u^t \right|^2 \mathrm{d}V_{B_r} = \int_{B_1 \setminus B_{r+\Delta r}} |y|^a \, \mu \left| \nabla u \right|^2 \mathrm{d}V_{B_r}$$

and consequently that $I_3(t)$ does not give contribution to the derivative of I(t). Next, we have

$$\begin{split} I_{1}(t) &= \int_{B_{rt}} \left| y \right|^{a} \mu \left| \nabla u^{t} \right|^{2} \mathrm{d} V_{B_{r}} \\ &= \int_{0}^{r} \int_{\partial B_{1}} t^{a} \left| (\rho, \theta_{n}) \right|^{a} \mu(t\rho, \theta) \partial_{\rho} u^{2}(\rho, \theta) \frac{\sqrt{g(t\rho, \theta)}}{t} \mathrm{d} \theta \mathrm{d} \rho \\ &+ \int_{0}^{r} \int_{\partial B_{1}} t^{a} \left| (\rho, \theta_{n}) \right|^{a} \mu(t\rho, \theta) b^{ij}(t\rho, \theta) \partial_{\theta_{i}} u(\rho, \theta) \partial_{\theta_{j}} u(\rho, \theta) t \sqrt{g(t\rho, \theta)} \mathrm{d} \theta \mathrm{d} \rho, \end{split}$$

where obviously b^{ij} are the entries of the inverse of $(b_{ij})_{ij}$ associated to the metric g. By (156), we get

$$\left|\frac{\partial}{\partial t}\mu(t\rho,\theta)\right| \leq \Lambda(n,\lambda,\Gamma)\rho.$$

Furthermore, we can rewrite

$$\begin{cases} \sqrt{g(t\rho,\theta)} = t^n \rho^n \sqrt{b(t\rho,\theta)} \\ b^{ij}(t\rho,\theta) \sqrt{g(t\rho,\theta)} = t^{n-2} \rho^{n-2} \left[\delta_{ij} + \varepsilon_{ij}(t\rho,\theta) \right] \end{cases}$$
(161)

for some $(\varepsilon_{ij}(t\rho,\theta))_{ij}$. Since that (154), we have

$$\left|\frac{\partial}{\partial t}\sqrt{b(t\rho,\theta)}\right| \le C(n,\Lambda)\rho, \quad \left|\frac{\partial}{\partial t}\sqrt{\varepsilon_{ij}(t\rho,\theta)}\right| \le C(n,\Lambda)\rho$$

which gives

$$I_{1}(t) = t^{n+a-1} \left[\int_{0}^{r} \int_{\partial B_{1}} |(\rho,\theta_{n})|^{a} \mu(t\rho,\theta) \partial_{\rho} u^{2}(\rho,\theta) \rho^{n} \sqrt{b(t\rho,\theta)} d\theta d\rho + \int_{0}^{r} \int_{\partial B_{1}} |(\rho,\theta_{n})|^{a} \mu(t\rho,\theta) \rho^{n-2} (\delta_{ij} + \varepsilon(t\rho,\theta)) \partial_{\theta_{i}} u(\rho,\theta) \partial_{\theta_{j}} u(\rho,\theta) d\theta d\rho \right],$$

and consequently

$$\frac{d}{dt}I_1(t)\bigg|_{t=1} = (n+a-1)\int_{B_r} |y|^a \,\mu \,|\nabla_g u|^2 \,\mathrm{d}V_g + O(r)\int_{B_r} |y|^a \,\mu \,|\nabla_g u|^2 \,\mathrm{d}V_g, \qquad (162)$$

with O(r) a function of (r, θ) whose absolute value is bounded by $C(n, \Lambda)r$.

Finally, in order to estimate the second term of I(t), we need to introduce the following notations. Hence, given $X \in B_{r+\Delta r} \setminus B_r$ and $Z = l_t(X) \in B_{r+\Delta r} \setminus B_{rt}$ let us consider their expression in the intrinsic geodesic polar coordinates associated to g, namely $X = (\rho, \theta)$ and $Z = (\gamma_t(\rho), \theta)$, where

$$\gamma_t(\rho) = \operatorname{dist}_g(Z, 0) = w_t(X)\rho = \rho \left[t \frac{r + \Delta r - \rho}{\Delta r} + \frac{\rho - r}{\Delta r} \right].$$

and

$$\frac{\partial}{\partial \rho} \gamma_t(\rho) = t \frac{r + \Delta r - 2\rho}{\Delta r} + \frac{2\rho - r}{\Delta r}.$$

Then, still using the polar coordinates, we have

$$\begin{split} \left| \nabla_g u^t(Z) \right|^2 &= \left| \partial_s u^t(s,\theta) \right|^2 + \frac{1}{s^2} b^{ij}(s,\theta) \partial_{\theta_i} u^t(s,\theta) \partial_{\theta_j} u^t(s,\theta) \right|_{s=\gamma_t(\rho)} \\ &= \left| \partial_\rho u(\rho,\theta) \right|^2 \left(\frac{\partial}{\partial s} \gamma_t^{-1}(s) \right|_{s=\gamma_t(\rho)} \right)^2 + \frac{1}{\gamma_t(\rho)^2} b^{ij}(\gamma_t(\rho),\theta) \partial_{\theta_i} u(\rho,\theta) \partial_{\theta_j} u(\rho,\theta), \\ &= \left| \partial_\rho u(\rho,\theta) \right|^2 h_t(\rho)^2 + \frac{1}{\gamma_t(\rho)^2} b^{ij}(\gamma_t(\rho),\theta) \partial_{\theta_i} u(\rho,\theta) \partial_{\theta_j} u(\rho,\theta), \end{split}$$

and similarly the volume element is given by

$$\mathrm{d}V_{B_r}(Z) = \gamma_t(\rho)^n \sqrt{g(\gamma_t(\rho), \theta)} \frac{\partial}{\partial \rho} \gamma_t(\rho) \mathrm{d}\rho \mathrm{d}\theta.$$

By the previous computations and the expansions in (161), we get

$$I_{2}(t) = \int_{B_{r+\Delta r} \setminus B_{rt}} |y|^{a} \mu \left| \nabla u^{t} \right|^{2} dV_{B_{r}}(Z)$$

$$= \int_{r}^{r+\Delta r} \int_{\partial B_{1}} s^{n} |(s,\theta_{n})|^{a} h_{t}(\rho) \mu(s,\theta) \partial_{\rho} u^{2}(\rho,\theta) \sqrt{b(s,\theta)} \left| \substack{\mathrm{d}\theta \mathrm{d}\rho} \right|_{s=\gamma_{t}(\rho)}$$

$$+ \int_{r}^{r+\Delta r} \int_{\partial B_{1}} s^{n-2} |(s,\theta_{n})|^{a} \frac{\partial}{\partial \rho} \gamma_{t}(\rho) \mu(s,\theta) (\delta_{ij} + \varepsilon(s,\theta)) \partial_{\theta_{i}} u(\rho,\theta) \partial_{\theta_{j}} u(\rho,\theta) \left| \substack{\mathrm{d}\theta \mathrm{d}\rho} \right|_{s=\gamma_{t}(\rho)}$$

Since

$$h_t(\rho) = \frac{\Delta r + t\rho - \rho}{t(r + \Delta r - \rho) + \rho - r}, \quad \frac{\partial}{\partial t}h_t(\rho)\Big|_{t=1} = -\frac{\Delta r + r - 2\rho}{\Delta r},$$

we can conclude

$$\frac{d}{dt}I_{2}(t)\Big|_{t=1} = \int_{B_{r+\Delta r}\setminus B_{r}} |y|^{a} \mu \left[(n+a+O(\rho)) \frac{r+\Delta r-\rho}{\Delta r} - \frac{r+\Delta r-2\rho}{\Delta r} \right] (\partial_{\rho}u)^{2} \mathrm{d}V_{B_{r}} \\
+ \int_{B_{r+\Delta r}\setminus B_{r}} |y|^{a} \mu \left[(n+a-2+O(\rho)) \frac{r+\Delta r-\rho}{\Delta r} + \frac{r+\Delta r-2\rho}{\Delta r} \right] \left(|\nabla_{g}u|^{2} - (\partial_{\rho}u)^{2} \right) \mathrm{d}V_{B_{r}}$$

Finally, by letting $\Delta r \rightarrow 0^+$ we get

$$\frac{d}{dt}I_2(t)\bigg|_{t=1} = 2r\int_{\partial B_r} |y|^a \,\mu(\partial_\rho u)^2 \mathrm{d}V_{\partial B_r} - r\int_{\partial B_r} |y|^a \,\mu \,|\nabla_g u|^2 \,\mathrm{d}V_{\partial B_r}.\tag{163}$$

From (160), (162) and (163), we obtain

$$r\frac{d}{dr}\int_{B_r}|y|^a\,\mu\,|\nabla_g u|^2\,\mathrm{d}V_{\partial B_r}-(n+a-1+O(r))\int_{B_r}|y|^a\,\mu\,(\nabla_g u)^2\,\mathrm{d}V_{B_r}=2r\int_{\partial B_r}|y|^a\,\mu(\partial_\rho u)^2\,\mathrm{d}V_{\partial B_r},$$

which implies with (159) that

$$\frac{d}{dr}\log N(X_0, u, r) = O(1) + \frac{2\int_{\partial B_r} |y|^a \,\mu(\partial_\rho u)^2 \mathrm{d}V_{\partial B_r}}{\int_{\partial B_r} |y|^a \,\mu u \partial_\rho u \mathrm{d}V_{\partial B_r}} - \frac{2\int_{\partial B_r} |y|^a \,\mu u \partial_\rho u \mathrm{d}V_{\partial B_r}}{\int_{\partial B_g(r)} |y|^a \,\mu u^2 \mathrm{d}V_{\partial B_r}} \ge -C(n, \Lambda),$$

where the inequality is a consequence of Schwarz's inequality. It follows immediately that the map $r \mapsto \exp(C(n, \Lambda)r)N_g(u, r)$ is a monotone nondecreasing function on $r \in (0, 1)$ as required.

Returning to the formulation of the problem in the euclidian metric, for $X_0 \in \Sigma$ and $r \in (0, 1 - |X_0|)$ we set

$$E(X_0, u, r) = \frac{1}{r^{n+a-1}} \int_{B_r(X_0)} |y|^a \langle \overline{A}(X) \nabla u, \nabla u \rangle \mathrm{d}X$$
$$H(X_0, u, r) = \frac{1}{r^{n+a}} \int_{\partial B_r(X_0)} |y|^a \mu_0 u^2 \mathrm{d}\sigma,$$

and consequently

$$N(X_0, u, r) = \frac{E(X_0, u, r)}{H(X_0, u, r)},$$

with μ_0 is a positive Lipschitz function bounded in B_1 satisfying (154) with Λ depending only on n, λ and Γ .

Corollary 3.8.12. Let $a \in (-1, 1)$ and u be a solution of (151) in B_1 symmetric with respect to Σ . Then there exist a constant C > 0 such that for every $X_0 \in B_1 \cap \Sigma$ the map

$$r \mapsto e^{Cr} N(X_0, u, r)$$

is absolutely continuous and monotone nondecreasing on $(0, 1 - |X_0|)$. Hence, there exists finite the Almgren frequency formula defined as

$$N(X_0, u, 0^+) = \lim_{r \to 0^+} N(X_0, u, r) = \inf_{r > 0} N(X_0, u, r).$$

Now, we can finally apply the previous analysis to the general case $(-L)^s$, by proving the validity of a doubling condition, a compactness result for blow-up sequences and a general Theorem on the structure of the nodal set itself.

Proposition 3.8.13. Let $a \in (-1, 1)$ and u be a solution of (151) in B_1 . Hence, there exists a constant $C = C(n, \Lambda)$ such that, for every $X_0 \in B_1 \cap \Sigma$,

$$H(X_0, u, r_2) \le CH(X_0, u, r_1) \left(\frac{r_2}{r_1}\right)^{2\tilde{C}}$$

for $0 < r_1 < r_2 < 1 - |X_0|$, where $\tilde{C} = N(X_0, u, R)e^{C(n,\Lambda)R}$.

Proof. Fixed $R = 1 - |X_0|$, by Corollary 3.8.12 we have that $N(X_0, u, r) \le e^{CR}N(X_0, u, R)$ for every $r \in (0, R)$. By (159) we get

$$\frac{d}{dr}\log H(X_0, u, r) = \frac{2}{r}N(X_0, u, r) + O(1)$$

$$\leq \frac{2}{r}N(X_0, u, R)e^{C(n,\Lambda)R} + C(n,\Lambda),$$

for every 0 < r < R. Now we integrate between $0 < r_1 < r_2 < R$, obtaining

$$\log \frac{H(X_0, u, r_2)}{H(X_0, u, r_1)} \le 2N(X_0, u, R)e^{C(n,\Lambda)R}\log \frac{r_2}{r_1} + C(n,\Lambda)(r_2 - r_1)$$

and finally

$$\frac{H(X_0, u, r_2)}{H(X_0, u, r_1)} \le e^{C(n, \Lambda)R} \left(\frac{r_2}{r_1}\right)^{2\tilde{C}}$$

with $\tilde{C} = N(X_0, u, R)e^{C(n,\Lambda)R}$.

Moreover, since we are dealing with the extension L_a^A of operator uniformly elliptic in divergence form with Lipschitz coefficent, we can easily extend Corollary 3.3.6 to our new class of operator following the technique developed in [85]. Indeed, since the lower bound on the Almgren frequency formula is based on the Hölder regularity of L_a^A -harmonic function, we easily get

Corollary 3.8.14. Let u be L_a^A -harmonic on B_1 , then for every $X_0 \in \Gamma(u) \cap \Sigma$ we have

$$N(X_0, u, 0^+) \ge \min\{1, 1 - a\}.$$
(164)

More precisely

- if u is symmetric with respect to Σ , we have $N(X_0, u, 0^+) \ge 1$,
- if u is antisymmetric with respect to Σ we have $N(X_0, u, 0^+) \ge 1 a$.

In particular, since in this Section we are focusing on the symmetric case, we directly get $N(X_0, u, 0^+) \ge 1$, for any $X_0 \in \Gamma(u) \cap \Sigma$. All techniques presented in this manuscript involve a local analysis of the solutions, which will be performed via a blow-up procedure. The following result are a generalization of the ones in Section 3.4. Fixed $a \in (-1, 1)$ and u an L_a^A -harmonic function in B_1 , for every $X_0 = (x_0,0) \in \Gamma(u) \cap \Sigma$ and $r_k \downarrow 0^+$ we define as the blow-up sequence the collection

$$u_k(X) = \frac{u(X_0 + r_k X)}{\sqrt{H(X_0, u, r_k)}} \quad \text{for } X \in X \in B_{X_0, r_k} = \frac{B_1 - X_0}{r_k},$$

such that $L_{a}^{A_{k}}u_{k} = 0$ and $||u_{k}||_{L^{2,a}(\partial B_{1})} = 1$, where

$$L_a^{A_k} = \operatorname{div}_{x,y}\left(|y|^a \,\overline{A}_k(x) \nabla_{x,y}\right), \quad \text{with } \overline{A}_k(x) = \overline{A}(x_0 + r_k x),$$

for every $X \in B_{X_0,r_h}$.

Proposition 3.8.15. Let $a \in (-1, 1)$. Given $X_0 \in \Gamma(u) \cap \Sigma$ and a blow-up sequence u_k centered in X_0 and associated to some $r_k \downarrow 0^+$, there exists $p \in H^{1,a}_{loc}(\mathbb{R}^n)$ such that, up to a subsequence, $u_k \to p$ in $C^{0,\alpha}_{loc}(\mathbb{R}^n)$ for every $\alpha \in (0, 1)$ and strongly in $H^{1,a}_{loc}(\mathbb{R}^n)$. In particular, the blow-up limit is and entire solution of following elliptic equation with constant coefficient

$$div_{x,y}\left(|y|^a \overline{A}(x_0)\nabla_{x,y}p\right) = 0 \quad \text{in } \mathbb{R}^{n+1}.$$

The proof of this result is a straightforward adaption of the one of Theorem 3.4.1. In particular, since the coefficient of \overline{A} are Lipschitz continuous and uniformly elliptic, all the computations of the blow-up argument follows the line of the local counterpart in [56, 66, 49, 48, 81, 4].

Moreover, since for every $X_0 \in \Gamma(u) \cap \Sigma$ the blow- up limit satisfies a degenerate-singular equation with constant coefficients, it is not restrictive to suppose that $\overline{A}(x_0) = \text{Id}$, since by trivial transformation we can rewrite the equation in a canonical form.

Therefore, all the results on the structure of the singular strata, proved in the previous part of the Section for the nodal set of *s*-harmonic functions, remain valid for the nodal set of fractional power of divergence form operator with Lipschitz leading coefficients. Indeed, as we already pointed out, in the proof of Theorem 3.7.7 and Theorem 3.7.8 we never used Proposition 3.2.5 in order to attain the result on the structure of the singular strata on Σ . The crucial idea is that the Whitney extension allows to study the structure of the nodal set just by using the generalized Taylor expansion (3.5.17) for symmetric function without the high-order differentiability of the function itself. In this way the results can be easily generalized to our class of operators.

Proposition 3.8.16. Given $s \in (0, 1)$, let u be a solution of

$$(-L)^s u = 0 \quad in B_1$$

with *L* a uniformly elliptic operator with Lipschitz coefficient defined as (149). Then the nodal set $\Gamma(u)$ splits into its regular and singular part

$$\mathcal{R}(u) = \{x \in \Gamma(u) \colon |\nabla u(x)| \neq 0\} \quad \textit{and} \quad \mathcal{S}(u) = \{x \in \Gamma(u) \colon |\nabla u(x)| = 0\}.$$

Moreover, if $u \in C^1(B_{1/2})$ on one hand $\mathcal{R}(u)$ is locally a smooth hypersurface and on the other one it holds

$$\mathcal{S}(u) = \mathcal{S}^*(u) \cup \mathcal{S}^s(u)$$

where $S^*(u)$ is contained in a countable union of (n-2)-dimensional C^1 manifolds and $S^s(u)$ is contained in a countable union of (n-1)-dimensional C^1 manifolds. Moreover

$$\mathcal{S}^*(u) = \bigcup_{j=0}^{n-2} \mathcal{S}^*_j(u) \quad \text{and} \quad \mathcal{S}^s(u) = \bigcup_{j=0}^{n-1} \mathcal{S}^s_j(u)$$

where both $S_i^*(u)$ and $S_i^s(u)$ are contained in a countable union of *j*-dimensional C^1 manifolds.

3.9 MEASURE ESTIMATES OF NODAL SETS OF *s*-harmonic functions

In this last Section, we estimate the measure of the nodal set $\Gamma(u)$ in the case of *s*-harmonic functions. This result can be seen as the nonlocal counterpart of a conjecture that Lin proposed in [66]. Indeed, following his strategy, we give an explicit estimate on the (n - 1)-Hausdorff measure of the nodal set in terms of the Almgren monotonicity formula of its L_a -extension.

As we already did in the previous Section, since the local structure of the nodal set $\Gamma(u)$ can be described using the results on the restriction of the nodal set of L_a -harmonic function on the characteristic manifold Σ , we will follow the notations previously introduced. More precisely, through this Section we will denote with $v \in H^{1,a}(B_1^+)$ the restriction on the unitary ball in \mathbb{R}^{n+1}_+ of the L_a -harmonic extension, defined by (144), symmetric with respect to Σ (see (145)).

Since the fractional Laplacian $(-\Delta)^s$ admits a representation formula, we directly have that the analyticity assumption, which is fundamental in order to apply a strategy developed in [66], is fully satisfied on every compact set $K \subset B_1$. Moreover, by Proposition 3.8.1 we already provide a quantitative doubling condition for *s*-harmonic functions strictly correlated to the one in the extended space \mathbb{R}^{n+1} .

In order to achieve the estimate on the Hausdorff measure of the nodal set $\Gamma(u)$ we use the following lemma relating the growth of a complex analytic function with the number of its zeros introduced in [41].

Lemma 3.9.1. Let $f: B_1 \subset \mathbb{C} \to \mathbb{C}$ be an analytic function such that

$$|f(0)| = 1$$
 and $\sup_{B_1} |f| \le 2^N$,

for some positive constant N. Then for any $r \in (0, 1)$

$$\# \{ z \in B_r \colon f(z) = 0 \} \le cN$$

and

$$\# \{ z \in B_{1/2} \colon f(z) = 0 \} \le N,$$

where C is a positive constant depending only on the radius r.

Before to state the main result on the measure of the nodal sets $\Gamma(u)$ in terms of the Almgren monotonicity formula of the L_a -harmonic extension, let us start with an example in the setting of tangent maps $\mathfrak{B}_k^s(\mathbb{R}^n)$ that emphasizes how the measure of the nodal set is strictly related to the class of tangent maps that we are considering. More precisely, the classes $\mathfrak{B}_k^*(\mathbb{R}^n)$ and $\mathfrak{B}_k^s(\mathbb{R}^n) \setminus \mathfrak{B}_k^*(\mathbb{R}^n)$ strictly affect the local measure of the nodal set.

First, it is not restrictive to assume that $\varphi \in \mathfrak{B}_k^s(\mathbb{R}^2)$ for some $k \in 1 + \mathbb{N}$. Hence, consider the case n = 2 with the notation $(x, z) \in \mathbb{R}^2$. Since every $\varphi \in \mathfrak{B}_k^*(\mathbb{R}^2)$ is harmonic in \mathbb{R}^2 , it is known that

$$\mathcal{H}^1\left(\Gamma(\varphi) \cap B_1\right) = 2k.$$

Instead, for $\varphi \in \mathfrak{B}_k^s(\mathbb{R}^2) \setminus \mathfrak{B}_k^*(\mathbb{R}^2)$, with $k \ge 2$, the previous bound turn to be not optimal. More precisely, given the constant $k' = \#\{t \in \mathbb{R} : \varphi(t, 1) = 0\}$, we get

$$\mathcal{H}^1\left(\Gamma(\varphi) \cap B_1\right) = 2k',$$

where, by the Fundamental Theorem of Algebra, it is obvious to see that $0 \le k' \le k$.

In general, we prove the following result which is based on an argument first introduced in [66] in the context of solution of second order elliptic equation with analytic coefficient.

More recently, in [6] the author constructs a similar estimate in a more general context connecting the Hausdorff measure of the nodal set of smooth functions with their finite vanishing order, which can be also applied to our case. Unfortunately, the remarkable difference between the case $\mathfrak{B}_k^*(\mathbb{R}^n)$ and $\mathfrak{B}_k^s(\mathbb{R}^n) \setminus \mathfrak{B}_k^*(\mathbb{R}^n)$ (or similarly $\mathfrak{sB}_k^*(\mathbb{R}^{n+1})$ and $\mathfrak{sB}_k^a(\mathbb{R}^{n+1}) \setminus \mathfrak{sB}_k^*(\mathbb{R}^{n+1})$) implies the not optimality of the result of Bär in our setting.

Theorem 3.9.2. Given $s \in (0, 1)$, let u be an s-harmonic function in B_1 and $0 \in \Gamma(u)$. Then

$$\mathcal{H}^{n-1}\left(\Gamma(u) \cap B_{\frac{1}{2}}\right) \le C(n,s)N,$$

where N = N(0, v, 1) is the frequency of the L_a -harmonic extension v in B_1^+ defined by

$$N = \frac{\int_{B_1^+} |y|^a |\nabla v|^2 \,\mathrm{d}X}{\int_{\partial B_1^+} |y|^a v^2 \mathrm{d}\sigma}.$$

Proof. Let $(B_R(p_i))_i$ be a finite cover of $B_{1/2}$ with R < 1/8 and $p_i \in B_{1/2}$. Moreover, up to a normalization, it is not restrictive to assume that

$$\int_{B_1} u^2 \mathrm{d}x = 1.$$

By Proposition 3.3.5 and Proposition 3.8.1, for every $p_i \in B_{1/2}$ we have

$$\int_{B_r(p_i)} u^2 \mathrm{d}x \ge 4^{-C(n,s)N} \int_{B_{2r}(p_i)} u^2 \mathrm{d}x,$$

with 0 < r < 1/4 and N = N(0, v, 1). Moreover, using the normalization hypothesi, we get

$$\int_{B_R(p_i)} u^2 \mathrm{d}x \ge 4^{-C(n,s)N}.$$

Given $p_1, \ldots, p_j \in B_{1/2}$ the collection of points associated to the covering, let us consider $(x_{p_i})_i \in B_R(p_i)$ such that

$$|u(x_{p_i})| \ge 2^{-C(n,s)N}$$
, for any $i = 1, \dots, j$.

In order to apply Lemma 3.9.1, for i = 1, ..., j consider the collection of analytic functions of one complex variable defined as

$$f_i(w, z) = u(x_{p_i} + 4Rzw), \text{ for } w \in S^{n-1}, z \in B_1^{\mathbb{C}}$$

Then, by construction, we have

$$|f_i(w,0)| \ge 2^{-C(n,s)N}$$
 and $|f_i(w,z)| \le C$,

for some positive dimensional constant C > 0. Since, by Lemma 3.9.1 we have

$$N_{i}(w) = \# \left\{ x \in B_{2R}(x_{p_{i}}) : u(x) = 0 \quad \text{for} (x - x_{p_{j}}) \parallel w \right\}$$

$$\leq \# \left\{ z \in B_{1/2}^{\mathbb{C}} : f_{i}(w, z) = 0 \right\}$$

$$\leq c(n, s, N)N,$$

for every $i = 1, \dots, j$, by the integral geometric formula in [46, Theorem 3.2.27], we finally obtain

$$\mathcal{H}^{n-1}(\Gamma(u) \cap B_{1/2}) \le \sum_{i=1}^{j} \mathcal{H}^{n-1}(\Gamma(v) \cap B_R(p_i)) \le c(n, s, N) \sum_{i=1}^{j} \int_{S^{n-1}} N_i(w) \mathrm{d}w \le C(n, s, N) N$$

where in the second inequality we used $B_R(p_i) \subset B_{2R}(x_{p_i})$ for every $i = 1, \ldots, j$.

In the end, since our estimate on the Hausdorff measure is deeply based on the existence of an L_a -harmonic extension of u and on the validity of an Almgren's type monotonicity result, we expect to improve Theorem ?? exploiting the connection between the Dirichlet energy associated to the L_a -extension and the Gagliardo seminorm introduced in (143). This improvement would show a purely nonlocal version of the result in [66].

REFERENCES

- F. J. Almgren Jr. "Dirichlet's problem for multiple valued functions and the regularity of mass minimizing integral currents". In: *Minimal submanifolds and geodesics (Proc. Japan-United States Sem., Tokyo, 1977)*. North-Holland, Amsterdam-New York, 1979, pp. 1–6.
- [2] H. W. Alt and L. Caffarelli. "Existence and regularity for a minimum problem with free boundary". In: *J. Reine Angew. Math.* 325 (1981), pp. 105–144.
- [3] H. W. Alt, L. Caffarelli, and A. Friedman. "Variational problems with two phases and their free boundaries". In: *Trans. Amer. Math. Soc.* 282.2 (1984), pp. 431–461.
- [4] N. Aronszajn, A. Krzywicki, and J. Szarski. "A unique continuation theorem for exterior differential forms on Riemannian manifolds". In: *Ark. Mat.* 4 (1962), 417–453 (1962).
- [5] R. Banuelos and K. Bogdan. "Symmetric stable processes in cones". English. In: Potential Analysis 21.3 (2004), pp. 263–288.
- [6] C. Bär. "Zero sets of solutions to semilinear elliptic systems of first order". In: *Invent. Math.* 138.1 (1999), pp. 183–202.
- [7] B. Barrios, A. Figalli, and X. Ros-Oton. "Global regularity for the free boundary in the obstacle problem for the fractional Laplacian". In: *preprint* to appear (2015).
- [8] K. Bogdan and T. Byczkowski. "Potential theory for the α -stable Schrödinger operator on bounded Lipschitz domains". In: *Studia Math.* 133.1 (1999), pp. 53–92.
- [9] K. Bogdan, B. Siudeja, and A. Stós. "Martin Kernel for Fractional Laplacian in Narrow Cones". English. In: *Potential Analysis* 42.4 (2015), pp. 839–859.
- [10] J. Bourgain, H. Brezis, and P. Mironescu. Another look at Sobolev spaces. Original research article appeared at in Optimal Control and Partial Differential Equations IOS Press ISBN 1 58603 096 5. 2001.
- S. P. Burke and T. E. W. Schumann. "Diffusion Flames". In: Industrial & Engineering Chemistry 20.10 (1928), pp. 998–1004.
- X. Cabré and X. Ros-Oton. "Sobolev and isoperimetric inequalities with monomial weights". In: J. Differential Equations 255.11 (2013), pp. 4312–4336.
- [13] X. Cabré and Y. Sire. "Nonlinear equations for fractional Laplacians, I: Regularity, maximum principles, and Hamiltonian estimates". In: *Ann. Inst. H. Poincaré Anal. Non Linéaire* 31.1 (2014), pp. 23–53.

- [14] L. Caffarelli, D. De Silva, and O. Savin. "The two membranes problem for different operators". In: Ann. Inst. H. Poincaré Anal. Non Linéaire 34.4 (2017), pp. 899–932.
- [15] L. Caffarelli and A. Friedman. "Partial regularity of the zero-set of solutions of linear and superlinear elliptic equations". In: *J. Differential Equations* 60.3 (1985), pp. 420–433.
- [16] L. Caffarelli, D. Jerison, and C. Kenig. "Some new monotonicity theorems with applications to free boundary problems". In: *Ann. of Math. (2)* 155.2 (2002), pp. 369–404.
- [17] L. Caffarelli, A. Karakhanyan, and F.-H. Lin. "The geometry of solutions to a segregation problem for nondivergence systems". In: J. Fixed Point Theory Appl. 5.2 (2009), pp. 319–351.
- [18] L. Caffarelli and F.-H. Lin. "Analysis on the junctions of domain walls". In: Discrete Contin. Dyn. Syst. 28.3 (2010), pp. 915–929.
- [19] L. Caffarelli and F.-H. Lin. "Singularly perturbed elliptic systems and multi-valued harmonic functions with free boundaries". In: J. Amer. Math. Soc. 21.3 (2008), pp. 847–862.
- [20] L. Caffarelli and J.-M. Roquejoffre. "Uniform Hölder estimates in a class of elliptic systems and applications to singular limits in models for diffusion flames". In: Arch. Ration. Mech. Anal. 183.3 (2007), pp. 457–487.
- [21] L. Caffarelli and S. Salsa. *A geometric approach to free boundary problems*. Vol. 68. Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2005, pp. x+270.
- [22] L. Caffarelli, S. Salsa, and L. Silvestre. "Regularity estimates for the solution and the free boundary of the obstacle problem for the fractional Laplacian". In: *Invent. Math.* 171.2 (2008), pp. 425–461.
- [23] L. Caffarelli and L. Silvestre. "An extension problem related to the fractional Laplacian". In: Comm. Partial Differential Equations 32.7-9 (2007), pp. 1245–1260.
- [24] L. Caffarelli and L. Silvestre. "Regularity theory for fully nonlinear integro-differential equations". In: *Comm. Pure Appl. Math.* 62.5 (2009), pp. 597–638.
- [25] L. Cafferelli and F.-H. Lin. "An optimal partition problem for eigenvalues". In: J. Sci. Comput. 31.1-2 (2007), pp. 5–18.
- [26] S.-M. Chang et al. "Segregated nodal domains of two-dimensional multispecies Bose-Einstein condensates". In: *Phys. D* 196.3-4 (2004), pp. 341–361.
- [27] S.-Y. A. Chang and M. del Mar González. "Fractional Laplacian in conformal geometry". In: Adv. Math. 226.2 (2011), pp. 1410–1432.
- [28] Z. Chen and R. Song. "Estimates on Green functions and Poisson kernels for symmetric stable processes". English. In: *Mathematische Annalen* 312.3 (1998), pp. 465–501.
- [29] M. Conti, S. Terracini, and G. Verzini. "A variational problem for the spatial segregation of reaction-diffusion systems". In: *Indiana Univ. Math. J.* 54.3 (2005), pp. 779–815.

- [30] M. Conti, S. Terracini, and G. Verzini. "An optimal partition problem related to nonlinear eigenvalues". In: *J. Funct. Anal.* 198.1 (2003), pp. 160–196.
- [31] M. Conti, S. Terracini, and G. Verzini. "Asymptotic estimates for the spatial segregation of competitive systems". In: *Adv. Math.* 195.2 (2005), pp. 524–560.
- [32] M. Conti, S. Terracini, and G. Verzini. "Asymptotic estimates for the spatial segregation of competitive systems". In: *Adv. Math.* 195.2 (2005), pp. 524–560.
- [33] M. Conti, S. Terracini, and G. Verzini. "Nehari's problem and competing species systems". In: Ann. Inst. H. Poincaré Anal. Non Linéaire 19.6 (2002), pp. 871–888.
- [34] E. N. Dancer, K. Wang, and Z. Zhang. "Dynamics of strongly competing systems with many species". In: *Trans. Amer. Math. Soc.* 364.2 (2012), pp. 961–1005.
- [35] E. N. Dancer, K. Wang, and Z. Zhang. "The limit equation for the Gross-Pitaevskii equations and S. Terracini's conjecture". In: *J. Funct. Anal.* 262.3 (2012), pp. 1087–1131.
- [36] E. N. Dancer, K. Wang, and Z. Zhang. "Uniform Hölder estimate for singularly perturbed parabolic systems of Bose-Einstein condensates and competing species". In: J. Differential Equations 251.10 (2011), pp. 2737–2769.
- [37] J. Dávila et al. "Concentration phenomena for the nonlocal Schrödinger equation with Dirichlet datum". In: *Anal. PDE* 8.5 (2015), pp. 1165–1235.
- [38] E. Di Nezza, G. Palatucci, and E. Valdinoci. "Hitchhiker's guide to the fractional Sobolev spaces". In: *Bull. Sci. Math.* 136.5 (2012), pp. 521–573.
- [39] S. Dipierro, O. Savin, and E. Valdinoci. "All functions are locally *s*-harmonic up to a small error". In: *J. Eur. Math. Soc. (JEMS)* 19.4 (2017), pp. 957–966.
- [40] S. Dipierro, O. Savin, and E. Valdinoci. "Definition of fractional Laplacian for functions with polynomial growth". In: *Rev. Mat. Iberoam* ().
- [41] H. Donnelly and C. Fefferman. "Nodal sets of eigenfunctions on Riemannian manifolds". In: *Invent. Math.* 93.1 (1988), pp. 161–183.
- [42] D. E. Edmunds, A. Kufner, and J. Rákosní k. "Embeddings of Sobolev spaces with weights of power type". In: *Z. Anal. Anwendungen* 4.1 (1985), pp. 25–34.
- [43] E. Fabes, C. Kenig, and R. Serapioni. "The local regularity of solutions of degenerate elliptic equations". In: *Comm. Partial Differential Equations* 7.1 (1982), pp. 77–116.
- [44] E. B. Fabes, C. E. Kenig, and D. Jerison. "Boundary behavior of solutions to degenerate elliptic equations". In: *Conference on harmonic analysis in honor of Antoni Zygmund, Vol. I, II (Chicago, Ill., 1981).* Wadsworth Math. Ser. Wadsworth, Belmont, CA, 1983, pp. 577–589.

- [45] M. M. Fall and V. Felli. "Unique continuation property and local asymptotics of solutions to fractional elliptic equations". In: *Comm. Partial Differential Equations* 39.2 (2014), pp. 354– 397.
- [46] H. Federer. *Geometric measure theory*. Die Grundlehren der mathematischen Wissenschaften, Band 153. Springer-Verlag New York Inc., New York, 1969, pp. xiv+676.
- [47] S. Friedland and W. K. Hayman. "Eigenvalue inequalities for the Dirichlet problem on spheres and the growth of subharmonic functions". In: *Comment. Math. Helv.* 51.2 (1976), pp. 133–161.
- [48] N. Garofalo and F.-H. Lin. "Monotonicity properties of variational integrals, A_p weights and unique continuation". In: *Indiana Univ. Math. J.* 35.2 (1986), pp. 245–268.
- [49] N. Garofalo and F.-H. Lin. "Unique continuation for elliptic operators: a geometric-variational approach". In: *Comm. Pure Appl. Math.* 40.3 (1987), pp. 347–366.
- [50] N. Garofalo and A. Petrosyan. "Some new monotonicity formulas and the singular set in the lower dimensional obstacle problem". In: *Invent. Math.* 177.2 (2009), pp. 415–461.
- [51] N. Garofalo and X. Ros-Oton. "Structure and regularity of the singular set in the obstacle problem for the fractional Laplacian". In: *preprint* to appear (2017).
- [52] N. Garofalo and D. Vassilev. "Strong unique continuation properties of generalized Baouendi-Grushin operators". In: *Comm. Partial Differential Equations* 32.4-6 (2007), pp. 643–663.
- [53] G. F. Gause. "Experimental Studies on the Struggle for Existence". In: 9.4 (1932), pp. 389-402.
- [54] D. S. Hall et al. Dynamics of Component Separation in a Binary Mixture of Bose-Einstein Condensates, pp. 515–518. eprint: https://www.worldscientific.com/doi/ pdf/10.1142/9789812813787_0071.
- [55] Q. Han. "Nodal sets of harmonic functions". In: *Pure Appl. Math. Q.* 3.3, Special Issue: In honor of Leon Simon. Part 2 (2007), pp. 647–688.
- [56] Q. Han. "Singular sets of solutions to elliptic equations". In: Indiana Univ. Math. J. 43.3 (1994), pp. 983-1002.
- [57] Q. Han, R. Hardt, and F.-H. Lin. "Geometric measure of singular sets of elliptic equations". In: *Comm. Pure Appl. Math.* 51.11-12 (1998), pp. 1425–1443.
- [58] R. Hardt and L. Simon. "Nodal sets for solutions of elliptic equations". In: J. Differential Geom. 30.2 (1989), pp. 505–522.
- [59] N. Humphries. "Environmental context explains Lévy and Brownian movement patterns of marine predators". In: *Nature* 465.7301 (2010), pp. 1066–1069.
- [60] D. Jerison and C. Kenig. "Boundary behavior of harmonic functions in nontangentially accessible domains". In: *Adv. in Math.* 46.1 (1982), pp. 80–147.

- [61] K. Kasamatsu, M. Tsubota, and M. Ueda. "Vortices in multicomponent Bose-Einstein condensates". In: *International Journal of Modern Physics B* 19.11 (2005), pp. 1835–1904.
- [62] J. Lamboley, Y. Sire, and E. V. Teixeira. "Free boundary problems involving singular weights". In: *ArXiv e-prints* (Nov. 2017). arXiv: 1711.08339 [math.AP].
- [63] N. S. Landkof. Foundations of modern potential theory. Translated from the Russian by A. P. Doohovskoy, Die Grundlehren der mathematischen Wissenschaften, Band 180. Springer-Verlag, New York-Heidelberg, 1972, pp. x+424.
- [64] N. Laskin. "Fractional quantum mechanics". In: Phys. Rev. E 62 (3 2000), pp. 3135-3145.
- [65] N. Laskin. "Fractional Schrödinger equation". In: Phys. Rev. E 66 (5 2002), p. 056108.
- [66] F.-H. Lin. "Nodal sets of solutions of elliptic and parabolic equations". In: Comm. Pure Appl. Math. 44.3 (1991), pp. 287–308.
- [67] K. Michalik. "Sharp estimates of the Green function, the Poisson kernel and the Martin kernel of cones for symmetric stable processes". In: *Hiroshima Math. J.* 36.1 (2006), pp. 1–21.
- [68] K. Miller. "Nonunique continuation for uniformly parabolic and elliptic equations in selfadjoint divergence form with Hölder continuous coefficients". In: Arch. Rational Mech. Anal. 54 (1974), pp. 105–117.
- [69] V. Millot and Y. Sire. "On a fractional Ginzburg-Landau equation and 1/2-harmonic maps into spheres". In: Arch. Ration. Mech. Anal. 215.1 (2015), pp. 125–210.
- [70] A. Nekvinda. "Characterization of traces of the weighted Sobolev space $W^{1,p}(\Omega, d_M^{\epsilon})$ on M". In: Czechoslovak Math. J. 43(118).4 (1993), pp. 695–711.
- [71] B. Noris et al. "Convergence of minimax structures and continuation of critical points for singularly perturbed systems". In: *J. Eur. Math. Soc.* (*JEMS*) 14.4 (2012), pp. 1245–1273.
- [72] B. Noris et al. "Uniform Hölder bounds for nonlinear Schrödinger systems with strong competition". In: *Comm. Pure Appl. Math.* 63.3 (2010), pp. 267–302.
- [73] A. Rüland. "On quantitative unique continuation properties of fractional Schrödinger equations: doubling, vanishing order and nodal domain estimates". In: *Trans. Amer. Math. Soc.* 369.4 (2017), pp. 2311–2362.
- [74] H. Shahgholian. "When does the free boundary enter into corner points of the fixed boundary?" In: *Zap. Nauchn. Sem. S.-Peterburg. Otdel. Mat. Inst. Steklov. (POMI)* 310.Kraev. Zadachi Mat. Fiz. i Smezh. Vopr. Teor. Funkts. 35 [34] (2004), pp. 213–225, 229.
- [75] N. Shigesada, K. Kawasaki, and E. Teramoto. "Spatial segregation of interacting species". In: J. Theoret. Biol. 79.1 (1979), pp. 83–99.
- [76] L. Silvestre. *Regularity of the obstacle problem for a fractional power of the Laplace operator.* Thesis (Ph.D.)–The University of Texas at Austin. ProQuest LLC, Ann Arbor, MI, 2005, p. 95.

- [77] L. Silvestre. *Regularity of the obstacle problem for a fractional power of the Laplace operator.* Thesis (Ph.D.)–The University of Texas at Austin. ProQuest LLC, Ann Arbor, MI, 2005, p. 95.
- [78] L. Simon. Lectures on geometric measure theory. Vol. 3. Proceedings of the Centre for Mathematical Analysis, Australian National University. Australian National University, Centre for Mathematical Analysis, Canberra, 1983, pp. vii+272.
- [79] L. Simon. Theorems on regularity and singularity of energy minimizing maps. Lectures in Mathematics ETH Zürich. Based on lecture notes by Norbert Hungerbühler. Birkhäuser Verlag, Basel, 1996, pp. viii+152.
- [80] P. R. Stinga and J. L. Torrea. "Extension problem and Harnack's inequality for some fractional operators". In: *Comm. Partial Differential Equations* 35.11 (2010), pp. 2092–2122.
- [81] H. Tavares and S. Terracini. "Regularity of the nodal set of segregated critical configurations under a weak reflection law". In: *Calc. Var. Partial Differential Equations* 45.3-4 (2012), pp. 273-317.
- [82] S. Terracini, G. Tortone, and S. Vita. "On s-harmonic functions on cones". In: Accepted in Anal. PDE (May 2017). arXiv: 1705.03717 [math.AP].
- [83] S. Terracini, G. Verzini, and A. Zilio. "Uniform Hölder bounds for strongly competing systems involving the square root of the laplacian". English. In: *Journal of the European Mathematical Society* 18.12 (2016), pp. 2865–2924.
- [84] S. Terracini, G. Verzini, and A. Zilio. "Uniform Hölder regularity with small exponent in competition-fractional diffusion systems". English. In: *Discrete and Continuous Dynamical Systems- Series A* 34.6 (2014), pp. 2669–2691.
- [85] S. Terracini and S. Vita. "On the asymptotic growth of positive solutions to a nonlocal elliptic blow-up system involving strong competition". In: Annales de l'Institut Henri Poincare (C) Non Linear Analysis (2017), pp. –.
- [86] G. Verzini and A. Zilio. "Strong competition versus fractional diffusion: the case of Lotka-Volterra interaction". In: *Comm. Partial Differential Equations* 39.12 (2014), pp. 2284–2313.
- [87] S. Vita. *Strong competition systems ruled by anaomalous diffusions*. Thesis (Ph.D.)– Politecnico di Torino and Università di Torino.
- [88] K. Wang and J. Wei. "On the uniqueness of solutions of a nonlocal elliptic system". English. In: *Mathematische Annalen* 365.1-2 (2016), pp. 105–153.
- [89] K. Wang and Z. Zhang. "Some new results in competing systems with many species". In: Ann. Inst. H. Poincaré Anal. Non Linéaire 27.2 (2010), pp. 739–761.
- [90] J. Wei and T. Weth. "Asymptotic behaviour of solutions of planar elliptic systems with strong competition". In: *Nonlinearity* 21.2 (2008), pp. 305–317.
- [91] G. Weiss. "A homogeneity improvement approach to the obstacle problem". In: *Invent. Math.* 138.1 (1999), pp. 23–50.
- [92] H. Whitney. "Analytic extensions of differentiable functions defined in closed sets". In: *Trans. Amer. Math. Soc.* 36.1 (1934), pp. 63–89.